SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-K

×	ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
	For the fiscal year ended December 31, 2004	
	or	
	TRANSITION REPORT PURSUANT TO SECTION 13 OR 1934	15(d) OF THE SECURITIES EXCHANGE ACT OF
	For the transition period from to	
	Commission File Numb	er: 1-14100
	IMPAC MORTGAGE (Exact name of registrant as speci	•
	Maryland (State or other jurisdiction of incorporation or organization)	33-0675505 (I.R.S. Employer Identification No.)
	1401 Dove Street, Newport Beac (Address of principal execu	
	(949) 475-360	_
	(Registrant's telephone number, in	cluding area code)
	Citi	12(h) of the A -t.
	Securities registered pursuant to Securities registered registered pursuant to Securities registered registered registered registered register	Name of each exchange on which registered
	Common Stock, \$0.01 par value	New York Stock Exchange
	Preferred Share Purchase Rights	New York Stock Exchange
	9.375% Series B Cumulative Redeemable Preferred Stock 9.125% Series C Cumulative Redeemable Preferred Stock	New York Stock Exchange New York Stock Exchange
the pre	e by check mark whether the registrant (1) has filed all reports required to be fileceding 12 months (or for such shorter period that the registrant was required to fit 90 days. Yes \boxtimes No \square	
	e by check mark if disclosure of delinquent filers pursuant to Item 405 of Regula ant's knowledge, in definitive proxy or information statements incorporated by re	
Indicat	e by check mark whether the registrant is an accelerated filer (as defined in Exch	ange Act Rule 12b-2) Yes ⊠ No □
sales p	une 30, 2004, the aggregate market value of the voting stock held by non-affiliatorice of common stock on the New York Stock Exchange on that date. For purpose ant have been deemed affiliates. There were 75,250,097 shares of common stock	es of the calculation only, all directors and executive officers of the
Portior annual	ns of our Definitive Proxy Statement issued in connection with our 2005 Annual I report.	Meeting of Stockholders are incorporated by reference into Part III of this

EXPLANATORY NOTE

As disclosed in previous reports, we restated our financial statements for each of the years ended December 31, 2003, 2002 and 2001, the three months ended March 31, 2004 and 2003, the three and six months ended June 30, 2003 and the three and nine months ended September 30, 2003. The Company also restated its consolidated statements of operations for the six months ended June 30, 2004 due to a correction to earnings per share as a result of a clerical error. The time and resources committed to the restatements delayed our internal timetable with respect to our documentation, assessment and evaluation of internal control over financial reporting, which were undertaken in order to comply with the requirements of Section 404 of the Sarbanes-Oxley Act of 2002, or "SOX 404." Due in large part to the foregoing, management's assessment of the effectiveness of our internal control over financial reporting was substantially delayed, thereby delaying the performance by KPMG LLP, or "KPMG," our independent registered public accounting firm, of its audit of management's assessment of the effectiveness of our internal control over financial reporting pursuant to the requirements of SOX 404. As a result, at the date of this filing, such audit and management's report on internal control over financial reporting is not yet complete. In addition, the audit of our financial statements is not complete. Therefore, the audit reports from KPMG on the financial statements and its audit of management's assessment of the effectiveness of internal control over financial reporting are not included in this annual report on Form 10-K. Consequently, this annual report on Form 10-K includes unaudited financial statements and statistical information for the year ended December 31, 2004. The financials statements as of December 31, 2003 and for the years ended December 31, 2003 and 2002 have been derived from the audited financials statements for such periods. Upon completion of our assessment of the effectiveness of internal control over fin

Management believes that the financial statements included in this report fairly present in all material respects our financial condition, results of operations and cash flows for the periods presented. However, because no independent registered public accounting firm has opined that the unaudited financial statements included in this annual report present fairly, in all material respects, the financial position, the results of operations, cash flows and the changes in shareholders' equity for the year ended December 31, 2004 reported in accordance with generally accepted accounting principles in the United States of America, or "GAAP," such statements could be subject to change or the financial information included therein may be materially different from audited financial information. There can be no assurance that such changes or differences would not be significant. Our discussion and analysis of our financial condition and results of operations are based upon these unaudited financial statements.

IMPAC MORTGAGE HOLDINGS, INC. 2004 FORM 10-K ANNUAL REPORT TABLE OF CONTENTS

PART I

ITEM 1.	<u>BUSINESS</u>	1
	Forward-Looking Statements	1
	Available Information	1
	General Overview	2
	Long-Term Investment Operations	3
	Mortgage Operations	6
	Warehouse Lending Operations	12
	<u>Regulation</u>	12
	<u>Competition</u>	12
	<u>Employees</u>	13
	Revisions in Policies and Strategies	13
	Risk Factors	14
ITEM 2.	<u>PROPERTIES</u>	30
ITEM 3.	LEGAL PROCEEDINGS	31
ITEM 4.	SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS	32
	PART II	
ITEM 5.	MARKET FOR REGISTRANT'S COMMON EQUITY AND RELATED STOCKHOLDER MATTERS	33
ITEM 6.	SELECTED UNAUDITED CONSOLIDATED FINANCIAL DATA	33
ITEM 7.	MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS	36
	Use of Unaudited Financial Statements	36
	Consolidation of IFC	36
	Critical Accounting Policies	37
	Financial Highlights for 2004	38
	Taxable Income	39
	<u>Issuance of Preferred Stock</u>	40
	Estimated Expenditures Related to Section 404 of the Sarbanes-Oxley Act of 2002	41
	Results of Operations and Financial Condition	41
	<u>Liquidity and Capital Resources</u>	51
	Off-Balance Sheet Arrangements	55
	Contractual Obligations	55
ITEM 7A.	QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK	55
	General Overview	55
	<u>Changes in Interest Rates</u>	56
ITEM 8.	UNAUDITED FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA	58
ITEM 9.	CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE	58

IMPAC MORTGAGE HOLDINGS, INC. 2004 FORM 10-K ANNUAL REPORT TABLE OF CONTENTS

PART II

ITEM 9A.	CONTROLS AND PROCEDURES	59
ITEM 9B.	OTHER INFORMATION	61
	<u>PART III</u>	
ITEM 10.	DIRECTORS AND EXECUTIVE OFFICERS OF THE REGISTRANT	62
ITEM 11.	EXECUTIVE COMPENSATION	62
ITEM 12.	SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT	62
ITEM 13.	CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS	62
ITEM 14.	PRINCIPAL ACCOUNTANT FEES AND SERVICES	62
	<u>PART IV</u>	
ITEM 15.	EXHIBITS AND FINANCIAL STATEMENT SCHEDULES	63
SIGNATUR	ES	68

PART I

ITEM 1. BUSINESS

Unless the context otherwise requires, the terms "Company," "we," "us," and "our" refer to Impac Mortgage Holdings, Inc. ("IMH"), a Maryland corporation incorporated in August 1995, and its wholly-owned subsidiaries, IMH Assets Corp., or "IMH Assets," Impac Warehouse Lending Group, Inc., or "IWLG," Impac Multifamily Capital Corporation, or "IMCC," and Impac Funding Corporation, or "IFC," together with its wholly-owned subsidiaries Impac Secured Assets Corp., or "ISAC," and Novelle Financial Services, Inc., or "Novelle."

Forward-Looking Statements

This report on Form 10-K contains certain forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. Forward-looking statements, some of which are based on various assumptions and events that are beyond our control, may be identified by reference to a future period or periods or by the use of forward-looking terminology, such as "may," "will," "believe," "expect," "likely," "should," "anticipate," or similar terms or variations on those terms or the negative of those terms. The forward-looking statements are based on current management expectations. Actual results may differ materially as a result of several factors, including, but not limited to, failure to achieve projected earnings levels; the ability to generate sufficient liquidity and conduct our operations as planned; the ability to access the equity markets; delays in raising, or the inability to raise, additional capital, either through equity offerings, lines of credit or otherwise as a result of, among other things, market conditions or the inability to provide, or delay in providing, audited financial statements and the auditor's report on our internal control over financial reporting; delays in the anticipated filing of, or the inability to file, audited financial statements and the audit report on our internal control over financial reporting due to unanticipated delays in completing the audit pursuant to SOX 404; the ability to generate taxable income and to pay dividends; interest rate fluctuations and changes in expectations of future interest rates; changes in prepayment rates on our mortgages; the availability of financing and, if available, the terms of any financing; continued ability to access the securitization markets or other funding sources; risks related to our ability to maintain an effective system of internal control over financial reporting and disclosure controls and procedures due to reported or potential material weaknesses and the ability to remediate any material weaknesses; changes in markets which the Company serves; risks relating to our ability to receive from our external auditors an audit report on our internal control over financial reporting due to unforeseen issues or delays; the effectiveness of risk management strategies; and changes in other general market and economic conditions. For a discussion of these and other risks and uncertainties that could cause actual results to differ from those contained in the forward-looking statements, see "Risk Factors" and Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations" in this report. We do not undertake, and specifically disclaim any obligation, to publicly release the results of any revisions that may be made to any forward-looking statements to reflect the occurrence of anticipated or unanticipated events or circumstances after the date of such statements.

Available Information

Our Internet website address is www.impaccompanies.com. We make our annual report on Form 10-K, quarterly reports on Form 10-Q, current reports on Form 8-K and proxy statement for our annual stockholders' meetings, as well as any amendments to those reports, available free of charge through our website as soon as reasonably practicable after we electronically file such material with, or furnish it to, the Securities and Exchange Commission, or "SEC." You can learn more about us by reviewing our SEC filings on our website by clicking on "Stockholder Relations" located on our home page and proceeding to "Financial Reports." We also make available on our website, under "Corporate Governance," charters for the committees of our board of directors, our Code of Business Conduct and Ethics, our Corporate Governance Guidelines and other company information, including amendments to such documents and waivers, if any to our Code. These documents will also be furnished, free of charge, upon written request to Impac Mortgage Holdings, Inc., Attention: Stockholder Relations, 1401 Dove Street, Newport Beach, California 92660. The SEC also maintains a website at www.sec.gov that contains reports, proxy statements and other information regarding SEC registrants, including the Company.

General Overview

We are a mortgage real estate investment trust, or "REIT," that is a nationwide acquirer, originator, seller and investor of non-conforming Alt-A mortgages, or "Alt-A mortgages," and to a lesser extent, small-balance, multi-family mortgages, or "multi-family mortgages" and sub-prime, or "B/C mortgages." We also provide warehouse and repurchase financing to originators of mortgages.

We operate three core businesses:

- the long-term investment operations that is conducted by IMH, IMH Assets and IMCC;
- the mortgage operations that is conducted by IFC, ISAC and Novelle; and
- the warehouse lending operations that is conducted by IWLG.

The long-term investment operations primarily invest in adjustable rate and fixed rate Alt-A mortgages that are acquired and originated by our mortgage operations. Alt-A mortgages are primarily first lien mortgages made to borrowers whose credit is generally within typical Fannie Mae and Freddie Mac guidelines, but have loan characteristics that make them non-conforming under those guidelines. Some of the principal differences between mortgages purchased by Fannie Mae and Freddie Mac and Alt-A mortgages are as follows:

- credit and income histories of the mortgagor;
- · documentation required for approval of the mortgagor; and
- loan balances in excess of maximum Fannie Mae and Freddie Mac lending limits.

For instance, Alt-A mortgages may not have certain documentation or verifications that are required by Fannie Mae and Freddie Mac and, therefore, in making our credit decisions, we are more reliant upon the borrower's credit score and the adequacy of the underlying collateral. We believe that Alt-A mortgages provide an attractive net earnings profile by producing higher yields without commensurately higher credit losses than other types of mortgages.

The long-term investment operations also originate and invest in multi-family mortgages that are primarily adjustable rate mortgages with initial fixed interest rate periods of two-, three-, five-, seven- and ten-years that subsequently adjust to adjustable rate mortgages, or "hybrid ARMs," with balances that generally range from \$500,000 to \$5.0 million. Multi-family mortgages have interest rate floors, which is the initial start rate, and prepayment penalty periods of three-, five-, seven- and ten-years. Multi-family mortgages provide greater asset diversification on our balance sheet as borrowers of multi-family mortgages typically have higher credit scores and multi-family mortgages typically have lower loan-to-value ratios, or "LTV ratios," and longer average term to payoff than Alt-A mortgages.

The long-term investment operations generate earnings primarily from net interest income earned on mortgages held for long-term investment, or "long-term mortgage portfolio." The long-term mortgage portfolio as reported on our consolidated balance sheet consists of mortgages held as collateralized mortgage obligations, or "CMO," and mortgages held-for-investment. Investments in Alt-A mortgages and multi-family mortgages are initially financed with short-term borrowings under reverse repurchase agreements which are subsequently converted to long-term financing in the form of CMO financing. Cash flow from the long-term mortgage portfolio and proceeds from the sale of capital stock also finance new Alt-A and multi-family mortgages.

The mortgage operations acquire, originate, sell and securitize primarily adjustable rate and fixed rate Alt-A mortgages and, to a lesser extent, B/C mortgages. The mortgage operations generate income by securitizing and selling mortgages to permanent investors, including the long-term investment operations. This business also earns revenue from fees associated with mortgage servicing rights, master servicing agreements and interest income earned on mortgages held for sale. The mortgage operations use warehouse facilities provided by the warehouse lending operations to finance the acquisition and origination of mortgages.

The warehouse lending operations provide short-term financing to mortgage loan originators, including our mortgage operations, by funding mortgages from their closing date until sale to pre-approved investors. This business earns fees from warehouse transactions as well as net interest income from the difference between its cost of borrowings and the interest earned on warehouse advances.

Our goal is to generate consistent and reliable taxable income for distribution as dividends to our stockholders primarily from earnings generated by our core operating businesses. For financial information relating to the long-term investment operations, mortgage operations and warehouse lending operations, please refer to Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations" and our consolidated financial statements beginning on page F-1.

Long-Term Investment Operations

The long-term investment operations invest primarily in Alt-A mortgages and, to a lesser extent, multi-family mortgages and generate revenue primarily from net interest income on its long-term mortgage portfolio. Net interest income represents the difference between income received on mortgages and the corresponding cost of financing. Net interest income also includes (1) amortization of acquisition costs on mortgages acquired from the mortgage operations, (2) accretion of loan discounts, which represents the amount allocated to mortgage servicing rights when they are sold to third parties and mortgages are transferred to the long-term investment operations from the mortgage operations and retained for long-term investment, (3) amortization of CMO securitization expenses and, to a lesser extent, (4) amortization of CMO bond discounts. Net cash payments on derivative instruments are included in gain (loss) on derivative instruments which is a component of non-interest income on our financial statements. For additional information regarding the classification of interest income, interest expense and non-interest income items refer to Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations—Results of Operations and Financial Condition."

The mortgage operations supports the investment objectives of the long-term investment operations by supplying mortgages at prices that are comparable to those available through mortgage bankers and brokers and other third parties. We believe that retaining mortgages acquired and originated by our mortgage operations gives us a competitive advantage because of our historical understanding of the underlying credit of these mortgages and the extensive information on the performance and prepayment patterns of these types of mortgages. We also believe that Alt-A mortgages provide an attractive net earnings profile by producing higher yields without commensurately higher credit risks than other types of mortgages.

Long-Term Mortgage Portfolio

Alt-A mortgages that we retain for long-term investment are primarily adjustable rate mortgages, or "ARMs," hybrid ARMs and, to a lesser extent, fixed rate mortgages, or "FRMs." The interest rate on ARMs are typically tied to an index, such as the six-month London Interbank Offered Rate, or "LIBOR," plus a spread and adjust periodically, subject to lifetime interest rate caps and periodic interest rate and payment caps. The initial interest rates on ARMs are typically lower than average comparable FRMs over the life of the mortgage. Hybrid ARMs are mortgages with maturity periods ranging from 15 to 30 years with initial fixed interest rate periods generally ranging from two to ten years, which subsequently adjust to ARMs. The majority of mortgages retained by the long-term investment operations have prepayment penalty features with prepayment penalty periods ranging from six months to seven years. Prepayment penalties may be assessed to the borrower refinances or, in some cases, sells the home.

During 2004, the long-term investment operations retained \$16.9 billion in principal balance of primarily adjustable rate Alt-A mortgages for long-term investment which were initially acquired and originated by the mortgage operations. In addition, the long-term investment operations originated \$458.5 million of multi-family mortgages. The retention and origination of Alt-A and multi-family mortgages increased the long-term mortgage portfolio to \$21.9 billion at year-end 2004.

The following table presents selected information on mortgages held as CMO collateral, which comprise a substantial portion of the long-term mortgage portfolio, for the periods indicated:

	As of December		
	2004	2003	2002
Percent Alt-A mortgages	99	99	99
Percent ARMs	90	86	85
Percent FRMs	10	14	15
Percent hybrid ARMs	70	48	35
Weighted average coupon	5.62	5.56	6.57
Weighted average margin	3.61	3.10	3.01
Weighted average original LTV ratio	76	79	82
Weighted average original credit score	696	694	683
Percent with prepayment penalty	76	81	76
Prior 12-month prepayment rate	29	28	25
Lifetime prepayment rate	21	21	33
Percent of mortgages in California	62	64	63
Percent purchase transactions	60	57	62
Percent owner occupied	81	87	93
Percent first lien	99	99	99

The following table presents mortgages retained and originated by the long-term investment operations by loan characteristic for the periods indicated (dollars in thousands):

		For the year ended December 31,					
	2004	2004 2003					
	Principal Balance	%	Principal Balance %		Principal Balance	%	
Mortgages by Type:							
Fixed rate first trust deeds	\$ 1,195,200	7	\$ 706,227	12	\$ 599,566	15	
Fixed rate second trust deeds	244,491	1	6,744	0	311	0	
Adjustable rate first trust deeds:	·		•				
LIBOR ARMs (1)	2,754,757	16	1,670,720	27	2,352,863	60	
LIBOR hybrid ARMs (1)	13,173,928	76	3,694,687	61	964,316	25	
Total adjustable rate first trust deeds	15,928,685	92	5,365,407	88	3,317,179	85	
Total mortgages retained	\$17,368,376	100	\$6,078,378	100	\$3,917,056	100	
		_		_		_	
Mortgages by Credit Quality:							
Alt-A mortgages	\$16,846,781	97	\$5,760,779	95	\$3,875,903	99	
Multi-family mortgages	458,532	3	290,527	5	25,799	1	
B/C mortgages (1)	63,063	0	27,072	0	15,354	0	
Total mortgages retained	\$17,368,376	100	\$6,078,378	100	\$3,917,056	100	
		_		_			
Mortgages by Purpose:							
Purchase	\$10,516,622	61	\$3,408,584	56	\$2,353,727	60	
Refinance	6,851,754	39	2,669,794	44	1,563,329	40	
Total mortgages retained	\$17,368,376	100	\$6,078,378	100	\$3,917,056	100	
Mortgages by Prepayment Penalty:							
With prepayment penalty	\$12,657,395	73	\$4,823,027	79	\$3,100,540	79	
Without prepayment penalty	4,710,981	27	1,255,351	21	816,516	21	
Total mortgages retained	\$17,368,376	100	\$6,078,378	100	\$3,917,056	100	

⁽¹⁾ Primarily includes mortgages indexed to one-, three- and six-month LIBOR and one-year LIBOR. Also includes minimal amounts of mortgages indexed to the prime lending rate and constant maturity Treasury index.

For additional information regarding the long-term mortgage portfolio refer to Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations," "Note C—CMO Collateral," and "Note D—Mortgages Held for Investment" in the accompanying notes to the consolidated financial statements.

Financing

We primarily finance our long-term mortgage portfolio as follows:

- issuance of CMO borrowings;
- · short-term borrowings under reverse repurchase agreements, prior to securitization as CMOs; and
- · proceeds from the sale of capital stock.

As we accumulate mortgages we may issue CMOs secured by such mortgages as a means of financing. The decision to issue CMOs is based on our current and future investment needs, market conditions and other factors. Each issue of CMOs is fully payable from the principal and interest payments on the underlying mortgages securing such debt and any cash or other collateral pledged as a condition of receiving the desired rating on the debt. We earn a net interest spread on interest income on mortgages held as CMO collateral less interest and other expenses associated with the acquisition or origination of the loans and with CMO financing. Net interest spreads may be directly impacted by levels of early prepayment of underlying mortgages and, to the extent each CMO class has variable rates of interest, may be affected by changes in short-term interest rates. Our CMOs typically are structured as adjustable rate securities that are indexed to one-month LIBOR and fixed rate securities with interest payable monthly.

When we issue CMOs for financing purposes, we seek an investment grade rating for our CMOs by nationally recognized rating agencies. To secure such ratings, it is often necessary to incorporate certain structural features that provide for credit enhancement. This can include the pledge of collateral in excess of the principal amount of the securities to be issued, a bond guaranty insurance policy for some or all of the issued securities, or additional forms of mortgage insurance. The need for additional collateral or other credit enhancements depends upon factors such as the type of collateral provided, the interest rates paid, the geographic concentration of the mortgaged property securing the collateral and other criteria established by the rating agencies. The pledge of additional collateral reduces our capacity to raise additional funds through short-term secured borrowings or additional CMOs, and diminishes the potential expansion of our long-term mortgage portfolio. As a result, our objective is to pledge additional collateral for CMOs only in the amount required to obtain an investment grade rating by nationally recognized rating agencies. Our total loss exposure is limited to total capital invested in the CMOs at any point in time.

For additional information regarding CMOs refer to Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity" and "Note H—CMO Borrowings" in the accompanying notes to the consolidated financial statements.

Prior to the issuance of CMOs, we use reverse repurchase agreements as short-term financing at interest rates that are consistent with our investment objectives. A reverse repurchase agreement acts as a financing vehicle under which we effectively pledge our mortgages as collateral to secure a short-term loan. Generally, the other party to the agreement makes the loan in an amount equal to a percentage of the market value of the pledged collateral. At maturity of the reverse repurchase agreement, we are required to pay interest and repay the loan and correspondingly in return, we receive our collateral. Our borrowing agreements require us to pledge cash, additional mortgages or additional investment securities backed by mortgages in the event the market value of existing collateral declines. We may be required to sell assets to reduce our borrowings to the extent that cash reserves are insufficient to cover such deficiencies in collateral.

For additional information regarding reverse repurchase agreements refer to Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity" and "Note G—Reverse Repurchase Agreements" in the accompanying notes to the consolidated financial statements.

Interest Rate Risk Management

Our primary objective is to manage exposure to the variability in future cash flows attributable to the variability of one-month LIBOR, which is the underlying index of our adjustable rate CMO borrowings. We also monitor on an ongoing basis the prepayment risks that arise in fluctuating interest rate environments. Our interest rate risk management program is formulated with the intent to mitigate the potential adverse effects of changing interest rates on cash flows on adjustable rate CMO borrowings.

To mitigate our exposure to the effect of changing interest rates on cash flows on our adjustable rate CMO borrowings, we acquire derivatives in the form of interest rate swaps, or "swaps," interest rate cap agreements, or "caps" and interest rate floor agreements, or "floors," collectively, "derivatives." For additional information regarding interest rate risk management activities refer to Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations," Item 7A. "Quantitative and Qualitative Disclosures About Market Risk" and "Note O—Derivative Instruments" in the accompanying notes to the consolidated financial statements.

Mortgage Operations

The mortgage operations acquire, originate, sell and securitize primarily adjustable rate and fixed rate Alt-A mortgages and, to a lesser extent B/C mortgages, from correspondents, mortgage bankers and brokers and retail customers.

Correspondent Acquisition Channel. The mortgage operations acquire adjustable rate and fixed rate Alt-A mortgages from its network of third party correspondents on a flow basis (loan-by-loan) or on a bulk basis (multiple loans) from approved correspondent mortgage companies. Correspondents originate and close mortgages under the mortgage operations' mortgage programs. Correspondents include savings and loan associations, commercial banks and mortgage bankers. The mortgage operations acts as an intermediary between the originators of mortgages that may not meet the guidelines for purchase by Fannie Mae and Freddie Mac and permanent investors in mortgage-backed securities secured by or representing an ownership interest in such mortgages. The mortgage operations also acquires Alt-A mortgages on a bulk basis from approved correspondent sellers that are underwritten to guidelines substantially similar to Alt-A loan programs, but not specific to those of the mortgage operations.

Wholesale and Retail Origination Channel. The mortgage operations market, underwrite, process and fund mortgages for wholesale and, to a lesser extent, retail customers. The wholesale origination channel works directly with mortgage bankers and brokers to originate, underwrite and fund their mortgages. Many wholesale customers cannot conduct business with the mortgage operations as correspondents because they do not have the necessary net worth or financing to close mortgages in their name. Through its retail channel, the mortgage operations markets mortgages directly to the public.

B/C Origination Channel. Novelle originates B/C mortgages through a network of wholesale mortgage brokers and sells its mortgages to third party investors for cash gains.

Marketing Strategy

We believe that we can compete effectively with other Alt-A mortgage conduits through our efficient loan purchasing process, flexible purchase commitment options, competitive pricing and by designing Alt-A mortgages that suit the needs of our correspondents and mortgage bankers and brokers and their borrowers. Our principle strategy is to expand our market position as a low-cost nationwide acquirer and originator of Alt-A mortgages, while continuing to emphasize an efficient centralized operating structure. To help accomplish this, we have developed a second-generation web-based automated underwriting and pricing system called Impac Direct Access System for Lending, or "iDASLg2." iDASLg2 substantially increases efficiencies for our customers and our mortgage operations by significantly decreasing the processing time for a mortgage while improving employee productivity and maintaining superior customer service.

iDASLg2 is an interactive Internet-based system that allows our customers to automatically underwrite mortgages, enabling our customers to pre-qualify borrowers for various mortgage programs and receive automated approval decisions. iDASLg2 is intended to increase efficiencies not only for our customers but also for the mortgage operations by significantly decreasing the processing time for a mortgage. We believe iDASLg2 improves employee production and maintains superior customer service, which together leads to higher closing ratios, improved profit margins and

increased profitability at all levels of our business operations. Most importantly, iDASLg2 allows us to move closer to our correspondents and mortgage bankers and brokers with minimal future capital investment while maintaining centralization, a key factor in the success of our operating strategy. All of our correspondents submit mortgages via iDASLg2 and all wholesale mortgages delivered by mortgage bankers and brokers are directly underwritten through iDASLg2. However, mortgages purchased on a bulk basis from approved correspondent sellers that are not underwritten specifically to our Alt-A mortgage guidelines are not underwritten through iDASLg2.

We also focus on expansion opportunities to attract correspondent originators and mortgage bankers and brokers to our nationwide network in order to increase mortgage acquisitions and originations in a controlled manner. This allows us to shift the high fixed costs of interfacing with the homeowner to our correspondents and mortgage bankers and brokers. This marketing strategy is designed to accomplish the following three objectives:

- attract a geographically diverse group of both large and small correspondent originators and mortgage bankers and brokers;
- establish relationships with correspondents and mortgage bankers and brokers that facilitate their ability to offer a variety of loan products designed by the mortgage operations; and
- purchase mortgages and securitize and sell them in the secondary market or to the long-term investment operations.

In order to accomplish our production objectives, we design and offer mortgage products that we believe are attractive to potential Alt-A borrowers and to end-investors in Alt-A mortgages and mortgage-backed securities. We have historically emphasized and continue to emphasize flexibility in our mortgage product mix as part of our strategy to attract and establish long-term relationships with our correspondents and mortgage bankers and brokers. We also maintain relationships with numerous investors so that we may develop mortgage products that may be of interest to them as market conditions change. In response to the needs of our correspondents, and as part of our strategy to facilitate the sale of our mortgages through the mortgage operations, our marketing strategy offers efficient response time in the purchase process, direct and frequent contact with our correspondents and mortgage bankers and brokers through a trained sales force and flexible commitment programs. Finally, due to the price sensitivity of most homebuyers, we are competitive in pricing our products in order to attract sufficient numbers of mortgages.

Underwriting

We have developed comprehensive purchase guidelines for the acquisition and origination of mortgages. Each mortgage underwritten assesses the borrower's credit score and ability to repay the mortgage obligation and the adequacy of the mortgaged property as collateral for the mortgage. Subject to certain exceptions and the type of mortgage product, each purchased mortgage generally conforms to the loan parameters and eligibility requirements specified in our seller/servicer guide with respect to, among other things, loan amount, type of property, compliance, LTV ratio, mortgage insurance, credit history, debt service-to-income ratio, appraisal and loan documentation.

All mortgages acquired or originated under our loan programs are underwritten either by our employees or by contracted mortgage insurance companies or delegated sellers. Under all of our underwriting methods, loan documentation requirements for verifying the borrower's income and assets vary according to LTV ratios and other factors. Generally, as the standards for required documentation are lowered, the borrowers' down payment requirements are increased and the required LTV ratios are decreased. The borrower is also required to have a stronger credit history, larger cash reserves and an appraisal of the property that may be validated by an enhanced desk or field review, depending on the loan program. Lending decisions are based on a risk analysis assessment after the review of the entire mortgage file. Each mortgage is individually underwritten with emphasis placed on the overall quality of the mortgage.

Seller Eligibility Requirements

Mortgages acquired by the mortgage operations are originated by various sellers, including mortgage bankers, savings and loan associations and commercial banks. Sellers are required to meet certain regulatory, financial, insurance and performance requirements established by us before they are eligible to participate in our mortgage purchase programs. Sellers must also submit to periodic reviews to ensure continued compliance with these

requirements. Our current criteria for seller participation generally includes a minimum tangible net worth requirement of \$500,000, approval as a Fannie Mae or Freddie Mac seller/servicer in good standing, a Housing and Urban Development, or "HUD," approved mortgagee in good standing or a financial institution that is insured by the Federal Deposit Insurance Corporation, or "FDIC," or comparable federal or state agency, and that the seller is examined by a federal or state authority.

In addition, sellers are required to have comprehensive mortgage origination quality control procedures. In connection with its qualification, each seller enters into an agreement that generally provides for recourse by us against the seller in the event of a breach of representations or warranties made by the seller with respect to mortgages sold to us, which includes but is not limited to any fraud or misrepresentation during the mortgage loan origination process or upon early payment default on mortgages.

Mortgage Acquisitions and Originations

Mortgages acquired and originated by the mortgage operations are adjustable rate and fixed rate Alt-A mortgages. A portion of Alt-A mortgages that are acquired and originated by the mortgage operations exceed the maximum principal balance for a conforming loan purchased by Fannie Mae or Freddie Mac, which is currently \$359,650, and are referred to as "jumbo loans." We generally do not acquire or originate Alt-A mortgages with principal balances above \$2.0 million. Alt-A mortgages generally consist of mortgages that are acquired and originated in accordance with underwriting or product guidelines that differ from those applied by Fannie Mae and Freddie Mac. Alt-A mortgages may involve greater risk as a result of different underwriting and product guidelines. Additionally, a portion of mortgages acquired and originated through the mortgage operations are B/C mortgages, which may entail greater credit risks than Alt-A mortgages. B/C mortgages represented 3% and 6% of total acquisitions and originations during 2004 and 2003, respectively.

We generally do not originate B/C mortgages with principal balances above \$650,000. In general, B/C mortgages are residential mortgages made to borrowers with lower credit ratings than borrowers of Alt-A mortgages. B/C mortgages are normally subject to higher rates of loss and delinquency than Alt-A mortgages acquired and originated by the mortgage operations. As a result, B/C mortgages normally bear a higher rate of interest and are typically subject to higher fees than Alt-A mortgages. In general, greater emphasis is placed upon the value of the mortgaged property and, consequently, the quality of appraisals, and less upon the credit history of the borrower in underwriting B/C mortgages than in underwriting Alt-A mortgages. In addition, B/C mortgages are generally subject to lower LTV ratios than Alt-A mortgages.

Mortgages acquired or originated by the mortgage operations are generally secured by first liens and, to a lesser extent, second liens on single-family residential properties with either adjustable rate or fixed rates of interest. FRMs have a constant interest rate over the life of the loan, which is generally 15 or 30 years. The interest rates on ARMs are typically tied to an index, such as six-month LIBOR, plus a spread and adjust periodically, subject to lifetime interest rate caps and periodic interest rate and payment caps. The initial interest rates on ARMs are typically lower than the average comparable FRM but may be higher than average comparable FRMs over the life of the loan.

We acquire and originate mortgages with the following most common loan characteristics, although we may purchase mortgages with other interest rate, prepayment and maturity characteristics:

- FRMs that have original terms to maturity ranging from 15 to 30 years with six-month to five-year prepayment penalty periods;
- ARMs that adjust based on one-, three- and six-month LIBOR and one-year LIBOR with terms to maturity ranging from 15 to 30 years with six-month to five-year prepayment penalty periods;
- two-, three-, five- and seven-year hybrid ARMs with terms to maturity ranging from 15 to 30 years that subsequently adjust to one-, three- and six-month LIBOR and one-year LIBOR with six-month to five-year prepayment penalty periods; and
- adjustable rate and fixed rate interest-only mortgages with 5 to 10 year interest-only periods and terms to maturity of 30 years with six-month to five-year prepayment penalty periods.

The following table presents the mortgage operations' acquisitions and originations by loan characteristic for the periods indicated (in thousands):

	For the year ended December 31,						
	2004	2004 2003			2002		
	Principal Balance	%	Principal Balance	%	Principal Balance	%	
Mortgages by Type:							
Fixed rate first trust deeds	\$ 1,968,502	9	\$3,812,952	40	\$2,159,696	36	
Fixed rate second trust deeds	755,913	3	181,173	2	82,145	2	
Adjustable rate first trust deeds:							
LIBOR ARMs (1)	3,382,978	15	1,611,392	17	2,426,865	41	
LIBOR hybrids (1)	16,105,711	73	3,919,604	41	1,276,792	21	
Total adjustable rate first trust deeds	19,488,689	88	5,530,996	58	3,703,657	62	
Total mortgage acquisitions and originations	\$22,213,104	100	\$9,525,121	100	\$5,945,498	100	
		_		_		_	
Mortgages by Channel:							
Correspondent acquisitions:							
Flow acquisitions	\$10,996,260	50	\$5,399,428	57	\$4,286,905	72	
Bulk acquisitions	8,537,504	38	2,159,116	23	164,636	3	
Total correspondent acquisitions	19,533,764	88	7,558,544	80	4,451,541	75	
Wholesale and retail originations	1,994,569	9	1,468,697	15	1,089,008	18	
B/C originations.	684,771	3	497,880	5	404,949	7	
Total mortgage acquisitions and originations	\$22,213,104	100	\$9,525,121	100	\$5,945,498	100	
Mortgages by Credit Quality:							
Alt-A mortgages	\$21,453,383	97	\$8,988,018	94	\$5,515,573	93	
B/C mortgages (2)	759,721	3	537,103	6	429,925	<i>9</i> 3	
Die mortgages (2)	755,721				423,323		
Total mortgage acquisitions and originations	\$22,213,104	100	\$9,525,121	100	\$5,945,498	100	
Mantee et a la Diagnatia							
Mortgages by Purpose:	¢12.272.040	CO	¢4.000.000	40	¢2.200.500		
Purchase	\$13,373,840	60	\$4,683,202	49	\$3,288,566	55	
Refinance	8,839,264	40	4,841,919	51	2,656,932	45	
Total mortgage acquisitions and originations	\$22,213,104	100	\$9,525,121	100	\$5,945,498	100	
Maria I Day (D. I.		_		_			
Mortgages by Prepayment Penalty:	¢1E 0CE 0E0	70	¢7.105.040	75	¢ 4 C77 O70	70	
With prepayment penalty	\$15,965,959	72	\$7,165,949	75 25	\$4,677,078	79	
Without prepayment penalty	6,247,145	28	2,359,172	25	1,268,420	21	
Total mortgage acquisitions and originations	\$22,213,104	100	\$9,525,121	100	\$5,945,498	100	

⁽¹⁾ Primarily includes mortgages indexed to one-, three- and six-month LIBOR and one-year LIBOR. Also includes minimal amounts of mortgages indexed to the prime lending rate and constant maturity Treasury index.

⁽²⁾ The year ended December 31, 2004, 2003 and 2002 includes \$684.8 million, \$497.9 million and \$404.9 million, respectively, of B/C mortgages originated by Novelle that were subsequently sold to third party investors for cash gains.

Our mortgage acquisition and origination activities focus on those regions of the country where higher volumes of Alt-A mortgages are originated including California, Florida, New York, Colorado, New Jersey, Maryland, Virginia, Illinois, Arizona and Nevada. During the years ended December 31, 2004 and 2003, 69% and 76%, respectively, of mortgage acquisitions and originations were secured by combined California and Florida properties.

Of the \$22.2 billion in principal balance of mortgages acquired and originated in 2004, \$9.1 billion, or 42%, were acquired from our top ten correspondents. No single correspondent, banker or broker accounted for more than 10% of the total mortgages acquired and originated by the mortgage operations in 2004.

Securitization and Sale

After acquiring mortgages from correspondents on a flow or bulk basis and originating mortgages through wholesale and retail channels, the mortgage operations sells and securitizes mortgages to permanent investors. The mortgage operations sells its ARM acquisitions to the long-term investment operations at prices comparable to prices available from third party investors at the date of sale. When a sufficient volume of FRMs with similar characteristics has been accumulated, generally \$100 million to \$350 million, the mortgage operations may (1) sell bulk packages, referred to as whole loan sales, to third party investors, (2) securitize mortgages through the issuance of mortgage-backed securities in the form of real estate mortgage investment conduits, or "REMICs," or (3) sell them to the long-term investment operations.

During 2004, the mortgage operations sold \$16.9 billion in principal balance of mortgages to the long-term investment operations, sold \$1.6 billion in principal balance of mortgages as REMICs. Generally, the mortgage operations sells all of its mortgage acquisitions and originations to third party investors as servicing released, which means that it does not retain primary mortgage servicing rights. However, the mortgage operations does retain rights as master servicer for its securitizations, see "Master Servicing" below.

The period of time between when we commit to purchase mortgages and the time we sell or securitize mortgages generally ranges from 15 to 45 days, depending on certain factors, including the length of the purchase commitment period, volume by product type and the securitization process. REMICs are accounted for as sales transactions. REMIC securities generally consist of one or more classes of "regular interests" and a single class of "residual interest." The regular interests are tailored to the needs of investors and may be issued in multiple classes with varying maturities, average lives and interest rates. REMICs created by us are structured so that one or more of the classes of securities are rated investment grade by at least one nationally recognized rating agency. The ratings for our REMICs are based upon the perceived credit risk by the applicable rating agency of the underlying mortgages, the structure of the securities and the associated level of credit enhancement. Credit enhancement is designed to provide protection to the security holders in the event of borrower defaults and other losses including those associated with fraud or reductions in the principal balances or interest rates on mortgages as required by law or a bankruptcy court.

Master Servicing

We retain master servicing rights on substantially all of our Alt-A and multi-family mortgage acquisitions and originations. Our function as master servicer includes collecting loan payments from loan servicers and remitting loan payments, less master servicing fees receivable and other fees, to a trustee or other purchaser for each series of mortgage-backed securities or loans master serviced. In addition, as master servicer, we monitor compliance with our servicing guidelines and are required to perform, or to contract with a third party to perform, all obligations not adequately performed by any loan servicer. We may also be required to advance funds or we may cause our loan servicers to advance funds to cover interest payments not received from borrowers depending on the status of their mortgages. We also earn income or incur expense on principal and interest payments we receive from our borrowers until those payments are remitted to the investors in those mortgages. Master servicing fees are generally 0.03% per annum on the declining principal balances of the loans serviced. At year-end 2004, we master serviced 120,889 mortgages with a principal balance of \$28.4 billion.

The following table presents the amount of delinquent mortgages in our master servicing portfolio for the periods indicated (dollars in thousands):

		As of December 31,				
	200	2004		3	200	2
	Principal Balance of Mortgages	% of Master Servicing Portfolio	Principal Balance of Mortgages	% of Master Servicing Portfolio	Principal Balance of Mortgages	% of Master Servicing Portfolio
Loans delinquent for:						
60-89 days	\$205,486	0.72%	\$105,455	0.76%	\$100,878	1.16%
90 days and over	87,277	0.31	87,297	0.63	71,466	0.82
Total 60 days and over	292,763	1.03	192,752	1.39	172,344	1.98
Foreclosures pending	258,189	0.91	158,261	1.14	212,309	2.44
Bankruptcies pending	23,807	0.08	19,912	0.14	26,402	0.30
	<u> </u>					
Total	\$574,759	2.02%	\$370,925	2.67%	\$411,055	4.72%

Servicing

We sell or subcontract all of our servicing obligations to independent third parties pursuant to sub-servicing agreements. We believe that the sale of servicing rights or the selection of third-party sub-servicers is more effective than establishing a servicing department within our mortgage operations. However, part of our responsibility is to continually monitor the performance of servicers or sub-servicers through performance reviews and regular site visits. Depending on our reviews, we may in the future rely on our internal default management group to take an ever more active role to assist servicers or sub-servicers in the servicing of our mortgages. Servicing includes collecting and remitting loan payments, making required advances, accounting for principal and interest, holding escrow or impound funds for payment of taxes and insurance, if applicable, making required inspections of the mortgaged property, contacting delinquent borrowers, and supervising foreclosures and property dispositions in the event of un-remedied defaults in accordance with our guidelines. Servicing fees are charged on the declining principal balances of loans services and generally range from 0.25% per annum for FRMs, 0.375% per annum for ARMs, 0.50% per annum for B/C mortgages and 0.75% per annum for properties secured by second liens. To the extent the mortgage operations finances the acquisition of mortgages with warehouse facilities provided by the warehouse lending operations, the mortgage operations pledges mortgages and the related servicing rights to the warehouse lending operations as collateral. As a result, the warehouse lending operations has an absolute right to control the servicing of such mortgages, including the right to collect payments on the underlying mortgages, and to foreclose upon the underlying real property in the case of default. Typically, the warehouse lending operations delegates its right to service the mortgages securing the warehouse line to the mortgage operations. The following table presents inf

	As of December 31,			
	2004	2003	2002	
Beginning servicing portfolio	\$ 1,402.1	\$ 2,653.4	\$ 1,754.4	
Add: Loan acquisitions and originations	22,213.1	9,525.1	5,945.5	
Less: Servicing transferred and principal repayment (1)	(21,924.4)	(10,776.4)	(5,046.5)	
Ending servicing portfolio	\$ 1,690.8	\$ 1,402.1	\$ 2,653.4	
Number of loans serviced	9,256	6,695	15,987	
Average loan size	\$ 183,000	\$ 209,000	\$ 166,000	
Weighted average coupon	6.62%	6.28%	7.26%	

⁽¹⁾ Includes the sale of mortgages on a servicing released basis, the sale of servicing rights on mortgages owned and scheduled and unscheduled principal repayments.

Interest Rate Risk Management

The mortgage operations manages interest rate risk and price volatility on its pipeline of rate-locked mortgage loans, or "mortgage pipeline," during the time it commits to acquire or originate mortgages at a pre-determined rate and the time it sells the mortgage loans. To mitigate interest rate and price volatility risks, the mortgage operations may enter into derivatives. The nature and quantity of derivatives are determined based on various factors, including market conditions and the expected volume of mortgage acquisitions and originations. For additional information regarding interest rate risk management activities refer to Item 7A. "Quantitative and Qualitative Disclosures About Market Risk" and "Note O—Derivative Instruments" in the accompanying notes to the consolidated financial statements.

Warehouse Lending Operations

The warehouse lending operations provide warehouse financing to affiliated companies and reverse repurchase financing to approved non-affiliated mortgage bankers, or "non-affiliated clients," some of which are correspondents of the mortgage operations, to finance mortgages during the time from the closing of the mortgages to sale or other settlement with pre-approved investors. The warehouse lending operations relies mainly on the sale or liquidation of the mortgages as a source of repayment. Any claim of the warehouse lending operations as a secured lender in a bankruptcy proceeding may be subject to adjustment and delay. Borrowings under warehouse facilities are presented on our balance sheet as finance receivables. Terms of non-affiliated clients' warehouse lines, including the commitment amount, are determined based upon the financial strength, historical performance and other qualifications of the borrower. As of December 31, 2004, the warehouse lending operations had approved warehouse lines to non-affiliated clients of \$738.7 million, of which \$471.8 million was outstanding, as compared to \$1.0 billion and \$630.0 million, respectively, as of December 31, 2003.

Regulation

We establish underwriting guidelines that include provisions for inspections and appraisals, require credit reports on prospective borrowers and determine maximum loan amounts. Our mortgage acquisition and origination activities are subject to, among other laws, the Equal Credit Opportunity Act, Federal Truth-in-Lending Act, Fair Credit Reporting Act, Fair and Accurate Credit Transaction Act, Fair Housing Act, Gramm-Leach-Bliley Act, Telephone Consumer Protection Act, Can Spam Act, Real Estate Settlement Procedures Act and Home Mortgage Disclosure Act and the regulations promulgated thereunder. These laws and regulations, among other things, prohibit discrimination and require the disclosure of certain basic information to mortgagors concerning credit terms and settlement costs, prohibit the payment of kickbacks for the referral of business incident to a real estate settlement service, limit payment for settlement services to the reasonable value of the services rendered and goods furnished, restrict the marketing practices we may use to find customers, require us to safeguard non-public information about our customers and require the maintenance and disclosure of information regarding the disposition of mortgage applications based on race, gender, geographical distribution, price and income level. Our mortgage acquisition and origination activities are also subject to state and local laws and regulations, including state licensing laws, anti-predatory lending laws, and may also be subject to applicable state usury statutes. IFC is an approved Fannie Mae seller/servicer and is an approved servicer of Freddie Mac. In addition, IFC is required annually to submit to Fannie Mae and Freddie Mac audited financial statements, or the equivalent, according to the financial reporting requirements of each regulatory entity for its sellers/servicers. IFC's affairs are also subject to examination by Fannie Mae and Freddie Mac at any time to assure compliance with applicable regulations, policies and procedures.

Competition

In acquiring and originating Alt-A mortgages and issuing securities backed by such loans, we compete with other established mortgage conduit programs, investment banking firms, savings and loan associations, banks, thrift and loan associations, finance companies, mortgage bankers and brokers, insurance companies, other lenders and other entities purchasing mortgage assets. As the Federal Reserve continues to raise interest rates at a measured pace and the number of mortgage refinance opportunities diminish, the mortgage industry may experience a consolidation that may reduce the number of current correspondents and independent mortgage bankers and brokers available to the mortgage operations, reducing our potential customer base and resulting in the mortgage operations acquiring and originating a larger percentage of mortgages from a smaller number of customers. In addition, until the consolidation occurs in the mortgage industry, price competition among competitors can affect the profitability on the sale of mortgage loans or the return on investments as mortgage lenders are willing to cut their profitability margins to maintain current production levels. Changes of this nature could negatively impact our businesses.

Mortgage-backed securities issued by the mortgage operations and the long-term investment operations face competition from other investment opportunities available to prospective investors. We face competition in our mortgage operations and warehouse lending operations from other financial institutions, including but not limited to banks and investment banks. Our main competitors include Countrywide Home Loans, IndyMac Bancorp, Inc., Greenpoint Financial Corporation, Residential Funding Corporation, Aurora Loan Services, Inc., Credit Suisse First Boston Corporation and Bear Stearns and Company, Inc.

Competition can take place on various levels, including convenience in obtaining a mortgage, service, marketing, origination channels and pricing. We depend primarily on correspondents and independent mortgage bankers and brokers for the acquisition and origination of mortgages. These independent mortgage bankers and brokers deal with multiple lenders for each prospective borrower. We compete with these lenders for the independent bankers and brokers' business on the basis of price, service, loan fees, costs and other factors. Our competitors also seek to establish relationships with such bankers and brokers, who are not obligated by contract or otherwise to do business with us. Many of the institutions with which we compete in our mortgage operations and warehouse lending operations have significantly greater financial resources than we have. However, we can compete effectively with other Alt-A mortgage conduits through our efficient loan purchasing process, flexible purchase commitment options and competitive pricing and by designing Alt-A mortgage programs that suit the needs of our correspondents and their borrowers, which is intended to provide sufficient credit quality to our investors.

Risk factors, as outlined below, provide additional information related to risks associated with competition in the mortgage banking industry.

Employees

As of December 31, 2004, we had a total of 752 full- and part-time employees and temporary and contract employees. Management believes that relations with its employees are good. We are not a party to any collective bargaining agreements.

Revisions in Policies and Strategies

Our board of directors has approved our investment and operating policies and strategies. Our core operations involve the acquisition and origination of mortgages and their subsequent securitization and sale. We also act as a warehouse lender providing financing facilities to mortgage originators. These operations and their associated policies and strategies, are further described herein. Our board of directors has delegated asset/liability management to the Asset/Liability Committee, or "ALCO," which reports to the board of directors at least quarterly. See a further discussion of ALCO in Item 7. "Management's Discussion of Financial Condition and Results of Operations" and Item 7A. "Quantitative and Qualitative Disclosures About Market Risk." Any of our policies, strategies and activities may be modified or waived by our board of directors without stockholder consent. Developments in the market, which affect the policies and strategies mentioned herein or which change our assessment of the market, may cause our board of directors to revise our policies and financing strategies.

We have elected to qualify as a REIT for tax purposes. We have adopted certain compliance guidelines, which include restrictions on the acquisition, holding and sale of assets. Prior to the acquisition of any asset, we determine whether the asset meets REIT requirements. Substantially all of the assets that we have acquired and will acquire for investment are expected to qualify as REIT assets. This REIT requirement limits our investment strategies.

The long-term investment operations primarily invest in Alt-A and multi-family mortgages and, to a lesser extent, B/C mortgages. The long-term investment operation does not limit the proportion of its assets that may be invested in each type of mortgage.

We closely monitor our acquisition and investment in mortgage assets and the sources of our income, including income or expense from interest rate risk management strategies, to ensure at all times that we maintain our qualifications as a REIT. We have developed certain accounting systems and testing procedures to facilitate our ongoing compliance with the REIT provisions of the Internal Revenue Code. No changes in our investment policies and operating strategies, including credit criteria for mortgage asset investments, may be made without the approval of our board of directors.

We may at times and on terms that our board of directors deems appropriate:

- Issue senior securities In May 2004, we issued 2,000,000 shares of our 9.375% Series B Cumulative Redeemable Preferred Stock, par value \$0.01 per share, liquidation preference \$25.00 per share. In November and December 2004, we issued an aggregate of 4,300,000 shares of our 9.125% Series C Cumulative Redeemable Preferred Stock, par value \$0.01 per share, liquidation preference \$25.00 per share;
- Borrow money We finance our operations in large part through the issuance of CMOs and short-term borrowings under reverse repurchase agreements;
- Make loans to other persons The warehouse lending operations provide reverse repurchase financing to affiliated companies and to approved nonaffiliated clients, some of which are correspondents of the mortgage operations, to finance mortgages during the time from the closing of the
 mortgages to their sale or other settlement with pre-approved investors;
- Engage in the purchase and sale of investments In connection with the issuance of mortgage-backed securities by our mortgage operations in the form of REMICs, our long-term investment operations may retain senior or subordinated securities on a short- or long-term basis;
- Repurchase or otherwise reacquire our shares or other securities in the future During 2000, we adopted a repurchase plan to repurchase up to \$3.0 million of our common stock in the open market. During 2001 and 2000, we repurchased 1,015,950 shares for \$2.3 million. All repurchased shares were cancelled. During 2002 and 2003, we did not repurchase any shares of common stock. In February of 2004, the share repurchase program was cancelled by our board of directors.

We may also offer securities in exchange of property, invest in securities of other issuers for the purpose of exercising control and underwrite the securities of other issuers, although we have not done so in the past three years and have no present intention to do so. Historically, we have and intend to continue to distribute annual reports to our stockholders, including financial statements audited by independent auditors, describing our current business and strategy.

Risk Factors

Some of the following risk factors relate to a discussion of our assets. For additional information on our asset categories refer to Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations," "Note B—Mortgages Held-for-Sale," "Note C— CMO Collateral," "Note D—Mortgages Held-for-Investment," and "Note E—Finance Receivables" and in the accompanying notes to the consolidated financial statements.

Risks Related To Our Businesses

A prolonged economic downturn or recession would likely result in a reduction of our mortgage origination activity which could adversely affect our financial results.

The United States economy has undergone in the past and may in the future, undergo, a period of economic slowdown, which some observers view as a recession. An economic downturn or a recession may have a significant adverse impact on our operations and our financial condition. For example, a reduction in new mortgages may adversely affect our ability to maintain or expand our long-term mortgage portfolio, our principal means of generating earnings. In addition, a decline in new mortgage activity may likely result in reduced activity for our warehouse lending operations and our long-term investment operations. In the case of our mortgage operations, a decline in mortgage activity may result in fewer loans that meet its criteria for purchase and securitization or sale, thus resulting in a reduction in interest income and fees and gain on sale of loans. We may also experience larger than previously reported losses on our long-term mortgage portfolio due to a higher level of defaults or foreclosures or higher loss rates on our mortgages.

If we are unable to generate sufficient liquidity we may be unable to conduct our operations as planned.

If we cannot generate sufficient liquidity, we may be unable to continue to grow our operations, grow our asset base, maintain our current interest rate risk management policies and pay dividends. We have traditionally derived our liquidity from the following primary sources:

- financing facilities provided to us by others to acquire or originate mortgage assets;
- whole loan sales and securitizations of acquired or originated mortgages;
- · our issuance of equity and debt securities;
- excess cash flow from our long-term mortgage portfolio; and
- · earnings from operations.

We cannot assure you that any of these alternatives will be available to us, or if available, that we will be able to negotiate favorable terms. Our ability to meet our long-term liquidity requirements is subject to the renewal of our credit and repurchase facilities and/or obtaining other sources of financing, including additional debt or equity from time to time. Any decision by our lenders and/or investors to make additional funds available to us in the future will depend upon a number of factors, such as our compliance with the terms of our existing credit arrangements, our financial performance, industry and market trends in our various businesses, the lenders' and/or investors' own resources and policies concerning loans and investments, and the relative attractiveness of alternative investment or lending opportunities. If we cannot raise cash by selling debt or equity securities, we may be forced to sell our assets at unfavorable prices or discontinue various business activities. Our inability to access the capital markets could have a negative impact on our growth of taxable income and also our ability to pay dividends.

Any significant margin calls under our financing facilities would adversely affect our liquidity and may adversely affect our financial results.

Prior to the fourth quarter of 1998, we generally had no difficulty in obtaining favorable financing facilities or in selling acquired mortgages. However, during the fourth quarter of 1998, the mortgage industry experienced substantial turmoil as a result of a lack of liquidity in the secondary markets. At that time, investors expressed unwillingness to purchase interests in securitizations due, in part, to:

- the lack of financing to acquire these securitization interests;
- · the widening of returns expected by institutional investors on securitization interests over the prevailing Treasury rate; and
- market uncertainty.

As a result, during this period many mortgage originators, including us, were unable to access the securitization market on favorable terms. This resulted in some companies declaring bankruptcy. Some companies, like us, were required to sell loans on a whole loan basis and liquidate holdings of mortgage-backed securities to repay short-term borrowings. However, the large amount of mortgages available for sale on a whole loan basis affected the pricing offered for these mortgages, which in turn reduced the value of the collateral underlying the financing facilities. Therefore, many providers of financing facilities initiated margin calls. Margin calls resulted when our lenders evaluated the market value of the collateral securing our financing facilities and required us to provide them with additional equity or collateral to secure our borrowings.

Our financing facilities were short-term borrowings and due to the turmoil in the mortgage industry during the latter part of 1998 many traditional providers of financing facilities were unwilling to provide facilities on favorable terms, or at all. Our current financing facilities continue to be short-term borrowings and we expect this to continue. If we cannot renew or replace maturing borrowings, we may have to sell, on a whole loan basis, the loans securing these facilities, which, depending upon market conditions may result in substantial losses.

We face risks related to our recent accounting restatements.

On July 22, 2004, we publicly announced that we had discovered accounting inaccuracies in previously reported financial statements. As a result, following consultation with our auditors, we decided to restate our financial statements for the three months ended March 31, 2004 and 2003, the three and six months ended June 30, 2003, the three and nine months ended September 30, 2003 and for each of the years ended December 31, 2003, 2002 and 2001. The restatements relate to a correction to our revenue recognition policy with respect to the cash sales of mortgage servicing rights to unrelated third parties when the mortgage loans are retained, our accounting for derivatives and interest rate risk management activities, the accounting for loan purchase commitments as derivatives and selected elimination entries to consolidate IFC with that of IMH. We also corrected a clerical error in the calculation of earnings per share for the six months ended June 30, 2004. The effect of this restatement on net earnings (loss) and the correction of basic and diluted earnings per share were as follows:

Net Effect of Restatement on Net Earnings (Loss) (in millions)

	Net Ea	nge in arnings oss)
For the three months ended March 31, 2004	\$	(36.7)
For the year ended December 31, 2003		21.7
For the nine months ended September 30, 2003		12.8
For the three months ended September 30, 2003		11.2
For the six months ended June 30, 2003		1.7
For the three months ended June 30, 2003		2.4
For the three months ended March 31, 2003		(0.7)
For the year ended December 31, 2002		(34.6)
For the year ended December 31, 2001		(35.4)

Net Effect of Correction on Earnings per Share

ret zintet of Gott tetam on zurimige per Gime		
	Char	nge in
	Basic EPS	Diluted EPS
For the six months ended June 30, 2004	\$0.14	\$ 0.13

The restatement of our financial statements could lead to litigation claims and/or regulatory proceedings against us. The defense of any such claims or proceedings may cause the diversion of management's attention and resources, and we may be required to pay damages if any such claims or proceedings are not resolved in our favor. Any litigation or regulatory proceeding, even if resolved in our favor, could cause us to incur significant legal and other expenses. We also may have difficulty raising equity capital or obtaining other financing, such as lines of credit or otherwise. We may not be able to effectuate our current operating strategy, including the ability to originate, acquire or securitize mortgage loans for retention or sale at projected levels. We may be subject to resignation of our current external auditors which may, among other things, cause a delay in the preparation of future financial statements and increase expenditures related to the retention of new external auditors and the lead time required to become familiar with our operations. The process of retaining new external auditors may limit our access to the capital markets for an extended period of time. Moreover, we may be the subject of negative publicity focusing on the financial statement inaccuracies and resulting restatement and negative reactions from our stockholders, creditors or others with which we do business. The occurrence of any of the foregoing could harm our business and reputation and cause the price of our securities to decline, and could result in a delisting of our securities from the New York Stock Exchange.

Since we are reporting in this annual report that our system of internal control over financial reporting and disclosure controls and procedures are ineffective, we may not be able to accurately report our financial results or prevent fraud, which could adversely affect the trading price of our securities or our ability to raise capital.

Effective internal control over financial reporting and disclosure controls and procedures are necessary for us to provide reliable financial reports and effectively prevent fraud and operate successfully as a public company. If we cannot provide reliable financial reports or prevent fraud, our reputation and operating results would be harmed. As a result, current and potential stockholders could lose confidence in our financial reporting which would harm our business and the trading price of our securities. We are reporting in this annual report, and may discover in the future, areas of our disclosure controls and procedures and internal control over financial reporting that need improvement. We have identified material weaknesses in some of our internal controls which we believe required remediation.

Furthermore, in connection with the restatement of our consolidated financial statements, we noted certain matters involving internal controls and operations that we considered to be a material weakness, as defined by the Public Company Accounting Oversight Board, or "PCAOB." We needed to improve the evaluation and documentation of accounting policies and procedures for complex transactions, such as transfers of financial assets, derivatives and hedge accounting and allowance for credit losses. We did not have a sufficient amount or type of staff in the financial reporting and accounting departments. Furthermore, we also noted significant deficiencies in that our internal audit function did not provide an adequate or effective monitoring of our controls and we needed to evaluate whether we have appropriate internal resources to manage and monitor work performed by our outsourced tax compliance function.

We cannot be certain that our efforts to improve our internal control over financial reporting and disclosure controls and procedures will be successful or that we will be able to maintain adequate controls over our financial processes and reporting in the future. Any failure to develop or maintain effective controls or difficulties encountered in their implementation or other effective improvement of our internal control over financial reporting and disclosure controls and procedures could harm our operating results or cause us to fail to meet our reporting obligations. If we are unable to adequately establish or improve our internal control over financial reporting, our external auditors will not be able to issue an unqualified opinion and will issue an adverse opinion on the effectiveness of our internal control over financial reporting. Ineffective internal control over financial reporting and disclosure controls and procedures and any material weakness in our internal control over financial reporting could also cause investors to lose confidence in our reported financial information, which would likely have a negative effect on the trading price of our securities or could affect our ability to access the capital markets and which could result in regulatory proceedings against us by, among others, the SEC.

In addition, a material weakness in internal control over financial reporting, which may lead to deficiencies in the preparation of financial statements, could lead to litigation claims and/or regulatory proceedings against us. The defense of any such claims or proceedings may cause the diversion of management's attention and resources, and we may be required to pay damages if any such claims or proceedings are not resolved in our favor. Any litigation or regulatory proceeding, even if resolved in our favor, could cause us to incur significant legal and other expenses. Such events could harm our business, affect our ability to raise capital and adversely affect the trading price of our securities.

Due in large part to delays in management's assessment of our internal control over financial reporting, our independent registered public accountant firm, KPMG, has not completed its audit of our internal control over financial reporting. We expect to receive an adverse opinion on our internal control over financial reporting. We cannot assure you that the audit will not identify any additional material weaknesses in our internal control over financial reporting or that we will not receive an adverse opinion on management's assessment of the effectiveness of our internal control over financial reporting.

Our failure to include audited financial statements and our reports on our internal control over financial reporting in this annual report may subject us to regulatory or litigation proceedings, may adversely impact our ability to obtain financing or raise capital and could adversely affect the trading price of our securities.

We are required under the Exchange Act, and the regulations promulgated thereunder, to include audited financial statements and an audit report of a registered public accounting firm on such financial statements in this annual report on Form 10-K. As a result of our failure to file in this annual report, audited financial statements, management's assessment of the effectiveness of internal control over financial reporting and the audit report thereon and related certifications, we may be subject to regulatory action by, among others, the SEC. Our failure to file these required items will cause us to lose our eligibility to use a registration statement on Form S-3 until that date which is twelve months from the date we amend this annual report on Form 10-K to include those audited financial statements. Such loss of eligibility would adversely affect our ability to generate liquidity through the public issuance of our equity and debt securities on a rapid basis, which could have a negative impact on our business activities, although we may be able to raise capital through other means. We contacted the SEC regarding our S-3 eligibility and are seeking relief from the SEC in this regard; however, we may not be successful in obtaining such relief.

We are also subject to the rules and requirements of the NYSE pertaining to annual reports of listed companies, including the requirement to provide our stockholders with audited financial statements within 120 days from the end of the prior fiscal year. Our failure to satisfy these requirements on a timely basis means that we will receive a late filer status from the NYSE and will have nine months to cure the deficient 10-K. If we are unable to comply in a timely manner we will be subject to actions by the NYSE, including the suspension and delisting of our securities, which would adversely affect the trading price of our securities and the ability of our investors to sell their shares. We have informed the NYSE regarding our failure to satisfy our annual report requirements.

Under our agreements relating to our repurchase facilities, we are required to provide our lenders with audited financial statements within 90 to 120 days from the end of the prior fiscal year. Any failure to furnish audited financial statements, absent a waiver, would cause us to be in default or breach under these agreements. While we have received waivers of any defaults resulting from our current failure to provide audited financial statements within the required time periods under these repurchase agreements, if we are unable to provide audited financial statements by the extended deadlines or if we fail to obtain any further waivers from our lenders, then we will be in default under those applicable agreements and repayments of our borrowings may be immediately accelerated by the lenders and the agreements may be terminated. As a result, we may need to seek alternate sources of financing, which may not be available on acceptable terms or at all, and we may have to sell, on a whole loan basis, the loans securing these facilities, which, depending upon market conditions may result in losses.

Our failure to include audited financial statements in this annual report or our auditor's attestation regarding our internal control over financial reporting may deter potential creditors from providing loans and financing facilities to us. Additionally, potential investors may be deterred from investing in our securities through private offering transactions. Such unwillingness to provide us with additional financing or to invest in our securities would adversely affect our ability to raise capital and generate liquidity. Current and potential stockholders could lose confidence in our financial reporting and operations as a result of our failure to include audited financial statements in this annual report, any loss of our Form S-3 eligibility, and any suspension or delisting of our securities by the NYSE. Such loss of confidence may adversely affect the trading price of our securities and our ability to access the capital markets for additional funding.

We cannot assure you that our auditors will be able to complete their assessment of our internal controls on a timely basis or that our auditors will not find additional material weaknesses in our internal controls. Furthermore, the content and information contained in the unaudited financial statements included in this report may not be identical to the content and information contained in the audited financial statements and any changes or differences in such financial statements could be material. Therefore, our failure to include our audited financial statements and the auditor's report on our internal controls could lead to litigation claims and/or regulatory proceedings under the federal securities laws and/or state laws relating to material misstatements and/or omissions, fraud and other causes of action. The defense of any such claims or proceedings may cause the diversion of management's attention and resources, and we may be required to pay damages if any such claims or proceedings are not resolved in our favor. Any litigation or regulatory proceeding, even if resolved in our favor, could cause us to incur significant legal or other expenses.

We incurred net losses in accordance with GAAP for fiscal years 1997, 1998, 2000 and 2001 and may incur losses in the future.

During the years ended December 31, 2001 and 2000, we experienced a net loss of \$2.2 million and \$54.5 million. The 2001 loss was related to a loss on derivatives and the 2000 loss was the result of write-downs of non-performing investment securities secured by mortgages and additional increases in the provision for loan losses to provide for the deterioration of the performance of collateral supporting specific investment securities for 2000. During the year ended December 31, 1998, we experienced a net loss of \$5.9 million primarily as the mortgage industry experienced substantial turmoil as a result of a lack of liquidity in the secondary markets, which caused us to sell mortgages at losses to meet margin calls on our financing facilities. During the year ended December 31, 1997, we experienced a net loss of \$16.0 million. The net loss incurred during 1997 included an accounting charge of \$44.4 million that was the result of expenses related to the termination and buyout of our management agreement with Imperial Credit Advisors, Inc. We cannot be certain that revenues will remain at current levels or improve or that we will generate net earnings in the future, which could prevent us from effectuating our business strategy.

If we are unable to complete securitizations or if we experience delayed mortgage loan sales or securitization closings, we could face a liquidity shortage which would adversely affect our operating results.

We rely significantly upon securitizations to generate cash proceeds to repay borrowings and replenish our borrowing capacity. If there is a delay in a securitization closing or any reduction in our ability to complete securitizations we may be required to utilize other sources of financing, which, if available at all, may not be on similar terms. In addition, delays in closing mortgage sales or securitizations of our mortgages increase our risk by exposing us to credit and interest rate risks for this extended period of time. Furthermore, gains on sales from certain of our securitizations represent a significant portion of the taxable income dividend up from our taxable REIT subsidiary, IFC. Several factors could affect our ability to complete securitizations of our mortgages, including:

- · conditions in the securities and secondary markets;
- credit quality of the mortgages acquired or originated through our mortgage operations;
- volume of our mortgage loan acquisitions and originations;
- our ability to obtain credit enhancements; and
- lack of investors purchasing higher risk components of the securities.

If we are unable to sell a sufficient number of mortgages at a premium or profitably securitize a significant number of our mortgages in a particular financial reporting period, then we could experience lower net earnings or a loss for that period, which could have a material adverse affect on our operations. We cannot assure you that we will be able to continue to profitably securitize or sell our loans on a whole loan basis, or at all.

The market for first loss risk securities, which are securities that take the first loss when mortgages are not paid by the borrowers, is generally limited. In connection with our REMIC securitizations, we endeavor to sell all securities subjecting us to a first loss risk. If we cannot sell these securities, we may be required to hold them for an extended period, subjecting us to a first loss risk.

Our borrowings and use of substantial leverage may cause losses.

Our use of CMOs may expose our operations to credit losses.

To grow our long-term mortgage portfolio, we borrow a substantial portion of the market value of substantially all of our investments in mortgages in the form of CMOs. There are no limitations on the amount of CMO borrowings we may incur, other than the aggregate value of the underlying mortgages. We currently use CMOs as financing vehicles to increase our leverage since mortgages held for CMO collateral are retained for investment.

Retaining mortgages as collateral for CMOs exposes our operations to greater credit losses than does the use of other securitization techniques that are treated as sales because as the equity holder in the security, we are allocated losses from the liquidation of defaulted loans first prior to any other security holder. Although our liability under a collateralized mortgage obligation is limited to the collateral used to create the collateralized mortgage obligation, we generally are required to make a cash equity investment to fund collateral in excess of the amount of the securities issued in order to obtain the appropriate credit ratings for the securities being sold, and therefore obtain the lowest interest rate available, on the CMOs. If we experience greater credit losses than expected on the pool of loans subject to the CMO, the value of our equity investment will decrease and we may have to increase the allowance for loan losses on our financial statements.

If we default under our financing facilities, we may be forced to liquidate collateral.

If we default under our financing facilities, our lenders could force us to liquidate the collateral. If the value of the collateral is less than the amount borrowed, we could be required to pay the difference in cash. Furthermore, if we default under one facility, it would generally cause a default under our other facilities. If we were to declare bankruptcy, some of our reverse repurchase agreements may obtain special treatment and our creditors would then be allowed to liquidate the collateral without any delay. On the other hand, if a lender with whom we have a reverse repurchase

agreement declares bankruptcy, we might experience difficulty repurchasing our collateral, or enforcing our claim for damages, and it is possible that our claim could be repudiated and we could be treated as an unsecured creditor. If this occurs, our claims would be subject to significant delay and we may receive substantially less than our actual damages or nothing at all.

If we are forced to liquidate, we may have few unpledged assets for distribution to unsecured creditors.

We have pledged a substantial portion of our assets to secure the repayment of CMO borrowings issued in securitizations and our financing facilities. We will also pledge substantially all of our current and future mortgages to secure borrowings pending their securitization or sale. The cash flows we receive from our investments that have not yet been distributed or pledged or used to acquire mortgages or other investments may be the only unpledged assets available to our unsecured creditors if we were liquidated.

Interest rate fluctuations may adversely affect our operating results.

Our operations, as a mortgage loan acquirer and originator, an investor in mortgage loans or a warehouse lender, may be adversely affected by rising and falling interest rates. Interest rates have been low over the past few years; however increases in interest rates may discourage potential borrowers from refinancing mortgages, borrowing to purchase homes or seeking second mortgages. This may decrease the amount of mortgages available to be acquired or originated by our mortgage operations and decrease the demand for warehouse financing provided by our warehouse lending operations, which could adversely affect our operating results. If short-term interest rates exceed long-term interest rates, there is a higher risk of increased loan prepayments, as borrowers may seek to refinance their fixed and adjustable rate mortgages at lower long-term fixed interest rates. Increased loan prepayments could lead to a reduction in the number of loans in our long-term mortgage portfolio and reduce our net interest income. Rising interest rates may also increase delinquencies, foreclosures and losses on our adjustable rate mortgages.

We are subject to the risk of rising mortgage interest rates between the time we commit to purchase mortgages at a fixed price through the issuance of individual, bulk or other rate-locks and the time we sell or securitize those mortgages. An increase in interest rates will generally result in a decrease in the market value of mortgages that we have committed to purchase at a fixed price, but have not been sold or securitized. As a result, we may record a smaller gain, or even a loss, upon the sale or securitization of those mortgages.

We may experience reduced net earnings or losses if our liabilities reprice at different rates than our assets.

Our principal source of revenue is net interest income or net interest spread from our long-term mortgage portfolio, which is the difference between the interest we earn on our interest earning assets and the interest we pay on our interest bearing liabilities. The rates we pay on our borrowings are independent of the rates we earn on our assets and may be subject to more frequent periodic rate adjustments. Therefore, we could experience a decrease in net earnings or a loss because the interest rates on our borrowings could increase faster than the interest rates on our assets, if the increased borrowing costs are not offset by reduced cash payments on derivatives recorded in other non-interest income. If our net interest spread becomes negative, we will be paying more interest on our borrowings than we will be earning on our assets and we will be exposed to a risk of loss.

Additionally, the rates paid on our borrowings and the rates received on our assets may be based upon different indices. Our long-term mortgage portfolio includes mortgages that are one-, three- and six-month LIBOR and one-year LIBOR hybrid ARMs. These are mortgages with fixed interest rates for an initial period of time, after which they begin bearing interest based upon short-term interest rate indices and adjust periodically. We generally fund mortgages with adjustable interest rate borrowings having interest rates that are indexed to short-term interest rates, typically one-month LIBOR, and adjust periodically at various intervals. To the extent that there is an increase in the interest rate index used to determine our adjustable interest rate borrowings and it increases faster than the indices used to determine the rates on our assets (i.e., the increase is not offset by a corresponding increase in the rates at which interest accrues on our assets) or by various cash payments on interest rate derivatives that we have in place at any given time, our net earnings will decrease or result in losses.

ARMs typically have interest rate caps, which limit interest rates charged to the borrower during any given period. Our borrowings are not subject to similar restrictions. As a result, in a period of rapidly increasing interest rates, the

interest rates we pay on our borrowings could increase without limitation, while the interest rates we earn on our ARMs would be capped. If this occurs, our net interest spread could be significantly reduced or we could suffer a net interest loss if not offset by a decrease in the cash payments on interest rate derivatives that we have in place at any given time.

Our operating results will be affected by the results of our interest rate risk management activities.

To mitigate risks associated with our mortgage and long-term investment operations, we enter into transactions designed to limit our exposure to interest rate risks. To mitigate the risks associated with adjustable rate borrowings, we attempt to match the interest rate sensitivities of our ARMs with the associated financing liabilities. Management determines the nature and quantity of derivative transactions based on various factors, including market conditions and the expected volume of mortgage acquisitions. While we believe that we properly manage our interest rate risk on an economic and tax basis, we have elected not to achieve hedge accounting, as established by the Financial Accounting Standards Board, or FASB," under the provisions of Statement of Financial Accounting Standards No. 133, or "SFAS 133," for our interest rate risk management activities in our financial statements. The effect of not applying hedge accounting means that our interest rate risk management activities may result in significant volatility in our quarterly net earnings as interest rates go up or down. It is possible that there will be periods during which we will incur losses on derivative transactions that may result in net losses, as was the case in 2001 after the restatement of our consolidated financial statements. In addition, if the counter parties to our derivative transactions are unable to perform according to the terms of the contracts, we may incur losses. While we believe we prudently manage interest rate risk, our derivative transactions may not offset the risk of adverse changes in our net interest margins.

Increased levels of early prepayments of mortgages may accelerate our expenses and decrease our net income.

Mortgage prepayments generally increase on our ARMs when fixed mortgage interest rates fall below the then-current interest rates on outstanding ARMs. Prepayments on mortgages are also affected by the terms and credit grades of the mortgages, conditions in the financial markets, housing appreciation and general economic conditions. If we acquire mortgages at a premium and they are subsequently repaid, we must expense the unamortized premium at the time of the prepayment. We could possibly lose the opportunity to earn interest at a higher rate over the expected life of the mortgage. Also, if prepayments on mortgages increase when interest rates are declining, our net interest income may decrease if we cannot reinvest the prepayments in mortgage assets bearing comparable net interest margins. If prepayment rates differ from our projections, we may experience a change in net earnings due to a change in the ratio of derivatives to loans being interest rate risk managed. This may result in a reduction of cash flows from our mortgage loans net of financing costs as we have a higher percentage of derivatives costs related to these loans than originally projected.

We generally acquire mortgages on a servicing released basis, meaning we acquire both the mortgages and the rights to service them. This strategy requires us to pay a higher purchase price or premium for the mortgages. If the mortgages that we acquire at a premium prepay faster than originally projected, generally accepted accounting principles, or "GAAP," require us to write down the remaining capitalized premium amounts at a faster speed than was originally projected, which would decrease our current net interest income.

We may be subject to losses on mortgages for which we do not obtain credit enhancements.

We do not obtain credit enhancements such as mortgage pool or special hazard insurance for all of our mortgages and investments. Generally, we require mortgage insurance on any mortgage with an LTV ratio greater than 80%. During the time we hold mortgages for investment, we are subject to risks of borrower defaults and bankruptcies and special hazard losses that are not covered by standard hazard insurance. If a borrower defaults on a mortgage that we hold, we bear the risk of loss of principal to the extent there is any deficiency between the value of the related mortgaged property and the amount owing on the mortgage loan and any insurance proceeds available to us through the mortgage insurer. In addition, since defaulted mortgages, which under our financing arrangements are mortgages that are generally 60 to 90 days delinquent in payments, may be considered ineligible collateral under our borrowing arrangements, we could bear the risk of being required to own these loans without the use of borrowed funds until they are ultimately liquidated or possibly sold at a loss.

Our mortgage products expose us to greater credit risks.

We are an acquirer and originator of Alt-A mortgages, and to a lesser extent, multi-family and B/C mortgages. These are mortgages that generally may not qualify for purchase by government-sponsored agencies such as Fannie Mae and Freddie Mac or "conforming loans". Our operations may be negatively affected due to our investments in these mortgages. Credit risks associated with these mortgages may be greater than those associated with conforming mortgages. The interest rates we charge on these mortgages are often higher than those charged for conforming loans in order to compensate for the higher risk and lower liquidity. Lower levels of liquidity may cause us to hold loans or other mortgage-related assets supported by these loans that we otherwise would not hold. By doing this, we assume the potential risk of increased delinquency rates and/or credit losses as well as interest rate risk. Additionally, the combination of different underwriting criteria and higher rates of interest leads to greater risk, including higher prepayment rates and higher delinquency rates and/or credit losses.

Lending to our type of borrowers may expose us to a higher risk of delinquencies, foreclosures and losses.

Our market includes borrowers who may be unable to obtain mortgage financing from conventional mortgage sources. Mortgages made to such borrowers generally entail a higher risk of delinquency and higher losses than mortgages made to borrowers who utilize conventional mortgage sources. Delinquency, foreclosures and losses generally increase during economic slowdowns or recessions. The actual risk of delinquencies, foreclosures and losses on mortgages made to our borrowers could be higher under adverse economic conditions than those currently experienced in the mortgage lending industry in general.

Further, any material decline in real estate values increases the LTV ratios of mortgages previously made by us, thereby weakening collateral coverage and increasing the possibility of a loss in the event of a borrower default. Any sustained period of increased delinquencies, foreclosures or losses after the mortgages are sold could adversely affect the pricing of our future loan sales and our ability to sell or securitize our mortgages in the future. In the past, certain of these factors have caused revenues and net earnings of many participants in the mortgage industry, including us, to fluctuate from quarter to quarter.

Our multi-family mortgages expose us to increased lending risks.

Generally, we consider multi-family mortgages to involve a higher degree of risk compared to first mortgages on one- to four-family, owner occupied residential properties. These mortgages have higher risks than mortgages secured by residential real estate because repayment of the mortgages often depends on the successful operations and the income stream of the borrowers. Furthermore, multi-family mortgages typically involve larger mortgage balances to single borrowers or groups of related borrowers compared to one- to four-family residential mortgages.

Our use of second mortgages exposes us to greater credit risks.

Our security interest in the property securing second mortgages is subordinated to the interest of the first mortgage holder and the second mortgages have a higher combined LTV ratio than does the first mortgage. If the value of the property is equal to or less than the amount needed to repay the borrower's obligation to the first mortgage holder upon foreclosure, our second mortgage loan will not be repaid.

The geographic concentration of our mortgages increases our exposure to risks in those areas.

We do not set limitations on the percentage of our long-term mortgage portfolio composed of properties located in any one area (whether by state, zip code or other geographic measure). Concentration in any one area increases our exposure to the economic and natural hazard risks associated with that area. Historically, a majority of our mortgage acquisitions and originations, long-term mortgage portfolio and finance receivables were secured by properties in California and, to a lesser extent, Florida. For instance, certain parts of California have experienced an economic downturn in past years and California and Florida have suffered the effects of certain natural hazards. Declines in those residential real estate markets may reduce the values of the properties collateralizing the mortgages, increase foreclosures and losses and have material adverse effect on our results of operations or financial condition.

Furthermore, if borrowers are not insured for natural disasters, which are typically not covered by standard hazard insurance policies, then they may not be able to repair the property or may stop paying their mortgages if the property is damaged. This would cause increased foreclosures and decrease our ability to recover losses on properties affected by such disasters. This would have a material adverse effect on our results of operations or financial condition.

Representations and warranties made by us in our loan sales and securitizations may subject us to liability.

In connection with our loan sales to third parties and our securitizations, we transfer mortgages acquired and originated by us to the third parties or into a trust in exchange for cash and, in the case of a CMO, residual certificates issued by the trust. The trustee or purchaser will have recourse to us with respect to the breach of the standard representations and warranties made by us at the time such mortgages are transferred. While we generally have recourse to our customers for any such breaches, there can be no assurance of our customers' abilities to honor their respective obligations. Also, we engage in bulk whole loan sales pursuant to agreements that generally provide for recourse by the purchaser against us in the event of a breach of one of our representations or warranties, any fraud or misrepresentation during the mortgage origination process, or upon early default on such mortgage. We generally limit the potential remedies of such purchasers to the potential remedies we receive from the customers from whom we acquired or originated the mortgages. However, in some cases, the remedies available to a purchaser of mortgages from us may be broader than those available to us against the sellers of the mortgages and should a purchaser enforce its remedies against us, we may not always be able to enforce whatever remedies we have against our customers. Furthermore, if we discover, prior to the sale or transfer of a loan, that there is any fraud or misrepresentation with respect to the mortgage and the originator fails to repurchase the mortgage, then we may not be able to sell the mortgage or we may have to sell the mortgage at a discount.

In the ordinary course of our business, we are subject to claims made against us by borrowers and trustees in our securitizations arising from, among other things, losses that are claimed to have been incurred as a result of alleged breaches of fiduciary obligations, misrepresentations, errors and omissions of our employees, officers and agents (including our appraisers), incomplete documentation and our failure to comply with various laws and regulations applicable to our business. Any claims asserted against us may result in legal expenses or liabilities that could have a material adverse effect on our results of operations or financial condition.

A substantial interruption in our use of iDASLg2 may adversely affect our level of mortgage acquisitions and originations.

We utilize the Internet in our business principally for the implementation of our automated mortgage origination program, iDASLg2. iDASLg2 allows our customers to pre-qualify borrowers for various mortgage programs based on criteria requested from the borrower and renders an automated underwriting decision by issuing an approval of the mortgage loan or a referral for further review or additional information. Substantially all of our correspondents submit mortgages through iDASLg2 and all wholesale mortgages delivered by mortgage bankers and brokers are directly underwritten through the use of iDASLg2. iDASLg2 may be interrupted if the Internet experiences periods of poor performance, if our computer systems or the systems of our third-party service providers contain defects, or if customers are reluctant to use or have inadequate connectivity to the Internet. Increased government regulation of the Internet could also adversely affect our use of the Internet in unanticipated ways and discourage our customers from using our services. If our ability to use the Internet in providing our services is impaired, our ability to originate or acquire mortgages on an automated basis could be delayed or reduced. Furthermore, we rely on a third party hosting company in connection with the use of iDASLg2. If the third party hosting company fails for any reason, and adequate back-up is not implemented in a timely manner, it may delay and reduce those mortgage acquisitions and originations done through iDASLg2. Any substantial delay and reduction in our mortgage acquisitions and originations will reduce our taxable income for the applicable period.

We are subject to risks of operational failure that are beyond our control.

Substantially all of our operations are located in Newport Beach, California. Our systems and operations are vulnerable to damage and interruption from fire, flood, telecommunications failure, break-ins, earthquake and similar events. Our operations may also be interrupted by power disruptions, including rolling black-outs implemented in California due to power shortages. We do not have alternative power sources in all of our locations. Furthermore, our security mechanisms may be inadequate to prevent security breaches to our computer systems, including from computer viruses, electronic break-ins and similar disruptions. Such security breaches or operational failures could expose us to liability, impair our operations, result in losses, and harm our reputation.

Competition for mortgages is intense and may adversely affect our operations.

We compete in acquiring and originating Alt-A, B/C and multi-family mortgages and issuing mortgage-backed securities with other mortgage conduit programs, investment banking firms, savings and loan associations, banks, thrift and loan associations, finance companies, mortgage bankers and brokers, insurance companies, other lenders, and other entities purchasing mortgage assets.

We also face intense competition from Internet-based lending companies where entry barriers are relatively low. Some of our competitors are much larger than we are, have better name recognition than we do, and have far greater financial and other resources. Government-sponsored entities, in particular Fannie Mae and Freddie Mac, are also expanding their participation in the Alt-A mortgage industry. These government-sponsored entities have a size and cost-of-funds advantage over us that allows them to price mortgages at lower rates than we are able to offer. This phenomenon may seriously destabilize the Alt-A mortgage industry. In addition, if as a result of what may be less-conservative, risk-adjusted pricing, these government-sponsored entities experience significantly higher-than-expected losses, it would likely adversely affect overall investor perception of the Alt-A and B/C mortgage industry because the losses would be made public due to the reporting obligations of these entities.

The intense competition in the Alt-A, B/C and multi-family mortgage industry has also led to rapid technological developments, evolving industry standards and frequent releases of new products and enhancements. As mortgage products are offered more widely through alternative distribution channels, such as the Internet, we may be required to make significant changes to our current retail and wholesale structure and information systems to compete effectively. Our inability to continue enhancing our current Internet capabilities, or to adapt to other technological changes in the industry, could have a material adverse effect on our business, financial condition, liquidity and results of operations.

The need to maintain mortgage loan volume in this competitive environment creates a risk of price competition in the Alt-A, B/C and multi-family mortgage industry. Competition in the industry can take many forms, including interest rates and costs of a loan, less stringent underwriting standards, convenience in obtaining a loan, customer service, amount and term of a loan and marketing and distribution channels. Our failure to maintain our customer service levels may affect our ability to effectively compete in the mortgage industry. Price competition would lower the interest rates that we are able to charge borrowers, which would lower our interest income and or our gain on sale of mortgage loans. Price-cutting or discounting reduces profits and will depress earnings if sustained for any length of time. If our competition uses less stringent underwriting standards we will be pressured to do so as well, resulting in greater loan risk without being able to price for that greater risk. Our competitors may lower their underwriting standards to increase their market share. If we do not relax underwriting standards in the face of competition, we may lose market share. Increased competition may also reduce the volume of our loan originations and acquisitions. Any increase in these pricing and credit pressures could have a material adverse effect on our business, financial condition, liquidity and results of operations.

We are exposed to potential credit losses in providing warehouse financing.

As a warehouse lender, we lend money to mortgage bankers on a secured basis and we are subject to the risks associated with lending to mortgage bankers, including the risks of fraud, borrower default and bankruptcy, any of which could result in credit losses for us. Fraud risk may include, but is not limited to, the financing of nonexistent loans or fictitious mortgage loan transactions or the delivery to us of fraudulent collateral that could result in the loss of all sums we have advanced to the borrower. For example, during 2004, the warehouse lending operations had a specific allowance for loan losses of \$10.7 million for impaired warehouse advances. Also, our claims as a secured lender in a bankruptcy proceeding may be subject to adjustment and delay.

A reduction in the demand for our loan products may adversely affect our operations.

The availability of sufficient mortgages meeting our criteria is dependent in part upon the size and level of activity in the residential real estate lending market and, in particular, the demand for residential mortgages, which is affected by:

- interest rates;
- national economic conditions;
- · residential property values; and
- · regulatory and tax developments.

If our mortgage acquisitions and originations decline, we may have:

- · decreased economies of scale;
- higher origination costs per loan;
- reduced fee income;
- · smaller gains on the sale of mortgages; and
- an insufficient volume of mortgages to generate securitizations which thereby causes us to accumulate mortgages over a longer period.

Our delinquency ratios and our performance may be adversely affected by the performance of parties who service or sub-service our mortgages.

We sell or contract with third-parties for the servicing of all mortgages, including those in our securitizations. Our operations are subject to risks associated with inadequate or untimely servicing. Poor performance by a servicer may result in greater than expected delinquencies and losses on our mortgages. A substantial increase in our delinquency or foreclosure rate could adversely affect our ability to access the capital and secondary markets for our financing needs. Also, with respect to mortgages subject to a securitization, greater delinquencies would adversely impact the value of our equity interest, if any, we hold in connection with that securitization.

In a securitization, relevant agreements permit us to be terminated as servicer or master servicer under specific conditions described in these agreements. If, as a result of a servicer or sub-servicer's failure to perform adequately, we were terminated as master servicer of a securitization, the value of any master servicing rights held by us would be adversely affected.

We are a defendant in purported class actions and may not prevail in these matters.

Class action lawsuits and regulatory actions alleging improper marketing practices, abusive loan terms and fees, disclosure violations, improper yield spread premiums and other matters are risks faced by all mortgage originators, particularly those in the Alt-A and B/C market. We are a defendant in purported class actions pending in different states. The class actions allege generally that the loan originator improperly charged fees in violation of various state lending or consumer protection laws in connection with mortgages that we acquired. Although the suits are not identical, they generally seek unspecified compensatory damages, punitive damages, pre- and post-judgment interest, costs and expenses and rescission of the mortgages, as well as a return of any improperly collected fees. These actions are in the early stages of litigation and, accordingly, it is difficult to predict the outcome of these matters. We believe we have meritorious defenses to the actions and intend to defend against them vigorously; however, an adverse judgment in any of these matters could have a material adverse effect on us.

Regulatory Risks

We may be subject to fines or other penalties based upon the conduct of our independent brokers or correspondents.

The mortgage brokers and correspondents from which we obtain mortgages have parallel and separate legal obligations to which they are subject. While these laws may not explicitly hold the originating lenders or an acquirer of the loan responsible for the legal violations of mortgage bankers and brokers, increasingly federal and state agencies have sought to impose such liability. Previously, for example, the United States Federal Trade Commission, or "FTC," entered into a settlement agreement with a mortgage lender where the FTC characterized a broker that had placed all of its loan production with a single lender as the "agent" of the lender; the FTC imposed a fine on the lender in part because, as "principal," the lender was legally responsible for the mortgage broker's unfair and deceptive acts and practices. The United States Justice Department in the past has sought to hold a sub-prime mortgage lender responsible

for the pricing practices of its mortgage bankers and brokers, alleging that the mortgage lender was directly responsible for the total fees and charges paid by the borrower under the Fair Housing Act even if the lender neither dictated what the mortgage banker could charge nor kept the money for its own account. Accordingly, we may be subject to fines or other penalties based upon the conduct of our independent mortgage bankers, brokers or correspondents.

Violation of various federal, state and local laws may result in losses on our loans.

Applicable state and local laws generally regulate interest rates and other charges, require certain disclosure, and require licensing of the mortgage broker, lender and purchaser. In addition, other state and local laws, public policy and general principles of equity relating to the protection of consumers, unfair and deceptive practices and debt collection practices may apply to the origination, servicing and collection of our loans. Mortgage loans are also subject to federal laws, including:

- the Federal Truth-in-Lending Act and Regulation Z promulgated there under, which require certain disclosures to the borrowers regarding the terms
 of the loans:
- the Equal Credit Opportunity Act and Regulation B promulgated there under, which prohibit discrimination on the basis of age, race, color, sex, religion, marital status, national origin, receipt of public assistance or the exercise of any right under the Consumer Credit Protection Act, in the extension of credit;
- the Fair Housing Act, which prohibits discrimination in housing on the basis of race, color, national origin, religion, sex, familial status, or handicap, in housing-related transactions;
- · the Fair Credit Reporting Act, which regulates the use and reporting of information related to the borrower's credit experience;
- the Fair and Accurate Credit Transaction Act, which regulates credit reporting and use of credit information in making unsolicited offers of credit;
- the Gramm-Leach-Bliley Act, which imposes requirements on all lenders with respect to their collection and use of nonpublic financial information and requires them to maintain the security of that information;
- the Real Estate Settlement Procedures Act, which requires that consumers receive disclosures at various times and outlaws kickbacks that increase
 the cost of settlement services;
- the Home Mortgage Disclosure Act, which requires the reporting of public loan data;
- the Telephone Consumer Protection Act and the Can Spam Act, which regulate commercial solicitations via telephone, fax, and the Internet;
- the Depository Institutions Deregulation and Monetary Control Act of 1980, which preempts certain state usury laws; and
- the Alternative Mortgage Transaction Parity Act of 1982, which preempts certain state lending laws which regulate alternative mortgage transactions.

Violations of certain provisions of these federal and state laws may limit our ability to collect all or part of the principal of or interest on the loans and in addition could subject us to damages and could result in the mortgagors rescinding the loans whether held by us or subsequent holders of the loans. In addition, such violations may cause us to be in default under our credit and repurchase facilities and could result in the loss of licenses held by us.

Our operations may be adversely affected if we are subject to the Investment Company Act.

We intend to conduct our business at all times so as not to become regulated as an investment company under the Investment Company Act. The Investment Company Act exempts entities that are primarily engaged in the business of purchasing or otherwise acquiring mortgages and other liens on and interests in real estate.

In order to qualify for this exemption we must maintain at least 55% of our assets directly in mortgages, qualifying pass-through certificates and certain other qualifying interests in real estate. Our ownership of certain mortgage assets may be limited by the provisions of the Investment Company Act. If the SEC adopts a contrary interpretation with respect to these securities or otherwise believes we do not satisfy the above exception, we could be required to restructure our activities or sell certain of our assets. To insure that we continue to qualify for the exemption we may be required at times to adopt less efficient methods of financing certain of our mortgage assets and we may be precluded from acquiring certain types of higher-yielding mortgage assets. The net effect of these factors will be to lower our net interest income. If we fail to qualify for exemption from registration as an investment company, our ability to use leverage would be substantially reduced, and we would not be able to conduct our business as described. Our business will be materially and adversely affected if we fail to qualify for this exemption.

New regulatory laws affecting the mortgage industry may increase our costs and decrease our mortgage origination and acquisition.

The regulatory environments in which we operate have an impact on the activities in which we may engage, how the activities may be carried out, and the profitability of those activities. Therefore, changes to laws, regulations or regulatory policies can affect whether and to what extent we are able to operate profitably. For example, recently enacted and proposed local, state and federal legislation targeted at predatory lending could have the unintended consequence of raising the cost or otherwise reducing the availability of mortgage credit for those potential borrowers with less than prime-quality credit histories, thereby resulting in a reduction of otherwise legitimate Alt-A or B/C lending opportunities. Similarly, recently enacted and proposed local, state and federal privacy laws and laws prohibiting or limiting marketing by telephone, facsimile, email and the Internet may limit our ability to market and our ability to access potential loan applicants. For example, the Can Spam Act of 2003 establishes the first national standards for the sending of commercial email allowing, among other things, unsolicited commercial email provided it contains certain information and an opt-out mechanism. We cannot provide any assurance that the proposed laws, rules and regulations, or other similar laws, rules or regulations, will not be adopted in the future. Adoption of these laws and regulations could have a material adverse impact on our business by substantially increasing the costs of compliance with a variety of inconsistent federal, state and local rules, or by restricting our ability to charge rates and fees adequate to compensate us for the risk associated with certain loans.

Some states and local governments have enacted, or may enact, laws or regulations that prohibit inclusion of some provisions in mortgage loans that have mortgage rates or origination costs in excess of prescribed levels, and require that borrowers be given certain disclosures prior to the consummation of such mortgage loans. Our failure to comply with these laws could subject us to monetary penalties and could result in the borrowers rescinding the mortgage loans, whether held by us or subsequent holders. Lawsuits have been brought in various states making claims against assignees of these loans for violations of state law.

Furthermore, various federal and state laws impose significant privacy or customer information security obligations which may subject us to additional costs and legal risks and we cannot assure you that we will not be subject to lawsuits or compliance actions under such requirements. Similarly various state and federal laws have been enacted to restrict unsolicited advertising using telephones, facsimile machines and electronic means of transmission. These laws and regulations could have a material adverse impact on our business by substantially increasing the costs of compliance or by subjecting us to lawsuits or compliance actions.

Risks Related To Our Status As A REIT

We may not pay dividends to stockholders.

REIT provisions of the Internal Revenue Code generally require that we annually distribute to our stockholders at least 90% of all of our taxable income, exclusive of the application of any tax loss carry forwards that may be used to offset current period taxable income. These provisions restrict our ability to retain earnings and thereby generate capital from our operating activities. We may decide at a future date to terminate our REIT status, which would cause us to be taxed at the corporate levels and cease paying regular dividends. In addition, for any year that we do not generate taxable income, we are not required to declare and pay dividends to maintain our REIT status. For instance, due to losses incurred in 2000, we did not declare any dividends from September 2000 until September 2001.

To date, a portion of our taxable income and cash flow has been attributable to our receipt of dividend distributions from the mortgage operations. The mortgage operations is not a REIT and is not, therefore, subject to the above-described REIT distribution requirements. Because the mortgage operations is seeking to retain earnings to fund the future growth of our mortgage operations business, its board of directors may decide that the mortgage operations should cease making dividend distributions in the future. This would materially reduce the amount of our taxable income and in turn, would reduce the amount we would be required to distribute as dividends.

If we fail to maintain our REIT status, we may be subject to taxation as a regular corporation.

We believe that we have operated and intend to continue to operate in a manner that enables us to meet the requirements for qualification as a REIT for federal income tax purposes. We have not requested, and do not plan to request, a ruling from the Internal Revenue Service that we qualify as a REIT.

Moreover, no assurance can be given that legislation, new regulations, administrative interpretations or court decisions will not significantly change the tax laws with respect to qualification as a REIT or the federal income tax consequences of such qualification. Our continued qualification as a REIT will depend on our satisfaction of certain asset, income, organizational and stockholder ownership requirements on a continuing basis.

If we fail to qualify as a REIT, we would not be allowed a deduction for distributions to stockholders in computing our taxable income and would be subject to federal income tax at regular corporate rates. We also may be subject to the federal alternative minimum tax. Unless we are entitled to relief under specific statutory provisions, we could not elect to be taxed as a REIT for four taxable years following the year during which we were disqualified. Therefore, if we lose our REIT status, the funds available for distribution to stockholders would be reduced substantially for each of the years involved. Failure to qualify as a REIT could adversely affect the value of our securities.

On October 22, 2004, President Bush signed the American Jobs Creation Act of 2004 (the "2004 Act"), which, among other things, amends the rules applicable to REIT qualification. In particular, the 2004 Act provides that a REIT that fails the quarterly asset tests for one or more quarters will not lose its REIT status as a result of such failure if either (i) such failure is regarded as a de minimis failure under standards set out in the 2004 Act, or (ii) the failure is greater than a de minimis failure but is attributable to reasonable cause and not willful neglect. In the case of a greater than de minimis failure, however, the REIT must pay a tax and must remedy the failure within 6 months of the close of the quarter in which such failure occurred. In addition, the 2004 Act provides relief for failures of other tests imposed as a condition of REIT qualification, as long as such failures are attributable to reasonable cause and not willful neglect. A REIT would be required to pay a penalty of \$50,000, however, in the case of each such failure. The above-described changes apply for taxable years of REITs beginning after the date of enactment.

Potential characterization of distributions or gain on sale as unrelated business taxable income to tax-exempt investors.

If (1) all or a portion of our assets are subject to the rules relating to taxable mortgage pools, (2) we are a "pension-held REIT," (3) a tax-exempt stockholder has incurred debt to purchase or hold our common stock, or (4) the residual REMIC interests we buy generate "excess inclusion income," then a portion of the distributions to and, in the case of a stockholder described in (3), gains realized on the sale of common stock by such tax-exempt stockholder may be subject to Federal income tax as unrelated business taxable income under the Internal Revenue Code.

Classification as a taxable mortgage pool could subject us or certain of our stockholders to increased taxation.

If we have borrowings with two or more maturities and, (1) those borrowings are secured by mortgages or mortgage-backed securities and, (2) the payments made on the borrowings are related to the payments received on the underlying assets, then the borrowings and the pool of mortgages or mortgage-backed securities to which such borrowings relate may be classified as a taxable mortgage pool under the Internal Revenue Code. If any part of our Company were to be treated as a taxable mortgage pool, then our REIT status would not be impaired, but a portion of the taxable income we recognize may, under regulations to be issued by the Treasury Department, be characterized as "excess inclusion" income and allocated among our stockholders to the extent of and generally in proportion to the distributions we make to each stockholder. Any excess inclusion income would:

not be allowed to be offset by a stockholder's net operating losses;

- be subject to a tax as unrelated business income if a stockholder were a tax-exempt stockholder;
- be subject to the application of federal income tax withholding at the maximum rate (without reduction for any otherwise applicable income tax treaty) with respect to amounts allocable to foreign stockholders; and
- be taxable (at the highest corporate tax rate) to us, rather than to our stockholders, to the extent the excess inclusion income relates to stock held by
 disqualified organizations (generally, tax-exempt companies not subject to tax on unrelated business income, including governmental organizations).

Based on advice of our tax counsel, we take the position that our existing financing arrangements do not create a taxable mortgage pool.

We may be subject to possible adverse consequences as a result of limits on ownership of our shares.

Our charter limits ownership of our capital stock by any single stockholder to 9.5% of our outstanding shares unless waived by the board of directors. Our board of directors may increase the 9.5% ownership limit. In addition, to the extent consistent with the REIT provisions of the Internal Revenue Code, our board of directors may, pursuant to our articles of incorporation, waive the 9.5% ownership limit for a stockholder or purchaser of our stock. In order to waive the 9.5% ownership limit our board of directors must require the stockholder requesting the waiver to provide certain representations to the Company to ensure compliance with the REIT provisions of the Internal Revenue Code. Our charter also prohibits anyone from buying shares if the purchase would result in us losing our REIT status. This could happen if a share transaction results in fewer than 100 persons owning all of our shares or in five or fewer persons, applying certain broad attribution rules of the Internal Revenue Code, owning more than 50% (by value) of our shares. If you or anyone else acquires shares in excess of the ownership limit or in violation of the ownership requirements of the Internal Revenue Code for REITs, we:

- will consider the transfer to be null and void;
- will not reflect the transaction on our books;
- may institute legal action to enjoin the transaction;
- will not pay dividends or other distributions with respect to those shares;
- will not recognize any voting rights for those shares;
- · may redeem the shares; and
- will consider the shares held in trust for the benefit of a charitable beneficiary as designated by us.

The trustee shall sell the shares held in trust and the owner of the excess shares will be entitled to the lesser of:

- (a) the price paid by the owner;
- (b) if the owner did not purchase the excess shares, the closing price for the shares on the national securities exchange on which IMH is listed on the day of the event causing the shares to be held in trust; or
- (c) the price received by the trustee from the sale of the shares.

Notwithstanding the above, our charter contains a provision which provides that nothing in the charter will preclude the settlement of transactions entered into through the facilities of the NYSE.

Limitations on acquisition and change in control ownership limit.

The 9.5% ownership limit discussed above may have the effect of precluding acquisition of control of our Company by a third party without consent of our board of directors.

Risks Related To Ownership of Our Securities

Our share prices have been and may continue to be volatile.

Historically, the market price of our securities has been volatile. The market price of our securities is likely to continue to be highly volatile and could be significantly affected by factors including:

- the amount of dividends paid;
- availability of liquidity in the securitization market;
- loan sale pricing;
- · termination of financing agreements;
- margin calls by warehouse lenders or changes in warehouse lending rates;
- · unanticipated fluctuations in our operating results;
- · prepayments on mortgages;
- valuations of securitization related assets;
- the effect of the restatement of our financial condition and results of operations;
- mark to market adjustments related to the fair value of derivatives;
- · cost of funds; and
- · general market conditions.

In addition, significant price and volume fluctuations in the stock market have particularly affected the market prices for the securities of mortgage REIT companies such as ours. These broad market fluctuations have adversely affected and may continue to adversely affect the market price of our common stock. If our results of operations fail to meet the expectations of securities analysts or investors in a future quarter, the market price of our securities could also be materially adversely affected and we may experience difficulty in raising capital.

Sales of additional common stock may adversely affect its market price.

To sustain our growth strategy we intend to raise capital through the sale of equity. The sale or the proposed sale of substantial amounts of our common stock in the public market could materially adversely affect the market price of our common stock or other outstanding securities. We do not know the actual or perceived effect of these offerings, the timing of these offerings, the potential dilution of the book value or earnings per share of our securities then outstanding and the effect on the market price of our securities then outstanding. For example, during 2004 we raised approximately \$366.3 million and \$152.2 million of net proceeds through offerings of our common and preferred stock, respectively. We also have shares reserved for future issuance under our stock plans. The sale of a large amount of shares or the perception that such sales may occur, could adversely affect the market price for our common stock or other outstanding securities.

ITEM 2. PROPERTIES

Our primary executive and administrative offices are located at 1401 Dove Street, Newport Beach, California where we have a premises lease expiring in May of 2008 to use approximately 74,000 square feet of office space. We also executed premises leases located at 1500 Quail Street, Newport Beach, California expiring in November of 2006 to use approximately 15,000 square feet of office space and 1301 Dove St., Newport Beach, California expiring in August

of 2008 to use approximately 16,000 square feet of office space to accommodate expansion. In addition, the mortgage operations has mortgage production offices located in various states with premises lease terms ranging from month to month or one to two years.

On March 4, 2005, we entered into a new lease for our business and corporate facilities. The lease is for a term of ten years and commences on the date construction of the premises is complete or the date we commence business operations on the premises. We have two options to extend the term for five-year periods for each option. The premises are to be located at 19500 Jamboree Road, Newport Beach, California and are anticipated to be ready for business operations during 2006. The premises will consist of a seven-story building containing approximately 200,000 square feet with an initial annual rental rate of \$31.80 per square foot, which amount increases every 30 months. We have options for additional space in the complex if needed. We anticipate moving our entire Orange County operations to this facility.

ITEM 3. LEGAL PROCEEDINGS

On September 1, 2000, a complaint captioned Michael P. and Shellie Gilmor v. Preferred Credit Corporation and Impac Funding Corporation, et al. was filed in the Circuit Court for Clay County, Missouri, Case No. CV100-4263-CC, as a purported class action lawsuit alleging that the defendants violated Missouri's Second Loans Act and Merchandising Practices Act. In July 2001, the Missouri complaint was amended to include IMH and other Impac-related entities. A plaintiffs class was certified on January 2, 2003. On June 22, 2004, the court issued an order to stay all proceedings pending the outcome of an appeal in a similar case in the Eighth Circuit.

On February 3, 2004, a complaint captioned <u>James and Jill Baker v. Century Financial Group, Inc, et al</u> was filed in the Circuit Court of Clay County, Missouri, Case No. CV100-4294-CC as a purported class action lawsuit alleging that the defendants violated Missouri's Second Loan Act and Merchandising Practices Act.

On October 2, 2001, a complaint captioned <u>Deborah Searcy, Shirley Walker, et al. v. Impac Funding Corporation, Impac Mortgage Holdings, Inc. et. al.</u> was filed in the Wayne County Circuit Court, State of Michigan, as a purported class action lawsuit alleging that the defendants violated Michigan's Secondary Mortgage Loan Act, Credit Reform Act and Consumer Protection Act. A motion to dismiss an amended complaint has been filed, but not yet ruled upon.

On November 30, 2001, a complaint captioned <u>Garry Lee Skinner and Judy Cooper Skinner, et al. v. Preferred Credit, et al.</u> was filed in the Superior Court of Durham County, North Carolina as Case No. 1CV-05596. This is stated as a purported class action alleging a violation of the North Carolina Interest Statutes and Unfair and Deceptive Trade Practices Act when the secondary mortgage loans were originated by the defendants. A motion to dismiss the complaint was filed and granted by an order of the court dated June 9, 2004. Plaintiffs have appealed the dismissal. On January 28, 2005, plaintiffs filed a motion to dismiss us from the appeal. On February 14, 2005, the court granted the plaintiffs their motion to dismiss us from the appeal.

On July 31, 2003, a purported class action complaint captioned <u>Frazier</u>, <u>et al v. Impac Funding Corp.</u>, <u>et al</u>, Case No. 03-2565 was filed in federal court in Tennessee. The causes of action in the action allege violations of Tennessee's usury statute and Consumer Protection Act. A motion to dismiss the complaint was filed and not yet ruled upon. The court agreed to administratively close the case on April 5, 2004 pending an appeal in a similar case. On April 29, 2004, the court issued its order administratively closing <u>Frazier</u>.

On November 25, 2003, a complaint captioned Michael and Amber Stallings v. Empire Funding Home Loan Owner Trust 1997-3; U.S. Bank, National Association; and Wilmington Trust Company was filed in the United States District Court for the Western District of Tennessee, Case No. 03-2548, as a purported class action lawsuit alleging that the defendants violated Tennessee predatory lending laws governing second mortgage loans. The complaint further alleges that certain assignees of mortgage loans, including two Impac-related trusts, should be included as defendants in the lawsuit. Like the Frazier matter this case was administratively closed on April 5, 2004 pending an appeal in a similar case.

All of the above purported class action lawsuits are similar in nature in that they allege that the mortgage loan originators violated the respective state's statutes by charging excessive fees and costs when making second mortgage loans on residential real estate. The complaints allege that IFC was a purchaser, and is a holder, along with other affiliated entities, of second mortgage loans originated by other lenders. The plaintiffs in the lawsuits are seeking damages that include disgorgement of interest paid, restitution, rescission, actual damages, statutory damages, exemplary damages, pre-judgement interest and punitive damages. No specific dollar amount of damages is specified in the complaints.

We believe that we have meritorious defenses to the above claims and intend to defend these claims vigorously. Nevertheless, litigation is uncertain and we may not prevail in the lawsuits and can express no opinion as to its ultimate outcome. An adverse judgment in any of these matters could have a material adverse affect on us, however, no judgment in any matter is probable to occur nor is any amount of any loss from such judgment reasonably estimable at this time. As of December 31, 2004 and 2003, no reserves regarding such litigation and claims have been recorded.

On October 14, 2003, an action was filed in the Circuit Court of Cook County, Illinois as Case No. 03 CH17085 entitled Fast Forward Solutions, LLC v. Novelle Financial Services, Inc. The complaint contained an allegation of a class action and alleged that the defendant sent out unsolicited faxes in violation of the Telephone Consumer Protection Act, the Illinois Consumer Fraud Act, and Illinois common law. The plaintiff was seeking statutory and treble damages. We settled this action for an immaterial amount and the matter was dismissed on January 31, 2005.

We are a party to other litigation and claims which are normal in the course of our operations. While the results of such other litigation and claims cannot be predicted with certainty, we believe the final outcome of such matters will not have a material adverse effect on our financial condition or results of operations.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

No matters were submitted to the security holders to be voted on during the fourth quarter of 2004.

PART II

ITEM 5. MARKET FOR REGISTRANT'S COMMON EQUITY AND RELATED STOCKHOLDER MATTERS

On June 11, 2003, our common stock was listed and began trading on the NYSE under the symbol "IMH." Prior to June 11, 2003, our common stock was listed on the American Stock Exchange under the same symbol. The following table summarizes the high, low and closing sales prices for our common stock for the periods indicated:

	2004			2003		
	High	Low	Close	High	Low	Close
First Quarter	\$27.20	\$18.25	\$27.20	\$13.23	\$11.05	\$12.99
Second Quarter	26.73	17.15	22.52	16.97	12.46	16.69
Third Quarter	27.91	21.07	26.30	16.55	12.50	16.19
Fourth Quarter	27.19	20.50	22.67	19.21	14.80	18.21

On March 24, 2005, the last reported sale price of our common stock on the NYSE was \$16.42 per share. As of March 24, 2005, there were 615 holders of record, including holders who are nominees for an undetermined number of beneficial owners, of our common stock.

Common Stock Dividend Distributions. To maintain our qualification as a REIT, we intend to make annual distributions to stockholders at an amount that maintains our REIT status in accordance with the Internal Revenue Code, which may not necessarily equal net earnings as calculated in accordance with GAAP. Our dividend policy is subject to revision at the discretion of the board of directors. All distributions in excess of those required to maintain our REIT status will be made at the discretion of the board of directors and will depend on our taxable income, financial condition and other factors as the board of directors deems relevant. The board of directors has not established a minimum distribution level. Distributions to stockholders will generally be taxable as ordinary income or qualified income, which is subject to a 15% tax rate, although a portion of such distributions may be designated by us as capital gain or may constitute a tax-free return of capital. We annually furnish to each of our stockholders a statement setting forth distributions paid during the preceding year and their characterization as ordinary income, qualified income, capital gain or return of capital.

The following table presents our common stock dividends paid or declared for the periods indicated:

Period Covered	Stockholder Record Date	Di	Per Share Dividend Amount	
Quarter ended March 31, 2003	April 4, 2003	\$	0.50	
Quarter ended June 30, 2003	July 3, 2003	\$	0.50	
Quarter ended September 30, 2003	October 3, 2003	\$	0.50	
Quarter ended December 31, 2003	December 15, 2003	\$	0.55	
Quarter ended March 31, 2004	April 05, 2004	\$	0.65	
Quarter ended June 30, 2004	July 06, 2004	\$	0.75	
Quarter ended September 30, 2004	October 08, 2004	\$	0.75	
Quarter ended December 31, 2004	December 15, 2004	\$	0.75	

ITEM 6. SELECTED UNAUDITED CONSOLIDATED FINANCIAL DATA

The following unaudited selected consolidated statements of operations data for the year ended December 31, 2004 and the unaudited consolidated balance sheet data as of December 31, 2004 were derived from the unaudited consolidated financial statements. The following selected consolidated statements of operations data for each of the years in the four-year period ended December 31, 2003 and the consolidated balance sheet data as of the year end for each of the years in the four-year period ended December 31, 2003 were derived from the audited consolidated financial statements. Such selected financial data should be read in conjunction with the consolidated financial statements and the notes to the consolidated financial statements starting on page F-1 and with Item 7. "Management's Discussion and Analysis of Financial Condition and Results of Operations."

IMPAC MORTGAGE HOLDINGS, INC.

(the 2004 financial statements are unaudited and the 2003, 2002, 2001 and 2000 financial statements have been derived from the audited financial statements)

(amounts in thousands, except per share data)

	For the year ended December 31,				
	2004	2003 (1)	2002	2001	2000
Statement of Operations Data:					
Net interest income:					
Interest income	\$755,616	\$385,716	\$230,267	\$141,563	\$119,277
Interest expense	412,533	209,009	127,801	108,183	124,071
Net interest income (expense)	343,083	176,707	102,466	33,380	(4,794)
Provision for loan losses	30,927	24,853	19,848	16,813	18,839
Net interest income (expense) after provision for loan losses	312,156	151,854	82,618	16,567	(23,633)
Non-interest income:					
Gain on sale of loans	25,134	39,022	_	_	_
Other income	10,948	9,995	2,864	5,295	3,857
Gain (loss) on derivative instruments	4,694	(16,021)		(33,391)	238
Equity in net earnings of IFC	_	11,537	11,299	19,499	24,461
Total non-interest income (expense)	40,776	44,533	(36,339)	(8,597)	28,556
Non-interest expense:					
Personnel expense	60,420	25,250	1,856	1,192	652
Other expense	17,797	12,571	1,898	1,669	2,435
General and administrative and other expense	17,797	7,660	985	1,686	904
Amortization of deferred charge	16,212	5,658		1,000 —	
Write-down on securities available-for-sale	1,120	298	1,039	2,217	53,576
(Gain) loss on disposition of real estate owned	(3,901)	(2,632)	1,039	(1,931)	1,814
(Galif) foss off disposition of feat estate owned	(3,901)	(2,032)		(1,951)	
Total non-interest expense	108,745	48,805	5,932	4,833	59,381
Earnings (loss) before extraordinary item and cumulative effect of change in accounting principle	244,187	147,582	40,347	3,137	(54,458)
Extraordinary item	<u> </u>	_	_	(1,006)	
Income taxes	(13,450)	(1,397)	_		_
Cumulative effect of change in accounting principle			_	(4,313)	_
Net earnings (loss)	\$257,637	\$148,979	\$ 40,347	\$ (2,182)	\$ (54,458)
Net earnings (loss) per share before extraordinary item and cumulative effect of change in accounting					
principle: Basic	\$ 3.79	\$ 2.94	\$ 1.01	\$ 0.07	\$ (2.71)
	5.75	—	4 1.01		(= 17.1)
Diluted	\$ 3.72	\$ 2.88	\$ 0.99	\$ 0.11	\$ (2.71)
Net earnings (loss) per share:					
Basic	\$ 3.79	\$ 2.94	\$ 1.01	\$ (0.16)	\$ (2.71)
Diluted	\$ 3.72	\$ 2.88	\$ 0.99	\$ (0.16)	\$ (2.71)
	ф. 2.06	Φ 2.05	ф. 1.70	ф. 2.66	ф
Dividends declared per share	\$ 2.90	\$ 2.05	\$ 1.76	\$ 0.69	\$ 0.36

IMPAC MORTGAGE HOLDINGS, INC.

(the 2004 financial statements are unaudited and the 2003, 2002, 2001 and 2000 financial statements have been derived from the audited financial statements)

(amounts in thousands, except operating data)

As of December 31,

	·				
	2004	2003 (1) 2002		2001	2000
Balance Sheet Data:					
CMO collateral and mortgages held-for-investment	\$21,895,592	\$ 9,296,893	\$5,215,731	\$2,242,036	\$1,367,573
Finance receivables	471,820	630,030	664,021	300,571	138,406
Mortgages held-for-sale	587,745	397,618	_	_	_
Investment in and advances to IFC (1)	_	_	531,032	210,134	304,578
Total assets	23,815,767	10,577,957	6,540,339	2,842,677	1,884,489
CMO borrowings	21,206,373	8,489,853	5,019,934	2,139,818	1,277,161
Reverse repurchase agreements	1,527,558	1,568,807	1,168,029	469,491	398,653
Total liabilities	22,771,692	10,105,170	6,256,814	2,646,847	1,706,274
Total stockholders' equity	1,044,075	472,787	283,525	195,830	178,214

(1) On July 1, 2003, IMH purchased 100% of the outstanding shares of common stock of IFC. The purchase of IFC's common stock combined with IMH's ownership of 100% of IFC's preferred stock resulted in the consolidation of IFC from July 1, 2003 through December 31, 2003. Prior to July 1, 2003, IFC was a non-consolidated subsidiary of IMH and 99% of the net earnings of IFC were reflected in IMH's financial statements as "Equity in net earnings (loss) of IFC."

	As of and for the year December 51,				
	2004	2003	2002	2001	2000
Operating Data (in millions):					
Mortgage acquisitions and originations for the year	\$22,213	\$ 9,525	\$5,945	\$3,204	\$2,113
Master servicing portfolio at period-end	28,404	13,919	8,694	5,569	4,043
Servicing portfolio at period-end	1,691	1,402	2,653	1,754	2,429

ITEM 7. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Management's discussion and analysis of financial condition and results of operations contain certain forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. Refer to Item 1. "Business—Forward-Looking Statements" for a complete description of forward-looking statements. All of our businesses actively work together to deliver comprehensive mortgage and lending services to our correspondents, mortgage bankers and brokers, retail customers and capital market investors through a wide array of mortgage loan programs using webbased technology and centralized operations so that we can provide high levels of customer service at low per loan operating costs. We elect to be taxed as a REIT for federal income tax purposes, which generally allows us to pass through income to stockholders without payment of federal income tax at the corporate level. Our goal is to generate consistent and reliable income for distribution to our stockholders primarily from the earnings of our core operating businesses, which include the long-term investment operations, mortgage operations and warehouse lending operations. Refer to Item 1. "Business" for additional information on our businesses and operating segments.

Use of Unaudited Financial Statements

As disclosed in previous reports, we restated our financial statements for each of the years ended December 31, 2003, 2002 and 2001, the three months ended March 31, 2004 and 2003, the three and six months ended June 30, 2003 and the three and nine months ended September 30, 2003. The Company also restated its consolidated statements of operations for the six months ended June 30, 2004 due to a correction to earnings per share as a result of a clerical error. The time and resources committed to the restatements delayed our internal timetable with respect to our documentation, assessment and evaluation of internal control over financial reporting, which were undertaken in order to comply with the requirements of Section 404 of the Sarbanes-Oxley Act of 2002, or "SOX 404." Due in large part to the foregoing, management's assessment of the effectiveness of our internal control over financial reporting was substantially delayed, thereby delaying the performance by KPMG LLP, or "KPMG," our independent registered public accounting firm, of its audit of management's assessment and audit of the effectiveness of our internal control over financial reporting pursuant to the requirements of SOX 404. As a result, at the date of this filing, such audit and management's report on internal control over financial reporting is not yet complete. In addition, the audit of our financial statements is not complete. Therefore, the audit reports from KPMG on the financial statements and its audit of management's assessment of the effectiveness of internal control over financial reporting are not included in this annual report on Form 10-K. Consequently, this annual report on Form 10-K includes unaudited financial statements and statistical information for the year ended December 31, 2004. The financials statements as of December 31, 2003 and for the years ended December 31, 2003 and 2002 have been derived from the audited financials statements for such periods. Upon completion of our assessment of the effectiveness of internal contro

Management believes that the financial statements included in this report fairly present in all material respects our financial condition, results of operations and cash flows for the periods presented. However, because no independent registered public accounting firm has opined that the unaudited financial statements included in this annual report present fairly, in all material respects, the financial position, the results of operations, cash flows and the changes in shareholders' equity for the year ended December 31, 2004 reported in accordance with generally accepted accounting principles in the United States of America, or "GAAP," such statements could be subject to change or the financial information included therein may be materially different from audited financial information. There can be no assurance that such changes or differences would not be significant. Our discussion and analysis of our financial condition and results of operations are based upon these unaudited financial statements.

Consolidation of IFC

On July 1, 2003, IMH entered into a stock purchase agreement with Joseph R. Tomkinson, our Chairman, Chief Executive Officer and a director, William S. Ashmore, our Chief Operating Officer, President and a director, and the Johnson Revocable Living Trust, of which Richard J. Johnson, our Executive Vice President and Chief Financial Officer is trustee, whereby we purchased all of the outstanding shares of voting common stock of IFC for aggregate consideration of \$750,000. Each of Messer's. Tomkinson and Ashmore and the Johnson Revocable Living Trust owned one-third of the outstanding common stock of IFC. The fairness opinion related to the purchase price of IFC, as rendered by an independent financial advisor, and the subsequent transaction was approved by our board of directors. Prior to the July 1, 2003 acquisition, the common stock of IFC represented 1% of the economic interest in IFC and the outstanding non-voting preferred stock of IFC represented 99% of the economic interest in IFC. As a result of acquiring 100% of IFC's common stock on July 1, 2003, IMH owns all of the common stock and preferred stock of IFC and began to consolidate IFC as of that date. Due to the

consolidation of IFC on July 1, 2003, the consolidated financial statements include the results of operations of the mortgage operations for the period from January 1, 2003 to June 30, 2003, or "non-consolidation period," as equity in net earnings of IFC and for the period from July 1, 2003 to December 31, 2003, or "consolidation period," on a consolidated basis. IMH has not consolidated the financial statements of IFC for the periods prior to July 1, 2003 to conform to the current presentation.

Critical Accounting Policies

We define critical accounting policies as those that are important to the portrayal of our financial condition and results of operations and require estimates and assumptions based on our judgment of changing market conditions and the performance of our assets and liabilities at any given time. In determining which accounting policies meet this definition, we considered our policies with respect to the valuation of our assets and liabilities and estimates and assumptions used in determining those valuations. We believe the most critical accounting issues that require the most complex and difficult judgments and that are particularly susceptible to significant change to our financial condition and results of operations include the following:

- allowance for loan losses;
- · derivative financial instruments; and
- securitization of financial assets as financing versus sale.

Allowance for Loan Losses

We provide an allowance for loan losses for mortgages held as CMO collateral, finance receivables and mortgages held-for-investment, or "loans provided for." In evaluating the adequacy of the allowance for loan losses, management takes several items into consideration. For instance, a detailed analysis of historical loan performance data is accumulated and reviewed. This data is analyzed for loss performance and prepayment performance by product type, origination year and securitization issuance. The results of that analysis are then applied to the current mortgage portfolio and an estimate is created. We believe that pooling of mortgages with similar characteristics is an appropriate methodology in which to evaluate the allowance for loan losses. Management also recognizes that there are qualitative factors that must be taken into consideration when evaluating and measuring inherent loss in our loan portfolios. These items include, but are not limited to, economic indicators that may affect the borrower's ability to pay, changes in value of collateral, projected loss curves, political factors and industry statistics. Specific valuation allowances may be established for loans that are deemed impaired, if default by the borrower is deemed probable, and if the fair value of the loan or the collateral is estimated to be less than the gross carrying value of the loan. Actual losses on loans are recorded as a reduction to the allowance through charge-offs. Subsequent recoveries of amounts previously charged off are credited to the allowance.

Derivative Financial Instruments

Rate Lock and Purchase Commitments

We enter into commitments to make loans whereby the interest rate on the loan is set prior to funding (rate lock commitments). We also enter into commitments to purchase mortgage loans through our correspondent channel (purchase commitments). Rate lock and purchase commitments that are considered to be derivatives are recorded at fair value in the consolidated statements of financial condition with changes in fair value recorded in gain (loss) on derivative instruments in the consolidated statements of operations.

Unlike most other derivative instruments, there is no active market for rate lock and purchase commitments that can be used to determine their fair value. Consequently, we have developed a method for estimating the fair value of our rate lock and purchase commitments. The fair value of rate lock and purchase commitments are determined by calculating the change in market value from the point of lock to the measurement date (estimated fund date) based upon changes in interest rates during the period, adjusted for an anticipated fallout factor for loan commitments that are not expected to fund. Under this fair value methodology, the lock has zero value on day one and all future value is the result of changes in interest rates, exclusive of any inherent servicing value.

Forward Commitments

The policy of recognizing the fair value of the rate lock and purchase commitments has the effect of recognizing a gain or loss on the related mortgage loans based on changes in the interest rate environment before the mortgage loans are funded and sold. As such, both rate lock and purchase commitments expose us to interest rate risk. We mitigate such risk by entering

into forward commitments and derivative transactions, such as mandatory commitments on U.S. Treasury bonds and mortgage-backed securities, call options and put options. These forward commitments and derivative transactions are treated as derivatives under the provisions of SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities," (SFAS 133), with the change in fair value reported as a gain (loss) on derivative instruments.

The fair values of our forward commitments and derivative transactions are generally based on market prices provided by dealers which make markets in these financial instruments.

Interest Rate Swaps, Caps, and Floors

Our primary objective is to limit the exposure to the variability in future cash flows attributable to the variability of one-month LIBOR, which is the underlying index of adjustable rate CMO and warehouse borrowings. We also monitor on an ongoing basis the prepayment risks that arise in fluctuating interest rate environments. Our interest rate risk management policies are formulated with the intent to offset the potential adverse effects of changing interest rates on CMO and warehouse borrowings.

To mitigate exposure to the effect of changing interest rates on cash flows on CMO and warehouse borrowings, we purchase derivative instruments primarily in the form of interest rate swap agreements (swaps) and, to a lesser extent, interest rate cap agreements (caps) and interest rate floor agreements (floors). The swaps, caps and floors are treated as derivatives under the provisions of SFAS 133, with changes in fair value reported as gain (loss) on derivative instruments in the consolidated statements of operations. Cash paid or received on swaps, caps and floors is recorded as a current period expense or income in gain (loss) on derivative instruments in the consolidated statements of operations.

The fair values of our interest rate swaps, caps, floors and other derivative transactions are generally based on market prices provided by dealers which make markets in these financial instruments.

Securitization of Financial Assets as Financing versus Sale

The mortgage operations recognize gains or losses on the sale of mortgages when the sales transaction settles or upon the securitization of the mortgages when the risks of ownership have passed to the purchasing party. Gains and losses may be increased or decreased by the amount of any servicing related premiums received and costs associated with the acquisition or origination of mortgages. A transfer of financial assets in which control is surrendered is accounted for as a sale to the extent that consideration other than a beneficial interest in the transferred assets is received in the exchange. The long-term investment operations structure CMO securitizations as financing arrangements and recognize no gain or loss on the transfer of mortgage assets. The CMO securitization trusts do not meet criteria within SFAS No. 140, "Accounting for Transfers and Servicing of Financial Assets and Extinguishments of Liabilities" (SFAS 140), to be qualifying special purpose entities, and further, are considered variable interest entities under FASB Interpretation No. 46R and, therefore, are consolidated by the long-term investment operations as the entities' primary beneficiary. The mortgage operations structure REMIC securitizations as sales, and gains and losses are recognized. Liabilities and derivatives incurred or obtained at the transfer of financial assets are required to be measured at fair value, if practicable. Also, servicing assets and other retained interests in the transferred assets must be measured by allocating the previous carrying value between the asset sold and the interest retained, if any, based on their relative fair values at the date of transfer. To determine the value of the securities and retained interest, management estimates future rates of prepayments, prepayment penalties to be received, delinquencies, defaults and default loss severity and their impact on estimated cash flows.

Financial Highlights for 2004

- Net earnings per diluted common share were \$3.72 compared to net earnings per diluted common share of \$2.88 for 2003
- Estimated taxable income per diluted common share increased 21% to \$2.97 compared to actual taxable income per diluted common share of \$2.46 for 2003
- Cash dividends declared increased 41% to \$2.90 per common share compared to \$2.05 per common share for 2003
- Total assets increased 125% to \$23.8 billion as of December 31, 2004 compared to \$10.6 billion as of prior year-end
- Book value per common share increased 41% to \$11.80 as of December 31, 2004 as compared to \$8.39 as of prior year-end primarily due to the issuance of new common equity resulting in net proceeds of \$366.3 million at an average gross price per share of \$20.78; in addition we raised net proceeds of \$152.2 million of preferred equity

- Total market capitalization was \$1.7 billion as of December 31, 2004 compared to \$1.0 billion as of prior year-end
- Dividend yield as of December 31, 2004 was 13.23%, based on an annualized fourth quarter dividend of \$0.75 per common share and a closing stock price of \$22.67
- For 2004, total return to common stockholders was 40.42% based on common stock price appreciation of \$4.46 per common share during the year as well as common stock dividends declared of \$2.90 per common share
- The mortgage operations acquired and originated \$22.2 billion of primarily Alt-A mortgages during 2004, a 134% increase over \$9.5 billion for 2003
- The long-term investment operations retained \$16.9 billion of primarily Alt-A mortgages and originated \$458.5 million of multi-family mortgages during 2004 compared to \$5.8 billion and \$290.5 million, respectively, for 2003
- For 2004, the long-term investment operations securitized \$17.7 billion of mortgages as CMO borrowings to finance the acquisition and origination and retention of Alt-A and multi-family mortgages for long-term investment.

Taxable Income

After adjusting for differences between net earnings and estimated taxable income, estimated taxable income available to common stockholders was \$202.9 million, or \$2.97 per diluted common share, for 2004 as compared to \$127.5 million, or \$2.46 per diluted common share, for 2003 and \$84.4 million, or \$2.07 per diluted common share, for 2002. When we file our annual tax returns there are certain adjustments that we make to net earnings and taxable income due to differences in the nature and extent that revenues and expenses are recognized under the two methods. As an example, to calculate estimated taxable income we deduct actual loan losses as compared to the determination of net earnings which requires a deduction of loan loss provisions that are determined based on estimated losses inherent in our mortgage portfolios. To maintain our REIT status, we are required to distribute a minimum of 90% of our annual taxable income to our stockholders. Because we pay dividends based on taxable income, dividends may be more or less than net earnings. As such, we believe that the disclosure of estimated taxable income available to common stockholders, which is a non-generally accepted accounting principle, or "GAAP," financial measurement, is useful information for our investors.

We paid total regular cash dividends of \$2.90 per common share during 2004, \$2.53 per common share during 2003 and \$1.76 per common share during 2002, which when combined with available tax loss carry-forwards met taxable income distribution requirements for each year. Upon the filing of our 2003 tax return, we had a federal net operating tax loss carry-forward of \$18.7 million, which expires in the year 2020 and which may or may not be used to offset taxable income in 2004 or in subsequent years. We expect to file our 2004 federal and state tax returns in September 2005 at which time changes to federal net operating loss carry-forwards, if any, will be determined.

The following table presents a reconciliation of net earnings to estimated and actual taxable income available to common stockholders for the periods indicated (in thousands, except per share amounts):

	For the year ended December 31,		
	2004 (1)	2003 (1)	2002 (1)
Net earnings	\$ 257,637	\$148,979	\$ 40,347
Adjustments to net earnings:			
Loan loss provision	30,927	24,853	19,848
Dividends from IFC	37,000	31,385	12,870
Cash received from previously charged-off assets	_	(5,533)	
Tax loss on sale of investment securities		(4,725)	
Tax deduction for actual loan charge-offs, net of recoveries	(16,252)	(12,859)	(4,938)
Unrealized fair value of derivatives (2)	(103,724)	(38,762)	22,141
Dividends on preferred stock	(3,750)	_	_
Net earnings of IFC (3)	(42,944)	(16,889)	
Equity in net earnings of IFC (3)	_	(11,537)	(11,299)
Net adjustments on inter-company loan sales transactions and inter-company income and expense eliminations	44,048	12,339	6,655
Net miscellaneous adjustments	_	215	(1,267)
Estimated and actual taxable income available to common stockholders (4)	\$ 202,942	\$127,466	\$ 84,357
Estimated and actual taxable income per diluted common share (4)	\$ 2.97	\$ 2.46	\$ 2.07
•			
Diluted weighted average common shares outstanding	68,244	51,779	40,773

- (1) Estimated taxable income include estimates of book to tax adjustments and can differ from actual taxable income as calculated when we file our annual corporate tax return for 2004. Since estimated taxable income is a non-GAAP financial measurement, the reconciliation of estimated taxable income available to common stockholders to net earnings meets the requirement of Regulation G as promulgated by the SEC for the presentation of non-GAAP financial measurements. The years ended December 31, 2003 and 2002 are actual taxable income per the filing of our corporate tax returns for those years.
- (2) The mark-to-market change for the valuation of derivatives is income or expense for GAAP financial reporting but is not included as an addition or deduction for taxable income calculations. Unrealized fair value of derivatives is a component of gain (loss) on derivatives in the financial statements.
- (3) Represents net earnings of IFC, a taxable REIT subsidiary, which may not necessarily equal taxable income. Dividend distributions from IFC to IMH may exceed IFC's net earnings, however, IMH can only recognize dividends received from IFC as taxable income to the extent that IFC's dividends are from current period taxable income. Any dividends paid to IMH by IFC in excess of IFC's taxable income would be recognized as return of capital by IMH.
- (4) Excludes the deduction for dividends paid and the availability of a deduction attributable to net operating loss carry-forwards.

Issuance of Preferred Stock

In May of 2004, we completed the sale of 2.0 million shares of 9.375% Series B Cumulative Redeemable Preferred Stock, par value \$0.01 per share, liquidation preference \$25.00 per share, or "series B preferred stock." Dividends on the series B preferred stock are payable quarterly in arrears on or before March 31, June 30, September 30 and December 31 of each year. The shares of series B preferred stock have no stated maturity, are not subject to any sinking fund or mandatory redemption and are not convertible into any other securities. Holders of shares of series B preferred stock generally have no voting rights, but will have limited voting rights if the Company fails to pay dividends for six or more quarters and in certain other events. The Company may not redeem the series B preferred stock until May 29, 2009 except in limited circumstance to preserve the Company's status as a real estate investment trust. On or after May 29, 2009, the Company may, at its option, redeem the series B preferred stock in whole or in part, at any time and from time to time, for cash at \$25.00 per share, plus accrued and unpaid dividends (whether or not declared), if any, to and including the redemption date.

In November and December, 2004, we completed the sale of an aggregate of 4.3 million shares of 9.125% Series C Cumulative Redeemable Preferred Stock, par value \$0.01 per share, liquidation preference \$25.00 per share, or "series C preferred stock." Dividends on the series C preferred stock are payable quarterly in arrears on or before March 31, June 30, September 30 and December 31 of each year. The shares of series C preferred stock have no stated maturity, are not subject to any sinking fund or mandatory redemption and are not convertible into any other securities. Holders of shares of series C preferred stock generally have no voting rights, but will have limited voting rights if the Company fails to pay dividends for six or more quarters and in certain other events. The Company may not redeem the series C preferred stock until November 23, 2009 except in limited circumstances to preserve the Company's status as a real estate investment trust. On or after November 23, 2009, the Company may, at its option, redeem the series C preferred stock in whole or in part, at any time and from time to time, for cash at \$25.00 per share, plus accrued and unpaid dividends (whether or not declared), if any, to and including the redemption date.

Estimated Expenditures Related to Section 404 of the Sarbanes-Oxley Act of 2002

As of December 31, 2004, we have spent approximately \$2.0 million to implement actions related to compliance with Section 404 of the Sarbanes-Oxley Act of 2002. Additionally, we anticipate expenditures of approximately \$1.0 million to \$1.5 million for additional actions during 2005 to support and complete our auditor's reports on management's assessment and internal control over financial reporting for 2004.

Results of Operations and Financial Condition

Results of Operations

For the Year Ended December 31, 2004 compared to the Year Ended December 31, 2003

Net earnings were \$257.6 million, or \$3.72 per diluted common share, for 2004 as compared to \$149.0 million, or \$2.88 per diluted common share, for 2003. The year-over-year increase in net earnings of \$108.6 million was primarily the result of the following:

- \$166.4 million increase in net interest income; which was partially offset by,
- \$59.9 million increase in non-interest expense.

These variances along with a discussion of other significant changes affecting results of operations on a comparative basis are discussed in further detail below.

Net Interest Income

Net interest income is primarily derived from interest income on mortgage assets which include CMO collateral, mortgages held-for-investment, mortgages held-for-sale, finance receivables and investment securities available-for-sale, or collectively, "mortgage assets," less interest expense from interest paid on borrowings on mortgage assets, which include CMO borrowings, reverse repurchase agreements and borrowings secured by investment securities available-for-sale. Net interest income also includes (1) amortization of acquisition costs on mortgages acquired from the mortgage operations, (2) accretion of loan discounts, which represents the amount allocated to mortgage servicing rights when they are sold to third parties and mortgages are transferred to the long-term investment operations from the mortgage operations and retained for long-term investment, (3) amortization of CMO securitization expenses and, to a lesser extent, (4) amortization of CMO bond discounts. Net cash received or paid on derivatives is a component of gain (loss) on derivative instruments on the consolidated statement of operations and is not a component of net interest income.

Net interest income increased 94% to \$343.1 million for 2004 as compared to \$176.7 million for 2003. The year-over-year increase in net interest income of \$166.4 million was primarily due to a 114% increase in average mortgage assets to \$16.7 billion for 2004 as compared to \$7.8 billion for 2003 as the long-term investment operations acquired \$16.9 billion of mortgages from the mortgage operations in addition to \$458.5 million of multi-family mortgages originated by the long-term investment operations. Adjusted net interest margins on mortgage assets, as defined in the yield table below, declined by 19 basis points to 1.17% during 2004 as compared to 1.36% during 2003. The decrease in adjusted net interest margins on mortgage assets was primarily due to (1) an increase in short-term interest rates, (2) an increase in the amortization of loan premiums, securitization costs and bond discounts as a result of higher than expected mortgage prepayments and, to a lesser extent, (3) higher leverage and lower net interest margins on certain CMOs completed during the second half of 2004.

The following table summarizes average balance, interest and weighted average yield on mortgage assets and borrowings on mortgage assets for the periods indicated (dollars in thousands):

For the year ended December 31,

		2004			2003			2002	
	Average Balance	Interest	Yield	Average Balance	Interest	Yield	Average Balance	Interest	Yield
MORTGAGE ASSETS									
Subordinated securities collateralized by									
mortgages	\$ 27,937	\$ 3,764	13.47%	\$ 31,479	\$ 3,839	12.20%	\$ 35,139	\$ 1,864	5.30%
Mortgages held as CMO collateral (1)	14,283,347	618,771	4.33%	6,620,727	317,434	4.79%	3,383,906	200,742	5.93%
Mortgages held-for-investment and held-for-sale									
(2)	1,837,347	105,742	5.76%	633,474	34,580	5.46%	114,519	6,781	5.92%
Finance receivables (3)	510,899	25,018	4.90%	557,553	28,969	5.20%	341,543	19,383	5.68%
Total mortgage assets	\$16,659,530	\$753,295	4.52%	\$7,843,233	\$384,822	4.91%	\$3,875,107	\$228,770	5. 90%
<u>BORROWINGS</u>									
CMO borrowings	\$14,072,852	\$354,547	2.52%	\$6,445,968	\$174,199	2.70%	\$3,288,379	\$102,366	3.11%
Reverse repurchase agreements	2,175,728	57,837	2.66%	1,379,749	32,382	2.35%	814,248	23,583	2.90%
Borrowings secured by investment securities (4)		_		2,709	2,316	85.49%	10,037	1,852	18.45%
							-		
Total borrowings on mortgage assets	\$16,248,580	\$412,384	2.54%	\$7,828,426	\$208,897	2.67%	\$4,112,664	\$127,801	3.11%
Net Interest Spread (5)			1.98%			2.24%			2.79%
Net Interest Margin (6)			2.05%			2.24%			2.61%
Less: Accretion of loan discounts (7)		\$ 54,867	0.33%		\$ 21,101	0.27%		\$ 11,538	0.30%
Less: Net cash payments on derivatives (8)		\$ 91,882	0.55%		\$ 47,846	0.61%		\$ 28,362	0.73%
Adjusted Net Interest Margin (9)			1.17%			1.36%			1.58%
Effect of amortization of loan premiums and									
securitization costs (10)		\$166,649	1.00%		\$ 69,573	0.89%		\$ 38,094	0.98%

- (1) Interest includes amortization of acquisition cost on mortgages acquired from the mortgage operations and accretion of loan discounts, which represents the amount allocated to MSRs when MSRs are sold to third parties and mortgages are transferred from the mortgage operations to the long-term investment operations and retained for long-term investment.
- (2) Includes mortgages held-for-sale which were acquired via the consolidation of Impac Funding Corporation on July 1, 2003.
- (3) Total average borrowings on mortgage assets exceed total average mortgage assets during 2002 as we eliminated warehouse borrowings between the warehouse lending operations and mortgage operations prior to July 1, 2003.
- (4) Payments and excess cash flows received from investment securities collateralizing these borrowings were used to pay down the outstanding borrowings. The payments were received from a collateral base that was in excess of the borrowings. Therefore, while the payment amounts remained relatively stable, the average balance of the borrowings continued to decline. These borrowings were paid off during the third quarter of 2003 and the yield for 2003 reflects discount and securitization costs that were recorded as interest expense upon repayment of the borrowings.
- (5) Net interest spread on mortgage assets is calculated by subtracting the weighted average yield on total borrowings on mortgage assets from the weighted average yield on total mortgage assets.
- (6) Net interest margin on mortgage assets is calculated by subtracting interest expense on total borrowings on mortgage assets from interest income on total mortgage assets and then dividing by total mortgage assets.
- (7) Yield represents income from the accretion of loan discounts, which represents the amount allocated to MSRs when MSRs are sold to third parties and mortgages are transferred from the mortgage operations to the long-term investment operations and retained for long-term investment, divided by total average mortgage assets.
- (8) Yield represents net cash payments on derivatives divided by total average mortgage assets.
- (9) Adjusted net interest margin on mortgage assets is calculated by subtracting interest expense on total borrowings on mortgage assets, accretion of loan discounts and net cash payments on derivatives from interest income on total mortgage assets and dividing by total average mortgage assets. Net cash payments on derivatives is a component of gain (loss) on derivatives on the consolidated statements of operations. Adjusted net interest margins on mortgage assets is a non-GAAP financial measurement, however, the reconciliation provided in this table meets the requirements of Regulation G as promulgated by the SEC for the presentation of non-GAAP financial measurements. We believe that the presentation of adjusted net interest margin on mortgage assets is useful information for our investors as it more closely reflects the true economics of net interest margins on mortgage assets.
- (10) Yield represents cost of amortization of net loan premiums and securitization costs divided by total average mortgage assets.

During 2004 the Federal Reserve Bank raised short-term interest rates, which effected movements in one-month LIBOR, a total of 125 basis points. This caused borrowing costs on adjustable rate CMO borrowings, which are tied to one-month LIBOR and re-price monthly without limitation, to rise at a faster pace than coupons on LIBOR ARMs securing CMO borrowings, which generally re-price every six months with limitation. LIBOR ARMs held in our long-term investment portfolio are subject to the following interest rate risks:

- interest rate adjustment limitations on mortgages held for long-term investment due to periodic and lifetime interest rate cap features as compared to borrowings which are not subject to adjustment limitations;
- mismatched interest rate re-pricing periods between mortgages held for long-term investment, which generally re-price every six months, and borrowings, which re-price every month in regards to CMO borrowings and daily in regards to reverse repurchase agreements; and
- uneven and unequal movements in the interest rate indices used to re-price mortgages held for long-term investment, which are generally indexed to one-, three- and six-month LIBOR and one-year LIBOR, and borrowings, which are generally indexed to one-month LIBOR.

Along with an increase in short-term interest rates, our expectation, based on past experience, was that we would see a corresponding decline in mortgage prepayment rates. However, mortgage prepayment rates accelerated during the latter part of 2004. There is recent mortgage industry evidence that documents a substantial increase in home appreciation rates over the last three years has been a significant factor affecting prepayment patterns of Alt-A borrowers. Borrowers appear more than willing to use home equity to pay loan prepayment penalties in order to obtain lower monthly payments by refinancing into other mortgage products, such as interest-only and high loan-to-value mortgage products.

Actual prepayment rates in excess of projected future prepayment rates resulted in a cumulative upward adjustment in both the amortization rate and amortization amount of loan premiums, securitization costs and bond discounts during the fourth quarter of 2004. As such, amortization of loan premiums and securitization expenses increased by 11 basis points to 1.00% of average mortgage assets during 2004 as compared to 0.89% of average mortgage assets during 2003. A substantial portion of our long-term mortgage investment portfolio consists of mortgages with prepayment penalty features that are primarily designed to help minimize the rate of early mortgage prepayments. However, if mortgages do prepay, a prepayment penalty is charged which helps offset additional amortization of loan premiums and securitization costs. During 2004, prepayment penalties received from borrowers was recorded as interest income and increased the yield on average mortgage assets by 6 basis points. Therefore, prepayment penalty income offset the effect of increased amortization of loan premiums and securitization expenses due to higher than expected prepayments by approximately 45%.

Adjusted net interest margins were also affected by the following during 2004:

- our interest rate risk management policies do not provide 100% coverage of the principal amount outstanding on mortgages held for investment at any given time; and
- actual mortgage prepayments and the corresponding repayment of CMO borrowings exceeded the pre-determined amortization schedule of the notional amount of derivative instruments.

Our interest rate risk management policies are formulated with the intent to offset the potential adverse effects of changing interest rates primarily associated with cash flows on adjustable rate CMO borrowings. However, as a result of the combination of the factors listed above, the interest rate spread differential between ARMs and adjustable rate CMO borrowings compressed, which compressed net interest margins on mortgage assets. By design, our current interest rate risk management program provides 20% to 25% coverage of the outstanding principal balance of our LIBOR ARMs and 75% to 85% coverage of the outstanding principal balance of intermediate, or hybrid, ARMs at the point in time that we securitize the mortgages.

Additionally, we primarily acquire a certain notional amount of interest rate swap agreements which correspond to the balance of CMO borrowings at the time we securitize mortgages. The interest rate swap agreements are generally acquired with a pre-determined amortization schedule of the notional amount of the interest rate swap agreements and is based upon the past prepayment experience of our mortgages. However, actual prepayment of mortgages and the corresponding repayment of CMO borrowings exceeded the amortization schedule of the notional amount of the interest rate swap agreements which resulted in greater net cash payments on derivatives than we originally anticipated. Even so, as interest rates rose during 2004, derivative costs declined by 6 basis points to 55 basis points of average mortgage assets during 2004 as compared to 61 basis points during 2003 as total net cash payments on derivatives relative to average mortgage assets

declined. Total net cash payments on derivatives during 2004 were \$91.9 million on average mortgage assets of \$16.7 billion as compared to \$47.8 million on average mortgage assets of \$7.8 billion during 2003. Net cash payments on derivative instruments along with the change in unrealized fair value of derivatives comprise substantially all of the gain (loss) on derivative instruments on our statement of operations.

Non-Interest Expense

Total non-interest expense was \$108.7 million for 2004 as compared to \$48.8 million for 2003. The year-over-year increase in non-interest expense of \$59.9 million was primarily the result of the following:

- \$50.1 million increase in operating expenses, and
- \$10.6 million increase in amortization of deferred charge.

Operating expenses. Operating costs for 2003 include only six months of operating expenses from the mortgage operations that were consolidated on the consolidated financial statements as the mortgage operations was acquired by the Company on July 1, 2003. Therefore, if a full year of operating costs from the mortgage operations would have been recorded on the financial statements on a consolidated basis for 2003, the year-over-year percentage change in operating costs would be lower. Operating expenses from the mortgage operations during the first six months of 2003 are a component of the mortgage operations' net earnings and are reflected on the consolidated financial statements as equity in net earnings of Impac Funding Corporation. Operating expenses include personnel expense, general and administrative and other expense, professional services, equipment expense, occupancy expense and data processing expense.

Operating expense increased 117% to \$92.8 million during 2004 as compared to \$42.7 million for 2003 primarily due to (1) an 134% increase in originations and acquisitions from the mortgage operations during 2004 and (2) operating expenses for 2003 include the consolidation of operating expenses from the mortgage operations for only the last six months of 2003 as the mortgage operations was consolidated on July 1, 2003.

Operating costs rose by \$50.1 million, or 117%, primarily as acquisitions and originations from the mortgage operations increased 134% to \$22.2 billion for 2004 as compared to \$9.5 billion for 2003. The increase in mortgage acquisitions and originations resulted in the addition of personnel during 2004 which increased personnel expense by \$35.1 million, or 139%, to \$60.4 million during 2004 as compared to \$25.3 million during 2003. In addition, an increase in staffing caused an increase of \$9.4 million, or 123%, in general and administrative and other expense while occupancy expense increased to \$3.7 million, or 131%, during 2004 as compared to \$1.6 million during 2003. In order to accommodate expansion, we entered into premises leases for office space directly surrounding our main corporate facility in Newport Beach, California. The expansion of our operations within a geographically centralized area allows us to maintain our centralized operating approach as we are able to leverage technology and operational expertise from our main headquarters to the new facilities.

On a cost per loan basis, operating costs were lower during 2004 as compared to 2003 primarily as we acquired a larger percentage of mortgages on a bulk basis during 2004 as compared to the prior year. During 2004 the mortgage operations acquired \$8.5 billion, or 38% of total mortgage acquisitions and originations, of mortgages through bulk purchase transactions as compared to \$2.2 billion, or 23% of total mortgage acquisitions and originations, during 2003. Mortgages acquired on a bulk basis generally require less staffing and personnel-related costs than mortgages acquired on a flow basis. However, premiums paid for acquiring mortgages on a bulk basis are generally higher than premiums paid for the acquisition of a mortgage on a flow basis as the higher premium paid for bulk packages factors in operating costs incurred by the mortgage originator.

Amortization of deferred charge. A deferred charge was recorded to eliminate the income tax effect resulting from gains on inter-company mortgage sales, which represents the amount allocated to MSRs when MSRs are sold to third parties and mortgages are transferred from the mortgage operations to the long-term investment operations and retained for long-term investment. The deferred charge is amortized to expense over the expected life of the mortgages. Amortization of deferred charge was \$16.2 million during 2004 as compared to \$5.7 million during 2003. The year-over-year increase in the amortization of the deferred charge was the result of the acquisition of \$16.9 billion of mortgages by the long-term investment operations from the mortgage operations and the subsequent sale of MSRs to third parties during 2004 as compared to the acquisition of \$5.8 billion of mortgages by the long-term investment operations from the mortgage operations and the subsequent sale of MSRs to third parties during 2003.

Gain (Loss) on Derivative Instruments

During 2004 gain on derivative instruments was \$4.7 million as compared to a loss on derivative instruments of \$16.0 million during 2003. A component of gain on derivative instruments is unrealized mark-to-market gain on derivative instruments which increased to \$96.6 million during 2004 as compared to a gain of \$31.8 million for 2003. The increase in unrealized mark-to-market gain on derivative instruments was the result of changes in future expectations of short term rates which positively affected the value of our derivatives. We enter into derivative contracts to manage the various interest rate risks associated with cash flows on CMO and warehouse borrowings. In our consolidated financial statements, we record a market valuation adjustment for derivatives and any subsequent cash payments paid or received on these derivatives is recorded as a current period expense or income in gain (loss) on derivative instruments on the consolidated financial statements. The unrealized mark-to-market gain on derivative instruments was partially offset by an increase in cash payments of derivative instruments of \$91.9 million during 2004 as compared to \$47.8 million during 2003. In addition, we account for certain forward purchase commitments on mortgage loans as derivatives and record the change in fair value of such instruments and the related derivatives used for interest rate risk management on the mortgage pipeline as current period expense or income.

Income Taxes

Income tax benefit increased to \$13.5 million during 2004 as compared to \$1.4 million during 2003 primarily due to an increase in operating losses at IFC when profits on inter-company loan sales are eliminated from IFC's net earnings. IFC is a taxable REIT subsidiary (TRS) and is therefore subject to corporate income taxes. We record a deferred tax charge to eliminate the expense recognition of income taxes paid on inter-company profits that result from the sale of mortgages from IFC to the long-term operations. The amortization of the deferred tax charge is recorded in other expense rather than income tax expense.

Results of Operations by Business Segment

We operate three core businesses:

- · the long-term investment operations;
- · the mortgage operations; and
- · the warehouse lending operations.

Long-Term Investment Operations. The long-term investment operations generate earnings primarily from net interest income earned on mortgages held as CMO collateral and mortgages held for investment. Net earnings from the long-term operations increased by \$109.9 million to \$215.3 million for 2004 as compared to \$105.4 million for 2003 primarily due to the following:

- \$101.4 million increase in net interest income; and
- \$14.7 million increase in non-interest income.

Net interest income. Net interest income increased 78% to \$231.9 million for 2004 as compared to \$130.5 million for 2003 primarily due to an increase in total average mortgage assets as the long-term investment operations acquired \$16.9 billion of mortgages from the mortgage operations and originated \$458.5 million of multi-family mortgages. The acquisition and origination of mortgages by the long-term investment operations was primarily financed by the securitization of \$17.7 billion of CMOs. The adjusted net interest margin on mortgages held as CMO collateral declined 30 basis points to 0.82% during 2004 as compared to 1.12% during 2003. The decline in adjusted net interest margin was primarily due to (1) an increase in short-term interest rates, (2) an increase in the amortization of loan premiums, securitization costs and bond discounts as a result of higher than expected mortgage prepayments and, to a lesser extent, (3) higher leverage and lower net interest margins on certain CMOs completed during the second half of 2004, as previously discussed. Adjusted net interest margin on mortgages held as CMO collateral is calculated by subtracting interest expense on CMO borrowings, accretion of loan discounts and cost of derivatives from interest income on mortgages held as CMO collateral and dividing by average mortgages held as CMO collateral in the yield table above.

Non-interest income. Non-interest income rose by \$14.7 million to \$16.3 million during 2004 as compared to \$1.6 million during 2003 primarily due to an increase in gain (loss) on derivative instruments as future expectations of higher one-month LIBOR rates positively affected the value of derivatives. Gain (loss) on derivative instruments increased to a gain of \$11.8 million during 2004 as compared to a loss of \$9.1 million during 2003.

Refer to Note I. "Segment Reporting" in the notes to consolidated financial statements for financial results of the long-term investment operations and see Item 1. "Business—Long-Term Investment Operations" for additional detail regarding the operating structure of the long-term investment operations.

Mortgage Operations. The mortgage operations generate income by securitizing and selling mortgages to permanent investors, including the long-term investment operations and, to a lesser extent, it earns revenue from fees associated with mortgage servicing rights, master servicing agreements and interest income earned on mortgages held for sale. Net earnings from the mortgage operations for 2004 include twelve months of results of operations, however, prior to the consolidation of the mortgage operation on July 1, 2003, its results of operations were reflected on a consolidated basis for 2003 for the consolidation period, see "Consolidation of IFC."

Net earnings from the mortgage operations increased by \$26.0 million to \$42.9 million during 2004 as compared to net earnings of \$16.9 million during the consolidation period. For a full year comparison, net earnings for the mortgage operations were \$42.9 million during 2004 as compared to \$35.4 million on a non-consolidated basis during 2003. The increase in net earnings was primarily due to an increase of \$73.8 million in non-interest income which was partially offset by a \$54.2 million increase in non-interest expense.

Non-interest income increased 129% during 2004 primarily due to a higher volume of mortgages sold to the long-term investment operations and third party investors due a higher volume of mortgages that were acquired and originated during 2004 as compared to 2003. As a result of an increase in gain on sale of loans, non-interest income increased to \$131.0 million during 2004 as compared to \$57.2 million during 2003. Gain on sale of loans includes the difference between the price at which we acquire or originate mortgages and the price we receive upon the sale or securitization of mortgages plus or minus direct mortgage origination revenue and costs, i.e. loan and underwriting fees, commissions, appraisal review fees and document processing expenses. Gain on sale of loans acquired or originated by the mortgage operations also includes a premium for the sale of mortgage servicing rights upon the sale or securitization of mortgages, including REMICs and CMOs. Substantially all mortgages sold or securitized during 2004 and 2003 were done so on a servicing released basis, which resulted in substantially all cash gains. In order to minimize risks associated with the accumulation of our mortgages, we seek to securitize or sell mortgages monthly thereby reducing our exposure to interest rate risk and price volatility during the accumulation period of mortgages.

Partially offsetting the increase in non-interest income was an increase in non-interest expense which increased 112% to \$102.8 million during 2004 as compared to \$48.6 million as mortgage acquisitions and originations rose by 134% to \$22.2 billion for 2004 as compared to \$9.5 billion for 2003. The increase in mortgage acquisitions and originations resulted in the addition of personnel by the mortgage operations and a corresponding increase in operating costs. In order to accommodate expansion, we entered into premises leases for office space directly surrounding our main corporate facility in Newport Beach, California. The expansion of our operations within a geographically centralized area allows us to maintain our centralized operating approach as we are able to leverage technology and operational expertise from our main headquarters to the new facilities.

Refer to Note I. "Segment Reporting" in the notes to consolidated financial statements for financial results of the mortage operations and see Item 1. "Business—Long-Term Investment Operations" for additional detail regarding the operating structure of the mortgage operations.

Warehouse Lending Operations. The warehouse lending operations primarily generates net earnings from net interest income earned from the difference between its cost of borrowings and the interest earned on warehouse advances and, to a lesser extent, fees from warehouse lending transactions. The warehouse lending operations provide warehouse financing to affiliated companies, including the mortgage operations and long-term investment operations, and to approved non-affiliated clients some of which are correspondents of the mortgage operations.

Net earnings from the warehouse lending operations were \$43.4 million for 2004 as compared to \$27.5 million for 2003. The increase in net earnings of \$15.9 million was primarily due to a \$16.8 million increase in net interest income to \$45.8 million during 2004 as compared to \$29.0 million during 2003. Net interest income rose 58% on year-over-year basis as total average finance receivables rose 64% to \$2.3 billion during 2004 as compared to \$1.4 billion during 2003.

Net earnings were negatively impacted during 2004 as the warehouse lending operations added \$6.1 million to loan loss provisions during 2004 as fraudulent warehouse advances were discovered in 2004 which were determined to be impaired. By year-end 2004, the warehouse lending operations had a specific allowance for loan losses of \$10.7 million for impaired warehouse advances. For calculation of estimated taxable income, deductions for permanently impaired mortgages were taken as a deduction to estimated taxable income for 2004

Refer to Note I. "Segment Reporting" in the notes to consolidated financial statements for financial results of the warehouse lending operations and see Item 1. "Business—Long-Term Investment Operations" for additional detail regarding the operating structure of the warehouse lending operations.

Results of Operations

For the Year Ended December 31, 2003 compared to the Year Ended December 31, 2002

Net earnings were \$149.0 million, or \$2.88 per diluted common share, as compared to \$40.3 million, or \$0.99 per diluted common share, for 2002. The year-over-year increase in net earnings of \$108.7 million was primarily due to the following:

- \$74.2 million increase in net interest income; and
- \$34.5 million decrease in loss on derivative instruments.

These variances are discussed in further detail below.

Net Interest Income

Net interest income increased \$74.2 million to \$176.7 million for 2003 as compared to \$102.5 million for 2002. The year-over-year increase in net interest income was primarily due to an increase in average mortgage assets, which increased to \$7.8 billion for 2003 as compared to \$3.9 billion for 2002 as the long-term investment operations retained \$5.8 billion of primarily Alt-A mortgages acquired and originated by the mortgage operations and originated \$290.5 million of multi-family mortgages.

Adjusted net interest margins on mortgage assets declined 22 basis points to 1.36% during 2003 as compared to 1.58% during 2002. The decrease in adjusted net interest margins on mortgage assets was primarily due interest rate adjustments on mortgage assets as follows:

- at the time mortgages are retained for long-term investment, ARMs have initial coupons above the fully-indexed rate, which is the margin on the mortgage plus the index. Upon the mortgage interest rate reset date, interest rates on ARMs adjust downward to the fully-indexed rate. The reset to lower coupons on these mortgages during low interest rate environments, which was the case during 2003, is consistent with our strategy to mitigate the borrower's propensity to refinance their mortgage, which we believe will extend the duration of the mortgages;
- hybrid ARMs acquired for long-term investment during 2001 with initial fixed interest rate periods of two years subsequently converted to ARMs, which resulted in those mortgages resetting to lower current interest rates during 2003; and
- the decline in interest rates during 2003 led to record refinancing activity, which in turn led to increased mortgage prepayments and resulted in
 mortgages with coupons above current market interest rates were repaid and replaced with mortgages at lower yielding current market rates.

Our interest rate risk management policies are formulated with the intent to offset the potential adverse effects of changing interest rates on cash flows on adjustable rate CMO borrowings. However, as a result of the combination of the factors listed above, the interest rate spread differential between ARMs and adjustable rate CMO borrowings compressed, which compressed net interest margins on mortgage assets. Net margin compression on mortgage assets was partially offset by a steep yield curve during 2003, which further steepened during the second half of 2003. A steep yield curve generally benefits net interest margins on our mortgage assets as coupons on mortgages are based, to some extent, on yields at the long end of the yield curve while our borrowing costs are indexed to yields at the short end of the yield curve, primarily one-month LIBOR.

The 22 basis point decline in adjusted net interest margins on mortgage assets was partially offset by the consolidation of the mortgage operations. The consolidation of the mortgage operations resulted in an increase in net interest margins on mortgage assets during 2003 as mortgages held-for-sale were consolidated with total average mortgage assets during the second half of 2003. During 2003 mortgages held-for-sale were not consolidated with total average mortgage assets as IFC was a non-consolidated entity prior to July 1, 2003. In addition, net interest margins on mortgages held-for-sale were favorable during 2003 due to a steep yield curve as compared to 2002. The yield curve represents the mathematical difference

between short-term interest rates and long-term interest rates. Because the mortgage operations establishes interest rates on its mortgages by indexing those interest rates to long-term market interest rates and finances mortgages with borrowings that are indexed to short-term interest rates, a steep yield curve benefits net interest margins.

Gain (Loss) on Derivative Instruments

During 2003 loss on derivative instruments decreased to \$16.0 million as compared to \$50.5 million for 2002. The decrease in loss on derivative instruments was the result of changes in future expectations of short-term rates which affected the value of our derivatives. We enter into derivative contracts to manage the various interest rate risks associated with cash flows on CMO and warehouse borrowings. In our consolidated financial statements, we now record a market valuation adjustment for economic hedges and any subsequent cash payments paid or received on these derivatives as a current period expense or income in loss on derivative instruments on the consolidated financial statements. In addition, we account for certain forward purchase commitments on mortgage loans as free-standing derivatives, and record the change in fair value of such instruments and the related derivatives used to hedge the mortgage pipeline as a current period expense or income.

Results of Operations by Business Segment

Long-Term Investment Operations. Net earnings from the long-term operations was \$105.4 million for 2003 as compared to \$17.2 million for 2002. The increase in net earnings of \$88.2 million was primarily due to a decrease in loss on derivative instruments and an increase in net interest income. Loss on derivative instruments declined by \$41.4 million to a loss of \$9.1 million for 2003 as compared to a loss of \$50.5 million for 2002. Loss on derivative instruments declined as future expectations of short-term rates positively affected the value of derivatives. Net interest income rose by \$39.4 million to \$130.5 million for 2003 as compared to \$91.1 million for 2002 primarily due to an increase in total average mortgage assets. The long-term investment operations retained \$6.1 billion of Alt-A and multi-family mortgages during 2003 which in turn led to an increase in average CMO collateral and mortgages held-for-investment as shown in the comparative yield table above. Refer to Note I. "Segment Reporting" in the notes to consolidated financial statements for additional detail. See Item 1. "Business—Long-Term Investment Operations" and Note I. "Segment Reporting" in the notes to consolidated financial statements for additional detail regarding the operating structure and financial results of the long-term investment operations.

Mortgage Operations. Net earnings from the mortgage operations include its results of operations for the consolidation period on a consolidated basis. See "Consolidation of IFC" above. Prior to the consolidation of IFC, its results of operations were included in equity in net earnings of IFC on the consolidated financial statements. Equity in net earnings of IFC is reflected in Inter-company for purposes of segment reporting as shown in Note I. "Segment Reporting" in the notes to consolidated financial statements. As such, net earnings from the mortgage operations was \$16.9 million during the consolidation period while equity in net earnings of IFC during the non-consolidation period was \$11.5 million for 2003 as compared to \$11.3 million for 2002. Net earnings from the mortgage operations were primarily earned from the sale of mortgage loans and the corresponding gains from the sale of those loans. See Item 1. "Business—Mortgage Operations" and Note I. "Segment Reporting" in the notes to consolidated financial statements for additional detail regarding the operating structure and financial results of the mortgage operations.

Warehouse Lending Operations. Net earnings from the warehouse lending operations was \$27.5 million for 2003 as compared to \$18.5 million for 2002. The increase in net earnings of \$9.0 million was primarily due to an increase in net interest income. Net interest income rose during 2003 as total average finance receivables rose 65% to \$1.4 billion as compared to \$850.1 million during 2002 as shown in the comparative yield table above. See Item 1. "Business—Warehouse Lending Operations" and Note I. "Segment Reporting" in the notes to consolidated financial statements for additional detail regarding the operating structure and financial results of the warehouse lending operations.

Financial Condition

Total assets grew 125% to \$23.8 billion as of December 31, 2004 as compared to \$10.6 billion as of prior year-end as the long-term investment operations retained \$16.9 billion of primarily Alt-A mortgages and originated \$458.5 million of multi-family mortgages. The retention of Alt-A and multi-family mortgages increased the long-term mortgage portfolio to \$21.9 billion as of December 31, 2004 as compared to \$9.3 billion as of prior year-end. The acquisition and origination of mortgages were primarily financed through the issuance of \$17.7 billion of CMOs and net proceeds of \$366.3 million in new common equity and net proceeds of \$152.2 million in new preferred equity. The following table presents selected financial data for the periods indicated (dollars in thousands, except per share data):

	As of and fo	As of and for the year ended December 31,					
	2004	2003	2002				
Book value per share	\$ 11.80	\$ 8.39	\$ 6.26				
Return on average assets	2.02%	1.80%	0.92%				
Return on average equity	47.50%	41.59%	16.22%				
Assets to equity ratio	22.81:1	22.35:1	23.07:1				
Debt to equity ratio	21.77:1	21.28:1	21.85:1				
Allowance for loan losses as a percentage of loans provided for	0.29%	0.39%	0.46%				
Prior 12-month constant prepayment rate, or "CPR"	29%	28%	25%				
Total non-performing assets	\$ 259,695	\$ 140,369	\$ 130,614				
Total non-performing assets to total assets	1.09%	1.33%	2.00%				
Mortgages owned 60+ days delinquent	\$ 381,290	\$ 175,313	\$ 161,260				
60+ day delinquency of mortgages owned	1.74%	1.79%	3.22%				

We believe that in order for us to generate positive cash flows and net earnings from our long-term mortgage portfolio we must successfully manage the following primary operational and market risks:

- · credit risk;
- prepayment risk;
- · liquidity risk; and
- · interest rate risk.

Credit Risk. We manage credit risk by primarily retaining high credit quality Alt-A mortgages and, to a lesser extent, multi-family mortgages, adequately providing for loan losses and actively managing delinquencies and defaults. We believe that by improving the overall credit quality of our long-term mortgage portfolio we can consistently generate stable future cash flow and net earnings. During 2004 we retained primarily Alt-A mortgages with an original weighted average credit score of 695 and an original weighted average LTV ratio of 76%. Alt-A mortgages are primarily first lien mortgages made to borrowers whose credit is generally within typical Fannie Mae and Freddie Mac guidelines, but that have loan characteristics that make them non-conforming under those guidelines. We primarily acquire non-conforming "A" or "A-" credit quality mortgages, collectively, Alt-A mortgages. As of December 31, 2004, the original weighted average credit score of mortgages held as CMO collateral was 696 and the original weighted average LTV ratio was 76%. For additional information regarding the long-term mortgage portfolio refer to Item 1. "—Long-Term Mortgage Portfolio," "Note C—CMO Collateral" and "Note D—Mortgages Held-for-Investment" in the accompanying notes to the consolidated financial statements.

In addition to retaining mortgages acquired and originated by our mortgage operations, the long-term investment operations originated \$458.5 million of multi-family mortgages through IMCC. IMCC was formed to primarily originate small balance, multi-family mortgages of high credit quality. IMCC primarily originates hybrid ARMs with balances generally ranging from \$500,000 to \$5.0 million. Multi-family mortgages provide greater asset diversification on our balance sheet as multi-family mortgages typically have longer lives than residential mortgages. All multi-family mortgages originated during 2004 had interest rate floors with prepayment penalty periods ranging from three to ten years.

We believe that we have adequately provided for loan losses as allowance for loan losses increased to \$64.0 million as of December 31, 2004 as compared to \$38.6 million as of prior year-end as we retained \$16.9 billion of mortgages from the mortgage operations and originated \$458.5 million of multi-family mortgages for long-term investment. Actual loan charge-offs, net of recoveries, on mortgages held for long-term investment decreased to \$5.6 million for 2004 as compared to \$12.9 million for 2003. Additionally, we provided specific loan loss allowances of \$10.7 million for impaired warehouse advances as previously discussed.

We monitor our servicers and sub-servicers to make sure that they perform loss mitigation, foreclosure and collection functions according to our servicing guidelines. This includes an effective and aggressive collection effort in order to minimize mortgages from becoming non-performing assets. However, when resolving issues related to non-performing assets, including potential disposition, servicers and sub-servicers are required to take timely and aggressive action. Servicers and sub-servicers are required to determine payment collection under various circumstances, which will result in maximum financial benefit. This is accomplished by either working with the borrower to bring the mortgage current or by foreclosing and liquidating the property. We perform ongoing review of mortgages that display weaknesses and believe that we maintain adequate loss allowance on our mortgages. When a borrower fails to make required payments on a mortgage and does not cure the delinquency within 60 days, we generally record a notice of default and commence foreclosure proceedings. If the mortgage is not reinstated within the time permitted by law for reinstatement, the property may then be sold at a foreclosure

sale. In foreclosure sales, we generally acquire title to the property. As of December 31, 2004, mortgages that we owned included 1.74% of mortgages that were 60 days or more delinquent as compared to 1.79% as of year-end 2003 and 3.22% as of year-end 2002. The following table summarizes mortgages that we own, including mortgages held for long-term investment and mortgages held-for-sale, that were 60 or more days delinquent for the periods indicated (in thousands):

	F	As of December 31,			
	2004	2003	2002		
60-89 days delinquent	\$139,872	\$ 51,173	\$ 41,762		
90 or more days delinquent	68,877	52,080	33,822		
Foreclosures	157,867	66,767	74,597		
Delinquent bankruptcies	14,674	5,293	11,079		
Total 60 or more days delinquent	\$381,290	\$175,313	\$161,260		

Non-performing assets consist of mortgages that are 90 days or more delinquent, including loans in foreclosure and delinquent bankruptcies. It is our policy to place a mortgage that is categorized as held for investment on our financial statements on non-accrual status when it becomes 90 days delinquent and any previously accrued interest will be reversed from revenue. When real estate is acquired in settlement of loans, or other real estate owned, the mortgage is writtendown to a percentage of the property's appraised value or broker's price opinion less anticipated selling costs. As of year-end 2004, non-performing assets as a percentage of total assets was 1.09% compared to 1.33% as of year-end 2003 and 2.00% as of year-end 2002. The following table summarizes mortgages that we own, including mortgages held for long-term investment and mortgages held-for-sale, that were non-performing for the periods indicated (in thousands):

	P	As of December 31,				
	2004	2003	2002			
90 or more days delinquent, foreclosures and delinquent bankruptcies	\$241,418	\$124,140	\$119,498			
Other real estate owned	18,277	16,229	11,116			
Total non-performing assets	\$259,695	\$140,369	\$130,614			

Prepayment Risk. During 2004, 73% of Alt-A mortgages acquired by the long-term investment operations from the mortgage operations 2004 had prepayment penalty features ranging from six-months to five years and, as of December 31, 2004, 76% of mortgages held as CMO collateral had prepayment penalties. As of December 31, 2004, the twelve-month CPR of mortgages held as CMO collateral was 29% as compared to a 28% twelve-month CPR as of December 31, 2003 and a 25% twelve-month CPR as of December 31, 2002. CPR increased during 2004 as compared to 2003 even as short-term interest rates rose 125 basis points and resulted in an increase in amortization of premiums during the fourth quarter of 2004. Prepayment penalties are charged to borrowers for mortgages that are repaid early and recorded as interest income on our consolidated financial statements. Interest income from prepayment penalties helps offset additional amortization of loan premiums and securitization costs. During 2004 prepayment penalties received from borrowers was recorded as interest income and increased the yield on average mortgage assets by 6 basis points, as previously discussed, and offset the effect of increased amortization of premiums and securitization costs by approximately 45%.

Liquidity Risk. We employ a leveraging strategy to increase assets by financing our long-term mortgage portfolio primarily with CMO borrowings, reverse repurchase agreements and capital and then using cash proceeds to acquire additional mortgage assets. We retain ARMs and FRMs that are acquired and originated from the mortgage operations and finance the acquisition of those mortgages, during this accumulation period, with reverse repurchase agreements. After accumulating a pool of mortgages, generally between \$200 million and \$2.5 billion, we securitize the mortgages in the form of CMOs. Our strategy is to securitize our mortgages every 15 to 45 days in order to reduce the accumulation period that mortgages are outstanding on short-term warehouse or reverse repurchase facilities, which reduces our exposure to margin calls on these facilities. CMO borrowings are classes of bonds that are sold to investors of mortgage-backed securities and as such are not subject to margin calls. In addition, CMOs generally require a smaller initial cash investment as a percentage of mortgages financed than does interim warehouse and reverse repurchase financing. For additional information regarding financing refer to Item 1. "—Financing."

Because of the historically favorable loss rates of our Alt-A mortgages, we have received favorable credit ratings on our CMO borrowings from credit rating agencies, which has increased the percentage of bonds issued and reduced our required initial capital investment. The ratio of total assets to total equity, or "leverage ratio," was 22.81 to 1 as of December 31, 2004 as compared to 22.35 to 1 as of prior year-end. This use of leverage at these historical levels allows us to grow our balance sheet by efficiently using available capital. We continually monitor our leverage ratio and liquidity levels to insure that we are adequately protected against adverse changes in market conditions. For additional information regarding liquidity refer to "—Liquidity and Capital Resources."

Interest Rate Risk. Refer to Item 7A. "Quantitative and Qualitative Disclosures About Market Risk."

Liquidity and Capital Resources

We recognize the need to have funds available for our operating businesses and our customer's demands for obtaining short-term warehouse financing until the settlement or sale of mortgages with us or with other investors. It is our policy to have adequate liquidity at all times to cover normal cyclical swings in funding availability and mortgage demand and to allow us to meet abnormal and unexpected funding requirements. We plan to meet liquidity through normal operations with the goal of avoiding unplanned sales of assets or emergency borrowing of funds. Toward this goal, ALCO is responsible for monitoring our liquidity position and funding needs.

ALCO is comprised of senior executives of the mortgage operations and warehouse lending operations. ALCO meets on a weekly basis to review current and projected sources and uses of funds. ALCO monitors the composition of the balance sheet for changes in the liquidity of our assets. Our primary liquidity consists of cash and cash equivalents and maturing mortgages, or "liquid assets."

We believe that current cash balances, currently available financing facilities, capital raising capabilities and excess cash flows generated from our long-term mortgage portfolio will adequately provide for projected funding needs and asset growth. However, if we are unable to raise capital in the future, we may not be able to grow as planned. Refer to Item 1. "Business—Risk Factors" for additional information regarding risks that could adversely affect our liquidity.

Our operating businesses primarily use available funds as follows:

- acquisition and origination of mortgages,
- investment in mortgages;
- provide short-term warehouse advances to affiliates and non-affiliates; and
- pay common and preferred stock dividends.

Acquisition and origination of mortgages. During 2004 the mortgage operations acquired \$22.2 billion of primarily Alt-A mortgages, of which \$16.9 billion was retained for long-term investment. Capital invested in mortgages is outstanding until we sell or securitize mortgages, which is one of the reasons we attempt to sell or securitize mortgages frequently. Initial capital invested in mortgages includes premiums paid when mortgages are acquired and originated and initial capital investment, or "haircut," required upon financing, which is generally determined by the type of collateral provided. The mortgage operations paid weighted average premiums of 2.29% on the principal balance of Alt-A mortgages acquired during 2004, which were financed with warehouse borrowings at a haircut generally between 2% to 10% of the outstanding principal balance of the mortgages.

When we finance mortgages with long-term CMO borrowings, we repay short-term warehouse financing and recoup our 2% to 10% haircut. Then, depending on credit ratings from national credit rating agencies on our CMOs, we are generally required to provide an initial capital investment of 0.50% to 1% of the principal balance of mortgages securing CMO financing. Therefore, our initial total capital investment in CMOs range from approximately 3% to 5% of the principal balance of mortgages, depending on premiums paid upon acquisition of mortgages, costs paid for completion of CMOs, costs to acquire derivatives and initial capital investment in CMOs required to achieve desired credit ratings.

In addition, the long-term investment operations originated \$458.5 million of multi-family mortgages which were initially financed with short-term warehouse financing that generally requires a 3% to 10% haircut. Multi-family mortgages are either sold as whole loan sales or are financed with CMO borrowings at lower haircut requirements than warehouse financing.

Provide short-term warehouse advances to affiliates and non-affiliates. We utilize uncommitted and committed warehouse facilities with various lenders to provide short-term warehouse financing to affiliates and non-affiliated clients of the warehouse lending operations.

The warehouse lending operations provides short-term financing to non-affiliated clients from the closing of mortgages to their sale or other settlement with investors, including the mortgage operations. The warehouse lending operations generally finances between 95% and 98% of the fair market value of mortgages, which equates to a haircut requirement of between 2% and 5%, at one-month LIBOR plus a spread. As of December 31, 2004, the warehouse lending operations had \$738.7 million of approved warehouse lines available to non-affiliated clients, of which \$471.8 million was outstanding.

Our ability to meet liquidity requirements and the financing needs of our customers is subject to the renewal of our credit and repurchase facilities or obtaining other sources of financing, if required, including additional debt or equity from time to time. Any decision our lenders or investors make to provide available financing to us in the future will depend upon a number of factors, including:

- our compliance with the terms of our existing credit arrangements;
- our ability to provide audited financial statements;
- our financial performance;
- industry and market trends in our various businesses;
- the general availability of and rates applicable to financing and investments;
- our lenders or investors resources and policies concerning loans and investments; and
- · the relative attractiveness of alternative investment or lending opportunities.

Pay common and preferred stock dividends. We paid common stock dividends of \$202.7 million and preferred stock dividends of \$3.8 million during 2004, which we generated from our operating activities. We are required to distribute a minimum of 90% of our taxable income to our common stockholders in order to maintain our REIT status. Because we pay dividends based on taxable income, dividends may be more or less than net earnings. We paid total regular cash dividends of \$2.90 per common share in 2004 which met taxable income distribution requirements for the year. See "Issuance of Preferred Stock" for a discussion of the terms of our outstanding series of preferred stock.

Because the mortgage operations is seeking to retain earnings to fund future growth of our mortgage operations business, the board of directors may decide that the mortgage operations should decrease or cease making dividend distributions in the future. This would materially reduce the amount of our taxable income and in turn, would reduce the amount we would be required to distribute as dividends.

For additional information regarding dividend distribution requirements refer to Item 5. "Market for Registrant's Common Equity and Related Stockholder Matters" and "Note A—Summary of Business and Significant Accounting Policies" in the accompanying notes to the consolidated financial statements.

Our operating businesses are primarily funded as follows:

- CMO borrowings and reverse repurchase agreements;
- excess cash flows from our long-term mortgage portfolio;
- sale and securitization of mortgages; and
- cash proceeds from the issuance of securities.

CMO borrowings and reverse repurchase agreements. We use CMO borrowings and reverse repurchase agreements to fund substantially all of our warehouse financing to affiliates and non-affiliated clients and the acquisition and origination of mortgages for long-term investment. As we accumulate mortgages, we finance the acquisition of mortgages primarily with reverse repurchase agreements with third party lenders. We primarily use uncommitted and, to a lesser extent, committed repurchase facilities with major investment banks to finance substantially all warehouse financing, as needed. During 2004 we added \$1.1 billion of new uncommitted and committed warehouse facilities to finance the acquisition and origination of mortgages. The new warehouse facilities provide us with a higher aggregate credit limit to fund the acquisition and

origination of mortgages at terms comparable to those we have received in the past. These reverse repurchase agreements may have certain covenant tests which we continue to satisfy. Furthermore, our agreements require that we provide audited financial statements within 90 to 120 days of year end. We have received from our lenders waivers of this provision and, in some cases, extensions of the delivery deadlines. However, if we are unable to provide audited financial statements during any extended deadline or fail to obtain any further waivers from our lenders, then we will be in default under those applicable agreements and repayments of our borrowings may be immediately accelerated.

As of December 31, 2004, the warehouse lending operations had a total of \$3.4 billion of reverse repurchase facilities. Committed facilities comprised \$525.0 million of the total available facilities with uncommitted facilities totaling \$2.8 billion. In addition, from time to time during 2004, we utilized term repurchase financing provided to us by an underwriter of some of our securitizations at interest rates similar to our other repurchase facilities. These transactions allow us to remove mortgages from our current reverse repurchase facilities for periods immediately prior to the settlement of securitizations and at rates that are at times more favorable than on our reverse repurchase facilities.

We expect to continue to use short-term warehouse facilities to fund the acquisition and origination of mortgages. If we cannot renew or replace maturing borrowings, we may be forced to sell, on a whole loan basis, the mortgages securing these facilities, which, depending upon market conditions may result in substantial losses. Additionally, if for any reason the market value of our mortgages securing warehouse facilities decline, our lenders may require us to provide them with additional equity or collateral to secure our borrowings, which may require us to sell mortgages at substantial losses.

In order to mitigate the liquidity risk associated with reverse repurchase agreements, we attempt to sell or securitize our mortgages between 15 to 45 days from acquisition or origination. Although securitizing mortgages more frequently adds operating and securitization costs, we believe the added cost is offset as liquidity is provided more frequently with less interest rate and price volatility, as the accumulation and holding period of mortgages is shortened. When we have accumulated a sufficient amount of mortgages, generally between \$200 million and \$2.5 billion, we create CMOs and convert short-term advances under reverse repurchase agreements to long-term CMO borrowings. The use of CMO borrowings provides the following benefits:

- allows us to lock in our financing cost over the life of the mortgages securing the CMO borrowings; and
- eliminates margin calls on the borrowings that are converted from reverse repurchase agreements to CMO borrowings.

During 2004, we completed \$17.7 billion of CMOs, of which \$17.1 billion were adjustable rate CMO borrowings and \$599.0 million were fixed rate CMO borrowings to provide long-term financing for the retention of \$16.9 billion of primarily Alt-A mortgages and the origination of \$458.5 million of multi-family mortgages. Additionally, restricted cash of \$253.4 million, which primarily consists of cash deposits in a CMO securitization trust, was subsequently used to finance the remaining mortgage loan collateral. Because of the credit profile, historical loss performance and prepayment characteristics of our Alt-A and multi-family mortgages, we have been able to borrow a higher percentage against the principal balance of mortgages held as CMO collateral, which means that we have to provide less investment upon initial securitization of our mortgage loans in the form CMOs. Capital investment in the CMOs is established at the time CMOs are issued at levels sufficient to achieve desired credit ratings on the securities from credit rating agencies. Total credit loss exposure is limited to the capital invested in the CMOs at any point in time.

For additional information regarding reverse repurchase agreements and CMO borrowings refer to Item 1. "—Financing," "Note G—Reverse Repurchase Agreements" and "Note H—CMO Borrowings" in the accompanying notes to the consolidated financial statements.

Excess cash flows from our long-term mortgage portfolio. We receive excess cash flows on mortgages held as CMO collateral after distributions are made to investors on CMO borrowings to the extent cash or other collateral required to maintain desired credit ratings on the CMOs is fulfilled and can be used to provide funding for some of the long-term investment operations' activities. Excess cash flows represent the difference between principal and interest payments on the underlying mortgages less the following:

- interest paid on CMO borrowings;
- pro-rata early principal prepayments paid on CMO borrowings;
- servicing and master servicing fees paid;

- · premiums paid to mortgage insurers; and
- actual losses, net of any gains incurred upon disposition of other real estate owned or acquired in settlement of defaulted mortgages.

Sale and securitization of mortgages. When the mortgage operations accumulate a sufficient amount of mortgages, generally between \$100 million and \$300 million, that are intended to be deposited into a CMO, it sells the mortgages to the long-term investment operations. When selling mortgages on a whole loan basis, the mortgage operations will accumulate generally between \$50 million and \$200 million of mortgages and enter into sales transactions with third party investors on a monthly basis. When the mortgage operations enter into a REMIC securitization, it generally accumulates between \$200 million and \$2.0 billion of mortgages and sells these loans on a monthly basis.

The mortgage operations sold \$16.9 billion of mortgages to the long-term investment operations during 2004, sold \$3.7 billion of mortgages as whole loan sales and \$1.6 billion was securitized as REMICs. The mortgage operations sold mortgage servicing rights on substantially all mortgages sold in 2004. The sale of mortgage servicing rights generate cash gains which can be used to acquire and originate additional mortgage assets.

In order to mitigate interest rate and market risk, the mortgage operations attempts to sell and securitize mortgages between 15 and 45 days from acquisition and origination. Since we rely significantly upon sales and securitizations to generate cash proceeds to repay reverse repurchase borrowings and to create credit availability, any disruption in our ability to complete sales and securitizations may require us to utilize other sources of financing, which, if available at all, may be on unfavorable terms. In addition, delays in closing sales and securitizations of our mortgages increase our risk by exposing us to credit and interest rate risk for this extended period of time.

Cash proceeds from the issuance of securities. We raised \$366.3 million in net proceeds from the issuance of 18.4 million new shares of common stock and \$152.2 million in net proceeds from the issuance of 6.3 million new shares of preferred stock during 2004.

Cash Flows

Operating Activities - Net cash (used in) provided by operating activities was \$(179.4) million for 2004 as compared to \$166.1 million for 2003 and \$(191.4) million for 2002. For 2004, net change in restricted cash of \$(253.0) million and purchase of mortgages, net of loan sales, of \$(175.2) million were primarily used in operating activities. Funds used in operating activities during 2004 was partially offset by net earnings of \$257.6 million. For 2003, loan sales, net of loan purchases, of \$88.3 million and net earnings of \$149.0 million were provided by operating activities. For 2002, net change in investment in and advances to IFC of \$(301.8) million were used in operating activities which was partially offset by net earnings of \$40.3 million.

Investing Activities - Net cash used in investing activities was \$12.6 billion for 2004 as compared to \$4.0 billion for 2003 and \$3.4 billion for 2002. For 2004, 2003 and 2002, net cash of \$12.8 billion, \$4.1 billion and \$3.0 billion, respectively, was used in investing activities to acquire mortgages, net of principal repayments, for long-term investment.

Financing Activities - Net cash provided by financing activities was \$13.0 billion for 2004 as compared to \$3.9 billion for 2003 and \$3.6 billion for 2002. For 2004, 2003 and 2002, net cash flows of \$12.7 billion, \$3.4 billion and \$2.9 billion, respectively, were provided by financing activities as a result of CMO financing, net of principal repayments.

Inflation

The consolidated financial statements and corresponding notes to the consolidated financial statements have been prepared in accordance with GAAP, which require the measurement of financial position and operating results in terms of historical dollars without considering the changes in the relative purchasing power of money over time due to inflation. The impact of inflation is reflected in the increased costs of our operations during each of 2002, 2003 and 2004. Unlike industrial companies, nearly all of our assets and liabilities are monetary in nature. As a result, interest rates have a greater impact on our performance than do the effects of general levels of inflation. Inflation affects our operations primarily through its effect on interest rates, since interest rates normally increase during periods of high inflation and decrease during periods of low inflation. During periods of increasing interest rates, demand for mortgages and a borrower's ability to qualify for mortgage financing in a purchase transaction may be adversely affected. During periods of decreasing interest rates, borrowers may prepay their mortgages, which in turn may adversely affect our yield and subsequently the value of our portfolio of mortgage assets.

Off-Balance Sheet Arrangements

On July 1, 2003, IMH acquired the assets and liabilities of IFC as IMH purchased 100% of the outstanding shares of voting common stock of IFC, which represented 1% of the economic interest in IFC. Prior to the July 1, 2003, IFC was a non-consolidating entity of IMH and was accounted for using the equity method of accounting by virtue of IMH's ownership of 100% of the outstanding non-voting preferred stock of IFC, which represented 99% of the economic interest in IFC. As a result of the acquisition, IMH began to consolidate IFC as of that date. We do not have any relationships with any unconsolidated entities or financial partnerships, such as entities often referred to as structured finance or special purpose entities, which would have been established for the purpose of facilitating off-balance sheet arrangements or other contractually narrow or limited purposes. Refer to Item 7A. "Quantitative and Qualitative Disclosures About Market Risk" and "Note O - Derivative Instruments" in the accompanying notes to the consolidated financial statements for detail regarding off-balance sheet arrangements of consolidated entities.

Contractual Obligations

As of December 31, 2004, we had the following contractual obligations (in thousands):

	Payments Due by Period								
	Total	Less Than One Year	One to Three Years	Three to Five Years	More Than Five Years				
CMO borrowings (1)	\$21,260,630	\$5,735,344	\$ 11,209,965	\$2,846,158	\$ 1,469,163				
Reverse repurchase agreements	1,527,558	1,527,558	_	_	_				
Rate locked mortgage pipeline	1,959,919	1,959,919	_	_	_				
Premises operating lease agreements	11,178	3,454	6,309	1,415					
Total Contractual Obligations	\$24,759,285	\$9,226,275	\$ 11,216,274	\$2,847,573	\$ 1,469,163				

Darmanta Dua hu Dariad

For additional information regarding our commitments refer to "Note H—CMO Borrowings" and "Note N—Commitments and Contingencies" in the accompanying notes to the consolidated financial statements.

ITEM 7A. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

General Overview

Although we manage credit, prepayment and liquidity risk in the normal course of business, we consider interest rate risk to be a significant market risk, which could potentially have the largest material impact on our financial condition and results of operations. Since a significant portion of our revenues and earnings are derived from net interest income, we strive to manage our interest-earning assets and interest-bearing liabilities to generate what we believe to be an appropriate contribution from net interest income. When interest rates fluctuate, profitability can be adversely affected by changes in the fair market value of our assets and liabilities and by the interest spread earned on interest-earning assets and interest-bearing liabilities. We derive income from the differential spread between interest earned on interest-earning assets and interest paid on interest-bearing liabilities. Any change in interest rates affects income received and income paid from assets and liabilities in varying and typically in unequal amounts. Changing interest rates may compress or widen our interest rate margins and affect overall earnings.

Interest rate risk management is the responsibility of ALCO, which reports results of interest rate risk analysis to the board of directors on a quarterly basis. ALCO establishes policies that monitor and coordinate sources, uses and pricing of funds. ALCO also attempts to reduce the volatility in net interest income by managing the relationship of interest rate sensitive assets to interest rate sensitive liabilities. In addition, various modeling techniques are used to value interest sensitive mortgage-backed securities, including interest-only securities. The value of investment securities available-for-sale is determined using a discounted cash flow model using prepayment rate, discount rate and credit loss assumptions. Our investment securities portfolio is available-for-sale, which requires us to perform market valuations of the securities in order

⁽¹⁾ Payments on CMO borrowings are based on anticipated receipts of principal on underlying mortgage loan collateral using expected prepayment rates. If actual mortgage prepayment rates differ from our estimates, the payment amounts will vary from the reported amounts.

to properly record the portfolio. We continually monitor the interest rates of our investment securities portfolio as compared to prevalent interest rates in the market. We do not currently maintain a securities trading portfolio and are not exposed to market risk as it relates to trading activities.

Changes in Interest Rates

ALCO follows interest rate risk management policies intended to limit our exposure to changes in interest rates primarily associated with cash flows on our adjustable rate CMO borrowings. Our primary objective is to limit our exposure to the variability in future cash flows attributable to the variability of one-month LIBOR, which is the underlying index of our adjustable rate CMO borrowings. We also monitor on an ongoing basis the prepayment risks that arise in fluctuating interest rate environments. Our interest rate risk management policies are formulated with the intent to offset the potential adverse effects of changing interest rates on cash flows on adjustable rate CMO borrowings.

We primarily acquire for long-term investment ARMs and hybrid ARMs and, to a lesser extent, FRMs. ARMs are generally subject to periodic and lifetime interest rate caps. This means that the interest rate of each ARM is limited to upwards or downwards movements on its periodic interest rate adjustment date, generally six months, or over the life of the mortgage. Periodic caps limit the maximum interest rate change, which can occur on any interest rate change date to generally a maximum of 1% per semiannual adjustment. Also, each ARM has a maximum lifetime interest rate cap. Generally, borrowings are not subject to the same periodic or lifetime interest rate limitations. During a period of rapidly increasing or decreasing interest rates, financing costs could increase or decrease at a faster rate than the periodic interest rate adjustments on mortgages would allow, which could affect net interest income. In addition, if market rates were to exceed the maximum interest rate limits of our ARMs, borrowing costs could increase while interest rates on ARMs would remain constant. We also acquire hybrid ARMs that have initial fixed interest rate periods generally ranging from two to seven years which subsequently convert to ARMs. During a rapidly increasing or decreasing interest rate environment financing costs would increase or decrease more rapidly than would interest rates on mortgages, which would remain fixed until their next interest rate adjustment date. In order to provide protection against potential resulting basis risk shortfall on the related liabilities, we purchase derivatives.

We measure the sensitivity of our net interest income to changes in interest rates affecting interest sensitive assets and liabilities using various simulations. These simulations take into consideration changes that may occur in investment and financing strategies, the forward yield curve, interest rate risk management strategies, mortgage prepayment speeds and the volume of mortgage acquisitions and originations. As part of various interest rate simulations, we calculate the effect of potential changes in interest rates on our interest-earning assets and interest-bearing liabilities and their affect on overall earnings. The simulations assume instantaneous and parallel shifts in interest rates and to what degree those shifts affect net interest income. First, we estimate net interest income along with net cash flows on derivatives for the next twelve months using balance sheet data and the notional amount of derivatives as of December 31, 2004 and 12-month projections of the following primary drivers affecting net interest income:

- future interest rates using forward yield curves, which are considered market consensus estimates of future interest rates;
- mortgage acquisition and originations;
- mortgage prepayment rate assumptions; and
- · forward swap rates.

We refer to the 12-month projection of net interest income along with the 12-month projection of net cash flows on derivatives as the "base case." For financial reporting purposes, net cash flows on derivative instruments are included in gain (loss) on derivative instruments on the consolidated financial statements. However, for purposes of interest rate risk analysis we include net cash flows on derivatives in our base case simulations as we acquire derivatives to offset the effect that changes in interest rates have on variable borrowing costs, such as CMO and warehouse borrowings. We believe that including net cash flows on derivatives in our interest rate risk analysis presents a more useful simulation of the effect of changing interest rates on net cash flows generated by our long-term mortgage portfolio.

Once the base case has been established, we "shock" the base case with instantaneous and parallel shifts in interest rates in 100 basis point increments upward and downward. Calculations are made for each of the defined instantaneous and parallel shifts in interest rates over or under the forward yield curve used to determine the base case and include any

associated changes in projected mortgage prepayment rates caused by changes in interest rates. The results of each 100 basis point change in interest rates are then compared against the base case to determine the estimated dollar and percentage change to base case. The simulations consider the affect of interest rate changes on interest sensitive assets and liabilities as well as derivatives. The simulations also consider the impact that instantaneous and parallel shift in interest rates have on prepayment rates and the resulting affect of accelerating or decelerating amortization of premium and securitization costs.

In the following table, the down 100 basis point scenario as of December 31, 2004 represents our projection of the net change from base case net interest income, which is derived from assumptions as previously discussed, if market interest rates were to immediately decline by 100 basis points. This means that we reduce interest rates at all data points along our projected forward yield curve by 100 basis points and recalculate our projection of net interest income over the next 12 months. In addition, based on changes in interest rates, or changes in our forward yield curve, our model adjusts mortgage prepayment rates and recalculates amortization of acquisition and securitization costs and net cash receipts or payments on derivates as part of the calculation of net interest income. Thus, if a 100 basis point decline occurred the projected volatility to net interest income is positively impacted through our use of derivatives.

Over the past year, the interest rate risk profile shifted from modestly asset sensitive to modestly liability sensitive. This occurred as part of a deliberate and long-term optimization strategy as mortgages having marginally longer duration than that of CMO borrowings were added to our balance sheet during 2004. Other factors contributing to the shift in the interest rate risk profile include the increase in the overall level of interest rates, the flattening of the yield curve and slower expected prepayment behavior. However, since our estimates are based upon numerous assumptions, actual sensitivity to interest rate changes could vary if actual experience differs from the assumptions used.

The following table estimates the financial impact to base case, including net cash flow from derivatives, from various instantaneous and parallel shifts in interest rates based on both our on-balance sheet structure and off-balance sheet structure, which refers to the notional amount of derivatives that are not recorded on our balance sheet as of December 31, 2004 and 2003 (dollar amounts in millions):

	Cha	Changes in base case as of December 31, 2004 (1)							
		Excluding net cash flow on derivatives		Including no flow on deri					
	(\$)	(%)	(\$)	(\$)	(%)				
Instantaneous and Parallel Change in Interest Rates (2)									
Up 300 basis points, or 3% (3)	(380.1)	(132)	328.3	(51.8)	(18)				
Up 200 basis points, or 2%	(258.3)	(90)	218.9	(39.4)	(14)				
Up 100 basis points, or 1%	(123.2)	(43)	109.4	(13.8)	(5)				
Down 100 basis points, or 1%	114.9	40	(109.4)	5.5	2				

Ch	Changes in base case as of December 31, 2003 (1)			
		Net cash flow on derivatives	Including net cash flow on derivatives	
(\$)	(%)	(\$)	(\$)	(%)
(162.3)	(75)	182.6	20.3	12
(99.8)	(46)	121.7	21.9	13
(44.3)	(20)	60.9	16.6	10
57.3	26	(60.7)	(3.4)	(2)
	(\$) (162.3) (99.8) (44.3)	Excluding net cash flow on derivatives (\$) (%) (162.3) (75) (99.8) (46) (44.3) (20)	Excluding net cash flow on derivatives	Excluding net cash flow on derivatives

- (1) The dollar and percentage changes represent base case for the next twelve months versus the change in base case using various instantaneous and parallel interest rate change simulations, excluding the effect of amortization of loan discounts to base case.
- (2) Instantaneous and parallel interest rate changes over and under the projected forward yield curve.
- (3) This simulation was added to our analysis as it is relevant in light of the interest rate environment as of December 31, 2004 and the projected forward yield curve for 2004.

The use of derivatives to manage risk associated with changes in interest rates is an integral part of our strategy. The amount of cash payments or cash receipts on derivatives is determined by (1) the notional amount of the derivative and (2) current interest rate levels in relation to the various strikes or coupons of derivatives during a particular time period. By using derivatives, we attempt to minimize the effect of both upward and downward interest rate changes on our long-term mortgage portfolio. Our goal is to minimize significant changes to base case net interest income, including net cash flows from derivatives, as interest rates change. We primarily acquire swaps to essentially convert our adjustable rate CMO borrowings into fixed rate borrowings. For instance, we receive one-month LIBOR on swaps, which offsets interest expense on adjustable rate CMO borrowings, and we pay a fixed interest rate.

The following table presents the extent to which changes in interest rates and changes in the volume of interest rate sensitive assets and interest rate sensitive liabilities have affected interest income and interest expense during the periods indicated. Information is provided on mortgage assets and borrowings on mortgage assets, only, with respect to the following:

Year Ended December 31, 2004 over 2003

- changes attributable to changes in volume (changes in volume multiplied by prior rate);
- changes attributable to changes in rate (changes in rate multiplied by prior volume);
- · changes in interest due to both rate and volume; and
- net change.

	fear	Tear Ended December 31, 2004 over 2003			
	Volume	Rate	Rate/ Volume	Net Change	
		(in thousands)			
Increase/(decrease) in:					
Subordinated securities collateralized by mortgages	\$ (432)	\$ 402	\$ (45)	\$ (75)	
Mortgages held as CMO collateral	367,388	(30,616)	(35,435)	301,337	
Mortgages held-for-investment and mortgages held-for-sale	65,717	1,877	3,568	71,162	
Finance receivables	(2,424)	(1,666)	139	(3,951)	
Change in interest income on mortgage assets	430,249	(30,003)	(31,773)	368,473	
CMO borrowings	206,112	(11,801)	(13,963)	180,348	
Reverse repurchase agreements	18,681	4,296	2,478	25,455	
Borrowings secured by investment securities	(2,316)			(2,316)	
Change in interest expense on borrowings on mortgage assets	222,477	(7,505)	(11,485)	203,487	
Change in net interest income on mortgage assets	\$207,772	\$(22,498)	\$(20,288)	\$164,986	
Change in net interest income on mortgage assets	<u> </u>	\$(22,498)			
Change in net interest income on mortgage assets	<u> </u>				
Change in net interest income on mortgage assets	Yea	nr Ended Decemb	er 31, 2003 over	2002 Net	
Increase/(decrease) in:	Yea	r Ended Decemb	er 31, 2003 over Rate/ Volume	2002 Net Change	
Increase/(decrease) in: Subordinated securities collateralized by mortgages	Volume \$ (194)	r Ended Decemb	er 31, 2003 over Rate/ Volume	2002 Net Change \$ 1,975	
Increase/(decrease) in: Subordinated securities collateralized by mortgages Mortgages held as CMO collateral	Volume \$ (194) 192,017	Rate (in tho \$ 2,421 (38,499)	Rate/ Volume usands) \$ (252) (36,826)	2002 Net Change \$ 1,975 116,692	
Increase/(decrease) in: Subordinated securities collateralized by mortgages Mortgages held as CMO collateral Mortgages held-for-investment	Volume \$ (194) 192,017 30,729	Rate (in tho \$ 2,421 (38,499) (530)	Rate/ Volume usands) \$ (252) (36,826) (2,400)	2002 Net Change \$ 1,975 116,692 27,799	
Increase/(decrease) in: Subordinated securities collateralized by mortgages Mortgages held as CMO collateral	Volume \$ (194) 192,017	Rate (in tho \$ 2,421 (38,499)	Rate/ Volume usands) \$ (252) (36,826)	2002 Net Change \$ 1,975 116,692	
Increase/(decrease) in: Subordinated securities collateralized by mortgages Mortgages held as CMO collateral Mortgages held-for-investment	Volume \$ (194) 192,017 30,729	Rate (in tho \$ 2,421 (38,499) (530)	Rate/ Volume usands) \$ (252) (36,826) (2,400)	2002 Net Change \$ 1,975 116,692 27,799	
Increase/(decrease) in: Subordinated securities collateralized by mortgages Mortgages held as CMO collateral Mortgages held-for-investment Finance receivables Change in interest income on mortgage assets CMO borrowings	Volume \$ (194) 192,017 30,729 12,259 234,811	Rate (in tho \$ 2,421 (38,499) (530) (1,637)	Rate/Volume usands) \$ (252) (36,826) (2,400) (1,036)	\$ 1,975 116,692 27,799 9,586 156,052 71,833	
Increase/(decrease) in: Subordinated securities collateralized by mortgages Mortgages held as CMO collateral Mortgages held-for-investment Finance receivables Change in interest income on mortgage assets	Volume \$ (194) 192,017 30,729 12,259	Rate (in tho \$ 2,421 (38,499) (530) (1,637) (38,245)	Rate/Volume usands) \$ (252) (36,826) (2,400) (1,036) (40,514)	Net Change \$ 1,975 116,692 27,799 9,586 156,052	
Increase/(decrease) in: Subordinated securities collateralized by mortgages Mortgages held as CMO collateral Mortgages held-for-investment Finance receivables Change in interest income on mortgage assets CMO borrowings	Volume \$ (194) 192,017 30,729 12,259 234,811	Rate (in tho \$ 2,421 (38,499) (530) (1,637) (38,245) (13,499) (4,473)	Rate/Volume usands) \$ (252) (36,826) (2,400) (1,036) (40,514) (12,962)	\$ 1,975 116,692 27,799 9,586 156,052 71,833	

ITEM 8. UNAUDITED FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA

Change in net interest income on mortgage assets

The information required by this Item 8 is incorporated by reference to Impac Mortgage Holdings, Inc.'s Consolidated Financial Statements and Independent Auditors' Report beginning at page F-1 of this Form 10-K. Please refer to the discussion under Part I. Item 1. "—Use of Unaudited Financial Statements" for additional information regarding the inclusion and use of unaudited financial statements in this annual report.

ITEM 9. CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE

None.

\$121,490

\$(27,002)

\$(19,532)

\$ 74,956

ITEM 9A. CONTROLS AND PROCEDURES

Evaluation of Disclosure Controls and Procedures

Disclosure controls and procedures are controls and other procedures of the Company that are designed to ensure that information required to be disclosed by the Company in the reports that it files or submits under the Securities Exchange Act of 1934 (the "Exchange Act") is properly recorded, processed, summarized and reported, within the time periods specified in the SEC's rules and forms. Disclosure controls and procedures include processes to accumulate and communicate relevant information to management, including our CEO and CFO, as appropriate, to allow for timely decisions regarding required disclosures.

As of December 31, 2004, our CEO and CFO, with the participation of other management of the Company, evaluated the effectiveness of our disclosure controls and procedures, as such term is defined under Rule 13a-15(e) or 15(d)-15(e) promulgated under the Exchange Act, and based upon that evaluation, our CEO and CFO concluded that they believe that these disclosure controls and procedures were not effective as a result of the identification of material weaknesses, which are further discussed, under Management's Report on Internal Control Over Financial Reporting, below.

Management's Report on Internal Control over Financial Reporting

Introduction

Management of the Company is responsible for establishing and maintaining adequate internal control over financial reporting (as defined in Section 13a-15(f) of the Securities Exchange Act of 1934, as amended). Internal control over financial reporting is a process designed by, or under the supervision of, our CEO and CFO to provide reasonable assurance regarding the reliability of financial reporting and the preparation of the Company's financial statements for external reporting purposes in conformity with U.S. generally accepted accounting principles and include those policies and procedures that (i) pertain to the maintenance of records that in reasonable detail accurately and fairly reflect the transactions and dispositions of the assets of the Company; (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the Company are being made only in accordance with authorizations of management and directors of the Company; and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of the Company's assets that could have a material effect on the financial statements.

All internal control systems, no matter how well designed, have inherent limitations. Therefore, even those systems determined effective could provide only reasonable, not absolute, assurance with respect to financial statement preparation and presentation.

Management's Assessment

As of December 31, 2004, management is required to conduct an assessment of the effectiveness of the Company's internal control over financial reporting based on the framework established in *Internal Control – Integrated Framework* issued by the Committee Sponsoring Organizations of the Treadway Commission ("COSO"). Based on the criteria established by COSO, management expects that the Company's internal control over financial reporting as of December 31, 2004 will not be effective as a result of the identification of the material weaknesses described below.

A material weakness is a significant deficiency (within the meaning of Public Company Accounting Oversight Board, or PCAOB, Auditing Standard No. 2), or combination of significant deficiencies, that results in there being more than a remote likelihood that a material misstatement of the annual or interim financial statements will not be prevented or detected. A significant deficiency is a control deficiency, or combination of control deficiencies, that adversely affects the Company's ability to initiate, authorize, record, process, or report external financial information reliably in accordance with generally accepted accounting principles, such that there is more than a remote likelihood that a misstatement of the Company's annual or interim financial statements, that is more than inconsequential, will not be prevented or detected.

The time and resources we committed to the restatement of certain financial statements during 2004 delayed our internal timetable with respect to our documentation, assessment and evaluation of internal control over financial reporting. Due in large part to the foregoing, management's assessment of the effectiveness of our internal control over financial reporting was substantially delayed, thereby delaying the performance by KPMG LLP, our independent registered public accounting firm, of its audit of management's assessment of the effectiveness of internal control over financial reporting. As a result, such audit and management's report on internal control over financial reporting are not yet complete. We cannot assure you that additional material weaknesses or significant deficiencies will not be identified, or that any deficiencies identified, either alone or in combination with others, will not be considered additional material weaknesses.

Material Weaknesses

The Company's management has identified the following material weaknesses in the Company's internal control over financial reporting:

- 1) The Company's internal controls intended to ensure the proper accounting for complex transactions were not designed or operating effectively as of December 31, 2004. Complex transactions include, but are not limited to, the transfer of financial assets, derivatives and hedging and income taxes. Specifically, the Company does not have an adequate number of personnel in the accounting and finance departments with an appropriate level of skills and expertise to ensure that the accounting for such transactions in the financial statements are in accordance with generally accepted accounting principles. Such personnel related matters preclude the Company from conducting the necessary evaluation, performance and monitoring of the relevant controls and procedures as part of the financial reporting process. As a result of these ineffective controls, the Company restated its financial statements to correct certain accounting errors for the years ended December 31, 2003, 2002 and 2001, and the three months ended March 31, 2004 and 2003, the three and six months ended June 30, 2004 and 2003 and the three and nine months ended September 30, 2003.
- 2) The Company's internal controls intended to ensure adequate access and change control over end-user computing spreadsheets were not designed properly as of December 31, 2004. In addition, the information technology general controls related to access and program change were deficient as of year end, resulting in a potential lack of reliability and integrity of the financial information used in these spreadsheets. As a result, although no actual misstatement has been identified, there is a more than remote likelihood that financial statements and related footnote disclosures could be materially misstated.

Auditor Reports

KPMG LLP's audit report on management's assessment of our internal control over financial reporting is not provided with this annual report. Upon completion of our assessment of the effectiveness of internal control over financial reporting and the receipt of KPMG LLP's audit report on our internal control over financial reporting and its audit report on our financial statements, we will file an amended annual report on Form 10-K/A to include such reports and our audited financial statements for the year ended December 31, 2004. We cannot assure you that KPMG LLP's report will express an unqualified opinion on management's assessment of the effectiveness of its internal control over financial reporting, and we expect KPMG LLP's report to include an adverse opinion on the effectiveness of our internal control over financial reporting based on the previously discussed material weaknesses.

Changes in Internal Control Over Financial Reporting

In connection with the restatement of our consolidated financial statements last year, we noted certain matters involving internal controls and operations that we considered to be a material weakness. We needed to improve the evaluation and documentation of accounting policies and procedures for complex transactions, such as transfers of financial assets, derivatives and hedge accounting and allowance for credit losses. We did not have a sufficient amount or type of staff in the financial reporting and accounting departments, at the time of the restatement. Furthermore, we noted significant deficiencies in that our internal audit function did not provide an adequate or effective monitoring of our controls. Further, we needed to evaluate whether we have appropriate internal resources to manage and monitor work performed by our outsourced tax compliance function.

As a result of the findings described above and in addition to our obligations and findings under Section 404 of the Sarbanes-Oxley Act of 2002, during 2004, including during the fourth quarter of 2004, we began implementing the following actions to address the two identified material weaknesses:

- we reviewed the material weaknesses with our Audit Committee and senior management;
- as part of our new internal control procedures, we enhanced our documentation of critical accounting policies;

- we hired outside consultants to assist our internal audit group in documenting our accounting and business processes and identifying areas that require control or process improvement;
- we established new internal control processes based on discussions with our consultants and our own management team seeking to remedy any deficiencies;
- in June 2004, we hired a Director of Internal Audit whose primary responsibilities are to monitor our system of internal controls and, in addition, to oversee the establishment of formal policies and procedures throughout our organization;
- we have instituted new control procedures around our quarterly reporting processes for accounting for significant or complex transactions, which are
 discussed and documented, reviewed with our Audit Committee, formally approved by our management and given timely effect in our books and
 records;
- in October 2004, we hired a Controller;
- in November 2004, we began hiring additional personnel, totaling six additional staff as of the current date, to support the internal audit department in their primary responsibilities, as discussed above;
- we have hired additional resources in the accounting and finance areas, including a Vice President Compliance, with expertise in technical accounting. SEC reporting and the design and assessment of internal controls over financial reporting; and
- we began implementation of policies and procedures with respect to authorization and monitoring of user access and with respect to the authorization and documentation requirements for program changes in order to ensure the effectiveness of these IT general controls.

Subsequent to December 31, 2004, we have also taken the following actions:

- effective February 2005, we appointed Gretchen Verdugo, a current executive officer of IMH, as Executive Vice President, Chief Accounting Officer;
- in March 2005, we hired a Tax Manager to assist the Company in planning and managing the Company's outside tax professionals; and
- we are evaluating and plan to implement an automated end-user computing tool to ensure proper access and data integrity and to address the material weakness related to end-user computing spreadsheets we utilized for the aggregation, analysis and reporting of data.

We believe that our disclosure controls and procedures, including our internal control over financial reporting, have improved since year-end due to the scrutiny of such matters by our management and Audit Committee and the changes described above. We believe we have hired a sufficient amount and type of staff in the accounting and finance departments and we may make additional changes in the future, as we deem necessary. The Company believes that as of December 31, 2004 a sufficient amount of time has not passed since the implementation of remediation efforts pertaining to personnel related matters to address the design and operating effectiveness of the relevant controls. We cannot assure you that, as circumstances change, any additional material weakness will not be identified.

ITEM 9B. OTHER INFORMATION

None.

PART III

ITEM 10. DIRECTORS AND EXECUTIVE OFFICERS OF THE REGISTRANT

The information required by this Item 10 is hereby incorporated by reference to Impac Mortgage Holdings, Inc.'s definitive proxy statement, to be filed pursuant to Regulation 14A within 120 days after the end of Impac Mortgage Holdings, Inc.'s 2004 fiscal year.

ITEM 11. EXECUTIVE COMPENSATION

The information required by this Item 11 is hereby incorporated by reference to Impac Mortgage Holdings, Inc.'s definitive proxy statement, to be filed pursuant to Regulation 14A within 120 days after the end of Impac Mortgage Holdings, Inc.'s 2004 fiscal year.

ITEM 12. SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

The information required by this Item 12 is hereby incorporated by reference to Impac Mortgage Holdings, Inc.'s definitive proxy statement, to be filed pursuant to Regulation 14A within 120 days after the end of Impac Mortgage Holdings, Inc.'s 2004 fiscal year.

ITEM 13. CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS

The information required by this Item 13 is hereby incorporated by reference to Impac Mortgage Holdings, Inc.'s definitive proxy statement, to be filed pursuant to Regulation 14A within 120 days after the end of Impac Mortgage Holdings, Inc.'s 2004 fiscal year.

ITEM 14. PRINCIPAL ACCOUNTANT FEES AND SERVICES

The information required by this Item 14 is hereby incorporated by reference to Impac Mortgage Holdings, Inc.'s definitive proxy statement, to be filed pursuant to Regulation 14A within 120 days after the end of Impac Mortgage Holdings, Inc.'s 2004 fiscal year.

PART IV

ITEM 15. EXHIBITS AND FINANCIAL STATEMENT SCHEDULES

- (a)(1) All financial statements and schedules have been omitted because they are either not applicable, not required or the information required has been disclosed in the Consolidated Financial Statements and related Notes to Consolidated Financial Statements at page F-1, or otherwise included in this Form 10-K.
- (a)(3) Exhibits

Exhibit Number	Description
3.1	Charter of the Registrant (incorporated by reference to the corresponding exhibit number to the Registrant's Registration Statement on Form S-11, as amended (File No. 33-96670), filed with the Securities and Exchange Commission on September 7, 1995).
3.1(a)	Certificate of Correction of the Registrant (incorporated by reference to exhibit 3.1(a) of the Registrant's 10-K for the year ended December 31, 1998).
3.1(b)	Articles of Amendment of the Registrant (incorporated by reference to exhibit 3.1(b) of the Registrant's 10-K for the year ended December 31, 1998).
3.1(c)	Articles of Amendment for change of name to Charter of the Registrant (incorporated by reference to exhibit number 3.1(a) of the Registrant's Current Report on Form 8-K, filed February 11, 1998).
3.1(d)	Articles Supplementary and Certificate of Correction for Series A Junior Participating Preferred Stock of the Registrant (incorporated by reference to exhibit 3.1(d) of the Registrant's 10-K for the year ended December 31, 1998).
3.1(e)	Articles Supplementary for Series B 10.5% Cumulative Convertible Preferred Stock of the Registrant (incorporated by reference to exhibit 3.1b of the Registrant's Current Report on Form 8-K, filed December 23, 1998).
3.1(f)	Articles Supplementary for Series C 10.5% Cumulative Convertible Preferred Stock of the Registrant (incorporated by reference to the corresponding exhibit number of the Registrant's Quarterly Report on Form 10-Q for the period ending September 30, 2000).
3.1(g)	Certificate of Correction for Series C Preferred Stock of the Registrant (incorporated by reference to the corresponding exhibit number of the Registrant's Quarterly Report on Form 10-Q for the period ending September 30, 2000).
3.1(h)	Articles Supplementary, filed with the State Department of Assessments and Taxation of Maryland on July 12, 2002, reclassifying Series C Preferred Stock of the Registrant (incorporated by reference to exhibit 9 of the Registrant's Form 8-A/A, Amendment No. 2, filed July 30, 2002).
3.1(i)	Articles of Amendment, filed with the State Department of Assessments and Taxation of Maryland on July 17, 2002, increasing authorized shares of Common Stock of the Registrant (incorporated by reference to exhibit 10 of the Registrant's Form 8-A/A, Amendment No. 2, filed July 30, 2002).
3.1(j)	Resolution to Change Principal Officer or Resident Agent, filed with the State Department of Assessments and Taxation of Maryland on September 11, 2002 (previously filed as an exhibit to the Annual Report on Form 10-K for the year ended December 21, 2003).
3.1(k)	Articles of Amendment, filed with the State Department of Assessments and Taxation of Maryland on June 22, 2004, amending and restating Article

VII of the Registrant's Charter (incorporated by reference to exhibit 7 of the Registrant's Form 8-A/A, Amendment No. 1, filed June 30, 2004).

- 3.1(l) Articles Supplementary designating the Company's 9.375% Series B Cumulative Redeemable Preferred Stock, liquidation preference \$25.00 per share, par value \$0.01 per share, filed with the State Department of Assessments and Taxation of Maryland on May 26, 2004 (incorporated by reference to exhibit 3.8 of the Registrant's Form 8-A/A, Amendment No. 1, filed June 30, 2004).
- 3.1(m) Articles Supplementary designating the Company's 9.125% Series C Cumulative Redeemable Preferred Stock, liquidation preference \$25.00 per share, par value \$0.01 per share, filed with the State Department of Assessments and Taxation of Maryland on November 18, 2004 (incorporated by reference to exhibit 3.10 of the Registrant's Form 8-A filed November 19, 2004).
- 3.2 Bylaws of the Registrant, as amended and restated (incorporated by reference to the corresponding exhibit number of the Registrant's Quarterly Report on Form 10-Q for the period ending March 31, 1998).
- 3.2(a) Amendment to Bylaws of the Registrant (incorporated by reference to exhibit 3.2(a) of the Registrant's Registration Statement of Form S-3 (File No. 333-111517) filed with the Securities and Exchange Commission on December 23, 2003).
- 4.1 Form of Stock Certificate of the Company (incorporated by reference to the corresponding exhibit number to the Registrant's Registration Statement on Form S-11, as amended (File No. 33-96670), filed with the Securities and Exchange Commission on September 7, 1995).
- 4.2 Rights Agreement between the Registrant and BankBoston, N.A. (incorporated by reference to exhibit 4.2 of the Registrant's Registration Statement on Form 8-A as filed with the Securities and Exchange Commission on October 14, 1998).
- 4.2(a) Amendment No. 1 to Rights Agreement between the Registrant and BankBoston, N.A. (incorporated by reference to exhibit 4.2(a) of the Registrant's Registration Statement on Form 8-A/A as filed with the Securities and Exchange Commission on December 23, 1998).
- 4.3 Specimen Certificate representing the 9.375% Series B Cumulative Redeemable Preferred Stock (incorporated by reference to Exhibit 4.1 of the Registrant's Form 8-A, filed with the Securities and Exchange Commission May 27, 2004).
- 4.4 Specimen Certificate representing the 9.125% Series C Cumulative Redeemable Preferred Stock (incorporated by reference to Exhibit 4.1 of the Registrant's Form 8-A, filed with the Securities and Exchange Commission November 19, 2004).
- 10.1 1995 Stock Option, Deferred Stock and Restricted Stock Plan, as amended and restated (incorporated by reference to exhibit 10.1 of the Registrant's Quarterly Report on Form 10-Q for the period ending March 31, 1998).
- 10.2(a) Form of 2002 Indemnification Agreement between the Registrant and its Directors and Officers (incorporated by reference to exhibit 10.1(a) of the Registrant's Quarterly Report on Form 10-Q for the period ended September 30, 2004).
- 10.2(b) Schedule of each officer and director that is a party to an Indemnification Agreement (incorporated by reference to exhibit 10.1(b) of the Registrant's Quarterly Report on Form 10-Q for the period ended September 30, 2004).
- 10.3 Form of Loan Purchase and Administrative Services Agreement between the Registrant and Impac Funding Corporation (incorporated by reference to exhibit 10.9 to the Registrant's Registration Statement on Form S-11, as amended (File No. 33-96670), filed with the Securities and Exchange Commission on September 7,1995).
- Servicing Agreement effective November 11, 1995 between the Registrant and Impac Funding Corporation (incorporated by reference to exhibit 10.14 to the Registrant's Registration Statement on Form S-11, as amended (File No. 333-04011), filed with the Securities and Exchange Commission on May 17, 1996).

10.13(a)

10.5* Impac Mortgage Holdings, Inc. 1996 Stock Option Loan Plan (incorporated by reference to exhibit 10.15 to the Registrant's Form 10-K for the vear ended December 31, 1996). Lease dated June 1, 1998 regarding 1401 Dove Street, Newport Beach California (incorporated by reference to exhibit 10.17 of the Registrant's 10.6 10-K for the year ended December 31, 1998). 10.6(a) Second Amendment to Lease dated October 1, 1999 between The Realty Associates Fund V, L.P., the Registrant and Impac Funding Corporation regarding 1401 Dove Street, Newport Beach California (incorporated by reference to exhibit number 10.4(d) of the Registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2000). 10.7 Office Lease, First Amendment to Office Lease, and Assignment, Assumption and Consent to Assignment of Lease with Property California OB One Corporation and Assignment to Impac Funding Corporation regarding 15050 Avenue of Science Suite 210 San Diego California. (incorporated by reference to exhibit number 10.10 of the Registrant's Annual Report on Form 10-K for the year ended December 31, 2001). Lease dated March 4, 2005 regarding 19500 Jamboree Road, Newport Beach California. 10.8 10.9* Employment Agreement between Impac Funding Corporation and Ronald M. Morrison dated September 1, 2001 (incorporated by reference to exhibit 10.8 of the Registrant's Annual Report on Form 10-K for the year ended December 31, 2003). Note dated June 30, 1999 between the Registrant and Impac Funding Corporation (incorporated by reference to exhibit 10.17 of the Registrant's 10.10 Annual Report on Form 10-K for the year ended December 31, 1999). 10.10(b) Amendment to Note dated June 30, 1999 between the Registrant and Impac Funding Corporation. (incorporated by reference to exhibit 10.1 of the Registrant's Quarterly Report on Form 10-Q for the period ended June 30, 2004). 10.11* Impac Mortgage Holdings, Inc. 2001 Stock Option Plan, Deferred Stock and Restricted Stock Plan (incorporated by reference to Appendix A of Registrant's Definitive Proxy Statement filed with the SEC on April 30, 2001). Amendment to Impac Mortgage Holdings, Inc. 2001 Stock Option Plan, Deferred Stock and Restricted Stock Plan (incorporated by reference to 10.11(a)* exhibit 4.1(a) of the Registrant's Form S-8 filed with the SEC on March 1, 2002). Amendment No. 2 to Impac Mortgage Holdings, Inc. 2001 Stock Option Plan, Deferred Stock and Restricted Stock Plan (incorporated by 10.11(b)* reference to exhibit 10.10(b) of the Registrant's Annual Report on Form 10-K for the year ended December 31, 2003). 10.11(c)* Form of Stock Option Agreement for 2001 Stock Option, Deferred Stock and Restricted Stock Plan (incorporated by reference to exhibit 10.2 of the Registrant's Quarterly Report on Form 10-Q for the period ended September 30, 2004). 10.12* The Impac Companies Deferred Compensation Plan (incorporated by reference to exhibit 10.17 of the Registrant's Form 10-K for the year ended December 31, 2002). 10.13 Equity Distribution Agreement, dated May 5, 2003, between Impac Mortgage Holdings, Inc. and UBS Warburg LLC (incorporated by reference to exhibit 1.1 of the Registrant's Current Report on Form 8-K, filed May 7, 2003).

(incorporated by reference to exhibit 1.1 of the Registrant's Current Report on Form 8-K, filed June 12, 2003).

Letter Agreement to Equity Distribution Agreement, dated June 10, 2003, between Impac Mortgage Holdings, Inc. and UBS Securities LLC

10, 2004).

10.24

10.14*

10.14(a)*

by reference to exhibit 10.1 of the Registrant's Current Report on Form 8-K, filed September 15, 2004). Employment Agreement, made as of April 1, 2003, between Impac Funding Corporation and William S. Ashmore (incorporated by reference to 10.15* exhibit 10.2 of the Registrant's Current Report on Form 8-K, filed July 15, 2003). 10.15(a)* Amendment to Employment Agreement, dated September 9, 2004, between Impac Funding Corporation and William S. Ashmore (incorporated by reference to exhibit 10.2 of the Registrant's Current Report on Form 8-K, filed September 15, 2004). 10.16* Employment Agreement, made as of April 1, 2003, between Impac Funding Corporation and Richard J. Johnson (incorporated by reference to exhibit 10.3 of the Registrant's Current Report on Form 8-K, filed July 15, 2003). Amendment to Employment Agreement, dated September 9, 2004, between Impac Funding Corporation and Richard J. Johnson (incorporated by 10.16(a)* reference to exhibit 10.3 of the Registrant's Current Report on Form 8-K, filed September 15, 2004). 10.17 Guaranty, dated April 1, 2003, granted by Impac Mortgage Holdings, Inc. in favor of Joseph R. Tomkinson (incorporated by reference to exhibit 10.4 of the Registrant's Current Report on Form 8-K, filed July 15, 2003). 10.18 Guaranty, dated April 1, 2003, granted by Impac Mortgage Holdings, Inc. in favor of William S. Ashmore (incorporated by reference to exhibit 10.5 of the Registrant's Current Report on Form 8-K, filed July 15, 2003). 10.19 Guaranty, dated April 1, 2003, granted by Impac Mortgage Holdings, Inc. in favor of Richard J. Johnson (incorporated by reference to exhibit 10.6 of the Registrant's Current Report on Form 8-K, filed July 15, 2003). Stock Purchase Agreement, dated July 1, 2003, among Impac Mortgage Holdings, Inc, Joseph R. Tomkinson, William S. Ashmore, and Johnson 10.20 Revocable Living Trust (incorporated by reference to exhibit 10.7 of the Registrant's Current Report on Form 8-K, filed July 15, 2003). Underwriting Agreement, dated December 17, 2003, by and among Impac Mortgage Holdings, Inc., JMP Securities LLC and Sandler O'Neill & 10.21 Partners, L.P. (incorporated by reference to exhibit 1.1 of the Registrant's Current Report on Form 8-K, filed December 18, 2003). 10.22 Underwriting Agreement, dated February 5, 2004, by and among Impac Mortgage Holdings, Inc., UBS Securities LLC, Friedman, Billings, Ramsey & Co., Inc., Sandler O'Neill & Partners, L.P. and JMP Securities LLC (incorporated by reference to exhibit 1.1 of the Registrant's Current Report on Form 8-K, filed February 6, 2004). 10.23 Underwriting Agreement, dated May 7, 2004, by and among Impac Mortgage Holdings, Inc., UBS Securities LLC, RBC Capital Markets

Employment Agreement, made as of April 1, 2003, between Impac Funding Corporation and Joseph R. Tomkinson (incorporated by reference to

Amendment to Employment Agreement, dated September 9, 2004, between Impac Funding Corporation and Joseph R. Tomkinson (incorporated

exhibit 10.1 of the Registrant's Current Report on Form 8-K, filed July 15, 2003).

to exhibit 1.1 of the Registrant's Current Report on Form 8-K filed May 13, 2004).

Corporation and Roth Capital Partners LLC (incorporated by reference to exhibit 1.1 of the Registrant's Current Report on Form 8-K filed May

Equity Distribution Agreement, dated May 12, 2004, between Impac Mortgage Holdings, Inc. and UBS Securities LLC (incorporated by reference

- 10.25 Underwriting Agreement, dated May 25, 2004, by and between Impac Mortgage Holdings, Inc., and Bear, Stearns & Co. Inc., Stifel, Nicolaus & Company, Incorporated, JMP Securities LLC, RBC Dain Rauscher Inc., Advest, Inc., and Flagstone Securities, LLC (incorporated by reference to exhibit 1.1 of the Registrant's Current Report on Form 8-K filed May 27, 2004).
- 10.26 Underwriting Agreement, dated May 25, 2004, by and between Impac Mortgage Holdings, Inc., and Bear, Stearns & Co. Inc., Stifel, Nicolaus & Company, Incorporated, JMP Securities LLC, RBC Dain Rauscher Inc., Advest, Inc., and Flagstone Securities, LLC (incorporated by reference to exhibit 1.1 of the Registrant's Current Report on Form 8-K filed May 27, 2004).
- 10.27 Impac Companies 401(k) Savings Plan (formerly known as the Impac Funding Corporation 401(k) Profit Sharing Plan) (incorporated by reference to exhibit 99.1 of the Registrant's Registration Statement on Form S-8 (File No. 333-117070) filed July 1, 2004).
- 10.28 Underwriting Agreement, dated November 18, 2004, by and between Impac Mortgage Holdings, Inc., and Bear, Stearns & Co. Inc., Stifel, Nicolaus & Company, Incorporated, and RBC Dain Rauscher Inc. (incorporated by reference to exhibit 1.1 of the Registrant's Current Report on Form 8-K filed November 19, 2004).
- 10.29 Underwriting Agreement, dated November 18, 2004, by and between Impac Mortgage Holdings, Inc., and UBS Securities LLC, Bear, Steams & Co. Inc., Deutsche Bank Securities Inc., and JMP Securities LLC (incorporated by reference to exhibit 1.1 of the Registrant's Current Report on Form 8-K filed November 19, 2004).
- 10.30* Letter from Impac Funding Corporation to Gretchen Verdugo regarding proposed Compensation Agreement Terms.
- 21.1 Subsidiaries of the Registrant (incorporated by reference to exhibit 21.1 of the Registrant's Quarterly Report on Form 10-Q for the period ended June 30, 2003).
- 23.1† Consent of KPMG LLP.
- 31.1 Certification of Chief Executive Officer pursuant to Item 601(b)(31) of Regulation S-K, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 31.2 Certification of Chief Financial Officer pursuant to Item 601(b)(31) of Regulation S-K, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 32.1 Certifications of Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. Section 1350 as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.**
- † This Annual Report on Form 10-K for the year ended December 31, 2004 does not include the audit reports of KPMG LLP relating to the consolidated financial statements and management's assessment of the effectiveness of internal control over financial reporting. Upon receipt from KPMG LLP, such audit reports will be filed in an amendment on Form 10-K/A and the consent of KPMG LLP will be filed as an exhibit thereto.
- * Denotes a management or compensatory plan or arrangement required to be filed as an exhibit pursuant to Item 601 of Regulation S-K
- ** This exhibit shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934 or otherwise subject to the liabilities of that section, nor shall it be deemed incorporated by reference in any filing under the Securities Act of 1933 or the Securities Exchange Act of 1934, whether made before or after the date hereof and irrespective of any general incorporation language in any filings.

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Newport Beach, State of California, on the 31st day of March 2005.

IMPAC MORTGAGE HOLDINGS, INC.

by /s/ Joseph R. Tomkinson

Joseph R. Tomkinson Chairman of the Board and Chief Executive Officer

Pursuant to the requirements of the Securities Act of 1934, this report has been signed by the following persons on behalf of the registrant and in the capacities and on the dates indicated.

Signature	Title	Date	
/s/ Joseph R. Tomkinson	Chairman of the Board, Chief Executive Officer and Director	March 31, 2005	
Joseph R. Tomkinson	(Principal Executive Officer)		
/s/ Richard J. Johnson	Executive Vice President and Chief Financial Officer (Principal Financial Officer)	March 31, 2005	
Richard J. Johnson	(Finicipal Finalicial Officer)		
/s/ William S. Ashmore	President and Director	March 31, 2005	
William S. Ashmore			
/s/ Todd R. Taylor	Senior Vice President and Controller	March 31, 2005	
Todd R. Taylor			
/s/ James Walsh	Director	March 31, 2005	
James Walsh			
/s/ Frank P. Filipps	Director	March 31, 2005	
Frank P. Filipps			
/s/ Stephan R. Peers	Director	March 31, 2005	
Stephan R. Peers	·		
/s/ William E. Rose	Director	March 31, 2005	
William E. Rose	·		
/s/ Leigh J. Abrams	Director	March 31, 2005	
Leigh J. Abrams	•		

CONSOLIDATED FINANCIAL STATEMENTS

INDEX

The consolidated financial statements as of and for the year ended December 31, 2004 are unaudited and the consolidated financial statements as of December 31, 2003 and for the years ended December 31, 2003 and 2002 have been derived from the audited financial statements for such periods.

Consolidated Balance Sheets as of December 31, 2004 and 2003	F-2
Consolidated Statements of Operations and Comprehensive Earnings for the years ended December 31, 2004, 2003 and 2002	F-3
Consolidated Statements of Changes in Stockholders' Equity for the years ended December 31, 2004 2003 and 2002	F-5
Consolidated Statements of Cash Flows for the years ended December 31, 2004, 2003 and 2002	F-6
Notes to Consolidated Financial Statements	F-8

The following consolidated financial statements do not include an audit report of KPMG LLP. Because no independent registered public accounting firm has opined that the unaudited consolidated financial statements included in this annual report present fairly, in all material respects, the financial position, the results of operations, cash flows and the changes in stockholders' equity for the year ended December 31, 2004 reported in accordance with generally accepted accounting principles in the United States of America, or "GAAP," such statements could be subject to change or the financial information included therein may be materially different from audited financial information. There can be no assurance that such changes or differences would not be significant. See "Management's Discussion and Analysis of Financial Condition and Results of Operations – Use of Unaudited Financial Statements" for a further discussion.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES

CONSOLIDATED BALANCE SHEETS

(dollar amounts in thousands, except share data)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

	At Dece	nber 31,
	2004	2003
ASSETS		
Cash and cash equivalents	\$ 324,351	\$ 125,153
Restricted cash	253,360	322
CMO collateral	21,308,906	8,644,079
Mortgages held-for-investment	586,686	652,814
Finance receivables	471,820	630,030
Allowance for loan losses	(63,955)	(38,596)
Mortgages held-for-sale	587,745	397,618
Accrued interest receivable	97,617	39,347
Other assets	249,237	127,190
Total assets	\$23,815,767	\$10,577,957
LIABILITIES AND STOCKHOLDERS' EQUITY		
CMO borrowings	\$21,206,373	\$ 8,489,853
Reverse repurchase agreements	1,527,558	1,568,807
Other liabilities	37,761	46,510
Total liabilities	22,771,692	10,105,170
Total Infollutes		
Commitments and contingencies		
Stockholders' equity:		
Series A junior participating preferred stock, \$0.01 par value; 2,500,000 shares authorized; none issued and outstanding as of December 31, 2004 and 2003	_	_
Series B 9.375% cumulative redeemable preferred stock, \$0.01 par value; liquidation value \$50,000; 2,000,000 shares authorized; 2,000,000 shares and none issued and outstanding as of December 31, 2004 and 2003, respectively	20	_
Series C 9.125% cumulative redeemable preferred stock, \$0.01 par value; liquidation value \$107,500; 5,500,000 shares		
authorized; 4,300,000 shares and none issued and outstanding as of December 31, 2004 and 2003, respectively	43	_
Common stock, \$0.01 par value; 200,000,000 shares authorized; 75,153,926 and 56,368,368 shares issued and outstanding		
as of December 31, 2004 and 2003, respectively	752	564
Additional paid-in capital	1,152,861	629,662
Accumulated other comprehensive income	979	4,356
Net accumulated deficit:		
Cumulative dividends declared	(513,453)	(307,031)
Retained earnings	402,873	145,236
Net accumulated deficit	(110,580)	(161,795)
Total stockholders' equity	1,044,075	472,787
Total liabilities and stockholders' equity	\$23,815,767	\$10,577,957

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE EARNINGS

(in thousands, except per share data)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

	For the	For the year ended December 31,		
	2004	2003	2002	
INTEREST INCOME:				
Mortgage assets	\$753,295	\$384,822	\$228,770	
Other assets	2,321	894	1,497	
Total interest income	755,616	385,716	230,267	
INTEREST EXPENSE:				
CMO borrowings	354,547	174,199	102,366	
Reverse repurchase agreements	57,837	32,382	23,583	
Other liabilities	149	2,428	1,852	
Total interest expense	412,533	209,009	127,801	
Net interest income	343,083	176,707	102,466	
Provision for loan losses	30,927	24,853	19,848	
110 (1010) 101 10111 100000				
Net interest income after provision for loan losses	312,156	151,854	82,618	
NON-INTEREST INCOME:				
Gain on sale of loans	25,134	39,022		
Other income	10,948	9,995	2,864	
Gain (loss) on derivative instruments	4,694	(16,021)	(50,502)	
Equity in net earnings of Impac Funding Corporation		11,537	11,299	
Total non-interest income	40,776	44,533	(36,339)	
NON-INTEREST EXPENSE:				
Personnel expense	60,420	25,250	1,856	
General and administrative and other expense	17,097	7,660	985	
Amortization of deferred tax charge	16,212	5,658		
Professional services	4,374	4,785	1,389	
Equipment expense	3,689	1,608	186	
Occupancy expense	3,658	1,560	161	
Data processing expense	3,608	1,829	162	
Amortization and impairment of mortgage servicing rights	2,063	1,290	_	
Impairment on investment securities available-for-sale	1,120	298	1,039	
Provision for repurchases	405	1,499	_	
(Gain) loss on sale of other real estate owned	(3,901)	(2,632)	154	
Total non-interest expense	108,745	48,805	5,932	
		4.5	40.0.5	
Net earnings before income taxes	244,187	147,582	40,347	
Income tax benefit	(13,450)	(1,397)		
Net earnings	257,637	148,979	40,347	
Less: Cash dividends on cumulative redeemable preferred stock	(3,750)			
Net earnings available to common stockholders	\$253,887	\$148,979	\$ 40,347	

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF OPERATIONS AND COMPREHENSIVE EARNINGS—(continued)

(in thousands, except per share data)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

	For the year ended December 31,			
	2004	2003	2002	
Net earnings	\$257,637	\$148,979	\$40,347	
Other comprehensive earnings:				
Unrealized gains (losses) on securities:				
Unrealized holding gains (losses) arising during year	71	2,272	389	
Less: Reclassification of losses (gains) included in earnings	(3,448)	(6,387)	(112)	
Net unrealized gains (losses) arising during year	(3,377)	(4,115)	277	
Comprehensive earnings	\$254,260	\$144,864	\$40,624	
Net earnings per share:				
Net earnings per share—basic	\$ 3.79	\$ 2.94	\$ 1.01	
Net earnings per share—diluted	\$ 3.72	\$ 2.88	\$ 0.99	
Dividends declared per common share	\$ 2.90	\$ 2.05	\$ 1.76	

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CHANGES IN STOCKHOLDERS' EQUITY

(dollar amounts in thousands, except share data)
(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

	Number of Preferred Shares Outstanding	Preferred Stock	Number of Common Shares Outstanding	Commo Stock		Additional Paid-In Capital	Accumulated Other Comprehensive Income		Notes leceivable Common Stock Sales	Cumulative Dividends Declared	Ea (Acc	etained arnings cumulated Deficit)	Stoc	Total ckholders' Equity
Balance, December 31, 2001	_	\$ —	32,001,997	\$ 32	20	\$ 359,279	\$ 8,194	\$	(920)	\$(126,952)	\$	(44,090)	\$	195,831
Dividends declared (\$1.76 per common share)	_		_	_	_	_	_		_	(74,002)		_		(74,002)
Common stock offering	_	_	11,197,000	11	12	98,004	_		_	_		_		98,116
Proceeds from exercise of stock options	_	_	150,824		1	992	_		_			_		993
Sale of stock via structured equity shelf	_	_	1,970,696	2	20	21,023	_		_	_		_		21,043
Advances on notes receivable from common stock sales	_	_	_	_	_	_	_		920	_		_		920
Net earnings, 2002	_	_	_	_	_	_	_		_	_		40,347		40,347
Other comprehensive gain	_	_	_	_	-	_	277		_	_				277
			45.000.545		_	450,000	0.474	_		(200.05.1)	_	(2.5.42)	_	202 525
Balance, December 31, 2002	_	_	45,320,517	45	53	479,298	8,471		_	(200,954)		(3,743)		283,525
Dividends declared (\$2.05 per common share)	_	_	_	_	_	_	_		_	(106,077)		_	(106,077)
Common stock offering	_		5,750,000	5	58	76,692	_		_	_		_	,	76,750
Proceeds from exercise of stock options	_	_	520,978		5	4,549	_		_	_		_		4,554
Sale of stock via equity distribution agreement.	_	_	4,769,186	2	18	68,998	_		_	_		_		69,046
Issuance of shares for the purchase of Impac Funding Corp.	_		7,687	_	_	125	_		_	_		_		125
Net earnings, 2003	_	_	_	_	-	_	_			_		148,979		148,979
Other comprehensive loss					_		(4,115)) _						(4,115)
Balance, December 31, 2003	_	_	56,368,368	56	54	\$ 629,662	4,356		_	(307,031)		145,236		472,787
Dividends declared (\$2.90 per common share)			_		_	_	_		_	(202,672)			(202,672)
Dividends declared on preferred shares	_	_	_	_	_	_	_		_	(3,750)		_	((3,750)
Series B and C preferred stock offering	6,300,000	63	_	_	_	152,186	_		_	_		_		152,249
Common stock offering	_	_	11,787,500	11	18	232,474	_			_		_		232,592
Proceeds and tax benefit from exercise of stock options	_	_	345,893		3	4,934	_		_	_		_		4,937
Sale of stock via equity distribution agreement.			6,652,165	f	67	133,605	_		_			_		133,672
Net earnings, 2004	_	_					_		_	_		257,637		257,637
Other comprehensive loss	_	_	_	_	_	_	(3,377)	_	_				(3,377)
comprehensive 1000							(5,577)	_						(3,377)
Balance, December 31, 2004	6,300,000	\$ 63	75,153,926	\$ 75	52	\$1,152,861	\$ 979	\$	_	\$(513,453)	\$ 4	402,873	\$1,	044,075

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS

(in thousands) (the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

	For the year ended December 31,					
	_	2004 2003		2003		2002
CASH FLOWS FROM OPERATING ACTIVITIES:						
Net earnings	\$	257,637	\$	148,979	\$	40,347
Adjustments to reconcile net earnings to net cash provided by operating activities:						
Equity in net earnings of Impac Funding Corporation				(11,537)		(11,299)
Provision for loan losses		30,927		24,853		19,848
Net change in deferred tax charge		(18,181)		(8,076)		_
Amortization of premiums and securitization costs		166,649		69,573		38,094
Net change in accrued interest receivable		(58,270)		(12,128)		(12,932)
Net change in investment in and advances to IFC		_		(21,319)		(301,805
Impairment of investment securities available-for-sale		1,120		298		1,039
Amortization and impairment of mortgage servicing rights		2,063		1,290		_
(Gain) loss on sale of other real estate owned		(3,901)		(2,632)		154
Gain on sale of loans		(25,134)		(39,022)		_
Purchase of mortgages held-for-sale	(2	22,213,104)	(5	5,960,645)		_
Sale and principal reductions on mortgages held-for-sale	2	22,037,869	ϵ	5,048,976		_
Depreciation and amortization		3,471		1,524		_
Unrealized (gain) loss on derivative instruments		(96,575)		(31,826)		22,141
Net change in deferred taxes		(3,061)		18,903		_
Recoveries and gain on sale of investment securities available-for-sale		(5,474)		(9,078)		_
Net change in restricted cash		(253,038)		(322)		(81
Net change in other assets and liabilities		(2,370)		(51,702)		13,141
	_		_		_	
Net cash provided by (used in) operating activities		(179,372)		166,109		(191,353
CASH FLOWS FROM INVESTING ACTIVITIES:						
Net change in CMO collateral	(1	2,827,524)	(3	3,529,784)	(2	2,972,873
Net change in finance receivables		158,210		33,991		(363,450
Purchase of premises and equipment		(6,312)		(1,816)		_
Cash received from acquisition of Impac Funding Corporation				23,510		_
Net change in mortgages held-for-investment		56,261		(595,860)		(44,384
Sale of investment securities available-for-sale		4,510		12,632		
Purchase of restricted investment securities available-for-sale, net		(2,563)		(2,206)		_
Purchase of investment securities available-for-sale		(3,920)		(15,252)		(10,269)
Dividends from Impac Funding Corporation				11,385		12,870
Net change in mortgage servicing rights		(887)		(5,620)		_
Principal reductions on investment securities available-for-sale		6,837		12,717		5,905
Proceeds from sale of other real estate owned		38,688		33,877		12,147
			_		_	
Net cash used in investing activities	(1	2,576,700)	(4	1,022,426)	(3	3,360,054
			_		_	
CASH FLOWS FROM FINANCING ACTIVITIES:						
Net change in reverse repurchase agreements and other borrowings		(41,249)		400,778		692,675
Net proceeds from CMO borrowings		7,644,706		5,925,794	3	3,851,024
Repayment of CMO borrowings	((4,963,984)	(2	2,480,966)		(985,658
Dividends paid common stock		(202,672)		(127,831)		(66,329
Dividends paid preferred stock		(3,750)		_		_
Net proceeds from sale of common stock		232,592		69,046		98,116
Net proceeds from sale of preferred stock		152,249		05,040		50,110

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF CASH FLOWS—(continued)

(in thousands)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

		For the	year ended Decemb	er 31,	
		2004	2003	2	2002
CASH FLOWS FROM FINANCING ACTIVITIES (continued):					
Net proceeds from sale of common stock via equity distribution agreement		133,672	76,750		21,043
Proceeds from exercise of stock options		3,706	4,554		993
Reduction on notes receivable from common stock sales			_		920
Net cash provided by financing activities	1	2,955,270	3,868,125	3,6	512,784
	<u> </u>	100 100	11 000		C1 277
Net change in cash and cash equivalents		199,198	11,808		61,377
Cash and cash equivalents at beginning of year	_	125,153	113,345		51,887
Cash and cash equivalents at end of year	\$	324,351	\$ 125,153	\$ 1	13,264
	_			_	
SUPPLEMENTARY INFORMATION:					
Interest paid	\$	368,123	\$ 193,494	\$ 1	26,185
Taxes paid		26,720	17,885		_
NON-CASH TRANSACTIONS:					
Accumulated other comprehensive gain (loss)	\$	(3,377)	\$ (4,115)	\$	277
Transfer of mortgages held-for-investment and held-for-sale to other real estate owned		4,215	5,776		1,839
Transfer of CMO collateral to other real estate owned		32,630	30,394		13,441
Transfer of finance receivables to other real estate owned		_	91		_
Dividends declared and unpaid		_	_		21,754
The following table presents the acquisition of the assets and liabilities of Impac Funding Corporat ASSETS ACQUIRED	ion as of July	1, 2003 (in t	housands):		
Cash and cash equivalents				\$	24,135
Mortgages held-for-sale				4	51,465
Accrued interest receivable					565
Other assets					91,962
				_	
Total assets				\$ 5	68,127
A VADAN MINING ACCOUNTED					
LIABILITIES ASSUMED Warehouse borrowings				¢ 1	47,951
Other liabilities					66,971
Deferred revenue					52,371
Deferred revenue					32,37
Total liabilities				5	67,293
Total stockholders' equity					834
Total liabilities and stockholders' equity				\$ 5	68,127
Net Assets Acquired:					
Investment in Impac Funding Corporation				\$	84
Cash paid for common stock				Φ	625
Cush paid 101 Collinoli Stock					
					104
Shares issued for common stock					125

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Note A—Summary of Business and Significant Accounting Policies

1. Business

Unless the context otherwise requires, the terms "Company," "we," "us," and "our" refer to Impac Mortgage Holdings, Inc. (IMH), a Maryland corporation incorporated in August 1995, and its subsidiaries, IMH Assets Corp. (IMH Assets), Impac Warehouse Lending Group, Inc. (IWLG), Impac Multifamily Capital Corporation (IMCC) and Impac Funding Corporation (IFC), together with its wholly-owned subsidiaries Impac Secured Assets Corp. (ISAC) and Novelle Financial Services, Inc. (Novelle).

We are a mortgage real estate investment trust (REIT) that is a nationwide acquirer, originator, seller and securitizer of non-conforming Alt-A mortgages (Alt-A mortgages). Alt-A mortgages are primarily first lien mortgages made to borrowers whose credit is generally within typical Fannie Mae and Freddie Mac guidelines, but have loan characteristics that make them non-conforming under those guidelines. Some of the principal differences between mortgages purchased by Fannie Mae and Freddie Mac and Alt-A mortgages are as follows:

- credit and income histories of the mortgagor;
- · documentation required for approval of the mortgagor; and
- loan balances in excess of maximum Fannie Mae and Freddie Mac lending limits.

For instance, Alt-A mortgages may not have certain documentation or verifications that are required by Fannie Mae and Freddie Mac and, therefore, in making our credit decisions, we are more reliant upon the borrower's credit score and the adequacy of the underlying collateral. We believe that Alt-A mortgages provide an attractive net earnings profile by producing higher yields without commensurately higher credit losses than other types of mortgages.

We operate three core businesses:

- long-term investment operations that is conducted by IMH, IMH Assets and IMCC;
- mortgage operations that is conducted by IFC, ISAC and Novelle; and
- warehouse lending operations that is conducted by IWLG.

The long-term investment operations generate earnings primarily from net interest income earned on mortgages held for long-term investment (long-term mortgage portfolio). The long-term mortgage portfolio as reported on our consolidated balance sheet consists of mortgages held as collateralized mortgage obligations (CMO) and mortgages held-for-investment. Investments in Alt-A mortgages and multi-family mortgages are initially financed with short-term borrowings under reverse repurchase agreements which are subsequently converted to long-term financing in the form of CMO financing. Cash flow from the long-term mortgage portfolio and proceeds from the sale of capital stock also finance the acquisition of new Alt-A and multi-family mortgages.

The mortgage operations acquire, originate, sell and securitize primarily Alt-A adjustable rate mortgages (ARMs) and fixed rate mortgages (FRMs) and, to a lesser extent, sub-prime mortgages (B/C mortgages) from correspondents, mortgage brokers and retail customers. Correspondents originate and close mortgages under their mortgage programs and then sell the closed loans to the mortgage operations on a flow (loan-by-loan) basis or through bulk sale commitments. Correspondents include savings and loan associations, commercial banks and mortgage bankers. The mortgage operations generate income by securitizing and selling mortgages to permanent investors, including the long-term investment operations. This business also earns revenue from fees associated with mortgage servicing rights, master servicing agreements and interest income earned on mortgages held-for-sale. The mortgage operations use warehouse facilities provided by the warehouse lending operations to finance the acquisition and origination of mortgages.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

The warehouse lending operations provide short-term financing to mortgage loan originators, including the mortgage operations, by funding mortgages from their closing date until sale to pre-approved investors. This business earns fees from warehouse transactions as well as net interest income from the difference between its cost of borrowings and the interest earned on warehouse advances.

2. Financial Statement Presentation

Principles of Consolidation

The financial condition and results of operations have been presented in the consolidated financial statements for the three-year period ended December 31, 2004 and include the financial results of IMH, IMH Assets, IWLG, IMCC and IFC (together with its wholly-owned subsidiaries Novelle and ISAC).

On July 1, 2003, IMH entered into a stock purchase agreement with Joseph R. Tomkinson, our Chairman, Chief Executive Officer and a director, William S. Ashmore, our Chief Operating Officer, President and a director, and the Johnson Revocable Living Trust, of which Richard J. Johnson, our Executive Vice President and Chief Financial Officer is trustee, whereby IMH purchased all of the outstanding shares of voting common stock of IFC for aggregate consideration of \$750 thousand. Messrs. Tomkinson and Ashmore and the Johnson Revocable Living Trust each owned one-third of the outstanding common stock of IFC. Mr. Tomkinson elected to receive \$125 thousand worth of his consideration for the sale of his IFC shares of common stock in the form of 7,687 shares of IMH common stock. The fairness opinion related to the purchase price of IFC, as rendered by an independent financial advisor, and the subsequent transaction was approved by our board of directors. As a result of acquiring 100% of IFC's common stock on July 1, 2003, IMH owns all of the common stock and preferred stock of IFC and began to consolidate IFC as of that date. Due to the consolidation of IFC on July 1, 2003, the consolidated financial statements include results of operations of the mortgage operations for the period from January 1, 2002 to June 30, 2003 (non-consolidation period) as equity in net earnings of IFC and the period from July 1, 2003 to December 31, 2003 (consolidated period) on a consolidated basis.

The accompanying consolidated financial statements include accounts of IMH and other entities in which the Company has a controlling financial interest. The usual condition for a controlling financial interest is ownership of a majority of the voting interests of an entity. However, a controlling financial interest may also exist in entities, such as special purpose entities (SPEs), through arrangements that do not involve voting interests.

There are two different accounting frameworks applicable to SPEs, depending on the nature of the entity and the Company's relation to that entity; the qualifying SPE (QSPE) framework under Statement of Financial Accounting Standards No. 140, "Accounting for Transfers and Servicing of Financial Assets and Extinguishment of Liabilities" (SFAS 140) and the variable interest entity (VIE) framework under the Financial Accounting Standards Board (FASB) Interpretation No. 46 (revised December 2003), "Consolidation of Variable Interest Entities" (FIN 46R).

The QSPE framework is applicable when an entity transfers (sells) financial assets to an SPE meeting certain criteria. These criteria are designed to ensure that the activities of the SPE are essentially predetermined in their entirety at the inception of the vehicle and that the transferor cannot exercise control over the entity, its assets or activities. Entities meeting these criteria are not consolidated by the Company. For further details, refer to Note 5—Mortgages Held-for-Sale.

When the SPE does not meet the QSPE criteria, consolidation is assessed pursuant to FIN 46R. A VIE is defined as an entity that (1) lacks enough equity investment at risk to permit the entity to finance its activities without additional subordinated financial support from other parties, (2) has equity owners who are unable to make decisions, and/or (3) has equity owners that do not absorb or receive the entity's losses and returns. QSPEs are excluded from the scope of FIN 46R

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

FIN 46R requires a variable interest holder (counterparty to a VIE) to consolidate the VIE if that party will absorb a majority of the expected losses of the VIE, receive a majority of the residual returns of the VIE, or both. This party is considered the primary beneficiary of the entity. The determination of whether the Company meets the criteria to be considered the primary beneficiary of a VIE requires an evaluation of all transactions (such as investments, liquidity commitments, derivatives and fee arrangements) with the entity.

Prior to the Company's adoption of FIN 46R, the decision of whether or not to consolidate an SPE depended on the applicable accounting principles for non-QSPEs, including a determination regarding the nature and amount of the investments made by third parties in the SPE. Consideration was given to, among other factors, whether a third party had made a substantive equity investment in the SPE; which party had voting rights, if any; who made decisions about the assets in the SPE; and who was at risk of loss. The SPE was consolidated if the Company retained or acquired control over the risks and rewards of the assets in the SPE.

The Company securitizes mortgages in the form of CMOs and real estate mortgage investment conduits (REMICs). The typical CMO securitization is designed such that the transferee (securitization trust) is not a QSPE and thus as the sole residual interest holder the Company consolidates such VIE. Amounts consolidated are classified as CMO collateral and CMO borrowings in the consolidated balance sheets. Generally, the typical REMIC securitization is designed so that the securitization trust is a QSPE and thus not consolidated by the Company. To the extent that our REMIC securitization trusts do not meet the QSPE criteria, consolidation is assessed pursuant to FIN 46R.

Investments in other companies in which the Company has significant influence over operating and financing decisions and holds more than a 20% voting interest, are accounted for in accordance with the equity method of accounting. Prior to July 1, 2003, IMH was entitled to 99% of the earnings or losses of IFC through its ownership of all of the non-voting preferred sock of IFC. Therefore, the Company has accounted for its 99% interest in IFC under the equity method for periods prior to July 1, 2003.

Inter-Company Eliminations

All significant inter-company balances and transactions have been eliminated in consolidation or under the equity method of accounting regarding transactions involving the mortgage operations prior to its consolidation.

Use of Estimates in the Preparation of Financial Statements

The accompanying consolidated financial statements of IMH and our subsidiaries (as defined above) have been prepared in accordance with accounting principles generally accepted in the United States of America (GAAP). In the opinion of management, all adjustments, consisting of normal recurring adjustments, considered necessary for a fair presentation have been included. Management has made a number of estimates and assumptions relating to the reporting of assets and liabilities, the disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period to prepare these consolidated financial statements in conformity with GAAP. Actual results could differ from those estimates.

Reclassifications

Certain amounts in the 2003 and 2002 consolidated financial statements have been reclassified to conform with the current year presentation.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

3. Cash and Cash Equivalents

For purposes of the consolidated statements of cash flows, cash and cash equivalents consist of cash and money market mutual funds. Investments with maturities of three months or less at date of acquisition are considered to be cash equivalents.

4. Restricted Cash

Restricted cash primarily consists of cash deposits in a CMO securitization trust that will be used to finance the remaining mortgage loan collateral that will be deposited into the trust within 15 to 30 days of the issuance of the CMO. In addition, restricted cash includes money market accounts held in the Company's deferred compensation plan and escrow accounts related to the Company's master servicing activities.

5. Investment Securities Available-for-Sale

Investment securities are classified as held-to-maturity, available-for-sale, and/or trading securities and are included in other assets on our consolidated financial statements. Held-to-maturity investment securities are reported at amortized cost, available-for-sale securities are reported at fair value with unrealized gains and losses reported in earnings. Gains and losses realized on the sale of available-for-sale investment securities and declines in value judged to be other-than-temporary are based on the specific identification method and reported in current earnings. Premiums or discounts obtained on investment securities are accreted or amortized to interest income over the estimated life of the investment securities using the effective interest method. Investment securities may be subject to credit, interest rate and/or prepayment risk. The amortized cost and estimated fair value of investment securities available-for-sale for the periods indicated are presented as follows:

	Amortized Cost	Gross Unrealized Gain	Gross Unrealized Loss	Estimated Fair Value
As of December 31, 2004:				
Subordinated securities secured by mortgages	\$ 24,851	\$ 905	\$ (329)	\$ 25,427
Restricted investments (1)	3,786	403	_	4,189
	\$ 28,637	\$ 1,308	\$ (329)	\$ 29,616
As of December 31, 2003:				
Subordinated securities secured by mortgages	\$ 24,534	\$ 4,310	\$ (77)	\$ 28,767
Restricted investments (1)	1,783	123	_	1,906
U.S. Treasuries	2,830	_		2,830
	\$ 29,147	\$ 4,433	\$ (77)	\$ 33,503

⁽¹⁾ Investments related to the Company's deferred compensation program are classified as available-for-sale. The deferred compensation investments are included in other assets in the Company's consolidated balance sheets.

Investment securities available-for-sale include securities placed with third parties of \$15.3 million and \$18.2 million as of December 31, 2004 and 2003, respectively, that are pledged as collateral for margin calls on derivative instruments, if necessary, depending on the change in the fair value of the derivative instruments. Gross realized gains from the sale of investment securities available-for-sale were \$5.1 million during the year ended December 31, 2004. There were no realized gains or losses from the sale of available-for-sale securities during the years ended December 31, 2003 and 2002. During the year ended December 31, 2004 and 2003, we received \$389 thousand and \$5.2 million, respectively, of recoveries on investment securities available-for-sale that were written-off in prior periods.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

6. Mortgages Held-for-Sale

Mortgages held-for-sale consist primarily of Alt-A mortgages, which are secured by one-to-four family residential real estate located throughout the United States. The mortgage operations acquire and originate mortgages generally with the intent to sell them in the secondary market (primarily in REMIC securitizations and to a lesser extent on a whole loan basis) or to the long-term investment operations. Mortgages held-for-sale are carried at the lower of aggregate cost, net of purchase discounts or premiums and deferred fees, or market. We determine the fair value of mortgages held-for-sale using current secondary market prices for loans with similar coupons, maturities and credit quality.

SFAS 140 requires that a transfer of financial assets in which we surrender control over the assets be accounted for as a sale to the extent that consideration other than beneficial interests in the transferred assets is received in exchange. SFAS 140 requires a "true sale" analysis of the treatment of the transfer under law as if the Company was a debtor under the bankruptcy code. A "true sale" legal analysis includes several legally relevant factors, such as the nature and level of recourse to the transferor and the nature of retained servicing rights. Once the legal isolation test has been met under SFAS 140, other factors concerning the nature and extent of the transferor's control over the transferred assets are taken into account in order to determine whether de-recognition of assets is warranted, including whether the SPE has complied with rules concerning QSPE's.

A legal opinion regarding legal isolation for each securitization has been obtained by the Company. The "true sale" opinion provides reasonable assurance the purchased assets would not be characterized as the property of the transferring Company's receivership or conservatorship estate in the event of insolvency and also states the transferor would not be required to substantively consolidate the assets and liabilities of the SPE with those of the transferor upon such event.

The REMIC securitization process involves the sale of the loans to one of our wholly-owned bankruptcy remote special-purpose entities which then sells the loans to a separate, transaction-specific securitization trust in exchange for cash and certain trust interests that we retain. The securitization trust issues and sells undivided interests to third party investors that entitle the investors to specified cash flows generated from the securitized loans. These undivided interests are usually represented by certificates with varying interest rates, and are secured by the payments on the loans acquired by the trust, and commonly include senior and subordinated classes. The senior class securities are usually rated "AAA" by at least two of the major independent rating agencies and have priority over the subordinated classes in the receipt of payments. We have no obligation to provide funding support to either the third party investors or the securitization trusts. The third party investors or the securitization trusts generally have no recourse to our assets or us and have no ability to require us to repurchase their securities other than standard representations and warranties. We do make certain representations and warranties concerning the loans, such as lien status or mortgage insurance coverage, and if we are found to have breached a representation or warranty we may be required to repurchase the loan from the securitization trust. We do not guarantee any certificates issued by the securitization trusts. Generally, the securitization trusts represent QSPE's under SFAS 140, and are therefore not consolidated for financial reporting purposes. To the extent that a securitization trust does not meet the QSPE criteria, the securitization trust is evaluated in accordance with SFAS 140 to determine if there are any constraints on pledging or exchanging the assets it received that provide more than a trivial benefit to the transferor. In addition, any securitization trust that does not meet the QSPE criteria is evalu

In addition to the cash the securitization trust pays to the Company for the loans, we may retain certain interests in the securitization trust as part of the trust's payment to us for the loans. These retained interests may include subordinated classes of securities, interest-only securities, residual securities and master servicing rights. These retained interests are included in investment securities available-for-sale and other assets on the consolidated balance sheets. Transaction costs associated with the securitizations are recognized as a component of the gain or loss at the time of sale.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

When the Company securitizes mortgage loans, the carrying value of the mortgages sold is allocated between the loans sold and the retained interests based on their relative fair values. Our recognition of gain or loss on the sale of loans from REMIC securitizations is accounted for in accordance with SFAS 140 and represents the difference between the cash proceeds and the allocated cost of the loans sold and interests retained. At the closing of each securitization, mortgages held-for-sale are removed from the consolidated balance sheets and cash received and any portion of the mortgages retained from the securitizations (retained interests) are added to the consolidated balance sheet.

7. Gain on Sale of Mortgage Servicing Rights

The sub-servicing of mortgage servicing rights created in our CMO and REMIC securitizations are generally sold to third parties concurrent with the securitization of the mortgages. We believe that the sale of sub-servicing is consistent with the accounting for the sale of servicing, therefore, the sales of mortgage servicing rights are recognized in accordance with AICPA Statement of Position 01-6, "Accounting by Certain Entities (Including Entities with Trade Receivables) that Lend to or Finance the Activities with Others" (SOP 01-6) and Emerging Issues Task Force No. 95-5, "Determination of What Risks and Rewards, If any, Can be Retained and Whether Any Unresolved Contingencies May Exist in a Sale of Mortgage Loan Servicing Rights" when the following conditions have been met: (1) title has passed, (2) substantially all risks and rewards of ownership have irrevocably passed to the buyer, and (3) any protection provisions retained by the seller are minor and can be reasonably estimated. The Company believes that based on the terms and conditions of the related sales agreements all of the above conditions have been met.

The gains or losses on sale of mortgage servicing rights to third parties in CMO securitizations are accounted for in accordance with the provisions in SOP 01-6. Under SOP 01-6, for sales of mortgage servicing rights with the loans being retained, the carrying value of the loan is allocated between the loan basis and the mortgage servicing rights basis consistent with the relative fair value method prescribed in SFAS 140. As a result, only a nominal gain is realized from the sale of mortgage servicing rights and a discount is recorded on the mortgages retained as CMO collateral and mortgages held-for-investment. The discount is amortized to interest income over the estimated life of the mortgages using the interest method as an adjustment to the yield of the mortgages. Management utilizes an estimate of the prepayment rate of the mortgages to forecast the remaining average life of the mortgages.

The gains or losses on sale of mortgage servicing rights to third parties in REMIC securitizations are accounted for in accordance with SFAS 140 and SOP 01-6. Since the sale of the mortgage servicing rights to third parties generally occurs concurrently with the REMIC securitization, the carrying value of the securitized mortgage loans is allocated between the mortgages sold, mortgage servicing rights to be sold, and retained interests (master servicing rights) based on their relative fair values. A gain or loss on sale of mortgage servicing rights is recorded based upon the difference between its sales price and associated relative fair value.

8. Master Servicing Rights

Generally, master servicing rights are retained when the sub-servicing of mortgage servicing rights are sold and the corresponding mortgages are retained in CMO securitizations. In addition, master servicing rights are generally retained when the sub-servicing of mortgage servicing rights are sold and the corresponding mortgages are sold in REMIC securitizations. The retained master servicing rights are recorded as a separate retained asset in accordance with SFAS 140 in the REMIC securitizations, while in the CMO securitizations such rights remain as part of the retained mortgage loans.

Master servicing rights retained in REMIC securitizations are recorded in other assets in the consolidated balance sheets. The Company records master servicing rights arising from the transfer of mortgages to the securitization trusts utilizing the relative fair value allocation method based upon an estimate of what a third party would pay for the master servicing rights. The master servicing rights are amortized in proportion to and over the estimated period of net servicing income. The Company subsequently evaluates and measures the master servicing rights for impairment using a discounted cash flows valuation model to estimate the fair value. The valuation model incorporates assumptions relating to market discount rates, float values, prepayment speeds, master servicing fees, and default rates. An

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

impairment loss is recognized for master servicing rights that have an unamortized balance in excess of the estimated fair value. Master servicing rights retained in CMO securitizations remain as part of the mortgage loan balance, and are accounted for as part of such loan.

The servicing fee income associated with the master servicing rights is reported in other income in the consolidated statements of operations. Also reported in other income is any subservicing expense incurred during the period prior to the securitization. The amortization and impairment of mortgage servicing rights are classified separately in the consolidated statements of operations.

Master servicing fees are generally 0.03% per annum on the declining principal balances of the mortgages serviced. The value of master servicing fees is subject to prepayment and interest rate risks on the transferred financial assets. Carrying value of master servicing rights were \$3.5 million and \$2.5 million for the year ended December 31, 2004, and 2003, respectively.

As of December 31, 2004, we master serviced mortgages for others of approximately \$6.5 billion that were primarily mortgages collateralizing REMIC securitizations. Related fiduciary funds are held in trust for investors in non-interest bearing accounts. We may also be required to advance funds or we may cause our loan servicers to advance funds to cover interest payments not received from borrowers depending on the status of their mortgages.

9. CMO Collateral and Mortgages Held-for-Investment

The long-term investment operations invest in primarily Alt-A ARMs, FRMs secured by first liens on single-family residential real estate properties acquired and originated by the mortgage operations, multi-family residential real estate properties originated by IMCC and, to a lesser extent, fixed rate second trust deeds secured by single-family residential real estate properties to be held for long-term investment. After accumulating a pool of mortgages of generally between \$200.0 million and \$2.5 billion, mortgages held-for-investment on our financial statements are securitized as CMOs and the mortgages are deposited in a trust and at that time we record the mortgages as CMO collateral. CMO collateral is recorded in IMH Assets, a special purpose financing subsidiary which is used to issue CMO financing. The typical CMO securitization is designed such that the securitization trust is not a QSPE and thus as the sole residual interest holder the Company consolidates the securitization trust. Generally, this is achieved by including terms in the securitization agreements that give the Company the ability to unilaterally cause the securitization trust to return specific mortgages, other than through a clean-up call.

CMO collateral and mortgages held-for-investment are recorded at cost, net of premiums, discounts and adjustments for derivative gains or losses during the commitment period. Premiums, discounts, and derivative gains or losses during the commitment period are amortized to interest income over the estimated lives of the mortgages using the interest method as an adjustment to the yield of the mortgages. Management utilizes an estimate of the prepayment rate of the mortgages to forecast the remaining average life of the mortgages.

CMO collateral and mortgages held-for-investment are continually evaluated for collectibility and, if appropriate, the mortgage may be placed on non-accrual status, generally when the mortgage is 90 days past due, and previously accrued interest is reversed from income.

10. Finance Receivables

Finance receivables represent transactions with customers involving residential real estate lending. As a warehouse lender, the warehouse lending operations is a secured creditor of the mortgage bankers and brokers to which it extends credit and is subject to the risks inherent in that status, including the risk of borrower fraud, default and bankruptcy. Any claim of the warehouse lending operations as a secured lender in a bankruptcy proceeding may be subject to adjustment and delay. Finance receivables represent warehouse lines of credit with mortgage bankers that are primarily collateralized by mortgages on single-family residential real estate. Terms of non-affiliated warehouse lines, including the maximum warehouse line amount and interest rate, are determined based upon the financial strength, historical

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

performance and other qualifications of the borrower. The warehouse lines have maturities that range from on-demand to one year. Finance receivables are stated at the principal balance outstanding. Interest income is recorded on the accrual basis.

At the end of the first quarter of 2004, we discovered that one client of the warehouse lending operations and certain of its officers had perpetrated a fraud pursuant to which they defrauded the warehouse lending operations into making advances pursuant to a warehouse line of credit. As of the date the fraud was discovered, an aggregate of \$12.6 million of fraudulent loan advances were outstanding. We immediately terminated the warehouse line of credit and have been cooperating with federal investigators in their ongoing investigation of the defrauding parties.

We retained an independent consultant to investigate the matter. The investigator reported that no principals of the warehouse lending operations had knowingly participated in the fraud. As a result of the fraud, during 2004 we established a specific allowance for loan losses in the amount of \$8.0 million to provide for anticipated losses on the fraudulent warehouse advances as we have deemed this amount to be non-collectible. Based on available information, we believe we will be able to recover the remaining \$4.6 million of related warehouse advances over time. To the extent that we believe that the actual losses will exceed the \$8.0 million allowance, we will make an additional allowance for loan losses when, or if, we determine it is appropriate to do so as events and circumstances dictate. However, we believe that this specific allowance is adequate to provide for anticipated loan losses based on currently available information.

During the year ended December 31, 2004, we terminated a warehouse lending client that sold mortgages to third party investors that were pledged as collateral to our warehouse lending operations, whereby, the sales proceeds from these loans were wired by the third party investor directly to our customer without the customer repaying their borrowings to us. The warehouse lending operations contacted the investors who purchased these loans to notify them of our interest in these loans. As a result of the termination of this client, we seized the remaining available loans that were secured as collateral in settlement of a portion of these borrowings. In certain cases, investors have released their interest in loans secured by our warehouse advances previously purchased by them and we are pursuing legal action on any remaining loans secured by our warehouse lending advances in order to perfect our ownership interest in these loans. As a result, management has provided for a specific write-down of \$2.7 million on these warehouse lending advances which, based upon management's judgment, is sufficient to cover any inherent losses on these warehouse lending advances.

The cumulative amount of specific write-downs on warehouse lending advances as of December 31, 2004 was \$10.7 million. Management believes that the aggregate specific allowance of \$10.7 million, which is provided for in the allowance for loan losses, is adequate to provide for future losses based on currently available information.

11. Allowance for Loan Losses

An allowance is maintained for losses on mortgages held-for-investment, mortgages held as CMO collateral and finance receivables (loans provided for) at an amount that management believes provides for losses inherent in those loan portfolios. We have implemented a methodology designed to analyze the performance of various loan portfolios, based upon the relatively homogeneous nature within these loan portfolios. The allowance for losses is also analyzed using the following factors:

- management's judgment of the net loss potential of mortgages in the long-term mortgage portfolio based on prior loan loss experience;
- changes in the nature and volume of the long-term mortgage portfolio;
- value of the collateral;
- expected losses as derived from rating agencies analyses;

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

- delinquency trends; and
- current economic conditions that may affect the borrowers' ability to pay.

In evaluating the adequacy of the allowance for loan losses, a detailed analysis of historical loan performance data is accumulated and reviewed. This data is analyzed for loss performance and prepayment performance by product type, origination year and securitization issuance. The results of that analysis are then applied to the current mortgage portfolio and an estimate is created. We believe that pooling of mortgages with similar characteristics is an appropriate methodology in which to evaluate the allowance for loan losses. Management also recognizes that there are qualitative factors that must be taken into consideration when evaluating and measuring inherent loss in our loan portfolios. These items include, but are not limited to, economic indicators that may affect the borrower's ability to pay, changes in value of collateral, political factors and industry statistics.

Additions to the allowance are provided through a charge to earnings. Specific valuation allowances may be established for loans that are deemed impaired, if default by the borrower is deemed probable, and if the fair value of the loan or the collateral is estimated to be less than the gross carrying value of the loan. Actual losses on loans are recorded as a reduction to the allowance through charge-offs. Subsequent recoveries of amounts previously charged off are credited to the allowance.

For loans on non-accrual status, cash receipts are applied, and interest income is recognized, on a cash basis. For all other impaired loans, cash receipts are applied to principal and interest in accordance with the contractual terms of the loan and interest income is recognized on the accrual basis. Generally, a loan may be returned to accrual status when all delinquent principal and interest are brought current in accordance with the terms of the loan agreement.

12. Premises and Equipment

Premises and equipment are stated at cost, less accumulated depreciation or amortization and are included in other assets on our consolidated financial statements. Depreciation on premises and equipment is recorded using the straight-line method over the estimated useful lives of individual assets, typically, three to twenty years. Premises and equipment consisted of the following for the periods indicated:

	At ember 31, 2004	Dec	ember 31, 2003
Premises and equipment	\$ 24,250	\$	17,984
Less: Accumulated depreciation	 (15,158)		(11,733)
Total premises and equipment	\$ 9,092	\$	6,251

13. CMO Borrowings

The decision to issue CMOs is based on our current and future investment needs, market conditions and other factors. CMOs, which are primarily secured by Alt-A mortgages on single-family and multi-family residential real properties, are issued as a means of financing our long-term mortgage portfolio. CMOs are carried at their outstanding principal balances, including securitization costs and accrued interest on such obligations. For accounting purposes, mortgages financed through the issuance of CMOs are treated as assets and the CMOs are treated as debt when the CMO qualifies as a secured borrowing arrangement.

Each issuance of a CMO is fully payable from the principal and interest payments on the underlying mortgages collateralizing such debt. CMOs typically are structured as one-month London Interbank Offered Rate (LIBOR) "floaters" and fixed rate securities with interest payable to certificate holders monthly. The maturity of each class of CMO is directly affected by the rate of principal prepayments on the related CMO collateral. Each CMO series is also subject to redemption according to specific terms of the respective indentures. As a result, the actual maturity of any class of a CMO series is likely to occur earlier than the stated maturities of the underlying mortgages.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

When we issue CMOs for financing purposes, we generally seek an investment grade rating for our CMOs by nationally recognized rating agencies. To secure such ratings, it is often necessary to incorporate certain structural features that provide for credit enhancement. This can include the pledge of collateral in excess of the principal amount of the securities to be issued, a bond guaranty insurance policy for some or all of the issued securities, or additional forms of mortgage insurance. The need for additional collateral or other credit enhancements depends upon factors such as the type of collateral provided, the interest rates paid, the geographic concentration of the mortgaged property securing the collateral and other criteria established by the rating agencies. The pledge of additional collateral reduces our capacity to raise additional funds through short-term secured borrowings or additional CMOs and diminishes the potential expansion of our long-term mortgage portfolio. Our total loss exposure is limited to the net economic investment in the CMOs at any point in time.

14. Income Taxes

We operate so as to qualify as a REIT under the requirements of the Internal Revenue Code (the Code). Requirements for qualification as a REIT include various restrictions on ownership of IMH's stock, requirements concerning distribution of taxable income and certain restrictions on the nature of assets and sources of income. A REIT must distribute at least 90% of its taxable income to its stockholders of which 85% must be distributed within the taxable year in order to avoid the imposition of an excise tax. The remaining balance may extend until timely filing of our tax return in the subsequent taxable year. Qualifying distributions of taxable income are deductible by a REIT in computing taxable income. If in any tax year IMH should not qualify as a REIT, we would be taxed as a corporation and distributions to stockholders would not be deductible in computing taxable income. If IMH were to fail to qualify as a REIT in any tax year, we would not be permitted to qualify for that year and the succeeding four years.

IFC is a taxable REIT subsidiary (TRS) and is therefore subject to corporate income taxes. Deferred tax assets and liabilities are recognized for the future tax consequences attributable to differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax base. Deferred tax assets and liabilities are measured using enacted tax rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled. The effect on deferred tax assets and liabilities of a change in tax rates is recognized in income in the period that includes the enactment date.

In accordance with Accounting Research Bulletin No. 51, "Consolidated Financial Statements," the Company records a deferred tax charge to eliminate the expense recognition of income taxes paid on inter-company profits that result from the sale of mortgages from IFC to IMH. The deferred tax charge is included in other assets in the consolidated balance sheets. The deferred tax charge is amortized as non-interest expense in the consolidated statements of operations over the estimated life of the mortgages retained in the long-term mortgage portfolio. Included in equity in net earnings of IFC in the consolidated statement of operations during the years ended December 31, 2003 and 2002 was \$4.3 million and \$2.9 million, respectively, of amortization of deferred tax charge.

15. Net Earnings per Share

Basic net earnings per share are computed on the basis of the weighted average number of shares outstanding for the year divided into net earnings available to common stockholders for the year. Diluted net earnings per share are computed on the basis of the weighted average number of shares and dilutive common equivalent shares outstanding for the year divided by net earnings available to common stockholders for the year.

16. Stock Options

Stock options and awards may be granted to the members of the board of directors, officers and key employees. The exercise price for any qualified incentive stock options (ISOs), non-qualified stock options (NQSOs) granted under our stock option plans may not be less than 100% (or 110% in the case of ISOs granted to an employee who is deemed to own in excess of 10% of the outstanding common stock) of the fair market value of the shares of common stock at the time the NQSO or ISO is granted.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

In December 2002 the FASB issued SFAS No. 148, "Accounting for Stock-Based Compensation – Transition and Disclosure" (SFAS 148), an amendment of FASB Statement No. 123, "Accounting for Stock-Based Compensation," (FASB 123). SFAS 148 amends FASB 123 to provide alternative methods of transition for a voluntary change to the fair value method of accounting for stock-based employee compensation. In addition, SFAS 148 amends the disclosure requirements of SFAS 123 to require prominent disclosures in both annual and interim financial statements. On January 1, 2003, IMH adopted the disclosure requirements of SFAS 148. This statement establishes financial accounting standards for stock-based employee compensation plans. SFAS 123 permits management to choose either a fair value based method or the Accounting Principals Board Opinion No. 25 "Accounting For Stock Issued to Employees" (APB 25) intrinsic value based method of accounting for its stock-based compensation arrangements. SFAS 123 requires pro forma disclosures of net earnings (loss) computed as if the fair value based method had been applied in financial statements of companies that continue to follow current practice in accounting for such arrangements under APB 25. SFAS 123 applies to all stock-based employee compensation plans in which an employer grants shares of its stock or other equity instruments to employees except for employee stock ownership plans. SFAS 123 also applies to plans in which the employer incurs liabilities to employees in amounts based on the price of the employer's stock, i.e., stock option plans, stock purchase plans, restricted stock plans and stock appreciation rights. The statement also specifies the accounting for transactions in which a company issues stock options or other equity instruments for services provided by non-employees or to acquire goods or services from outside suppliers or vendors.

The Company applies APB 25 in accounting for stock-based awards to employees. No compensation cost has been recognized for stock-based awards to employees as the stock option exercise price is equal to the fair market value of the underlying common stock as of the stock option grant date. Summarized below are the pro forma effects on net earnings and earnings per share data, as if the Company had elected to use the fair value approach prescribed by SFAS 123 to account for its employee stock-based compensation plans:

	For the y	ıber 31,	
	2004	2003	2002
Net earnings available to common stockholders	\$253,887	\$148,979	\$40,347
Less: Total stock-based employee compensation expense using the fair value method	(1,705)	(1,158)	(764)
Pro forma net earnings	\$252,182	\$147,821	\$39,583
Net earnings per share as reported:			
Basic	\$ 3.79	\$ 2.94	\$ 1.01
Diluted	\$ 3.72	\$ 2.88	\$ 0.99
Pro forma net earnings per share:			
Basic	\$ 3.77	\$ 2.91	\$ 0.99
Diluted	\$ 3.71	\$ 2.85	\$ 0.97

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

The fair value of options granted, which is amortized to expense over the option vesting period in determining pro forma net earnings, is estimated on the date of grant using the Black-Scholes option pricing model with the following weighted average assumptions:

	For t	For the year ended December 31,				
	2004	2003	2002			
Risk-free interest rate	2.16%-4.50%	1.56%-4.18%	2.32%-4.47%			
Expected lives (in years)	3-4	3	1-3			
Expected volatility	42.26%	28.83%	35.07%			
Expected dividend yield	10.00%	10.00%	10.00%			
Fair value per share	\$3.71	\$1.09	\$1.19			

During the periods in which the mortgage operations was accounted for under the equity method, grants of stock options by IMH to IFC employees were not accounted for under APB 25 but were accounted for at fair value consistent with the provisions specified under SFAS 123.

17. Derivative Instruments

In June 1998, FASB issued SFAS 133, "Accounting for Derivative Instruments and Hedging Activities," as amended by SFAS No. 137, "Accounting for Derivative Instruments and Hedging Activities—Deferral of the Effective Date of FASB Statement No. 133—an amendment of FASB Statement No. 133," and SFAS No. 138, "Accounting for Certain Derivative Instruments and Certain Hedging Activities-an amendment of FASB Statement No. 133," collectively, (SFAS 133). SFAS 133, subsequently amended by SFAS 149, "Amendment of Statement 133 on Derivative Instruments and Hedging Activities," establishes accounting and reporting standards for derivative instruments, including a number of derivative instruments embedded in other contracts, and for hedging activities. It requires that an entity recognize all derivatives as either assets or liabilities in the balance sheet and measure those instruments at fair value. If specific conditions are met, a derivative may be specifically designated as (1) a hedge of the exposure to changes in the fair value of a recognized asset or liability or an unrecognized firm commitment; (2) a hedge of the exposure to variable cash flows of a forecasted transaction; or (3) a hedge of the foreign currency exposure of a net investment in a foreign operation, an unrecognized firm commitment, an available-for-sale security or a foreign-currency-denominated forecasted transaction. For derivatives that are not designated as a hedging instrument, any change in fair value is recorded as an expense or income in the current period.

Rate Lock and Purchase Commitments

We enter into commitments to make loans whereby the interest rate on the loan is set prior to funding (rate lock commitments). We also enter into commitments to purchase mortgage loans through our correspondent channel (purchase commitments). Rate lock commitments on mortgage loans that are intended to be sold are considered to be derivatives. In addition, purchase commitments for mortgage loans that are intended to be sold and those that will be held for investment purposes can qualify as derivatives. Both types of commitments to purchase loans are evaluated under the definition of a derivative to determine whether SFAS 133 is applicable. Rate lock and purchase commitments that are considered to be derivatives are recorded at fair value in the consolidated statements of financial condition with changes in fair value recorded in gain (loss) on derivative instruments in the consolidated statements of operations. Subsequent to the April 1, 2004 issuance of Staff Accounting Bulletin No. 105, "Application of Accounting Principles to Loan Commitments," (SAB 105), when measuring the fair value of rate lock and purchase commitments, the amount of the expected servicing rights is not included in the valuation. The fair value is calculated and adjusted using an anticipated fallout factor for loan commitments that are not expected to be funded.

Forward Commitments

The policy of recognizing the fair value of the rate lock and purchase commitments has the effect of recognizing a gain or loss on the related mortgage loans based on changes in the interest rate environment before the mortgage loans are funded and sold. As such, both rate lock and purchase commitments expose us to interest rate risk. We mitigate such risk by entering into forward commitments and derivative transactions, such as mandatory commitments on U.S. Treasury bonds and mortgage-backed securities, call options and put options. These forward commitments and derivative transactions are treated as derivatives under the provisions of SFAS 133 with the change in fair value reported as a gain (loss) on derivative instruments.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Interest Rate Swaps, Caps, and Floors

Our primary objective is to limit the exposure to the variability in future cash flows attributable to the variability of one-month LIBOR, which is the underlying index of adjustable rate CMO and warehouse borrowings. We also monitor, on an ongoing basis, the prepayment risks that arise in fluctuating interest rate environments. Our interest rate risk management policies are formulated with the intent to mitigate the potential adverse effects of changing interest rates on CMO and warehouse borrowings.

To mitigate exposure to the effect of changing interest rates on cash flows on CMO and warehouse borrowings, we purchase derivative instruments, primarily in the form of interest rate swap agreements (swaps) and, to a lesser extent, interest rate cap agreements (caps) and interest rate floor agreements (floors). A swap is generally a contractual agreement that obligates one party to receive or make cash payments based on an adjustable rate index and the other party to receive or make cash payments based on a fixed rate. Swaps have the effect of fixing borrowing costs on a similar amount of debt and, as a result, can reduce the interest rate variability of borrowings. A cap or floor is a contractual agreement for which we may receive or pay a fee. If prevailing interest rates reach levels specified in the cap or floor agreement, we may either receive or pay cash. Our objective is to lock in a reliable stream of cash flows when interest rates fall below or rise above certain levels. For instance, when interest rates rise, borrowing costs may increase at greater speeds than the underlying collateral supporting the borrowings. These derivative instruments limit exposure to the variability of forecasted cash flows attributable to CMO and warehouse borrowings and protect net interest income by providing cash flows at certain triggers during changing interest rate environments. The swaps, caps and floors are treated as derivatives under the provisions of SFAS 133 with changes in fair value reported as gain (loss) on derivative instruments in the consolidated statements of operations. Cash paid or received on swaps, caps and floors is recorded as a current period expense or income in gain (loss) on derivative instruments in the consolidated statements of operations.

Credit Risk

The Company's exposure to credit risk on derivative instruments is limited to the cost of replacing contracts should the counterparty fail. The Company seeks to minimize credit risk through the use of credit approval and review processes, the selection of only the most creditworthy counterparties, continuing review and monitoring of all counterparties, exposure reduction techniques and thorough legal scrutiny of agreements.

18. Assets Acquired Through Foreclosure

Real estate owned, which consists of residential real estate acquired in satisfaction of loans, is carried at the lower of cost or estimated fair value less estimated selling costs. Adjustments to the loan carrying value required at the time of foreclosure are charged off against the allowance for loan losses. Losses or gains from the ultimate disposition of real estate owned are recorded as (gain) loss on sale of other real estate owned in the consolidated statement of operations.

19. Recent Accounting Pronouncements

In December 2003, the Accounting Standards Executive Committee of the American Institute of Certified Public Accountants (AICPA) issued SOP 03-3, "Accounting for Certain Loans or Debt Securities Acquired in a Transfer" (SOP 03-3). SOP 03-3 addresses the accounting for acquired impaired loans, which are loans that show evidence of having deteriorated in terms of credit quality since their origination. SOP 03-3 is effective for loans acquired after December 31, 2004. Because the Company acquires newly originated and loans generally underwritten to "A quality" standards, the Company does not expect the adoption of SOP 03-3 to have a material effect on the financial condition, results of operations, or liquidity of the Company. In June 2004, the FASB reached consensus opinions detailed in EITF

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

03-01, "The Meaning of Other-Than-Temporary Impairment and Its Application to Certain Investments" (EITF 03-1). Such consensuses require an investor to determine when an investment is considered impaired, evaluate whether that impairment is other than temporary, and, if the impairment is other than temporary, recognize an impairment loss equal to the difference between the investment's cost and its fair value. The guidance also includes accounting considerations subsequent to the recognition of an other-than-temporary impairment and requires certain disclosures about unrealized losses that have not been recognized as other-than-temporary impairments. The impairment loss recognition and measurement guidance was to be applicable to other-than-temporary impairment evaluations in reporting periods beginning after June 15, 2004. In September 2004, the FASB proposed additional guidance related to debt securities that are impaired because of interest rate and/or sector spread increases, and delayed the effective date of EITF 03-01. The Company does not expect the adoption of EITF 03-01 to have a material effect on the financial condition, results of operations, or liquidity of the Company. EITF 03-01 also includes disclosure requirements for investments in an unrealized loss position for which other-than-temporary impairments have not been recognized.

In December 2004, the FASB issued Statement No. 123(R), "Share-Based Payment" (SFAS 123R). It requires all public companies to report share-based compensation expense at the grant date fair value of the related share-based awards. The Company is required to adopt the provisions of the SFAS 123R effective for periods beginning after June 15, 2005. The impact of adoption of SFAS 123R cannot be predicted at this time because it will depend on levels of share-based payments granted in the future. However, had we adopted SFAS 123R in prior periods, the impact of SFAS 123R would have approximated the impact of SFAS 123 as described in the disclosure of pro forma net earnings per share in Note 15 Stock Options.

Note B-Mortgages Held-for-Sale

Mortgages held-for-sale for the periods indicated consisted of the following:

	December 31, 2004	December 31, 2003
		ф. 20 5 100
Mortgages held-for-sale	\$ 576,777	\$ 385,108
Net premiums on mortgages held-for-sale	10,968	12,510
Total mortgages held-for-sale	\$ 587,745	\$ 397,618

Included in other liabilities as of December 31, 2004, was a liability for mortgage repurchases of \$2.2 million. The allowance for mortgage repurchases is maintained for the purpose of purchasing previously sold mortgages for various reasons, including early payment defaults or breach of representations or warranties, which may be subsequently sold at a loss. In determining the adequacy of the liability for mortgage repurchases, management considers such factors as specific requests for repurchase, known problem loans, underlying collateral values, recent sales activity of similar loans and other appropriate information. In the opinion of management, the potential exposure related to these representations and warranties will not have a material adverse effect on our financial condition and results of operations. Activity of the liability for mortgage repurchases was as follows:

	For the year ended December 31, 2004	For the six months ended December 31, 2003 (1)
Beginning balance	\$ 2,327	\$ 1,730
Provision for loan repurchases	405	1,499
Loss on sale of repurchased mortgages	(549)	(902)
Total provision for repurchases	\$ 2,183	\$ 2,327

Represents the consolidation period.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Note C—CMO Collateral

CMO collateral for the periods indicated consisted of the following:

	At Decen	nber 31,
	2004	2003
Adjustable- and fixed rate mortgages secured by single-family residential real estate	\$ 20,428,144	\$8,366,883
Mortgages secured by multi-family residential real estate	604,934	200,427
Unamortized net premiums on mortgages	275,828	76,769
Total CMO collateral	\$ 21,308,906	\$8,644,079

Generally, CMO collateral is not placed on non-accrual status as the sub-servicer remits the interest payments to the Company regardless of the delinquency status of the underlying mortgage loans.

Note D-Mortgages Held-for-Investment

Mortgages held-for-investment for the periods indicated consisted of the following:

	At Dece	mber 31,
	2004	2003
Adjustable- and fixed rate mortgages secured by single-family residential real estate	\$ 497,756	\$ 597,685
Mortgages secured by multi-family residential real estate	77,809	47,778
Unamortized net premiums and deferred costs on mortgages	11,121	7,351
Total mortgages held-for-investment	\$ 586,686	\$ 652,814

As of December 31, 2004 and 2003, there were \$14.9 million and \$10.0 million, respectively, of mortgages held-for-investment, which were not accruing interest due to the delinquent nature of the mortgages. We expect to recover a substantial portion of these mortgages upon disposition, if necessary.

Note E—Allowance for Loan Losses

Activity for allowance for loan losses for the periods indicated was as follows:

	For the y	For the year ended December 31,			
	2004	2003	2002		
Beginning balance	\$38,596	\$ 26,602	\$11,692		
Provision for loan losses (1)	30,927	24,853	19,848		
Charge-offs, net of recoveries	(5,568)	(12,859)	(4,938)		
Total allowance for loan losses	\$63,955	\$ 38,596	\$26,602		

⁽¹⁾ The provision for loan losses for the year ended December 31, 2004 includes a specific impairment on warehouse advances of \$10.7 million.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Note F—Other Assets

Other assets consisted of the following for the periods presented:

	At Dece	ember 31,
	2004	2003
Derivative assets	\$103,290	\$ 15,407
Deferred tax charge	48,211	30,030
Prepaid and other assets	35,423	23,503
Investment securities available-for-sale	25,427	31,597
Restricted investments	4,189	1,906
Real estate owned	18,277	16,229
Premises and equipment	9,092	6,251
Deferred income taxes	5,328	2,267
Total other assets	\$ 249,237	\$ 127,190

Note G—Reverse Repurchase Agreements

We enter into reverse repurchase agreements with major brokerage firms to finance our warehouse lending operations and to fund the purchase of mortgages. Reverse repurchase agreements consist of uncommitted lines, which may be withdrawn at any time by the lender, and committed lines.

A reverse repurchase agreement, although structured as a sale and repurchase obligation, acts as a financing vehicle under which we effectively pledge mortgages as collateral to secure a short-term loan. Generally, the other party to the agreement makes the loan in an amount equal to a percentage of the market value of the pledged collateral. At the maturity of the reverse repurchase agreement, we are required to repay the loan and correspondingly receive our collateral. Under reverse repurchase agreements, we retain the beneficial ownership, including the right to distributions on the collateral and the right to vote on matters as to which certificate holders vote. Upon payment default, the lending party may liquidate the collateral. Our borrowing agreements require us to pledge cash, additional mortgages or additional assets in the event the market value of existing collateral declines. We may be required to sell assets to reduce our borrowings to the extent that cash reserves are insufficient to cover such deficiencies in collateral.

As of December 31, 2004, the warehouse lending operations had a total of \$3.4 billion of reverse repurchase facilities. Committed facilities comprised of \$525.0 million of the total available facilities, with uncommitted facilities totaling \$2.8 billion. As of December 31, 2004 and 2003, reverse repurchase agreements include accrued interest payable of \$5.2 million and \$2.0 million, respectively. Generally, the reverse repurchase agreements require that the Company provide audited financial statements to the lending party within 90 to 120 days of year end. The Company has obtained a waiver of event of default from each of its lenders in the event that the audit of the annual financial statements is not completed within the time period specified in the lending arrangements. The following tables present certain information on reverse repurchase agreements for the periods indicated:

	Maximum Borrowing Capacity	Rate Range	Range of Allowable Advances	Balance Outstanding	Maturity Date
December 31, 2004					
Short-term borrowings (indexed to one-month LIBOR):					
Repurchase agreement 1	\$ 250,000	0.75-1.50%	90 - 97%	\$ 62,480	No Expiration
Repurchase agreement 2	700,000	0.88-1.50	93 - 98	485,041	September 13, 2005
Repurchase agreement 3	700,000	0.93-1.13	95.5 - 99	212,996	August 31, 2005
Repurchase agreement 4	1,200,000	0.70-1.00	70 - 98	539,233	No Expiration
Repurchase agreement 5	500,000	0.93	90 - 98	227,808	No Expiration
Total short-term borrowings	\$3,350,000			\$1,527,558	

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Maximum Borrowing Capacity	Rate Range	Range of Allowable Advances	Balance Outstanding	Maturity Date
\$ 250,000	0.75-1.50%	90-97%	\$ 86,955	No Expiration
25,000	1.00	103	7,949	January 30, 2004
400,000	0.88-1.13	93-98	372,601	September 14, 2004
1,200,000	0.70-1.00	70-98	1,101,302	No Expiration
\$1,875,000			\$1,568,807	
	\$ 250,000 25,000 400,000 1,200,000	\$ 250,000 0.75-1.50% 25,000 1.00 400,000 0.88-1.13 1,200,000 0.70-1.00	Borrowing Capacity Rate Range Allowable Advances \$ 250,000 0.75-1.50% 90-97% 25,000 1.00 103 400,000 0.88-1.13 93-98 1,200,000 0.70-1.00 70-98	Borrowing Capacity Rate Range Allowable Advances Balance Outstanding \$ 250,000 0.75-1.50% 90-97% \$ 86,955 25,000 1.00 103 7,949 400,000 0.88-1.13 93-98 372,601 1,200,000 0.70-1.00 70-98 1,101,302

The following table presents certain information on reverse repurchase agreements for the periods indicated:

	For the year ende	d December 31,
	2004	2003
Maximum month-end outstanding balance during period	\$ 2,253,540	\$ 1,789,393
Average balance outstanding for period	2,175,728	1,374,884
Underlying collateral (mortgage loans)	1,629,486	1,663,592
Weighted average rate for period	2.66%	2.35%

Note H—CMO Borrowings

The following table presents CMOs issued, CMOs outstanding for the periods indicated and certain interest rate information on CMOs by year of issuance for the periods indicated (dollars in millions):

	Original			Range of Fixed	Range of Interest Rate Margins Over	Range of Interest Rate Margins After
Year of Issuance	Issuance Amount	12/31/04	12/31/03	Interest Rates (%)	One-Month LIBOR (%)(1)	Adjustment Date (%)(1)(2)
2001	\$ 1,500.9	\$ —	\$ 444.9	N/A	0.28 - 2.30	0.56 - 3.45
2002	3,876.1	1,237.3	2,491.0	5.25 - 12.00	0.27 - 2.75	0.54 - 3.68
2003	5,966.1	3,615.8	5,583.5	4.34 - 12.75	0.27 - 3.00	0.54 - 4.50
2004	17,710.7	16,407.5	_	3.58 - 5.56	0.25 - 2.50	0.50 - 3.75
Sub-total CMO borrowings		21,260.6	8,519.4			
Accrued interest expense		12.9	7.4			
Unamortized securitization costs		(67.1)	(36.9)			
Total CMO borrowings		\$21,206.4	\$8,489.9			
-						

⁽¹⁾ One-month LIBOR was 2.40% as of December 31, 2004.

(2) Interest rate margins are generally adjusted when the unpaid principal balance is reduced to less than 10-20% of the original issuance amount.

Expected principal maturity of the CMO borrowings, which is based on expected prepayment rates, is as follows:

	Payments Due by Period				
Total	Less Than One Year	One to Three Years	Three to Five Years	More Than Five Years	
\$21,260,630	\$5,735,344	\$ 11,209,965	\$ 2,846,158	\$ 1,469,163	

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Note I—Segment Reporting

Management internally reviews and analyzes its operating segments as follows:

- long-term investment operations that is conducted by IMH, IMH Assets, and IMCC invest primarily in Alt-A ARMs and, to a lesser extent, Alt-A FRMs acquired from the mortgage operations and multi-family mortgages;
- warehouse lending operations that is conducted by IWLG provide warehouse and repurchase financing to affiliated companies and to approved
 mortgage bankers and brokers, some of which are clients of the mortgage operations, to finance mortgages; and
- mortgage operations that is conducted by IFC and its wholly-owned subsidiaries ISAC and Novelle acquire and originate primarily ARMs and FRMs and, to a lesser extent, B/C mortgages from its network of third party correspondents, mortgage brokers and retail customers.

The accounting policies of the operating segments are the same as those described in the summary of significant accounting policies except for the elimination of inter-company profits and the related tax effect that result from the sale of mortgages from the mortgage operations to the long-term investment operations. Rent expense related to the facilities are allocated among the operating segments based on square footage. Personnel, legal, and marketing costs are allocated among the operating segments based upon their estimated usage.

The following table presents reporting segments as of and for the year ended December 31, 2004:

	Long-Term Investment Operations	Warehouse Lending Operations	Mortgage Operations (IFC)	Inter- Company (1)	Consolidated
Balance Sheet Items as of December 31, 2004:					
Cash and cash equivalents	\$ 272,908	\$ 43,821	\$ 34,355	\$ (26,733)	\$ 324,351
CMO collateral and mortgages held-for-investment	22,018,119	_	_	(122,527)	21,895,592
Allowance for loan losses	(53,272)	(10,683)	_	_	(63,955)
Mortgages held-for-sale	_	1,154	586,591	_	587,745
Finance receivables	_	1,605,642	_	(1,133,822)	471,820
Other assets	363,031	50,456	51,377	135,350	600,214
Total assets	22,600,786	1,690,390	672,323	(1,147,732)	23,815,767
Total liabilities	21,695,469	1,528,221	636,527	(1,088,525)	22,771,692
Total stockholders' equity	905,317	162,169	35,796	(59,207)	1,044,075
Income Statement Items for the Year Ended December 31, 2004:					
Net interest income	\$ 231,944	\$ 45,822	\$ 14,744	\$ 50,573	\$ 343,083
Provision for loan losses	24,851	6,076	_	_	30,927
Non-interest income	16,311	10,592	130,969	(117,096)	40,776
Non-interest expense, incl. income taxes	8,102	6,899	102,769	(22,475)	95,295
Net earnings	\$ 215,302	\$ 43,439	\$ 42,944	\$ (44,048)	\$ 257,637

⁽¹⁾ Income statement items are net of adjustments on inter-company sales transactions.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

The following table presents reporting segments as of and for the year ended December 31, 2003:

Mortgage perations (1) (IFC)	Inter- Company (2)	Consolidated
27,454 \$	(25,843)	\$ 125,153
269,679	(67,289)	9,296,893
_		(38,596)
394,994	_	397,618
_	(1,000,949)	630,030
48,250	37,467	166,859
740,377	(1,056,614)	10,577,957
712,037	(1,041,456)	10,105,170
28,340	(15,158)	472,787
8,262 \$	8,966	\$ 176,707
_	_	24,853
_	11,537	11,537
57,222	(31,836)	32,996
48,595	(10,531)	47,408
	 .	
16,889 \$	(802)	\$ 148,979
p	27,454 \$ 269,679 — 394,994 — 48,250 740,377 712,037 28,340 8,262 \$ — 57,222 48,595	27,454 \$ (25,843) 269,679 (67,289) — — — — — — — — — — — — — — — — — — —

⁽¹⁾ Income statement items reflect the consolidation period.

The following table presents reporting segments for the year ended December 31, 2002:

	Long-Term Investment Operations	Warehouse Lending Operations	Inter- Company	Consolidated
Balance Sheet Items as of December 31, 2002:				
Cash and cash equivalents	\$ 90,823	\$ 22,522	\$ —	\$ 113,345
CMO collateral and mortgages held-for-investment	5,215,731	_	_	5,215,731
Allowance for loan losses	(25,098)	(1,504)		(26,602)
Finance receivables	_	1,189,001	(524,980)	664,021
Other assets	35,593	56,208	482,043	573,844
Total assets	5,317,049	1,266,227	(42,937)	6,540,339
Total liabilities	5,130,680	1,174,966	(48,832)	6,256,814
Total stockholders' equity	186,369	91,261	5,895	283,525
Income Statement Items for the Year Ended December 31, 2002:				
Net interest income	\$ 91,143	\$ 18,829	\$ (7,506)	\$ 102,466
Provision for loan losses	18,424	1,424		19,848
Equity in net earnings of IFC (1)	_	_	11,299	11,299
Non-interest income	(50,759)	4,765	(1,644)	(47,638)
Non-interest expense	4,756	3,671	(2,495)	5,932
Net earnings	\$ 17,204	\$ 18,499	\$ 4,644	\$ 40,347
	_			

⁽¹⁾ Income statement items include 99% of earnings from the mortgage operations as equity in net earnings of IFC.

⁽²⁾ Income statement items include 99% of net earnings from the mortgage operations as equity in net earnings of IFC for non-consolidation period within non-interest income. Also, includes net adjustments on inter-company sales transactions.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Note J—Fair Value of Financial Instruments

The estimated fair value amounts have been determined by management using available market information and appropriate valuation methodologies. Considerable judgment is required to interpret market data to develop the estimates of fair value. Accordingly, the estimates presented are not necessarily indicative of the amounts that could be realized in a current market exchange. The use of different market assumptions and/or estimation methodologies may have a material effect on the estimated fair value amounts.

	December 31, 2004		December 31, 2003				
	Fair Carrying of Fin		Estimated Fair Value of Financial Carrying Instruments Amount				
<u>Assets</u>							
Cash and cash equivalents	\$	324,351	\$	324,351	\$ 125,153	\$	125,153
Restricted cash		253,360		253,360	322		322
Investment securities available-for-sale		25,427		25,427	31,597		31,597
Restricted investments		4,189		4,189	1,906		1,906
CMO collateral	2	1,308,906	2	1,595,622	8,644,079	1	8,799,700
Mortgages held-for-investment		586,686		612,394	652,814		652,813
Finance receivables		471,820		471,820	630,030		630,030
Mortgages held-for-sale		587,745		601,203	397,618		395,124
Derivative assets		103,290		103,290	15,407		15,407
<u>Liabilities</u>							
CMO borrowings, excluding accrued interest	\$2	1,193,494	\$2	1,163,573	\$8,482,423	\$ 8	8,453,583
Reverse repurchase agreements		1,527,558		1,527,558	1,568,807		1,568,807
Derivative liabilities		4,417		4,417	14,690		14,690

The fair value estimates as of December 31, 2004 and 2003 are based on pertinent information available to management as of that date. Although management is not aware of any factors that would significantly affect the estimated fair value amounts, such amounts have not been comprehensively revalued for purposes of these consolidated financial statements since those dates and, therefore, current estimates of fair value may differ significantly from the amounts presented.

The following describes the methods and assumptions used by management in estimating fair values:

Cash and Cash Equivalents and Restricted Cash

Fair value approximates carrying amounts as these instruments are demand deposits and money market mutual funds and do not present unanticipated interest rate or credit concerns.

Investment Securities Available-for-Sale

Fair value is estimated using a discounted cash flow model, which incorporates certain assumptions such as prepayment, yield and losses.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

CMO Collateral

Fair value is estimated based on quoted market prices from independent dealers and brokers for similar types of mortgages.

Finance Receivables

Fair value approximates carrying amounts due to the short-term nature of the assets and do not present unanticipated interest rate or credit concerns.

Mortgages Held-for-Investment

Fair value is estimated based on estimates of proceeds that could be received from the sale of the underlying collateral of each mortgage.

Mortgages Held-for-Sale

Fair value is estimated based on estimates of proceeds that could be received from the sale of the underlying collateral of each mortgage.

CMO Borrowings

Fair value of CMO borrowings is estimated based on the use of a bond model, which incorporates certain assumptions such as prepayment, yield and losses.

Reverse Repurchase Agreements

Fair value approximates carrying amounts due to the short-term nature of the liabilities and do not present unanticipated interest rate or credit concerns.

Derivative Assets and Liabilities

Fair value is estimated based on quoted market prices from independent dealers and brokers.

Short-term Commitments to Extend Credit

Associated warehouse lines of credit are not charged a fee, accordingly, these commitments do not have an estimated fair value.

Note K-Income Taxes

Income taxes were as follows for the period indicated:

	For the year ended December 31, 2004	For the year ended December 31, 2003
Current income taxes:		
Federal	\$ 18,869	\$ 7,947
State	5,135	913
Total current income taxes	24,004	8,860
Deferred income taxes:		
Federal	396	(3,748)
State	(3,457)	(851)
Total deferred income taxes	(3,061)	(4,599)
Total income taxes at TRS	20,943	4,261
Elimination of income taxes on inter-company profits	(34,393)	(5,658)
Total income tax benefit	\$ (13,450)	\$ (1,397)

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Effective income taxes differ from the amount determined by applying the statutory Federal rate of 35% for the period indicated as follows:

	For the year ended December 31, 2004	For the year ended December 31, 2003	
Income taxed at Federal tax rate	\$ 85,465	\$ 51,654	
State tax, net of Federal income tax	(1,329)	567	
Exclusion of REIT income and IFC income prior to consolidation	(90,559)	(50,046)	
Amortization of deferred tax charge	(5,674)	(1,980)	
Other adjustment	(1,353)	(1,592)	
Total income taxes	\$ (13,450)	\$ (1,397)	

The tax affected cumulative temporary differences that give rise to deferred tax assets and liabilities for the periods indicated are as follows:

	For the year ended December 31, 2004		the year ended ember 31, 2003
Deferred tax assets:			
Depreciation and amortization	\$	675	\$ 354
Salary accruals		4,604	2,557
Other accruals		1,736	1,011
Non-accrual loans		_	197
State franchise tax		_	3,145
Provision for repurchases		947	942
REMIC interest		31	_
FAS 133 valuation		3,601	_
Other		2,251	_
Total gross deferred tax assets		13,845	8,206
<u>Deferred tax liabilities</u> :			
Mortgage servicing rights		(6,867)	(4,394)
Depreciation and amortization		(1,554)	(758)
Deferred gain		_	(73)
REMIC interest		_	(207)
FAS 133 valuation			(507)
Non-accrual loans		(96)	_
Total gross deferred tax liabilities		(8,517)	(5,939)
Net deferred tax asset	\$	5,328	\$ 2,267

Management believes that the deferred tax asset will more likely than not be realized due to the reversal of the deferred tax liability and expected future taxable income. In determining the possible future realization of deferred tax assets, future taxable income from the following sources are taken into account: (a) the reversal of taxable temporary differences and (b) future operations exclusive of reversing temporary differences.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

As of December 31, 2004, the REIT had an estimated federal net operating loss tax carry-forward of \$18.7 million, which expires in the year 2020, that is available to offset future taxable income.

Note L—Employee Benefit Plans

401(k) Plan

After meeting certain employment requirements, employees can participate in our 401(k) plan. Under the 401(k) plan, employees may contribute up to 25% of their salaries, pursuant to certain restrictions. We will match 50% of the first 4% of employee contributions. Additional contributions may be made at the discretion of the board of directors. During the years ended December 31, 2004, 2003 and 2002, we recorded approximately \$775 thousand, \$445 thousand and \$19 thousand, respectively, for matching and discretionary contributions. Matching and discretionary 401(k) contributions for 2004 and 2003 include contributions to employees of the mortgage operations, which were reported on a consolidated basis as of July 1, 2003.

Deferred Compensation Plan

The Company maintains a nonqualified deferred compensation plan (the "Deferred Compensation Plan") for certain executives of the Company. Under the Deferred Compensation Plan, eligible participants may defer receipt of up to 100% of their base compensation and bonuses on a pretax basis until specified future dates, upon retirement or death. The deferred amounts are placed in a trust and invested by the Company. Participants recommend investment vehicles for the funds, subject to approval by the trustees. The balance due each participant increases or decreases as a result of the related investment gains and losses. The trust and the investments therein are assets of the Company and the participants of the Deferred Compensation Plan are general creditors of the Company with respect to benefits due and are recorded in the accompanying consolidated balance sheets. Included in accrued liabilities in the accompanying consolidated balance sheets at December 31, 2004 and December 31, 2003 was \$5.2 million and \$2.2 million respectively, relating to amounts owed by the Company to the plan participants

Note M-Related Party Transactions

Transactions with the Mortgage Operations

During the non-consolidation period, the long-term investment operations purchased mortgages from the mortgage operations having a principal balance of \$2.2 billion with premiums of \$45.3 million. Master servicing rights on substantially all mortgages purchased by the long-term investment operations were retained by the mortgage operations.

The mortgage operations acts as a master servicer of mortgages acquired on a "servicing-released" basis by the long-term investment operations from the mortgage operations pursuant to the terms of a servicing agreement, which became effective on November 20, 1995. The mortgage operations subcontracts or sells substantially all of its servicing obligations under such loans to independent third parties pursuant to sub-servicing agreements.

Indebtedness of Management

On February 27, 2003, IFC provided William S. Ashmore, President and Chief Operating Officer of IMH, with a \$296 thousand fixed rate mortgage to provide financing with an initial rate of 5.50%. In the opinion of management, the loan was in the ordinary course of business, substantially on the same terms, including interest rates and collateral, as those prevailing at the time for comparable transactions with unrelated persons. Subsequently, this loan was sold to a non-affiliate.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

On December 10, 2001, IFC provided William S. Ashmore, President and Chief Operating Officer of IMH, with a \$600 thousand adjustable rate mortgage to provide financing with an initial rate of 4.13%. In the opinion of management, the loan was in the ordinary course of business, substantially on the same terms, including interest rates and collateral, as those prevailing at the time for comparable transactions with unrelated persons. As of December 31, 2004 and 2003, the outstanding principal balance on the mortgage was \$323 thousand and \$558 thousand, respectively.

Note N—Commitments and Contingencies

We are a party to financial instruments with off-balance sheet risk in the normal course of business. Such instruments include short-term commitments to extend credit to borrowers under warehouse lines of credit, which involve elements of credit risk, lease commitments, and exposure to credit loss in the event of nonperformance by the counter-parties to the various agreements associated with loan purchases. Unless noted otherwise, we do not require collateral or other security to support such commitments. We use the same credit policies in making commitments and conditional obligations as we do for on-balance sheet instruments.

Short-Term Loan Commitments

The warehouse lending operations provide secured short-term revolving financing to small- and medium-size mortgage originators to finance mortgages from the closing of the mortgages until sold to permanent investors. As of December 31, 2004, the warehouse lending operations had approved warehouse lines to non-affiliated clients of \$738.7 million, of which \$471.8 million was outstanding, as compared to \$1.0 billion and \$630.0 million, respectively, as of December 31, 2003.

Lease Commitments

The Company leases office space under various operating lease agreements. Minimum premises rental commitments under non-cancelable leases are as follows:

Year 2005	\$ 3,454
Year 2006	3,288
Year 2007	3,021
Year 2008 and thereafter	1,415
Total lease commitments	\$11,178

Total rental expense for the years ended December 31, 2004, 2003, and 2002 was \$3.2 million, \$1.4 million, and \$147 thousand, respectively.

Mortgage Repurchase Commitments

In the ordinary course of business, the mortgage operation is exposed to liability under representations and warranties made to purchasers and insurers of mortgages and the purchasers of servicing rights. Under certain circumstances, the mortgage operations are required to repurchase mortgages if there had been a breach of representations or warranties.

Legal Proceedings

On September 1, 2000, a complaint captioned Michael P. and Shellie Gilmor v. Preferred Credit Corporation and Impac Funding Corporation, et al. was filed in the Circuit Court for Clay County, Missouri, Case No. CV100-4263-CC, as a purported class action lawsuit alleging that the defendants violated Missouri's Second Loans Act and Merchandising Practices Act. In July 2001, the Missouri complaint was amended to include IMH and other Impac-related entities. A plaintiffs class was certified on January 2, 2003. On June 22, 2004, the court issued an order to stay all proceedings pending the outcome of an appeal in a similar case in the Eighth Circuit.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

On February 3, 2004, a complaint captioned <u>James and Jill Baker v. Century Financial Group, Inc, et al</u> was filed in the Circuit Court of Clay County, Missouri, Case No. CV100-4294-CC as a purported class action lawsuit alleging that the defendants violated Missouri's Second Loan Act and Merchandising Practices Act.

On October 2, 2001, a complaint captioned <u>Deborah Searcy, Shirley Walker, et al. v. Impac Funding Corporation, Impac Mortgage Holdings, Inc. et. al.</u> was filed in the Wayne County Circuit Court, State of Michigan, as a purported class action lawsuit alleging that the defendants violated Michigan's Secondary Mortgage Loan Act, Credit Reform Act and Consumer Protection Act. A motion to dismiss an amended complaint has been filed, but not yet ruled upon.

On November 30, 2001, a complaint captioned <u>Garry Lee Skinner and Judy Cooper Skinner, et al. v. Preferred Credit, et al.</u> was filed in the Superior Court of Durham County, North Carolina as Case No. 1CV-05596. This is stated as a purported class action alleging a violation of the North Carolina Interest Statutes and Unfair and Deceptive Trade Practices Act when the secondary mortgage loans were originated by the defendants. A motion to dismiss the complaint was filed and granted by an order of the court dated June 9, 2004. Plaintiffs have appealed the dismissal. On January 28, 2005, plaintiffs filed a motion to dismiss us from the appeal.

On July 31, 2003, a purported class action complaint captioned <u>Frazier</u>, <u>et al v. Impac Funding Corp.</u>, <u>et al</u>, Case No. 03-2565 was filed in federal court in Tennessee. The causes of action in the action allege violations of Tennessee's usury statute and Consumer Protection Act. A motion to dismiss the complaint was filed and not yet ruled upon. The court agreed to administratively close the case on April 5, 2004 pending an appeal in a similar case. On April 29, 2004, the court issued its order administratively closing <u>Frazier</u>.

On November 25, 2003, a complaint captioned Michael and Amber Stallings v. Empire Funding Home Loan Owner Trust 1997-3; U.S. Bank, National Association; and Wilmington Trust Company was filed in the United States District Court for the Western District of Tennessee, Case No. 03-2548, as a purported class action lawsuit alleging that the defendants violated Tennessee predatory lending laws governing second mortgage loans. The complaint further alleges that certain assignees of mortgage loans, including two Impac-related trusts, should be included as defendants in the lawsuit. Like the Frazier matter this case was administratively closed on April 5, 2004 pending an appeal in a similar case.

All of the above purported class action lawsuits are similar in nature in that they allege that the mortgage loan originators violated the respective state's statutes by charging excessive fees and costs when making second mortgage loans on residential real estate. The complaints allege that IFC was a purchaser, and is a holder, along with other affiliated entities, of second mortgage loans originated by other lenders. The plaintiffs in the lawsuits are seeking damages that include disgorgement of interest paid, restitution, rescission, actual damages, statutory damages, exemplary damages, pre-judgement interest and punitive damages. No specific dollar amount of damages is specified in the complaints.

We believe that we have meritorious defenses to the above claims and intend to defend these claims vigorously. Nevertheless, litigation is uncertain and we may not prevail in the lawsuits and can express no opinion as to its ultimate outcome. An adverse judgment in any of these matters could have a material adverse affect on us, however, no judgment in any matter is probable to occur nor is any amount of any loss from such judgment reasonably estimable at this time. As of December 31, 2004 and 2003, no reserves regarding such litigation and claims have been recorded.

On October 14, 2003, an action was filed in the Circuit Court of Cook County, Illinois as Case No. 03 CH17085 entitled Fast Forward Solutions, LLC v. Novelle Financial Services, Inc. The complaint contained allegation of a class action and alleged that the defendant sent out unsolicited faxes in violation of the Telephone Consumer Protection Act, the Illinois Consumer Fraud Act, and Illinois common law. The plaintiff was seeking statutory and treble damages. We settled this action for an immaterial amount and the matter was dismissed on January 31, 2005.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

We are a party to other litigation and claims which are normal in the course of our operations. While the results of such other litigation and claims cannot be predicted with certainty, we believe the final outcome of such matters will not have a material adverse effect on our financial condition or results of operations.

Master Commitments

The mortgage operations establishes mortgage purchase commitments (master commitments) with sellers that, subject to certain conditions, entitle the seller to sell and obligate the mortgage operations to purchase a specified dollar amount of mortgages over a period generally ranging from six months to one year. The terms of each master commitment specify whether a seller may sell mortgages to the mortgage operations on a mandatory, best efforts or optional basis. Master commitments generally do not obligate the mortgage operations to purchase mortgages at a specific price, but rather provide the seller with a future outlet for the sale of its originated mortgages based on quoted prices at the time of purchase. As of December 31, 2004, the mortgage operations had outstanding short-term master commitments with 212 sellers to purchase mortgages in the aggregate principal amount of \$11.6 billion over periods ranging from one month to one year, of which \$7.0 billion had been purchased or committed to be purchased pursuant to rate locks. These rate-locks were made pursuant to master commitments, bulk rate-locks and other negotiated rate-locks. There is no exposure to credit loss in this type of commitment until the loans are funded, and interest rate risk associated with the short-term commitments is mitigated by the use of forward contracts to sell loans to investors.

Sellers who have entered into master commitments may sell mortgages to the mortgage operations by executing individual, bulk or other rate-locks, or "rate-lock." Each rate-lock, in conjunction with the related master commitment, specifies the terms of the related sale, including the quantity and price of the mortgages or the formula by which the price will be determined, the rate-lock type and the delivery requirements. Historically, the up-front fee paid by a seller to obtain a master commitment on a mandatory delivery basis is often refunded pro rata as the seller delivers loans pursuant to rate-locks. We retain any remaining fee after the master commitment expires.

Following the issuance of a specific rate-lock the mortgage operations are subject to the risk of interest rate fluctuations and enter into derivatives to diminish such risk. Interest rate risk management transactions may include mandatory or optional forward sales of mortgages or mortgage-backed securities, interest rate caps, floors and swaps, mandatory forward sales, mandatory or optional sales of futures and other financial futures transactions. The nature and quantity of derivative transactions are determined by management based on various factors, including market conditions and the expected volume of mortgage purchases.

Bulk and Other Rate-Locks

The mortgage operations also acquire mortgages from sellers that are not purchased pursuant to master commitments. These purchases may be made on an individual rate-lock basis. Bulk rate-locks may obligate the seller to sell and the mortgage operations to purchase a specific group of mortgages, generally ranging from \$500 thousand to \$125.0 million in aggregate committed principal amount, at set prices on specific dates. Bulk rate-locks enable the mortgage operations to acquire substantial quantities of mortgages on a more immediate basis. The specific pricing, delivery and program requirements of these purchases are determined by negotiation between the parties but are generally in accordance with the provisions of our seller/servicer guide. Due to the active presence of investment banks and other substantial investors in this area, bulk pricing is extremely competitive. Mortgages are also purchased from individual sellers, typically smaller originators of mortgages, who do not wish to sell pursuant to either a master commitment or bulk rate-lock. The terms of these individual purchases are based primarily on our seller/servicer guide and standard pricing provisions.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Mandatory, Best-Efforts and Optional Rate-Lock

Mandatory rate-locks require the seller to deliver a specified quantity of mortgages over a specified period of time regardless of whether the mortgages are actually originated by the seller or whether circumstances beyond the seller's control prevent delivery. The mortgage operations are required to purchase all mortgages covered by the rate-lock at prices established at the time of rate-lock. If the seller is unable to deliver the specified mortgages, it may instead deliver comparable mortgages approved by the mortgage operations within the specified delivery time. Failure to deliver the specified mortgages or acceptable substitute mortgages under a mandatory rate-lock obligates the seller to pay a penalty. In contrast, mortgages sold on a best efforts basis must be delivered to the mortgage operations only if they are actually originated by the seller. The best-efforts rate-lock provides sellers with an effective way to sell mortgages during the origination process without any penalty for failure to deliver. Optional rate-locks give the seller the option to deliver mortgages to us at a fixed price on a future date and require the payment of up-front fees. The mortgage operations retain any up-front fees paid in connection with optional rate-locks if the mortgages are not delivered.

Other Commitments

As of December 31, 2004, the mortgage operations had \$47.3 million in outstanding commitments to sell mortgages through mortgage-backed securities. These commitments allow the mortgage operations to enter into mandatory commitments when the mortgage operations notify the investor of its intent to exercise a portion of the forward delivery contracts. The mortgage operations were not obligated under mandatory commitments to deliver loans to such investors as of December 31, 2004. The credit risk of forward contracts relates to the counter-parties ability to perform under the contract. We evaluate counter-parties based on their ability to perform prior to entering into any agreements.

As of December 31, 2004, the mortgage operations had written option contracts and swaps with an outstanding principal balance of \$18.7 million and \$34.9 million, respectively. The mortgage operations may sell, call or buy put options on U.S. Treasury bonds and mortgage-backed securities. The risk in writing a call option is that the mortgage operations give up the opportunity for profit if the market price of the mortgages increases and the option is exercised. The mortgage operations also has the additional risk of not being able to enter into a closing transaction if a liquid secondary market does not exist. The risk of buying a put option is limited to the premium paid for the put option.

Note O—Derivative Instruments

Our primary objective is to limit exposure to the variability in future cash flows attributable to the variability of one-month LIBOR, which is the underlying index of adjustable rate CMO borrowings, and in the variability of the value of mortgage loans held-for-sale as we enter into rate lock commitments and purchase commitments. We also monitor on an ongoing basis the prepayment risks that arise in fluctuating interest rate environments. Our interest rate risk management program is formulated with the intent to mitigate the potential adverse effects of changing interest rates on cash flows on CMO borrowings and the value of mortgages held-for-sale. To mitigate exposure to the effect of changing interest rates, we purchase derivative instruments primarily in the form of swaps and, to a lesser extent, caps, and floors.

Included in other assets on the consolidated balance sheets as of December 31, 2004 and 2003 are \$103.3 million and \$15.4 million of derivative assets, respectively. Derivative assets include cash margin balances placed with third parties of \$7.9 million and \$9.6 million as of December 31, 2004 and 2003, respectively. Included in other liabilities on the consolidated balance sheets as of December 31, 2004 and 2003 are \$4.4 million and \$14.7 million of derivative liabilities, respectively.

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Note P-Stock Option Plans

Grants under stock option plans are made and administered by the board of directors. We currently have a 1995 Stock Option, Deferred Stock and Restricted Stock Plan (1995 Plan) and a 2001 Stock Option, Deferred Stock and Restricted Stock Plan (2001 Plan), collectively, (the stock plans). Each stock plan provides for the grant of ISOs, NQSOs, deferred stock, and restricted stock, and, in the case of the 2001 Plan, dividend equivalent rights and, in the case of the 1995 Plan, stock appreciation rights and limited stock appreciation rights awards (awards). The total number of shares initially reserved and available for issuance under the 2001 Plan was 1.0 million shares. However, on the beginning of each calendar year the maximum number of shares available for issuance may increase by 3.5% of the total number of shares of stock outstanding or a lesser amount determined by the board of directors. Pursuant to this provision, in 2004, 2003, and 2002, under the 2001 Plan an additional, 2.0 million, 1.5 million, and 1.1 million shares, respectively, were available for grant. Unless previously terminated by the board of directors, no options or awards may be granted. At December 31, 2004, the total number of shares reserved and available for issuance under the 1995 Plan is 11,891 shares and under the 2001 Plan is 989,382 shares. Unless previously terminated by the board of directors, no options or awards may be granted under the 1995 Plan after August 31, 2005 and under the 2001 Plan after March 27, 2011

Options granted under the stock plans will become exercisable in accordance with the terms of the grant made by the board of directors. Awards will be subject to the terms and restrictions of the award made by the board of directors. The board of directors has discretionary authority to select participants from among eligible persons and to determine at the time an option or award is granted and, in the case of options, whether it is intended to be an ISO or a NQSO, and when and in what increments shares covered by the option may be purchased. Option transactions for the periods indicated are summarized as follows:

For the year ended December 31, 2004 2003 2002 Weighted-Weighted-Weighted-Number Number Number Average Average Average Shares Shares Price Price Shares Price Options outstanding at beginning of year 3,395,445 \$ 10.59 2,446,427 \$ 7.88 1,634,501 5.99 Options granted 1.536,000 22.91 1,548,000 14.27 962,750 10.88 8.74 (150,824)Options exercised (345,893)10.71 (520,978)6.59 Options forfeited/cancelled 19.90 10.68 (151,668)(78,004)Options outstanding at end of year \$ 10.59 7.88 4,433,884 \$ 14.53 3,395,445 2,446,427

The following table presents information about fixed stock options outstanding at December 31, 2004:

		Stock Options Outstanding		Option	s Exercisable
Exercise Price Range	Number Outstanding	Weighted- Average Remaining Contractual Life in Years	Weighted- Average Exercise Price (\$)	Number Exercisable	Weighted- Average Exercise Price (\$)
\$ 3.85	22,500	6.08	\$ 3.85	22,500	\$ 3.85
4.18	652,500	6.24	4.18	652,500	4.18
4.44 -9.42	374,094	2.54	7.58	374,094	7.58
10.54	22,668	1.33	10.54	6,000	10.54
10.95	550,286	1.58	10.95	278,591	10.95
11.80	2,500	0.73	11.80	2,500	11.80
14.27	1,372,836	2.58	14.27	401,826	14.27
21.77	40,000	9.47	21.77	_	_
22.83	841,500	3.58	22.83	_	_
23.10	555,000	3.59	23.10		_
3.85 -23.10	4,433,884	3.38	14.53	1,738,011	8.36

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Note Q—Reconciliation of Earnings Per Share

The following table presents the computation of basic and diluted net earnings per share, including the dilutive effect of stock options and cumulative redeemable preferred stock outstanding for the periods indicated:

	For the ye	For the year ended December 31,			
	2004	2003	2002		
Numerator for basic earnings (loss) per share:					
Net earnings	\$257,637	\$148,979	\$40,347		
Less: Cash dividends on cumulative redeemable preferred stock	(3,750)	_	_		
Net earnings available to common stockholders	\$253,887	\$148,979	\$40,347		
Denominator for basic earnings (loss) per share:					
Basic weighted average number of common shares outstanding during the period	66,967	50,732	40,099		
Denominator for diluted earnings (loss) per share:					
Diluted weighted average number of common shares outstanding during the period	66,967	50,732	40,099		
Net effect of dilutive stock options	1,277	1,047	674		
Diluted weighted average common shares	68,244	51,779	40,773		
Net earnings per share:					
Basic	\$ 3.79	\$ 2.94	\$ 1.01		
Diluted	\$ 3.72	\$ 2.88	\$ 0.99		

The anti-dilutive effects of stock options outstanding as of December 31, 2004, 2003 and 2002 was 612 thousand, none, and 406 thousand, respectively.

Note R—Quarterly Financial Data (unaudited)

Selected quarterly financial data for 2004 follows:

		For the Three Months Ended,			
	December 31,	September 30,	June 30,	March 31,	
Interest income	\$ 250,372	\$ 210,388	\$160,719	\$134,137	
Interest expense	160,683	114,967	75,269	61,614	
Net interest income	89,689	95,421	85,450	72,523	
Provision (benefit) for loan losses	6,149	(229)	15,282	9,725	
Non-interest income (expense)	66,034	(89,608)	98,268	(33,918)	
Non-interest expense	31,836	24,795	28,094	24,020	
Income taxes	3,371	(9,436)	(2,872)	(4,513)	
Net earnings (loss)	114,367	(9,317)	143,214	9,373	
Net earnings (loss) per share – diluted (1)	1.52	(0.15)	2.17	0.15	
Dividends declared per share	0.75	0.75	0.75	0.65	

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated) (the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Selected quarterly financial data for 2003 follows:

		For the Three Months Ended,		
	December 31,	September 30,	June 30,	March 31,
Interest income	\$ 112,684	\$ 100,995	\$89,667	\$82,370
Interest expense	56,825	52,695	53,772	45,717
Net interest income	55,859	48,300	35,895	36,653
Provision for loan losses	3,490	7,820	7,059	6,484
Non-interest income (expense)	15,038	27,922	5,090	(3,517)
Non-interest expense	25,338	19,846	1,778	1,843
Income taxes	(5,381)	3,984	_	_
Net earnings	47,450	44,572	32,148	24,809
Net earnings per share – diluted (1)	0.86	0.84	0.63	0.52
Dividends declared per share	0.55	0.50	0.50	0.50

Diluted earnings per share are computed independently for each of the quarters presented. Therefore, the sum of the quarterly earnings per share may not equal the total for the year.

Note S—Schedule of Mortgage Loans on Real Estate

	For the	For the year ended December 31,		
	2004	2003	2002	
Beginning Balance		\$ 5,215,731	\$ 2,242,035	
Additions:				
Loans retained and originated	17,368,376	6,078,378	3,917,056	
Additions of premiums	333,669	51,859	64,507	
Loans transferred from mortgages held-for-sale	-	269,679	_	
Total additions	17,702,045	6,399,916	3,981,563	
Deductions:				
Principal paydowns	(4,666,671)	(2,148,153)	(969,243)	
Loans transferred to mortgages held-for-sale	(269,679)			
Loans sold to third parties	<u> </u>	(89,949)	_	
Amortization of premiums	(130,851)	(44,482)	(23,344)	
Transfers to other real estate owned	(36,145)	(36,170)	(15,280)	
	<u></u> -			
Total deductions	(5,103,346)	(2,318,754)	(1,007,867)	
Ending Balance		\$ 9,296,893	\$ 5,215,731	

Characteristics of our CMO collateral and loans held-for-investment at December 31, 2004, which consisted primarily of Alt-A mortgages (dollar amounts in thousands):

Original Loan Amounts	Number of Mortgage Loans	Aggregate Principal Balance	Maturity Date	Percent of Total
\$50,000 or less	2,480	\$ 79,490	6/07 - 1/35	% 0.37
\$50,001 to \$100,000	8,277	652,622	10/10 - 1/35	3.02
\$100,001 to \$150,000	14,889	1,852,949	10/03 - 1/35	8.57
\$150,001 to \$200,000	14,071	2,444,224	11/03 -1/35	11.30
\$200,001 to \$250,000	11,522	2,573,413	2/12 - 1/35	11.90
\$250,001 to \$300,000	10,015	2,734,930	2/12 - 1/35	12.65
\$300,001 to \$350,000	7,614	2,452,943	3/12 - 1/35	11.34
\$350,001 to \$400,000	5,553	2,075,062	6/17 - 1/35	9.60
\$400,001 to \$450,000	3,356	1,420,290	4/14 - 1/35	6.57
\$450,001 to \$500,000	2,805	1,329,507	11/34 -12/34	6.15
\$500,001 to \$550,000	1,592	833,683	9/09 - 5/32	3.86
\$550,001 to \$600,000	1,277	732,545	11/17 - 1/35	3.39
\$600,001 to \$650,000	1,183	743,768	6/17 - 1/35	3.44
\$650,000 or more	1,829	1,699,502	11/17 - 1/35	7.84
				-
	86,463	21,624,928	_	%100.00
Unamortized net premiums on mortgages		286,949		

REO transfers pending (16,285)

Total CMO collateral and mortgages held-for-investment

\$21,895,592

Table of Contents

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated) (the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Interest Rate Ranges	Number of Mortgage Loans	Aggregate Principal Balance	Percent Total
4% or less	2,787	\$ 845,336	%3.91
4.01% to 4.5%	7,987	2,304,904	10.66
4.51% to 5.0%	15,058	4,340,837	20.07
5.01% to 5.5%	15,419	4,471,241	20.68
5.51% to 6.0%	14,373	3,775,223	17.46
6.01% to 6.5%	9,633	2,305,715	10.66
6.51% to 7.0%	7,395	1,653,356	7.65
7.01% to 7.5%	3,988	781,543	3.61
7.51% to 8.0%	2,594	456,547	2.11
8.01% to 8.5%	1,306	202,367	0.94
8.51% to 9.0%	1,226	156,387	0.72
9.01% to 9.5%	721	69,604	0.32
9.51% or more	3,976	261,868	1.21
	86,463	21,624,928	%100.00
	· · · · · · · · · · · · · · · · · · ·		
Unamortized net premiums on mortgages		286,949	
REO transfers pending		(16,285)	
Total CMO collateral and mortgages held-for-investment		\$21,895,592	

The geographic distribution of the Company's CMO collateral and loans held-for-investment at December 31, 2004 was as follows:

Geographic Location	Number of Mortgage Loans	Aggregate Principal Balance	Percent Total
CA	43,431	\$13,418,271	% 62.05
FL	9,866	1,764,466	8.16
VA	2,299	555,170	2.57
NV	2,376	503,833	2.33
AZ	2,843	495,113	2.29
CO	2,005	414,446	1.92
MD	1,828	414,064	1.91
NY	1,321	388,031	1.79
NJ	1,579	372,332	1.72
IL	1,830	363,562	1.68
Other	17,085	2,935,640	13.58
			
	86,463	21,624,928	%100.00
Unamortized net premiums on mortgages		286,949	
REO transfers pending		(16,285)	
Total CMO collateral and mortgages held-for-investment		\$21,895,592	

Table of Contents

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Note T - Impac Funding Corporation

The following condensed financial information summarizes the results of operations of Impac Funding Corporation for the period indicated:

Condensed Consolidated Statement of Operations (in thousands)

	For the year December 31	
Net interest income:		
Total interest income	\$ 30	0,393
Total interest expense	27	2,125
Net interest income		8,268
Non-interest income:		
Gain on sale of loans	6'	5,930
Loan servicing income		5,774
Other income		276
Total non-interest income	7	1,980
Non-interest expense:		
General and administrative and other expense	4.	3,748
Amortization of mortgage servicing rights		4,059
Provision for repurchases	:	2,806
Impairment of mortgage servicing rights		855
Write-down of securities available-for-sale		_
Total non-interest expense	5:	1,468
Earnings before income taxes	28	8,780
Income taxes	17	2,147
Net earnings	\$ 10	6,633

Note U - Redeemable Preferred Stock

On May 28, 2004, the Company sold 2.0 million shares of Series B Cumulative Redeemable Preferred Stock, raising \$48.3 million in net proceeds. The shares have a liquidation value of \$25.00 per share and will pay an annual coupon of 9.375%. The shares are redeemable at the Company's option, in whole or in part, on or after May 28, 2009 except in limited circumstances to preserve the Company's REIT status.

On November 18, 2004, the Company sold 4.0 million shares of Series C Cumulative Redeemable Preferred Stock, raising \$96.6 million in net proceeds. The shares have a liquidation value of \$25.00 per share and will pay an annual coupon of 9.125%. The shares are redeemable at the Company's option, in whole or in part, on or after November 23, 2009 except in limited circumstances to preserve the Company's REIT status. The Company granted its underwriters an option, exercisable for 30 days, to purchase up to an additional 300,000 shares to cover over-allotments, if any. On December 7, 2004 the underwriters exercised their options for 300,000 shares in over-allotments resulting in net proceeds of \$7.3 million.

Table of Contents

IMPAC MORTGAGE HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(dollars in thousands, except per share data or as otherwise indicated)

(the 2004 financial statements are unaudited and the 2003 and 2002 financial statements have been derived from the audited financial statements)

Note V – Subsequent Events

In March 2005 the Company entered into a lease agreement under which its corporate headquarters will move from its present location to a new, larger facility in Irvine, California. The lease agreement includes an option to lease another building and a right of first offer to lease additional buildings. The lease term is a period of ten years beginning after the completion of the first building and related tenant improvements, which is anticipated to be in the first quarter of 2006. The aggregate rent for the term of the lease for the first building is approximately \$69 million.

OFFICE LEASE (BUILDING II)

BETWEEN

SCHOLLE JAMBOREE PROPERTY DEVELOPMENT I LLC

AND

IMPAC FUNDING CORPORATION

LEASE EXHIBITS

Exhibit A (Site Plan)

Exhibit B (Rules and Regulations)

Exhibit C (Commencement Letter)

Exhibit D (Work Letter)

Exhibit E (Janitorial Specifications)

Exhibit F (Access Control Specifications)

Exhibit G (Recognition of Covenants, Conditions and Restrictions)

Exhibit H (Form of SNDA)

Exhibit I (Signage Maintenance Specifications)

Exhibit J (Form of Guaranty to Lease)

Exhibit K (Form of Telecommunications Agreement)

Exhibit L (Form of Memorandum of Lease)

Exhibit M (Form of Letter of Credit)

1.

2.

3.

		- ugc
Basic L	ease Information.	1
1.1	Notice Addresses:	1
1.2	"Building I"	1
1.3	"Building II"	1
1.4	"Building III"	1
1.5	"Building IV"	1
1.6	"Building V"	1
1.7	"Project"	1
1.8	"Premises"	1
1.9	"Term"	1
1.10	"Base Rent":	2
1.11	"Tenant's Share"	2
1.12	"Base Year"	2
1.13	"Tenant Improvement Allowance"	2
1.14	"Security Deposit"	2
1.15	"Permitted Use"	2
1.16	"Broker(s)"	2
1.17	"Business Day(s)"	2
1.18	"Normal Business Hours"	2
1.19	"Law(s)"	2
1.20	"Parking Pass Ratio"	2
1.21	"Guarantor"	2
Grant o	of Tenancy; Measurement.	3
2.1	Grant of Tenancy; Common Areas.	3
2.2	Measurement.	3
	ion; Renewal Options; Special Termination Right; First Right to Purchase; Rights of First Offer and	2
Lease.	Delinear of December 1	3
3.1	Delivery of Possession. Turnover of Premises.	3
3.2		4
3.3 3.4	Pre-Term Occupancy.	4
5.4	Renewal Options.	4
	3.4.1. Option Right.3.4.2. Option Rent.	4
		4
		4
	3.4.4. Determination of Option Rent.	5
	3.4.5. Fair Market Rental Rate.	5

			Page
	3.4.6.	Terms and Conditions.	6
	3.4.7.	Renewal Amendment.	6
3.5	Special	Termination Option.	6
	3.5.1.	Exercise of Special Termination Option.	6
	3.5.2.	Special Termination Payment.	6
	3.5.3.	Special Termination Amendment.	7
	3.5.4.	Other Terms.	7
3.6	Right of	of First Offer to Purchase.	7
	3.6.1.	Acceptance.	8
	3.6.2.	Time of the Essence and Failure to Accept or Purchase.	8
	3.6.3.	Re-Offer Right.	8
3.7	Right of	of First Offer to Lease Building I Space.	9
	3.7.1.	Procedure for Offer.	9
	3.7.2.	Procedure for Acceptance.	9
	3.7.3.	Construction In Building I First Offer Space.	9
	3.7.4.	New Lease.	10
3.8	Option	to Lease Building III.	10
3.9	Right of	of First Offer to Lease Building III Space.	10
	3.9.1.	Procedure for Offer.	10
	3.9.2.	Procedure for Acceptance.	10
	3.9.3.	Construction In Building III First Offer Space.	11
	3.9.4.	New Lease.	11
3.10	Right of	of First Offer to Lease Building IV.	11
	3.10.1.	Procedure for Offer.	11
	3.10.2.	Procedure for Acceptance.	11
	3.10.3.	Construction In Building IV First Offer Space.	12
	3.10.4.	New Lease.	12
3.11	Referen	nces to Landlord.	12
Rent.			12
4.1	Paymen	nt of Rent.	12
4.2	Late Pa	ayment.	13
Opera	ting Expe	enses and Real Estate Taxes.	13
5.1	General	l Terms.	13
5.2	Certain	Definitions Relating to Additional Rent.	13
	5.2.1.	"Base Year"	13

4.

5.

		Page
	5.2.2. "Direct Expenses"	13
	5.2.3. "Expense Year"	13
	5.2.4. "Operating Expenses"	13
	5.2.5. Taxes.	16
5.3	Allocation of Direct Expenses.	17
5.4	Calculation and Payment of Additional Rent.	17
	5.4.1. Statement of Actual Direct Expenses and Payment by Tenant.	17
	5.4.2. Statement of Estimated Direct Expenses.	18
5.5	Taxes and Other Charges for Which Tenant Is Directly Responsible.	18
5.6	Tenant's Payment of Certain Tax Expenses.	18
	5.6.1. The Tax Increase.	18
	5.6.2. Protection.	19
	5.6.3. Landlord's Right to Purchase the Proposition 13 Protection Amount Attributable to a Particular Reassessm	ent. 19
5.7	Landlord's Books and Records.	19
	5.7.1. In General.	19
	5.7.2. Termination of Rights.	20
Secu	curity Deposit; Letter of Credit.	20
Servi	vices to be Furnished by Landlord.	20
7.1	Utilities and Services.	20
7.2	Excess Usage; Tenant's Computer Room.	21
7.3	Interruption of Use.	22
7.4	Tenant HVAC System.	22
7.5	Emergency Generators.	22
	7.5.1. Auxiliary Space.	22
	7.5.2. Manner of Installation.	22
	7.5.3. Repair and Maintenance of Generator.	23
	7.5.4. Generator Use.	23
7.6	LEED Requirements.	23
	7.6.1. LEED Standards.	23
	7.6.2. Compliance with LEED Conditions.	23
	7.6.3. Reimbursement.	23
Use;	e; Occupancy Level.	23
8.1	Permitted Use.	23
8.2	Prohibited Uses.	23

6.7.

8.

8.3

Occupancy Level.

24

			rage
9.	Repairs	s and Alterations.	24
	9.1	Tenant's Repair Obligations.	24
	9.2	Landlord's Repair Obligations.	25
	9.3	Tenant's Right to Make Repairs.	25
	9.4	Alterations.	26
	9.5	Landlord's Property.	26
10.	Liens.		26
11.	Entry b	y Landlord.	27
12.	Assign	nent and Subletting.	27
	12.1	Transfers.	27
	12.2	Landlord's Consent.	28
	12.3	Transfer Premium.	29
	12.4	Effect of Transfer.	29
	12.5	Permitted Occupants.	29
	12.6	Occurrence of Default.	29
	12.7	Permitted Transfers.	30
13.	Insurar	nce.	30
	13.1	Landlord's Fire and Casualty Insurance.	30
	13.2	Tenant's Insurance.	30
	13.3	Form of Policies.	30
	13.4	Additional Insurance Obligations.	31
	13.5	Waiver of Claims and Subrogation.	31
14.	Indemn	nity and Waiver of Claims.	31
	14.1	Indemnification of Landlord.	31
	14.2	Indemnification of Tenant.	31
	14.3	Landlord Not Liable.	32
	14.4	Survival.	32
15.	Casualt	y Damage.	32
	15.1	Repair of Damage to Premises by Landlord.	32
	15.2	Landlord's Option to Repair.	32
	15.3	Waiver of Statutory Provisions.	33
16.	Conden	nnation.	33
17.	Events	of Default.	33
	17.1	Events of Default.	33

34

17.2

Remedies Upon Default.

			Page
	17.3	Subleases of Tenant.	34
	17.4	Waiver of Jury Trial.	34
	17.5	Landlord's Cure.	34
	17.6	Landlord Default.	35
		17.6.1. General.	35
		17.6.2. Abatement of Rent.	35
	17.7	Efforts to Relet.	35
18.	Limita	ntion of Liability.	35
19.	Holdin	ng Over.	35
20.	Surren	nder of Premises.	36
	20.1	Surrender; Removal of Tenant's Property and Required Removables.	36
	20.2	Removal of Lines.	36
21.	Suboro	dination; Estoppel Certificate.	36
	21.1	Subordination.	36
	21.2	Estoppels.	37
22.	Landlo	ord's Development Rights.	37
	22.1	Development of Project.	37
	22.2	Subdivision of Project.	37
	22.3	Telecommunications Equipment.	37
23.	Signag	ge.	38
	23.1	Project Name and Project Signage.	38
	23.2	Full Floors.	38
	23.3	Multi-Tenant Floors.	38
	23.4	Prohibited Signage and Other Items.	38
	23.5	Building Directory.	38
	23.6	Monument Signage.	38
	23.7	Building Top Signage.	39
	23.8	Transferability.	39
	23.9	Signage Maintenance.	39
24.	Parkin	ng.	39
25.	Compl	liance With Laws.	40
26.	Notices	s.	40
27.	Miscell	laneous.	40

40

41

27.1

27.2

Governing Law.

Recordation.

		rage —				
27.3	Attorneys' Fees.	41				
27.4	Force Majeure.	41				
27.5	Landlord Transfer.	41				
27.6	Brokers.	41				
27.7	Confidentiality.	41				
27.8	Hazardous Materials.	41				
27.9	Communications and Computer Lines.	42				
27.10	Binding Effect.	42				
27.11	No Air Rights.	42				
27.12	Modification of Lease.	42				
27.13	Relationship of Parties.	42				
27.14	Application of Payments.	42				
27.15	Time of Essence.	42				
27.16	Right to Lease.	42				
27.17	Intentionally Omitted.	43				
27.18	Counterparts.	43				
27.19	Transportation Management.					
27.20	Building Renovations.					
27.21	No Violation.	43				
27.22	Calendar Days.	43				
27.23	Assistance in Litigation.					
27.24	Survival of Provisions Upon Termination of Lease.	43				
27.25	Financial Statements.	43				
27.26	Independent Covenants.	43				
27.27	Quiet Enjoyment.	44				
27.28	Arbitration.	44				
	27.28.1. General Submittals to Arbitration.	44				
	27.28.2. JAMS.	44				
	27.28.3. Arbitration Procedure.	44				
27.29	Entire Agreement.	45				
27.30	Modification.					
27.31	No Offer to Lease.	45				
27.32	Anti-Terrorism Representations.					
27.33	Guaranty of Lease.	45				

OFFICE LEASE (BUILDING II)

THIS OFFICE LEASE (BUILDING II) (this "Lease") is made and entered into as of March 1, 2005, by and between SCHOLLE JAMBOREE PROPERTY DEVELOPMENT I LLC, a California limited liability company ("Landlord"), and IMPAC FUNDING CORPORATION, a California corporation ("Tenant").

1. Basic Lease Information.

1.1 Notice Addresses:

1. Landlord Address:

Scholle Jamboree Property Development I LLC 19500 Jamboree Road Irvine, CA 92612-2457 Attention: General Counsel

Fax: 949-476-2967

With a copy to: Dorn-Platz Properties

3444 North Central Glendale, California 91203

Attention: Mark Knapp Fax: 818-240-9179

2. Tenant Address:

On and after the Commencement Date, notices shall be sent to Tenant at the Premises. Prior to the Commencement Date, notices shall be sent to Tenant at the following address:

Impac Funding Corporation 1401 Dove Street

Newport Beach, CA 92660 Attention: General Counsel

Fax: 949-475-3967

- 1.2 "Building I" means that certain mixed use building to be constructed in the Project in the location shown on **Exhibit A** (sometimes referred to herein as the "Site Plan") as "Building I", containing four (4) floors and approximately 139,000 rentable square feet (and including an adjacent buildable area), and comprising a portion of the Project.
- 1.3 "Building II" or the "Building" means that certain office building to be constructed in the Project in the location shown on the Site Plan as "Building II" in accordance with this Lease, containing seven (7) floors and approximately 200,000 rentable square feet (and including the subterranean parking area beneath such building), and comprising a portion of the Project.
- 1.4 "Building III" means that certain office building that may be constructed in the Project in the location shown on Site Plan as "Building III", expected to contain four (4) floors and approximately 100,000 rentable square feet (and including the subterranean parking area beneath such building), and which, if constructed, shall comprise a portion of the Project.
- 1.5 "Building IV" means that certain office building that may be constructed in the Project in the location shown on the Site Plan as "Building IV", expected to contain four (4) floors and approximately 100,000 rentable square feet (and including any subterranean parking area beneath such building), and which, if constructed, shall comprise a portion of the Project.
- 1.6 "Building V" means that certain freestanding building, currently anticipated to be occupied for restaurant and retail uses, to be constructed in the Project in the location shown on the Site Plan as "Building V", containing approximately 7,000 rentable square feet, and comprising a portion of the Project.
- 1.7 "<u>Project"</u> means (i) Building II and the Common Areas (defined in Section 2.1), (ii) the parcel(s) of land upon which Building II and the Common Areas are located, and (iii) Building I and, if and when constructed, Building III, Building IV and Building V (collectively, the "<u>Adjacent Buildings</u>") and the parcel(s) of land upon which the Adjacent Buildings are located. Collectively, Building II and the Adjacent Buildings are called the "<u>Project Buildings</u>".
- 1.8 "Premises" means all of Building II, subject to Landlord's rights and reservations with respect thereto as set forth in this Lease.

1.9 "Term" means a period of one hundred twenty (120) months. The Term is subject to extension as set forth in Section 3.4. The Term shall commence on the date (the "Commencement Date") that is the earlier of (i) the date on which the Premises are "Ready for Occupancy." as provided in the Work Letter attached hereto as Exhibit D (the "Work Letter"), or (ii) the date on which Tenant commences business operations from the Premises. Unless terminated early or extended in accordance with this Lease, the Term shall end one hundred twenty (120) months after the Commencement Date (the "Termination Date"); provided, however, that if the Termination Date would occur on a day other than the last day of a calendar month, the Termination Date shall be extended such that it occurs on the last day of the month in which the Termination Date otherwise occurs. At any time during the Term, Landlord may deliver to Tenant a Commencement Letter in the form attached hereto as Exhibit C (the "Commencement Letter"), which Tenant shall execute and return to Landlord within ten (10) Business Days of receipt thereof; provided that if said notice is not factually correct, then Tenant shall make such changes as are necessary to make the notice factually correct and shall thereafter execute and return such notice to Landlord within such ten (10) Business Day period, and thereafter the dates set forth on such notice shall be conclusive and binding upon Tenant and Landlord, unless Landlord, within ten (10) Business Days following receipt of Tenant's changes, sends a notice to Tenant rejecting Tenant's changes, whereupon this procedure shall be repeated until the parties either (a) mutually agree upon the contents of Exhibit C, or (b) the contents are determined by arbitration pursuant to Section 27.28.

1.10 "Base Rent":

Months of Term	Annua	Annual Rate Per Rentable Square Foot		Monthly Base Rent	
	-				
1-30	\$	31.80	\$	530,000	
31-60	\$	33.60	\$	560,000	
61-90	\$	35.40	\$	590,000	
91-120	\$	37.20	\$	620,000	

The monthly Base Rent set forth above is based on an estimated 200,000 rentable square feet within the Premises and is subject to adjustment pursuant to Section 2.2 below.

- 1.11 "Tenant's Share" of Operating Expenses and Real Estate Taxes: 100%, subject to Section 5.3 and 5.6.
- 1.12 "Base Year" for Operating Expenses and Real Estate Taxes: Calendar year 2007.
- 1.13 "Tenant Improvement Allowance": \$43.00 per rentable square foot of the Premises.
- 1.14 "Security Deposit": Six Hundred Twenty-Thousand Dollars (\$620,000) (based on the last month's Base Rent amount).
- 1.15 "Permitted Use": General office uses and other uses approved by Landlord as described in Section 8.1 and Section 8.2.
- 1.16 "Broker(s)": Cushman & Wakefield of California, Inc. (Rick Kaplan), representing Landlord, and Meridian Pacific (Bill Anderson), representing Tenant.
- 1.17 "Business Day(s)" are Monday through Friday of each week, exclusive of New Year's Day, Memorial Day, Independence Day, Labor Day, Thanksgiving Day and Christmas Day ("Holidays"). Landlord may designate up to two (2) additional nationally recognized Holidays, provided that other office buildings in the area where the Building is located commonly recognize the additional Holidays and Landlord has provided Tenant with at least one hundred twenty (120) days prior written notice of such additional Holidays.
- 1.18 "Normal Business Hours" for the Building are 8:00 A.M. to 6:00 P.M. on Business Days and 9:00 A.M. to 12:00 P.M. on Saturdays.
- 1.19 "Law(s)" means all applicable statutes, codes, ordinances, orders, rules and regulations of the City of Irvine, the County of Orange, the State of California and any other federal, state, municipal or other governmental entity, agency or authority with jurisdiction over the Project or the operations thereon.
- 1.20 <u>"Parking Pass Ratio"</u> means five (5) parking passes for every 1,000 rentable square feet of the Premises, a portion of which shall be allocated to reserved parking in the Building Parking Facility (as defined in Section 24) and the remainder shall be allocated to unreserved parking in the Project Parking Facility (as defined in Section 24), subject to the terms and conditions of Section 24.
- 1.21 "Guarantor" means IMPAC MORTGAGE HOLDINGS, INC., a Maryland corporation.

2. Grant of Tenancy; Measurement.

2.1 Grant of Tenancy; Common Areas. Landlord leases the Premises to Tenant and Tenant leases the Premises from Landlord, together with the right in common with others to use those portions of the Project and all appurtenant areas and improvements that are designated by Landlord for the common use, convenience or benefit of tenants and their respective employees, invitees, licensees, and other visitors, including without limitation streets, driveways, sidewalks, the Parking Facilities (as defined in Section 24), landscaped and planted areas and walkways, designated break and seating areas, designated smoking areas (such areas, together with such other portions of the Project as shown on the Site Plan, or if not shown on the Site Plan, as designated by Landlord, in its good faith discretion, including certain areas to be shared by Landlord and certain tenants, are collectively referred to herein as the "Common Areas"). As used herein, "Building Common Areas" shall consist of hallways and interior stairwells within the applicable Project Buildings not located within the premises of any tenant, common entrances and lobbies, elevators, and restrooms not located within the premises of any tenant and subterranean automobile parking areas and accessways, if any, designated by Landlord for use by tenants or visitors of the applicable Project Building. Landlord and Tenant acknowledge that the purpose of Exhibit A is to show the approximate location of the to-be-constructed Building I and Building II, and currently anticipated other Project Buildings and amenities only, and such Exhibit is not meant to constitute an agreement, representation or warranty as to the Project or the construction of any of the Project Buildings, the precise area thereof or the specific location of the Common Areas, or the elements thereof or of the accessways to the Premises or the Project; provided, however, that (i) if Landlord (or an affiliate of Landlord) leases Building V for restaurant operations, Landlord shall cause to be included in the lease covering such restaurant operations (and Landlord shall use commercially reasonable efforts to enforce) a requirement that the restaurant tenant open and operate, for so long as such operation remains economically viable for such restaurant tenant, a deli-type sandwich operation within Building V that is open to public, including the employees, guests and business invitees of Tenant, during lunch hours on non-Holiday weekdays; and (ii) Landlord shall not, without Tenant's prior written approval, make any material change to the Site Plan that materially adversely affects Tenant's ingress to or egress from the Project, Premises or Parking Facilities or if such change would substantially and adversely alter the Project from a first-class office building project. Without limitation, examples of the types of changes which would violate the provisions of clause (ii) of the foregoing sentence would be conversion of a substantial portion of the floor area of the Project or Parking Facilities to residential, manufacturing or hotel usage or outlet type retail operations. The manner in which the Common Areas are maintained and operated shall be at the reasonable discretion of Landlord, provided that Landlord shall maintain and operate the same in a manner consistent with that of other first-class, mid-rise and high-rise office buildings located in the John Wayne Airport-South Coast Metro-Newport Beach areas of Orange County, California, which buildings are comparable to the Building in terms of quality of construction, level of services, amenities, age, size and appearance (the "Comparable Buildings"). The use of the Building and Common Areas shall be subject to such reasonable Rules and Regulations as Landlord may make from time to time as provided in Section 8.2. Landlord reserves the right to close temporarily, make alterations or additions to, or change the location of elements of the Project and the Common Areas, as long as such changes do not change the nature of the Project to something other than a first class office building project or materially, adversely affect Tenant's use of the Premises for the Premises for the Premises or the Premises of the Premises or the Premise the Parking Facilities servicing the same.

2.2 Measurement. Landlord shall, prior to delivery of possession of the Premises to Tenant, have its architect or a space measurement firm selected by Landlord and reasonably acceptable to Tenant (the "Measurement Firm") (Stevenson Systems and Pace Compumetrics are each hereby pre-approved by Tenant as an acceptable Measurement Firm) measure the rentable square footage of the Building in accordance with the guidelines for measuring rentable area (for a single tenant building if Tenant leases the entire Project Building and for a multi-tenant building if Tenant leases less than the entire Project Building) specified in the American National Standard Institute Publication ANSI Z65.1-1996 as adopted by the Building Owners and Managers Association, as interpreted and applied by the Measurement Firm (the "Measurement Standard"). If such measurement by the Measurement Firm indicates that the rentable square footage set forth in Section 1.3 is different than the rentable square footage numbers for the Building resulting from utilization of the Measurement Standard, then the rentable square footage set forth in Section 1.3 shall be adjusted to reflect the number of rentable square feet within the Building as measured under the Measurement Standard and all amounts, percentages and figures appearing or referred to in this Lease based upon such incorrect amount (including, without limitation, the amount of the "Rent" and any "Security Deposit," as those terms are defined in this Lease) shall be modified in accordance with such determination and any payments due to Landlord from Tenant based upon such rentable square feet shall be proportionally, retroactively and prospectively increased or decreased as applicable. The Measurement Firm shall also measure each of the Adjacent Buildings within a reasonable time after the same are completed, as well as the remaining Premises within a reasonable time following Tenant's exercise of the Special Termination Right under Section 3.5, as well as any space additionally leased by Tenant that is less than an entire Project Building pursuant to Sections 3.8, 3.9 or 3.10, in each case using the Measurement Standard. Landlord will confirm all such measurements in writing to Tenant and the same shall be binding upon Tenant for all purposes under this Lease; provided, however, that if Tenant disagrees with any of the Measurement Firm's measurements under this Section 2.2 and notifies Landlord of same within twenty (20) days after Landlord notifies Tenant of the Measurement Firm's applicable measurement, and if a dispute occurs regarding the final accuracy of such measurement using the Measurement Standard that is not resolved between Landlord and Tenant, using good faith efforts, within sixty (60) days after Landlord's receipt of Tenant's disagreement notice, such dispute will be resolved by arbitration pursuant to Section 27.28, below, and judgment upon the award rendered by the arbitrator may be entered in any court having iurisdiction thereof.

3. Possession; Renewal Options; Special Termination Right; First Right to Purchase; Rights of First Offer and Lease.

3.1 <u>Delivery of Possession.</u> Landlord shall deliver possession of the Premises to Tenant upon the date that the Premises are Ready for Occupancy as provided in the Work Letter. If Landlord, for any reason, cannot deliver

possession of the Premises to Tenant on the "Estimated Commencement Date" set forth in the Work Letter, this Lease shall not be void or voidable, nor shall Landlord be liable to Tenant for any loss or damage resulting therefrom. However, if date that the Premises are Ready for Occupancy is delayed beyond the Estimated Commencement Date, Tenant shall not be liable for any rent until Landlord tenders possession of the Premises to Tenant with the Premises Ready for Occupancy (or until Tenant commences business operations from the Premises). Notwithstanding the foregoing if the Commencement Date has not occurred within two hundred ten (210) days (subject to extension for delays attributable to Force Majeure Events, but in no event shall such extension for delays attributable to Force Majeure Events extend the 210-day period beyond an additional 120 days) after the Estimated Commencement Date (the "Outside Date"), Tenant may, as its sole remedy, terminate this Lease by delivering written notice thereof to Landlord within fifteen (15) days after the Outside Date (unless the Commencement Date occurs prior to Landlord's receipt of such notice). However, to the extent Landlord's inability to deliver possession of the Premises to Tenant in accordance with (or earlier than provided for in) the Work Letter is caused by Tenant's negligence or breach of this Lease or of the Work Letter, or by other Tenant Delay(s) (as defined in the Work Letter), the Commencement Date for all purposes under this Lease shall be accelerated by the number of days of those Tenant Delays.

- 3.2 <u>Turnover of Premises</u>. The taking of possession of the Premises by Tenant shall conclusively establish that the Premises and the Building are at such time in good order, condition and repair, subject only to Landlord's obligations specifically set forth in this Lease with respect to the repair and maintenance of the Base Building (as defined in Section 9.2), Landlord's obligations to provide services and utilities set forth in Section 7.1, Landlord's obligations regarding restoration set forth in Sections 15 and 16, and matters which are Landlord's responsibility under the Work Letter.
- 3.3 <u>Pre-Term Occupancy.</u> Tenant shall have the right to enter upon and occupy the Premises (or any portion thereof) before the Commencement Date in accordance with the terms of the Work Letter for the sole purpose of performing improvements and/or installing furniture, equipment or other personal property as permitted by the Work Letter, and such entry/occupancy shall be subject to the terms and conditions of this Lease except that Tenant shall not be required to pay Rent (defined in Section 4 below) for such time periods prior to the Commencement Date so long as Tenant's entry/occupancy complies with the terms therefor set forth in the Work Letter.

3.4 Renewal Options.

- 3.4.1. Option Right. Landlord hereby grants Tenant two (2) options to extend the Term as to the entire Premises for a period of five (5) years each (the "First Option Term" and the "Second Option Term" respectively) (the foregoing option terms shall be referred to hereinafter sometimes individually or collectively as the "Option Term"), which options shall be exercisable only by written notice delivered by Tenant to Landlord as provided below, provided that, as of the date of delivery of such notice, Tenant is not in Default under this Lease. The rights contained in this Section 3.4 shall be personal to the original Tenant executing this Lease and any Tenant Affiliate or Successor (as those terms are defined in Section 12.1) (collectively, "Original Tenant"), and may only be exercised by Original Tenant if Original Tenant occupies at least 100,000 rentable square feet of the Building. Upon the proper exercise of such option to extend the Term, as it applies to the entire Premises, shall be extended for a period of five (5) years (provided, however, that if, as of the end of the initial Term or the First Option Term, as the case may be, Tenant is in Default under this Lease, Landlord shall have the right to void the applicable option by notice to Tenant, even if the same has been exercised).
- 3.4.2. Option Rent. The rent payable by Tenant during any Option Term (the "Option Rent") shall be equal to the "Fair Market Rental Rate," as that term is defined in Section 3.4.5, below.
- 3.4.3. Exercise of Option. The option contained in this Section 3.4 shall be exercised by Tenant, if at all, only by Tenant's delivery of a written notice executed by a duly authorized officer of Tenant (the "Option Exercise Notice") to Landlord not more than eighteen (18) months nor less than twelve (12) months prior to the expiration of the initial Term or the First Option Term, as the case may be, stating that Tenant is irrevocably exercising its option for the entire Premises. Landlord, after receipt of the Option Exercise Notice, shall deliver notice (the "Landlord's Rent Determination") to Tenant not more than thirty (30) days after Landlord's receipt of any Option Exercise Notice, setting forth Landlord's determination of the Fair Market Rental Rate for the Option Term. Tenant, within thirty (30) days after the date on which Landlord delivers the Option Rent Notice, shall either (i) notify Landlord that Tenant accepts Landlord's Rent Determination (the "Rent Acceptance Notice"), or (ii) if Tenant disagrees with Landlord's Rent Determination, provide Landlord with written notice of rejection (the "Rejection Notice") accompanied by Tenant's determination of the Fair Market Rental Rate ("Tenant's Rent Determination"). If Tenant fails to provide Landlord with either a Rent Acceptance Notice or Rejection Notice within such thirty (30) day period, Tenant's options shall be null and void and of no further force and effect. If Tenant provides Landlord with a Rent Acceptance Notice, Landlord and Tenant shall enter into the Renewal Amendment (hereinafter defined) upon the terms and conditions set forth herein. If Tenant provides Landlord with a Rejection Notice without a Tenant's Rent Determination, Tenant's options shall be null and void and of no further force and effect. If Tenant provides Landlord with a Rejection Notice that includes Tenant's Rent Determination, Landlord and Tenant shall work together in good faith to agree upon the Fair Market Rental Rate for the applicable Option Term. Upon agreement, Landlord and Tenant shall enter into the Renewal Amendment in accordance with the terms and conditions hereof. Notwithstanding the foregoing, if Landlord and Tenant are unable to agree upon the Option Rent for the Premises within ten (10) Business Days after the date on which Tenant provides Landlord with both a Rejection Notice and Tenant's Rent Determination (the "Outside Agreement Date"), the parties shall follow the procedure, and the Option Rent shall be determined, as set forth in Section 3.4.4 below.

- 3.4.4. <u>Determination of Option Rent.</u> If Landlord and Tenant fail to reach agreement (i) with regard to the Option Rent by the Outside Agreement Date, then each party shall make a separate determination of the Option Rent within five (5) Business Days after the Outside Agreement Date, and such determinations shall be submitted to arbitration in accordance with Sections 3.4.4.1 through 3.4.4.7 below.
- 3.4.4.1 Landlord and Tenant shall each appoint one arbitrator who shall be a real estate appraiser who is a current member of the American Institute of Real Estate Appraisers, with at least ten (10) years of experience (and who shall have been active over the five (5) year period ending on the date of such appointment) appraising office space comparable to the Premises in the John Wayne Airport-South Coast Metro-Newport Beach area of California ("Market Area"). The determination of the arbitrators shall be limited solely to the issue of whether Landlord's or Tenant's submitted Option Rent, as the case may be, is the closest to the actual Fair Market Rental Rate, as determined by the arbitrators, taking into account the requirements of Section 3.4.5. Each party shall appoint an arbitrator meeting the foregoing requirements and shall notify the other party in writing of the name and contact information of their selected arbitrator within fifteen (15) days after the applicable Outside Agreement Date. Landlord and Tenant may consult with their selected arbitrators prior to appointment. The arbitrators so selected by Landlord and Tenant shall be deemed "Party Arbitrators."
- 3.4.4.2 The two (2) Party Arbitrators so appointed shall be specifically required pursuant to an engagement letter within ten (10) days of the date of the appointment of the last appointed Party Arbitrator to agree upon and appoint a third arbitrator ("Neutral Arbitrator") who shall be qualified under the same criteria set forth hereinabove for qualification of the two Party Arbitrators except that neither the Landlord or Tenant or either Party Arbitrator may, directly or indirectly, consult with the Neutral Arbitrator prior or subsequent to his or her appearance. The Neutral Arbitrator shall be retained via an engagement letter jointly prepared by Landlord's counsel and Tenant's counsel.
- 3.4.4.3 The three (3) arbitrators shall within thirty (30) days of the appointment of the Neutral Arbitrator reach a decision as to whether the parties shall use Landlord's or Tenant's submitted Option Rent, and shall notify Landlord and Tenant thereof. Whichever Option Rent is selected by the arbitrators shall become the then applicable Option Rent.
 - 3.4.4.4 The decision of the majority of the three (3) arbitrators shall be binding upon Landlord and Tenant.
- 3.4.4.5 If any party fails to notify the other party in writing of the name and contact information of their selected Party Arbitrator within fifteen (15) days after the applicable Outside Agreement Date, then the Party Arbitrator appointed by the other party shall solely decide whether the parties shall use Landlord's or Tenant's submitted Option Rent, and such decision shall be binding on the parties.
- 3.4.4.6 If the two (2) Party Arbitrators fail to agree upon and appoint the Neutral Arbitrator, then either party may petition JAMS (as defined in Section 27.28), or if JAMS refuses to act, either party may petition the American Arbitration Association, to appoint the Neutral Arbitrator, subject to criteria in Section 3.4.4.1 of this Lease.
- 3.4.4.7 Each party shall bear the costs of its own Party Arbitrator, and the cost of the Neutral Arbitrator and the arbitration shall be paid by Landlord and Tenant equally.
- 3.4.4.8 Once the Option Rent is determined by the arbitrators, Landlord and Tenant shall enter into the Renewal Amendment (as hereinafter defined) upon the terms and conditions set forth herein.
- 3.4.5. Fair Market Rental Rate. The "Fair Market Rental Rate" shall be equal to the rent (including additional rent and considering any "base year" or "expense stop" applicable thereto), including escalations, at which third party tenants, extrapolated to the commencement of the applicable Option Term, are, pursuant to transactions completed within the twenty-four (24) month period prior to the commencement of the applicable Option Term, leasing non-sublease, non-encumbered, non-equity, non-expansion renewal and non-renewal space comparable in size, location and quality to the Premises, for a similar lease term, in an arms length transaction, which comparable space is located in the Project or in the Comparable Buildings ("Comparable Transactions"), in either case taking into consideration the differences in Project amenities, parking rates and type of lease, as well as the following concessions: (a) rental abatement concessions, if any, being granted such tenants in connection with such comparable space, (b) tenant improvements or allowances provided or to be provided for such comparable space, taking into account, and deducting the value of, the existing improvements in the Premises, as the case may be, such value to be based upon the age, design, quality of finishes, and layout of the improvements and the extent to which the same could be utilized by a general office user, and (c) all other monetary and non-monetary concessions, if any, being granted such tenants in connection with such comparable space. The Fair Market Rental Rate shall in all events provide for (i) a new "Base Year," which shall be the calendar year in which the Option Term commences if the Option Term commences in any month from January through June, inclusive, and which shall be the calendar year following the calendar year in which the Option Term commences if the Option Term commences in any month from July through December, inclusive, and (ii) Tenant shall not be required to pay any Direct Expenses for the first twelve (12) consecutive calendar months (commencing on the date the Option Term commences); provided, however, that the foregoing Base Year and pass-through concessions shall be factored into the determination of the Fair Market Rental Rate. The Fair Market Rental Rate shall additionally include a determination as to whether, and if so to what extent, Tenant must provide Landlord with financial security, such as a letter of credit or guaranty, for Tenant's obligations during the Option Term. Such determination shall be made by reviewing the extent of financial security then generally being imposed in Comparable Transactions from tenants of comparable financial condition and credit history to the then existing financial condition and credit history of Tenant (with appropriate adjustments to account for differences in the then-existing financial condition of Tenant and such other tenants). If in determining the Fair Market Rental Rate for an Option Term, Tenant is deemed to be entitled to a tenant

improvement, refurbishment or comparable allowance for the improvement or refurbishment of the Premises (the total dollar value of such allowance, the "Option Term TI Allowance"), Landlord may, at Landlord's sole option, elect any or a portion of the following: (A) to grant some or all of the Option Term TI Allowance to Tenant as a lump sum payment to Tenant, and/or (B) in lieu of making a lump sum payment (or portion thereof) to Tenant, to reduce the rental rate component of the Fair Market Rental Rate to be an effective rental rate which takes into consideration that Tenant will not receive a payment of such Option Term TI Allowance, or portion thereof, evidenced in the effective rental rate shall not be paid to Tenant).

- 3.4.6. <u>Terms and Conditions</u>. Other than the Option Rent and the new Base Year as described in Section 3.4.5, the terms and conditions of this Lease during the Option Term shall be on the same terms, covenants and conditions as provided for in this Lease during the initial Term, except that Tenant shall have no additional options to extend or renew the Term beyond the Second Option Term, and Sections 3.5 through 3.10 and Section 5.6 shall not apply during any Option Term.
- 3.4.7. <u>Renewal Amendment.</u> If Tenant is entitled to and properly exercises an option, Landlord shall prepare and deliver to Tenant an amendment (the "<u>Renewal Amendment</u>") to reflect changes in the Base Rent, Base Year, Term, Termination Date, and other appropriate terms. Tenant shall be entitled to review same for compliance with this Lease. If accurate, Tenant shall execute and return the Renewal Amendment to Landlord within thirty (30) days after receipt thereof.
- 3.5 Special Termination Option. Provided that Tenant has not previously exercised its option to lease Building IV as described in Section 3.10 (in which case this Section 3.5 shall be null and void and of no further force or effect), then Tenant shall have the one-time special option (the "Special Termination Option") upon and subject to the terms and conditions set forth below, to terminate this Lease and, if applicable, the other leases, if any, entered into by Tenant with respect to any of the other Project Buildings (collectively, the "Other Leases"), effective as of midnight on the last day of the seventy-second (72nd) full calendar month of the initial Term of this Lease (the "Special Termination Date"), as to not more than fifty percent (50%) of the entire rentable square footage of all space being leased by Tenant in the Project at the time of the Special Termination Notice (hereinafter defined); provided, however, that the portion of the space being terminated by Tenant must be in full floor increments, i.e., following the Special Termination Date, the space being leased by Tenant in the Project may not include any partial floors (other than those partial floors that may have been leased by Tenant pursuant to Section 3.7); and provided further that the Special Termination Portion may not include any space under an Other Lease that will have been leased by Tenant, as of the Special Termination Date, for a period of less than 36 full calendar months (as measured from the commencement date of such Other Lease). Tenant shall specify the portion of the space being terminated by Tenant pursuant to this Section 3.5 (the "Special Termination Portion") in the Special Termination Notice. Notwithstanding the foregoing, if Tenant does not terminate the ground floor(s) within all Project Buildings that Tenant is then entirely leasing, then the Special Termination Portion shall be deemed to include that portion of the ground floor(s) as may reasonably be required by Landlord to enable Landlord to convert the appli
- 3.5.1. Exercise of Special Termination Option. To exercise the Special Termination Option, (i) Tenant must give Landlord written notice (the "Special Termination Notice") of its exercise of the Special Termination Option, which Special Termination Notice must be delivered to Landlord at least 365 days prior to the Special Termination Date; and (ii) at the time of the Special Termination Notice, Tenant shall not be in Default under this Lease.
- 3.5.2. Special Termination Payment. In consideration of Landlord's granting the Special Termination Option to Tenant, if Tenant elects to terminate this Lease as to the Special Termination Portion pursuant to this Section 3.5, as a condition thereto and as a covenant herein Tenant shall pay to Landlord on or before the Special Termination Date in immediately available funds the Special Termination Payment (hereinafter defined). If Tenant fails to deliver the Special Termination Payment to Landlord on or before the Special Termination Date, then at the sole election of Landlord by written notice to Tenant prior to Landlord's receipt of the Special Termination Payment, the Special Termination Notice shall be deemed cancelled and rescinded, and Landlord and Tenant shall continue to be obligated under all the terms and conditions of this Lease. As used herein the term "Special Termination Payment" means an amount equal to (i) the product of Thirty-Six and 75/100 Dollars (\$36.75) multiplied by the number of rentable square feet of the Special Termination Portion, plus (ii) the unamortized cost, as of the Special Termination Date of (1) all tenant improvement allowances, free rent and other monetary concessions, if any, actually received by Tenant in connection with this Lease and, if applicable, the Other Leases covering any space included within the Special Termination Portion, and (2) all leasing commissions and other transaction costs actually paid by or incurred Landlord in connection with this Lease and, if applicable, the Other Leases covering space included within the Special Termination Portion. The unamortized cost described in (ii) above (A) shall be determined separately for each part of the Special Termination Portion contained within different Project Buildings, (B) shall be determined on a straight-line basis over the initial term of the applicable affected space, and (C) shall be prorated based on the proportion that the terminated space in the affected Project Building bears to all of the space previously leased by Tenant within such Project Building (by way of example only, if as of the Special Termination Date Tenant is leasing all of Building II and all of Building III, and if Tenant exercises its Special Termination Option to terminate one-third (1/3) of the space in Building II and one-fourth (1/4) of the space in Building III, the unamortized costs shall be determined separately for all of the space within Building II and all of the space within Building III, but only one-third (1/3) of such unamortized costs for Building II and one-fourth (1/4) of such unamortized costs for Building III shall be included in the Special Termination Payment. At the time of exercising the Special Termination Option, Tenant shall notify Landlord of any subleases covering space contained

within the Special Termination Portion and shall provide Landlord with a true and correct copy of the affected subleases. If Landlord elects, in its sole discretion, to recognize and have one or more subtenants of Tenant within the Special Termination Portion attorn to Landlord as provided in Section 12.6 (the "Continuing Sublease(s)"), then Landlord shall notify Tenant and the sublessees under the Continuing Sublease(s) of such recognition/attornment within sixty (60) days after Landlord's receipt of the Special Termination Notice and copies of the affected subleases as described above. Provided that the sublessees under the Continuing Sublease(s) affirm to Landlord such recognition/attornment in writing prior to the Special Termination Date (the "Sublessee Affirmation"), the portion of the Special Termination Payment determined pursuant to subclause (i) above shall be reduced (but never below zero) by an amount equal to the monthly base rental rate (on a rentable square foot basis) specified in the Continuing Sublease(s) for the month in which the Special Termination Date occurs, multiplied by 12 (to arrive at an annual base rental rate), multiplied by the applicable number of rentable square feet within the applicable Continuing Sublease(s).

- 3.5.3. Special Termination Amendment. As soon as reasonably practicable after Landlord's receipt of a Special Termination Notice from Tenant, Landlord and Tenant shall execute an amendment (the "Special Termination Amendment") to this Lease and, if applicable, to the Other Leases, covering any portion of the Special Termination Portion, effective as of the Special Termination Date, reducing the premises then leased by Tenant by the Special Termination Portion upon the terms and conditions set forth herein. If as a result of Tenant's exercise of the Special Termination Option, the remaining premises shall constitute less than all of the rentable square footage within any of the Project Buildings, then the Special Termination Amendment shall contain such terms and conditions as may be reasonably necessary or appropriate to convert this Lease (and/or the Other Leases, as applicable) to a multi-tenant lease with respect to such premises, including, without limitation, (i) any alterations required by applicable Law or otherwise deemed reasonably necessary by Landlord to make the Building suitable for multi-tenant occupancy, (ii) such alterations and installations as Landlord may reasonably determine are necessary to complete the Lobby Conversion(s), (iii) security measures reasonably approved by Landlord and reasonably satisfactory to Tenant to ensure that access to tenant floors may be restricted to authorized personnel, and (iv) the removal of internal (non-Base Building) stairwells between floors. Item (i) in the immediately preceding sentence shall be at Landlord's sole cost and expense, and items (ii), (iii) and (iv) in the immediately preceding sentence shall be at Landlord may have the Measurement Firm re-measure the remaining Premises in accordance with the Measurement Standard to account for the conversion to a multi-tenant Building (and adjust Rent accordingly).
- 3.5.4. Other Terms. The rights set forth in this Section 3.5 are personal to Original Tenant and may only be exercised by Original Tenant if Tenant is not in Default under this Lease at the time Tenant delivers the Special Termination Notice. After the Special Termination Date and so long as Tenant has paid the Special Termination Payment and surrendered the Premises in the condition required under this Lease, neither party shall have any rights, liabilities or obligations under this Lease with respect to the Special Termination Portion for the period accruing after the Special Termination Date, except those which, by the provisions of this Lease, expressly survive the termination of this Lease. Landlord and Tenant agree that time is expressly of the essence as to every term and provision where time is a factor in this Section 3.5.
- 3.6 <u>Right of First Offer to Purchase</u>. Except with respect to Exempt Transactions (hereinafter defined), if Landlord elects to sell Landlord's entire interest in the Project or if Landlord or a Landlord Affiliate (hereinafter defined) elects to sell Landlord's or such Landlord Affiliate's entire interest in one or more of the Project Buildings ("<u>Landlord's Interest</u>") during the initial Term, Landlord (which for purposes of this Section 3.6 shall hereinafter be deemed to be a reference to Landlord or the applicable Landlord Affiliate(s) owning the applicable Landlord's Interest, or both/all if applicable) shall first offer Landlord's Interest to Tenant by notifying Tenant (the "<u>Offer Notice</u>") of the following economic terms and conditions on which Landlord is willing to sell Landlord's Interest (collectively, the "<u>Economic Terms</u>"):
 - (a) a description of the offered Landlord's Interest;
 - (b) the purchase price and payment terms (including the parties respective obligations in connection with any loan prepayment penalties, brokerage commissions and any tenant improvement allowances or other concessions payable to tenants following the closing);
 - (c) any deposit amount required;
 - (d) the length of any due diligence period;
 - (e) the condition of title;
 - (f) the closing date;
 - (g) the proposed representations and warranties, if any, which Landlord is willing to provide, as well as the "as-is" nature of the transaction and any required waiver of claims against Landlord;
 - (h) the escrow and title company;
 - (i) the allocation of escrow, title and closing costs; and
 - (j) the closing conditions.

The Economic Terms, together with those set forth in this Section 3.6, shall comprise the terms upon which Landlord is willing to sell Landlord's Interest to Tenant. Tenant's right of first offer contained in this Section 3.6 shall not apply with respect to any sale or transfer of Landlord's Interest ("Exempt Transactions") (i) to any entity which directly or indirectly controls, is controlled by or is under common control with Landlord or any of the principals or members of Landlord or transfers among existing principals within Landlord's ownership structure, entities controlled by such principals or immediate family members of such principals (collectively, a "Landlord")

Affiliate"), (ii) to any entity resulting from the merger of or consolidation with Landlord or any Landlord Affiliate, (iii) to any entity acquiring all or substantially all of the assets of Landlord or a Landlord Affiliate; provided, however, that if Landlord's Interest is the sole asset (or constitutes substantially all the assets) of Landlord or the applicable Landlord Affiliate, then this exemption shall not apply unless the acquisition of such assets is part of a larger corporate transaction involving Landlord or the applicable Landlord Affiliate (the intent of the parties being that Landlord or the applicable Landlord Affiliate may not use this exemption as a subterfuge to avoid Tenant's right of first offer provided in this Section 3.6), (iv) any sale, sale and leaseback or other transaction in connection with Landlord's financing of the Project, or (v) pursuant to any foreclosure sale or deed in lieu of foreclosure. Tenant's first offer right pursuant to this Section 3.6 shall continue to apply after a sale or other transfer as described in items (i) through and including (iv) in the immediately preceding sentence above, but shall not apply after a sale or transfer as described in item (v) of the immediately preceding sentence above and, in such case, Tenant's right of first offer shall terminate and shall be of no further force or effect. The rights set forth in this Section 3.6 are personal to Original Tenant and may only be exercised by Original Tenant if Tenant is not in Default under this Lease at the time Tenant delivers the Acceptance Notice (hereinafter defined).

3.6.1. Acceptance. Tenant may exercise Tenant's right of first offer only by giving to Landlord, within twenty (20) days after Tenant's receipt of the Offer Notice (the "Response Period"), written notice of Tenant's agreement to purchase the entire Landlord's Interest on the Economic Terms stated in the Offer Notice or on such other Economic Terms as Landlord and Tenant may mutually agree in writing during the Response Period (the "Acceptance Notice"). During the Response Period, Landlord shall make itself reasonably available to Tenant to discuss the Economic Terms, but shall be under no obligation to negotiate or modify any of the Economic Terms, and no such discussions shall effect a modification of the Economic Terms unless such modification is set forth in a writing signed by Landlord. In that regard, Tenant may, at least two (2) business days prior to the expiration of the Response Period, deliver to Landlord a notice (the "Response Notice") stating that Tenant is interested in purchasing Landlord's Interest, but rejecting some or all of the Economic Terms, in which event Tenant shall include in the Response Notice changes to such rejected Economic Terms that Tenant would accept. If prior to the end of the Response Period Landlord accepts in writing any of Tenant's proposed changes to the Economic Terms set forth in the Response Notice, then the Economic Terms, as so changed, shall be deemed the "Economic Terms" for purposes hereof; provided, however, that Landlord shall have no obligation to accept any such proposed changes and shall have the right in its sole and absolute discretion to reject any or all of such proposed changes (and if Landlord fails to accept Tenant's proposed changes in writing prior to the end of the Response Period, such proposed changes shall be deemed rejected). If Tenant fails to deliver an Acceptance Notice within the Response Period, Tenant's right of first offer as specified in this Section 3.6 shall terminate and shall be of no further force or effect except as set forth in Section 3.6.3. If Tenant delivers an Acceptance Notice within the Response Period, then within ten (10) days after Tenant's delivery thereof, Landlord shall cause to be prepared and delivered to Tenant a draft purchase agreement and escrow instructions (the "Purchase Agreement") setting forth the Economic Terms and such other terms and conditions (the "Non-Economic Terms") as Landlord may consider to be appropriate. During the twenty (20) days after Landlord's delivery of the Purchase Agreement (the "Negotiation Period"), Landlord and Tenant and/or their counsel shall meet (in person or by telephone) as often as is commercially reasonable in a good faith effort to negotiate the Non-Economic Terms based on the then-current standards in the Market Area for similar transactions and to finalize the Purchase Agreement.

3.6.2. Time of the Essence and Failure to Accept or Purchase. If (i) Tenant shall fail to deliver the Acceptance Notice within the Response Period or (ii) Tenant timely delivers the Acceptance Notice but the parties fail to finalize and execute the Purchase Agreement within the Negotiation Period, then, except as set forth in Section 3.6.3, Tenant's right of first offer contained in this Section 3.6 shall automatically terminate and be of no further force or effect, and Landlord may sell and convey Landlord's Interest to any prospective purchaser of its choosing free and clear of any rights of Tenant under this Section 3.6. In addition, if Landlord and Tenant enter into the Purchase Agreement and Tenant fails to consummate the transaction contemplated by the Purchase Agreement within the time periods specified in the Purchase Agreement for any reason other than Landlord's default, then Tenant's right of first offer contained in this Section 3.6 shall automatically terminate and be of no further force or effect, and Landlord may sell and convey Landlord's Interest to any prospective purchaser of its choosing free and clear of any rights of Tenant under this Section 3.6. Notwithstanding the preceding sentence, if Tenant fails to consummate the transaction contemplated by the Purchase Agreement due to a failure of an express condition for the benefit of Tenant set forth in the Purchase Agreement or, if permitted by the Purchase Agreement, Tenant properly terminates the transaction contemplated by the Purchase Agreement due to Tenant's disapproval of title or the physical or legal condition of the property covered by the Purchase Agreement during any contingency period, then the provisions of Section 3.6.3 shall apply.

3.6.3. Re-Offer Right. If (A) Tenant does not timely deliver the Acceptance Notice, but does timely deliver a Response Notice, or (B) Tenant timely delivers the Acceptance Notice but the parties fail to finalize and execute the Purchase Agreement within the Negotiation Period, or (C) the parties enter into the Purchase Agreement but Tenant fails to consummate the transaction contemplated by the Purchase Agreement due to a failure of an express condition for the benefit of Tenant set forth in the Purchase Agreement or, if permitted by the Purchase Agreement, Tenant properly terminates the transaction contemplated by the Purchase Agreement due to Tenant's disapproval of title or the physical or legal condition of the property covered by the Purchase Agreement during the contingency period, if any, set forth in the Purchase Agreement, then Landlord shall re-offer Landlord's Interest to Tenant by delivery of a new Offer Notice ("Re-Offer Right") only in the event that (i) Landlord reduces the purchase price to a prospective purchaser below, in the case of (A), above, the greater of 93% of the purchase price set forth in the Offer Notice or the purchase price that Tenant indicated in the Response Notice was acceptable to Tenant, or, in the case of (B) or (C), above, 93% of the Purchase Price set forth in the Acceptance Notice, or (ii) Landlord materially changes the other Economic Terms set forth in the Offer Notice (in the case of (A), above) or the Acceptance Notice (in the case of (B) or (C), above) in a manner which materially changes the economics of the proposed transaction so as to make the economic value of the transaction materially more favorable to the prospective purchaser than set forth in Tenant's Response Notice (in the case of (A), above) or the Acceptance

Notice (in the case of (B) or (C), above). Pursuant to its Re-Offer Right, Tenant shall have the right to purchase Landlord's Interest at such reduced purchase price or materially changed Economic Terms in accordance with the terms of this right of first offer, except that Tenant shall have ten (10) days from its receipt of the new Offer Notice in which to deliver to Landlord Tenant's Acceptance Notice. If Landlord delivers a new Offer Notice to Tenant pursuant to this Section 3.6.3, and if Tenant fails to deliver Tenant's Acceptance Notice to Landlord within said ten (10) day period, Tenant's rights under this Section 3.6 shall automatically become null and void and of no further force or effect except as set forth hereinbelow. Notwithstanding anything to the contrary set forth in this Section 3.6.3, in any case in which Tenant's rights under this Section 3.6 become null and void pursuant to Section 3.6.1 or this Section 3.6.3, such rights shall nevertheless be automatically reinstated if Landlord fails to complete the sale or transfer of Landlord's Interest to a prospective purchaser within two hundred seventy (270) days after the later to occur of the expiration of the Response Period or, if applicable, the expiration of the Negotiation Period.

- 3.7 <u>Right of First Offer to Lease Building I Space</u>. Landlord hereby grants to the Tenant a one-time right of first offer (the "<u>Building I First Offer Right</u>") during the initial Term to lease space in Building I (collectively, the "<u>Building I First Offer Space</u>") on the terms and conditions set forth hereinbelow. Notwithstanding the foregoing, the Building I First Offer Right shall in no event apply with respect to space leased to any Landlord Affiliate, Landlord's property management, or with respect to space leased pursuant to any lease of the Building I First Offer Space existing as of the date hereof, and shall be subordinate to all rights of existing tenants of the Project to lease the Building I First Offer Space (whether pursuant to rights of first offer, expansion options or must take requirements) (collectively, the "<u>Building I Superior Right Holders</u>"); provided, however, that Landlord shall notify Tenant upon request from time to time of the rights of any Building I Superior Right Holders.
- 3.7.1. <u>Procedure for Offer</u>. During the initial Term, prior to entering into a new lease for the Building I First Offer Space with any third party other than a Building I Superior Right Holder, Landlord shall notify Tenant (the "<u>Building I First Offer Notice</u>") when and if the Building I First Offer Space will become available for lease. Pursuant to such Building I First Offer Notice, Landlord shall offer to lease to Tenant the Building I First Offer Space. The Building I First Offer Notice shall describe the space so offered to Tenant (and, at Landlord's election, also shall designate configurations in which Landlord is willing to lease such Building I First Offer Space), the anticipated date of availability, the rent (the "<u>Building I First Offer Rent</u>"), the estimated rentable square footage, the term and the other economic terms upon which Landlord is willing to lease such space to Tenant.
- 3.7.2. Procedure for Acceptance. If Tenant wishes to exercise Tenant's Building I First Offer Right with respect to the Building I First Offer Space described in the Building I First Offer Notice, then within ten (10) Business Days of delivery of the Building I First Offer Notice to Tenant, Tenant shall deliver notice (the "Building I First Offer Exercise Notice") to Landlord of Tenant's election to exercise its Building I First Offer Right with respect to the entire Building I First Offer Space (or designated configurations, if any) described in the Building I First Offer Notice on the terms contained in the Building I First Offer Notice; provided, however, if, concurrent with Tenant's exercise of the Building I First Offer Right, Tenant notifies Landlord that it does not accept the Building I First Offer Rent set forth in the Building I First Offer Notice, and if Landlord and Tenant cannot agree on the Building I First Offer Rent within ten (10) days after receipt of such notice, then the Building I First Offer Rent shall be determined in accordance with the same procedures as set forth in Sections 3.4.4 and 3.4.5 (i.e., at the Fair Market Rental Rate). If Landlord has designated configurations with respect to the Building I First Offer Space, and if Tenant elects to exercise its Building I First Offer Right on a designated configuration only, as opposed to the entire Building I First Offer Space set forth in the Building I First Offer Notice, then Landlord shall be free to lease any or all of the remaining space described in the Building I First Offer Notice to anyone to whom Landlord desires on any terms Landlord desires and Tenant's Building I First Offer Right shall terminate and be null and void with respect to the remaining Building I First Offer Space described in the Building I First Offer Notice. If Tenant does not deliver the Building I First Offer Exercise Notice to Landlord within the ten (10) Business Day notice period, then Landlord shall be free to lease any or all of the space described in the Building I First Offer Notice to anyone to whom Landlord desires on any terms Landlord desires and Tenant's Building I First Offer Right shall terminate and be null and void with respect to the Building I First Offer Space described in the Building I First Offer Notice. Notwithstanding the foregoing, if Landlord shall in its offer to lease to third parties materially increase or decrease the size of the Building I First Offer Space from that offered to Tenant, Landlord shall re-offer such space to Tenant in accordance with the terms of this Section 3.7 (and Tenant shall deliver the Building I First Offer Exercise Notice, if at all, in response to any such re-offer of space by Landlord within ten (10) Business Days). Notwithstanding anything to the contrary contained herein, Tenant must elect to exercise its Building I Right of First Offer, if at all, with respect to all of the Building I First Offer Space (or designated configurations, if any) offered by Landlord to Tenant at any particular time.
- 3.7.3. Construction In Building I First Offer Space. Tenant shall take the Building I First Offer Space in its "as is" condition, and, unless the terms of the Building I First Offer Notice provide otherwise, the construction of improvements in the Building I First Offer Space shall be performed by Tenant in compliance with the terms of Section 9.4 of this Lease. Unless the terms of the Building I First Offer Notice specify otherwise, and notwithstanding if the Fair Market Rental Rate is used to determine any Building I First Offer Rent, Tenant agrees that it shall commence payment of rent for any Building I First Offer Space, and the term of the Building I First Offer Space shall commence, upon the date which is the earlier to occur of (i) the date Tenant commences to conduct business in any Building I First Offer Space, and (ii) the date which is one hundred twenty (120) days after the date Landlord delivers any such Building I First Offer Space to Tenant, and Tenant's lease of the Building I First Offer Space shall terminate upon the termination date set forth in the Building I First Offer Notice. If the Fair Market Rental Rate is used to determine any Building I First Offer Rent, and if such Fair Market Rental Rate has not been finally determined by the Building I First Offer Commencement Date, then Tenant shall pay rent for the Building I First Offer Space in the amount set forth in the Building I First Offer Notice until the Fair Market Rental Rate is determined, at which time an appropriate retroactive adjustment shall be made for any overpayment or underpayment of rent prior to such determination. The rights set forth in this Section 3.7 are personal to Original Tenant and may only be exercised by Original Tenant if Tenant is not in Default under this Lease at the time Tenant delivers the Building I Offer Exercise Notice.

- 3.7.4. New Lease. If Tenant timely exercises Tenant's right to lease the Building I First Offer Space as set forth herein, Tenant shall within thirty (30) days after receipt from Landlord execute a new lease for such Building I First Offer Space upon the terms and conditions as set forth in the Building I First Offer Notice and this Section 3.7, and otherwise on substantially the same non-economic terms and conditions as this Lease other than (i) the provisions of Sections 3.4 through 3.10 (provided, however, that such new lease shall nevertheless recognize Tenant's Special Termination Option under Section 3.5), Section 5.6, Section 7.5, Sections 23.1, Section 23.6 (unless the Building I First Offer Space constitutes at least twenty-five percent (25%) of Building I), Section 23.7 and the Work Letter, and (ii) those other changes that Landlord may deem necessary or appropriate to reflect that the Building I First Offer Space is in a different (and multitenant) Project Building than the Premises under this Lease (for example, Building I will have differing allocations of Direct Expenses, may have different access, parking and security procedures, will have a separate "Tenant's Share" of Direct Expenses, may have a different Base Year, etc.) and that Building I may be or become owned by a Landlord Affiliate or third party.
- 3.8 Option to Lease Building III. Landlord hereby grants to the Tenant during the initial Term a one-time option (the "Building III Lease Option") to lease all (but not less than all) of the entire rentable space within Building III on the same terms and conditions as set forth herein with respect to Building II (including the Tenant Improvement Allowance and the subterranean parking rights) and the term shall be coterminous with the Term of this Lease, except (i) Tenant's rental obligations with respect to the Building III space shall commence on the date (the "Building III Commencement Date") that is the first day of the twelfth (12th) month anniversary of the Commencement Date of this Lease (unless the Building III space is not substantially completed by then, in which case the rental obligations shall commence on "substantial completion" of the Building III space, as defined in the Work Letter), but shall otherwise match the rental and economic obligations (on a rentable square foot basis) set forth in this Lease, including without limitation the then existing Base Rent rate under this Lease, Base Rent increases under this Lease, Security Deposit (except that the Security Deposit shall be based on the last month's rent for the Building III space) and the 2007 Base Year (provided, however if the Building III space is not substantially completed by July 1, 2007, then the Base Year shall be 2008; if the Building III space is not substantially completed by July 1, 2008, then the Base Year shall be 2009, and so on), (ii) all provisions which vary based on rentable square feet shall be adjusted to reflect the rentable square footage of the Building III space, (iii) for changes that Landlord may deem necessary or appropriate to reflect that the Building III space is in a different Project Building than the Premises under this Lease and may be or become owned by a Landlord Affiliate or third party, (iv) Sections 3.5 through 3.10 (provided, however, that the new lease shall nevertheless recognize Tenant's Special Termination Option under Section 3.5), Section 7.5 and Section 23.1, and (v) all construction dates, milestones and completion dates shall be extended appropriately to reflect the extended commencement date for such space and any material change in construction conditions. Subject to (v) above, Landlord shall construct Building III and the tenant improvements therein on substantially the same terms and conditions as set forth herein with respect to Building II. The Building III Lease Option shall be exercisable only by written notice executed by a duly authorized officer of Tenant and delivered by Tenant to Landlord not later than the Commencement Date for Building II (the "Building III Option Exercise Notice"). If Tenant exercises the Building III Lease Option, Tenant shall within thirty (30) days after receipt from Landlord execute a new lease for the Building III Space upon the terms and conditions as set forth in this Section 3.8. If Tenant fails to exercise the Building III Lease Option as set forth herein, then the Building III Lease Option shall expire and be of no further force or effect and the space within Building III, if and when Building III is constructed, shall be deemed to be the space subject to the right of first offer set forth in Section 3.9. The rights set forth in this Section 3.8 are personal to Original Tenant and may only be exercised by Original Tenant if Tenant is not in Default under this Lease at the time Tenant delivers the Building III Option Exercise Notice.
- 3.9 Right of First Offer to Lease Building III Space. If Tenant fails to exercise the Building III Lease Option as described in Section 3.8, and if Building III is nevertheless constructed, Landlord hereby grants to the Tenant a one-time right of first offer (the "Building III First Offer Right") during the initial Term to lease space in Building III (collectively, the "Building III First Offer Space") on the terms and conditions set forth hereinbelow. Notwithstanding the foregoing, the Building III First Offer Right shall in no event apply with respect to space leased to any Landlord Affiliate or Landlord's property manager, or space leased to any third party or parties prior to substantial completion of Building III (a "First Generation Lease"), the intent of the parties being that if Tenant fails to exercise the Building III Lease Option as described in Section 3.8, Landlord shall have the right to market and lease any or all of the space within Building III to one or more third parties free and clear of Tenant's rights under this Section 3.9 at any time prior to substantial completion of Building III, and the Building III First Offer Right shall be subordinate to all rights of tenants under First Generation Leases to lease the Building III First Offer Space (whether pursuant to rights of first offer, expansion options or must take requirements) (collectively, the "Building III Superior Right Holders"); provided, however, that Landlord shall notify Tenant upon request from time to time of the rights of any Building III Superior Right Holders.
- 3.9.1. Procedure for Offer. During the initial Term, and after substantial completion of Building III, prior to entering into a new lease for the Building III First Offer Space with any third party other than a Building III Superior Right Holder, Landlord shall notify Tenant (the "Building III First Offer Notice") when and if the Building III First Offer Space will become available for lease. Pursuant to such Building III First Offer Notice, Landlord shall offer to lease to Tenant the Building III First Offer Space. The Building III First Offer Notice shall describe the space so offered to Tenant (and, at Landlord's election, also shall designate configurations in which Landlord is willing to lease such Building III First Offer Space), the anticipated date of availability, the rent (the "Building III First Offer Rent"), the estimated rentable square footage, the term and the other economic terms upon which Landlord is willing to lease such space to Tenant.
- 3.9.2. <u>Procedure for Acceptance</u>. If Tenant wishes to exercise Tenant's Building III First Offer Right with respect to the Building III First Offer Space described in the Building III First Offer Notice, then within ten (10) Business Days of delivery of the Building III First Offer Notice to Tenant, Tenant shall deliver notice (the

"Building III First Offer Exercise Notice") to Landlord of Tenant's election to exercise its Building III First Offer Right with respect to the entire Building III First Offer Space (or designated configurations, if any) described in the Building III First Offer Notice on the terms contained in the Building III First Offer Notice; provided, however, if, concurrent with Tenant's exercise of the Building III First Offer Right, Tenant notifies Landlord that it does not accept the Building III First Offer Rent set forth in the Building III First Offer Notice, and if Landlord and Tenant cannot agree on the Building III First Offer Rent within ten (10) days after receipt of such notice, then the Building III First Offer Rent shall be determined in accordance with the same procedures as set forth in Sections 3.4.4 and 3.4.5 (i.e., at the Fair Market Rental Rate). If Landlord has designated configurations with respect to the Building III First Offer Space, and if Tenant elects to exercise its Building III First Offer Right on a designated configuration only, as opposed to the entire Building III First Offer Space set forth in the Building III First Offer Notice, then Landlord shall be free to lease any or all of the remaining space described in the Building III First Offer Notice to anyone to whom Landlord desires on any terms Landlord desires and Tenant's Building III First Offer Right shall terminate and be null and void with respect to the remaining Building III First Offer Space described in the Building III First Offer Notice. If Tenant does not deliver the Building III First Offer Exercise Notice to Landlord within the ten (10) Business Day notice period, then Landlord shall be free to lease any or all of the space described in the Building III First Offer Notice to anyone to whom Landlord desires on any terms Landlord desires and Tenant's Building III First Offer Right shall terminate and be null and void with respect to the Building III First Offer Space described in the Building III First Offer Notice. Notwithstanding the foregoing, if Landlord shall in its offer to lease to third parties materially increase or decrease the size of the Building III First Offer Space from that offered to Tenant, Landlord shall re-offer such space to Tenant in accordance with the terms of this Section 3.9 (and Tenant shall deliver the Building III First Offer Exercise Notice, if at all, in response to any such re-offer of space by Landlord within ten (10) Business Days). Notwithstanding anything to the contrary contained herein, Tenant must elect to exercise its Building III Right of First Offer, if at all, with respect to all of the Building III First Offer Space (or designated configurations, if any) offered by Landlord to Tenant at any particular time

- 3.9.3. Construction In Building III First Offer Space. Tenant shall take the Building III First Offer Space in its "as is" condition, and, unless the terms of the Building III First Offer Notice provide otherwise, the construction of improvements in the Building III First Offer Space shall be performed by Tenant in compliance with the terms of Section 9.4 of this Lease. Unless the terms of the Building III First Offer Notice specify otherwise, and notwithstanding if the Fair Market Rental Rate is used to determine any Building III First Offer Rent, Tenant agrees that it shall commence payment of rent for any Building III First Offer Space, and the term of the Building III First Offer Space shall commence, upon the date which is the earlier to occur of (i) the date Tenant commences to conduct business in any Building III First Offer Space, and (ii) the date which is one hundred twenty (120) days after the date Landlord delivers any such Building III First Offer Space to Tenant, and Tenant's lease of the Building III First Offer Space shall terminate upon the termination date set forth in the Building III First Offer Notice. If the Fair Market Rental Rate is used to determine any Building III First Offer Rent, and if such Fair Market Rental Rate has not been finally determined by the Building III First Offer Commencement Date, then Tenant shall pay rent for the Building III First Offer Space in the amount set forth in the Building III First Offer Notice until the Fair Market Rental Rate is determined, at which time an appropriate retroactive adjustment shall be made for any overpayment or underpayment of rent prior to such determination. The rights set forth in this Section 3.9 are personal to Original Tenant and may only be exercised by Original Tenant if Tenant is not in Default under this Lease at the time Tenant delivers the Building III Offer Exercise Notice.
- 3.9.4. New Lease. If Tenant timely exercises Tenant's right to lease the Building III First Offer Space as set forth herein, Tenant shall within thirty (30) days after receipt from Landlord execute a new lease for such Building III First Offer Space upon the terms and conditions as set forth in the Building III First Offer Notice and this Section 3.9, and otherwise on substantially the same non-economic terms and conditions as this Lease other than (i) the provisions of Sections 3.4 through 3.10 (provided, however, that such new lease shall nevertheless recognize Tenant's Special Termination Option under Section 3.5), Section 5.6, Section 7.5, Sections 23.1, Section 23.6 (unless the Building III First Offer Space constitutes at least twenty-five percent (25%) of Building III), Section 23.7 and the Work Letter, and (ii) those other changes that Landlord may deem necessary or appropriate to reflect that the Building III First Offer Space is in a different (and multi-tenant) Project Building than the Premises under this Lease (for example, Building III will have differing allocations of Direct Expenses, may have different access, parking and security procedures, will have a separate "Tenant's Share" of Direct Expenses, may have a different Base Year, etc.) and that Building III may be or become owned by a Landlord Affiliate or third party.
- 3.10 Right of First Offer to Lease Building IV. Landlord hereby grants to the Tenant a one-time right of first offer (the "Building IV First Offer Right") during the initial Term to lease space in Building IV (collectively, the "Building IV First Offer Space") on the terms and conditions set forth in this Section 3.10. Notwithstanding the foregoing, the Building IV First Offer Right shall in no event apply with respect to space leased to any Landlord Affiliate or Landlord's property management.
- 3.10.1. Procedure for Offer. Prior to entering into a new lease for the Building IV First Offer Space with any third party other than a Building IV Superior Right Holder, Landlord shall notify Tenant (the "Building IV First Offer Notice") when and if Landlord determines that the Building IV First Offer Space shall become available for lease. Pursuant to such Building IV First Offer Notice, Landlord shall offer to lease to Tenant the Building IV First Offer Space. The Building IV First Offer Notice shall describe the space so offered to Tenant (and, at Landlord's election, also shall designate configurations in which Landlord is willing to lease such Building IV First Offer Space), the anticipated date of availability, the rent (the "Building IV First Offer Rent"), the estimated rentable square footage, the term and the other economic terms upon which Landlord is willing to lease such space to Tenant.
- 3.10.2. <u>Procedure for Acceptance</u>. If Tenant wishes to exercise Tenant's Building IV First Offer Right with respect to the Building IV First Offer Space described in the Building IV First Offer Notice, then within thirty (30) days after delivery of the Building IV First Offer Notice to Tenant, Tenant shall deliver notice (the "<u>Building IV First Offer Exercise Notice</u>") to Landlord of Tenant's election to exercise its Building IV First Offer Right with respect to the entire Building IV First Offer Space (or designated configurations, if any) on the terms contained in

such notice; provided, however, if, concurrent with Tenant's exercise of the Building IV First Offer Right, Tenant notifies Landlord that it does not accept the Building IV First Offer Rent set forth in the Building IV First Offer Notice, and if Landlord and Tenant cannot agree on the Building IV First Offer Rent within ten (10) days after receipt of such notice, then the Building IV First Offer Rent shall be determined in accordance with the same procedures as set forth in Sections 3.4.4 and 3.4.5 (i.e., at the Fair Market Rental Rate). If Landlord has designated configurations with respect to the Building IV First Offer Space, and if Tenant elects to exercise its Building IV First Offer Right on a designated configuration only, as opposed to the entire Building IV First Offer Space set forth in the Building IV First Offer Notice, then Landlord shall be free to lease any or all of the remaining space described in the Building IV First Offer Notice to anyone to whom Landlord desires on any terms Landlord desires and Tenant's Building IV First Offer Right shall terminate and be null and void with respect to the remaining Building IV First Offer Space described in the Building IV First Offer Notice. If Tenant does not deliver the Building IV First Offer Exercise Notice to Landlord within the thirty (30) day notice period, then Landlord shall be free to lease any or all of the space described in the Building IV First Offer Notice to anyone to whom Landlord desires on any terms Landlord desires and Tenant's Building IV First Offer Right shall terminate and be null and void with respect to the Building IV First Offer Space described in the Building IV First Offer Notice. Notwithstanding the foregoing, if Landlord shall its offer to lease to third parties materially increase or decrease the size of the Building IV First Offer Space from that offered to Tenant, Landlord shall re-offer such space to Tenant in accordance with the terms of this Section 3.10 (and Tenant shall deliver the Building IV First Offer Exercise Notice, if at all, in response to any such re-offer of space by Landlord within ten (10) Business Days). Notwithstanding anything to the contrary contained herein, Tenant must elect to exercise its Building IV Right of First Offer, if at all, with respect to all of the Building IV First Offer Space (or designated configurations, if any) offered by Landlord to Tenant at any particular time.

- 3.10.3. Construction In Building IV First Offer Space. Tenant shall take the Building IV First Offer Space in its "as is" condition, and, unless the terms of the Building IV First Offer Notice provide otherwise, the construction of improvements in the Building IV First Offer Space shall be performed by Tenant in compliance with the terms of Section 9.4 of this Lease. Unless the terms of the Building I First Offer Notice specify otherwise, and notwithstanding if the Fair Market Rental Rate is used to determine any Building I First Offer Rent, Tenant agrees that it shall commence payment of rent for any Building IV First Offer Space, and the term of the Building IV First Offer Space shall commence, upon the date which is the earlier to occur of (i) the date Tenant commences to conduct business in any Building IV First Offer Space, and (ii) the date which is one hundred twenty (120) days after the date Landlord delivers any such Building IV First Offer Space to Tenant, and Tenant's lease of the Building IV First Offer Space shall terminate upon the termination date set forth in the Building IV First Offer Notice. If the Fair Market Rental Rate is used to determine any Building IV First Offer Rent, and if such Fair Market Rental Rate has not been finally determined by the Building IV First Offer Commencement Date, then Tenant shall pay rent for the Building IV First Offer Space in the amount set forth in the Building IV First Offer Notice until the Fair Market Rental Rate is determined, at which time an appropriate retroactive adjustment shall be made for any overpayment or underpayment of rent prior to such determination. The rights set forth in this Section 3.7 are personal to Original Tenant and may only be exercised by Original Tenant if Tenant is not in Default under this Lease at the time Tenant delivers the Building IV Offer Exercise Notice
- 3.10.4. New Lease. If Tenant timely exercises Tenant's right to lease the Building IV First Offer Space as set forth herein, Tenant shall within thirty (30) days after receipt from Landlord execute a new lease for such Building IV First Offer Space upon the terms and conditions as set forth in the Building IV First Offer Notice and this Section 3.10, and otherwise on substantially the same non-economic terms and conditions as this Lease other than (i) the provisions of Sections 3.4 through 3.10 (provided, however, that such new lease shall nevertheless recognize Tenant's Special Termination Option under Section 3.5), Section 5.6, Section 7.5, Sections 23.1, Section 23.6 (unless the Building I First Offer Space constitutes at least twenty-five percent (25%) of Building I), Section 23.7 and the Work Letter, and (ii) those other changes that Landlord may deem necessary or appropriate to reflect that the Building IV First Offer Space is in a different (and potentially multi-tenant) Project Building than the Premises under this Lease (for example, Building IV will have differing allocations of Direct Expenses, may have different access, parking and security procedures, will have a separate "Tenant's Share" of Direct Expenses, may have a different Base Year, etc.) and that Building IV may be or become owned by a Landlord Affiliate or third party.
- 3.11 References to Landlord. For purposes of the foregoing Sections 3.6 through 3.10, if the applicable Project Building to which the applicable Sections apply become owned by a Landlord Affiliate (or, for purposes of Sections 3.7 through 3.10, by a third party), then references to "Landlord" in those Sections shall be deemed to be a reference to the applicable Landlord Affiliate (or, for purposes of Sections 3.7 through 3.10, the applicable third party) owning the applicable Project Building. As part of Landlord's sale or transfer of any of the Project Buildings to a Landlord Affiliate or a third party, Landlord will notify such Landlord Affiliate or third party of Tenant's rights under the applicable Sections with respect to the Project Building in question and shall require such Landlord Affiliate or third party to recognize the rights of Tenant and assume the obligations of "Landlord" under such applicable Sections with respect thereto. At Tenant's request, such recognition and assumption shall be memorialized pursuant to a commercially reasonable document prepared by Landlord, executed by Landlord and the Landlord Affiliate or third party, as applicable, and delivered to Tenant. At such time as any of Tenant's rights under the foregoing Sections 3.6 through 3.10 terminate or become null and void, Tenant shall, at the request of Landlord or the applicable Landlord Affiliate or third party, execute and deliver to Landlord or the applicable Landlord Affiliate or third party, execute and deliver to Landlord or null/void rights.

4. Rent.

4.1 <u>Payment of Rent.</u> Tenant shall pay, without prior notice or demand, to Landlord or Landlord's agent at the management office of the Project, or, at Landlord's option, at such other place as Landlord may from time to time designate in writing, base rent ("<u>Base Rent</u>") as set forth in Section 1.10, in equal monthly installments as set forth

in Section 1.10, in advance, on or before the first day of each and every calendar month during the Term, without any setoff or deduction whatsoever (except as otherwise expressly set forth in this Lease); provided, however, that at the time of Tenant's execution of this Lease, Tenant shall make an initial payment of Base Rent equal to Five Hundred Thirty Thousand Dollars (\$530,000.00), which amount shall be held by Landlord and credited against the Base Rent first coming due under this Lease. If any Rent payment date (including the Commencement Date) falls on a day of the month other than the first day of such month or if any payment of Rent is for a period which is shorter than one month, the Rent for any fractional month shall accrue on a daily basis for the period from the date such payment is due to the end of such calendar month or to the end of the Term at a rate per day which is equal to 1/365 of the applicable annual Rent. All other payments or adjustments required to be made under the terms of this Lease that require proration on a time basis shall be prorated on the same basis. "Additional Rent" means all sums (exclusive of Base Rent) that Tenant is required to pay Landlord under this Lease. Additional Rent and Base Rent are sometimes collectively referred to in this Lease as "Rent". Recurring monthly charges of Additional Rent shall be due and payable in advance on the first day of each calendar month together with Base Rent, without notice or demand. Unless otherwise specified in this Lease, all other items of Rent shall be due and payable by Tenant on or before thirty (30) days after billing by Landlord. All payments of Rent shall be by good and sufficient check (drawn on a bank having a branch office in Irvine or Newport Beach, California) or by other means (such as automatic debit or electronic transfer) acceptable to Landlord for currency which, at the time of payment, is legal tender for private or public debts in the United States of America. Landlord's acceptance of less than the correct amount of Rent shall be considered a payment on account of the earliest Rent due. No endorsement or statement on a check or letter accompanying a check or payment shall be considered an accord and satisfaction, and either party may accept the check or payment without prejudice to that party's right to recover the balance or pursue other available remedies. Tenant's covenant to pay Rent is independent of every other covenant in this Lease.

4.2 <u>Late Payment.</u> If Tenant fails to pay any item or installment of Rent within five (5) days of when due (provided that Tenant shall be entitled to a grace notice of five (5) days for the first two (2) late payments of Rent in a given calendar year), then Tenant shall pay to Landlord a late charge equal to three percent (3%) of the overdue amount plus any reasonable attorneys' fees incurred by Landlord by reason of Tenant's failure to pay Rent and/or other charges when due hereunder. The late charge shall be deemed Additional Rent and the right to require it shall be in addition to all of Landlord's other rights and remedies hereunder or at law and shall not be construed as liquidated damages or as limiting Landlord's remedies in any manner. In addition to the late charge described above, any Rent or other amounts owing hereunder which are not paid by Landlord or Tenant to the other within ten (10) Business Days after the date they are due shall bear interest from the date when due until paid at a rate per annum equal to the lesser of (i) the annual "Bank Prime Loan" rate cited in the Federal Reserve Statistical Release Publication G.13(415), published on the first Tuesday of each calendar month (or such other comparable index as Landlord and Tenant shall reasonably agree upon if such rate ceases to be published) plus two (2) percentage points, and (ii) the highest rate permitted by Applicable Law (the "<u>Interest Rate</u>").

5. Operating Expenses and Real Estate Taxes.

- 5.1 <u>General Terms.</u> In addition to paying the Base Rent specified in Section 4 of this Lease, Tenant shall pay, as Additional Rent, Tenant's Share of the annual Direct Expenses (as those terms are hereinafter defined), which are in excess of the amount of Direct Expenses applicable to the Base Year (hereinafter defined); provided, however, that in no event shall any decrease in Direct Expenses for any Expense Year (hereinafter defined) below Direct Expenses for the Base Year entitle Tenant to any decrease in Base Rent or any credit against sums due under this Lease. Without limitation on other obligations of Tenant and Landlord which survive the expiration of the Term, the obligations of Tenant to pay such Additional Rent attributable to the period of time prior to the Termination Date, as the same may be extended pursuant to the provisions of this Lease, or earlier termination of this Lease, and Landlord's obligation to refund to Tenant any overpayments of such Additional Rent, shall survive the expiration of the Term. Notwithstanding anything to the contrary set forth in this Lease, in no event shall Tenant be obligated to pay Tenant's Share of Direct Expenses (i) for any period prior to the commencement of the second (2nd) Lease year (the parties' intent being that Tenant shall have 365 days following the Commencement Date free of pass-throughs of Direct Expenses) or (ii) attributable to any period of time after the Term has ended except to the extent Tenant continues to occupy the Premises.
- 5.2 <u>Certain Definitions Relating to Additional Rent.</u> As used in this Section 5, the following terms shall have the meanings hereinafter set forth:
 - 5.2.1. "Base Year" shall mean the period set forth in Section 1.1.2.
 - 5.2.2. "Direct Expenses" shall mean "Operating Expenses" and "Tax Expenses."
- 5.2.3. "Expense Year" shall mean each calendar year in which any portion of the Term falls, through and including the calendar year in which the Term expires, provided that Landlord, upon notice to Tenant, may change the Expense Year from time to time to any other twelve (12) consecutive month period, and, in the event of any such change, Tenant's Share of Direct Expenses shall be equitably adjusted for any Expense Year involved in any such change.
- 5.2.4. "Operating Expenses" shall mean, except as otherwise provided in this Section 5.2.4 or otherwise in this Lease, all expenses, costs and amounts of every kind and nature which Landlord pays or accrues during any Expense Year in connection with the ownership, management, maintenance, security, repair, replacement, restoration or operation of the Project, or any portion thereof, subject to the allocation thereof as set forth in Section 5.3, below. Without limiting the generality of the foregoing, Operating Expenses shall specifically include any and all of the following: (i) the cost of supplying all utilities and services, the cost of operating, repairing, maintaining, and renovating the utility, telephone, mechanical, sanitary, and storm drainage systems, and the cost of maintenance and service contracts in connection therewith; (ii) the cost of licenses, certificates, permits and inspections, the cost of contesting any governmental enactments which are reasonably anticipated to reduce Operating Expenses, and the

costs incurred in connection with a transportation system management program or similar program to the extent required by applicable Law; (iii) the cost of all insurance carried by Landlord in connection with the Project; (iv) the cost of landscaping, re-lamping, and all supplies, tools, equipment and materials used in the operation, repair and maintenance of the Project, or any portion thereof; (v) costs incurred in connection with the maintenance, repair, replacement and operation of the Parking Facilities and Common Areas (including, without limitation, resurfacing, repainting, restriping and cleaning); (vi) fees and other costs reasonably incurred, including management fees, consulting fees, legal fees and accounting fees, of all persons engaged by Landlord in connection with the management, operation, maintenance and repair of the Project; (vii) payments under any equipment rental agreements and the fair rental value of any management office space; (viii) wages, salaries and other compensation and benefits, including taxes levied thereon, of all persons engaged in the operation, maintenance and security of the Project, including the Project manager and Project engineer(s) and all employees at levels below the Project manager; (ix) operation, repair, maintenance and replacement of all systems and equipment and components thereof of the Building; (x) the cost of janitorial, alarm, security and other services, and repair of wall and floor coverings, ceiling tiles and fixtures in Building Common Areas, (xi) maintenance and replacement of curbs and walkways, repair to roofs and reroofing; (xii) amortization over its reasonable useful life (including interest on the unamortized cost at the Interest Rate) of the cost of acquiring or the rental expense of personal property used in the maintenance, operation and repair of the Project, or any portion thereof; (xiii) the cost of capital improvements, repairs, replacements or other costs incurred in connection with the Project (A) which are intended to effect economies in the operation or maintenance of the Project, or any portion thereof, to the extent of cost savings reasonably anticipated by Landlord at the time of such expenditure to be incurred in connection therewith, or (B) that are required under any Law, or (C) are required to replace improvements, systems or equipment as necessary to maintain the Project in the condition required by this Lease; provided, however, that any such permitted capital expenditure shall be amortized (including interest at the Interest Rate) over its reasonable useful life (or, at Landlord's election, the anticipated cost recovery period), as determined by Landlord in accordance with generally accepted real estate accounting practices; (xiv) costs, fees, charges or assessments imposed by, or resulting from any mandate imposed on Landlord by, any federal, state or local government for fire and police protection, trash removal, community services, or other services which do not constitute "Tax Expenses" as that term is defined in Section 5.2.5, below; and (xiv) payments under any easement, license, operating agreement, declaration, covenants, conditions and restrictions, or instrument pertaining to the sharing of costs by the Building with other Project Buildings which does not conflict with the terms of this Lease. Only as provided hereinafter in this Section 5.2.4 below in items [i] [ii] and [iii], if Landlord incurs costs or expenses associated with or relating to separate items or categories or subcategories of Operating Expenses which were not part of Operating Expenses during the entire Base Year, Operating Expenses for the Base Year shall be deemed increased by the amounts Landlord would have incurred during the Base Year with respect to such costs and expenses had such separate items or categories or subcategories of Operating Expenses been included in Operating Expenses during the entire Base Year. The foregoing shall only apply as follows: [i] in the event and to the extent any portion of the Project is covered by a warranty or service agreement which provides warranty-type protection at any time during the Base Year and is not covered by such warranty or such warranty-type protection under such service agreement in a subsequent Expense Year to the same extent, Operating Expenses for the Base Year shall be deemed increased by the amount Landlord would have incurred during the Base Year with respect to the items or matters covered by the subject warranty or warranty-type protection, had such warranty or such service agreement not been in effect during the Base Year; [ii] any insurance premium resulting from any new forms of insurance, including earthquake or terrorism insurance, carried in any Expense Year but not in the Base Year shall be deemed to be included in Operating Expenses for the Base Year; and [iii] in the event any new categories of costs relating directly and solely to the Project Parking Facilities are first included in Operating Expenses in an Expense Year subsequent to the Base Year, Operating Expenses for the Base Year shall be retroactively increased (with respect to Expense Years containing such new category of costs) by the amount Landlord would have incurred had such category of costs been included in Operating Expenses during the Base Year. Notwithstanding anything in this Section 5.2.4 to the contrary, for purposes of this Lease, Operating Expenses and Tax Expenses shall not, however, include the following:

- (a) costs incurred in connection with the original construction of the Project or any of the Project Buildings; the cost of any "tap fees" or one-time lump sum sewer or water connection fees for the Project payable in connection with the original construction of the Project and the cost of any major changes in the Project, such as additions or deletions of floors; the cost to replace major Building systems (such as the HVAC system) because the same have reached the end of their useful life (irrespective of whether such replacement results in reducing Operating Expenses); and the cost to repair any damage to the Building or the Project in connection with any type of casualty, event of damage or destruction to the extent covered by Landlord's insurance (or to the extent covered by Landlord's Contribution, as hereinafter defined, with respect to any uninsured casualty); provided, however, that Operating Expenses shall include the commercially reasonable amount of any deductible;
- (b) costs of the design and construction of tenant improvements to the Premises or the premises of other tenants or other occupants and the amount of any allowances or credits paid to or granted to tenants or other occupants for any such design or construction;
 - (c) except as set forth in items (xii) and (xiii), above, depreciation, interest and principal payments on mortgages and other debt costs, if any;
- (d) marketing costs, legal fees, space planners' fees, advertising and promotional expenses, and brokerage fees incurred in connection with the original development, subsequent improvement, or original or future leasing of the Project (including tenant relation programs and events, unless consented to by an authorized representative of Tenant in writing after Tenant has received an estimate of Tenant's Share of the cost of such tenant relation program or event);
- (e) costs for which the Landlord is reimbursed, or would have been reimbursed if Landlord had carried the insurance Landlord is required to carry pursuant to this Lease or would have been reimbursed if Landlord had used commercially reasonable efforts to collect such amounts, by any tenant or occupant of the Project or by insurance from its carrier or any tenant's carrier;

- (f) any bad debt loss, rent loss, or reserves for bad debts or rent loss;
- (g) costs associated with the operation of the business of the partnership or entity which constitutes the Landlord, as the same are distinguished from the costs of operation of the Project, including partnership accounting and legal matters, Landlord's general corporate overhead and general corporate administrative expenses (except to the extent those expenses are included in the management fees), costs of defending any lawsuits with any mortgagee (except as the actions of Tenant may be in issue), costs of selling, syndicating, financing, mortgaging or hypothecating any of the Landlord's interest in the Project, and costs incurred in connection with any disputes between Landlord and its employees, between Landlord and Project management, or between Landlord and other tenants or occupants (except as the actions of Tenant may be in issue);
- (h) the wages and benefits of any employee who does not devote substantially all of his or her employed time to the Project unless such wages and benefits are prorated to reflect time spent on operating and managing the Project vis-à-vis time spent on matters unrelated to operating and managing the Project; provided, that in no event shall Operating Expenses for purposes of this Lease include wages and/or benefits attributable to personnel above the level of Project manager or Project engineer;
 - (i) except as set forth in items (xii) and (xiii), above, fines, late charges, penalties, liquidated damages, and interest;
 - (j) amount paid as ground rental or as rental for the Project by the Landlord;
- (k) costs, including permit, license and inspection costs, incurred with respect to the installation of tenant improvements made for new tenants or other occupants in the Project or incurred in renovating or otherwise improving, decorating, painting or redecorating vacant space for tenants or other occupants of the Project (excluding, however, such costs relating to any Common Areas);
 - (l) costs of capital repairs and alterations, capital improvements and equipment except as set forth in items (xii) and (xiii), above;
- (m) any amount paid by Landlord or to the parent organization or a subsidiary or affiliate of the Landlord for supplies and/or services in the Project to the extent the same exceeds the costs of such supplies and/or services rendered by qualified, first-class unaffiliated third parties on a competitive basis;
 - (n) any compensation paid to clerks, attendants or other persons in commercial concessions operated by or on behalf of the Landlord;
- (o) rentals and other related expenses incurred in leasing air conditioning systems, elevators or other equipment which if purchased the cost of which would be excluded from Operating Expenses as a capital cost, except equipment not affixed to the Project which is used in providing janitorial or similar services and, further excepting from this exclusion such equipment rented or leased to remedy or ameliorate an emergency condition in the Project;
- (p) all items and services for which Tenant or any other tenant in the Project is required to reimburse Landlord pursuant to the applicable lease (other than by payment of Direct Expenses), or which Landlord provides selectively to one or more tenants (other than Tenant) without reimbursement;
- (q) electric power costs or costs for natural gas for which any tenant (including Tenant) directly contracts with a public service company, or any costs for electricity, water, heat, air conditioning or other utilities provided by Landlord to any tenant free of charge in excess of the costs for utilities offered by Landlord to Tenant free of charge;
 - (r) costs, other than those incurred in ordinary maintenance and repair, for sculpture, paintings, fountains or other objects of art;
- (s) fees and reimbursements payable to Landlord (including its parent organization, subsidiaries and/or affiliates) or by Landlord for management of the Project which materially exceed the amount which would normally be paid to a company, in connection with the management of the Comparable Buildings, with a general reputation for excellence and integrity used at "arms length" and which is not, directly or indirectly, affiliated with Landlord ("Independent Company Test");
- (t) rent for any office space occupied by Project management personnel to the extent the size or rental rate of such office space exceeds the size or fair market rental value of office space occupied by management personnel of the Comparable Buildings, with adjustment where appropriate for the size of the applicable project;
- (u) all assessments and premiums which are not specifically charged to Tenant because of what Tenant has done, which can be paid by Landlord in installments, shall be paid by Landlord in the maximum number of installments permitted by law (except to the extent inconsistent with the general practice of the Comparable Buildings) and shall be included as Operating Expenses in the year in which the assessment or premium installment is actually paid;

- (v) costs incurred by Landlord in connection with the correction of defects in the initial design and or construction of the Project, or any portion thereof, and any costs arising from the gross negligence or willful misconduct of Landlord or Landlord Parties (as defined in Section 14.1);
- (w) costs arising from the presence, removal, remediation, cleanup or abatement of any Hazardous Material in, on or under the Premises, Building or Project that was not placed in, on or under the Premises, Building or Project by Tenant, Tenant Parties (defined in Section 14.1) or Tenant's contractors, visitors, patrons, guests, invitees, licensees or subtenants;
 - (x) costs arising from Landlord's charitable or political contributions, industry associations or similar organizations;
- (y) any finders fees, brokerage commissions, job placement costs or job advertising cost, other than with respect to a receptionist or secretary in the Project office, once per year;
 - (z) legal fees and costs concerning the negotiation and preparation of any lease for the Project;
- (aa) any rental payments and related costs pursuant to any ground lease of land underlying all or any portion of the Building or Project, and any costs related to the preparation of any reciprocal easement agreement, and/or covenant, condition and restriction agreement;
- (bb) costs incurred because the Building or Project violate any applicable Law (including without limitation the Americans With Disabilities Act, fire-life safety codes and similar codes) in effect on or prior to the Commencement Date, as interpreted by governmental authorities prior to the Commencement Date, including fines, penalties, interest, and the costs of repairs, replacements or improvements necessary to make the Building and Project comply with applicable past Laws in effect and as interpreted by governmental authorities prior to the Commencement Date;
 - (cc) expenses directly resulting from the gross negligence of Landlord or any Landlord Parties; and
 - (dd) assessments, charges and dues under any CC&Rs that would result in a duplicate charge to Tenant.

If Landlord is not furnishing any particular work or service (the cost of which, if performed by Landlord, would be included in Operating Expenses) to a tenant who has undertaken to perform such work or service in lieu of the performance thereof by Landlord, Operating Expenses shall be deemed to be increased by an amount equal to the additional Operating Expenses which would reasonably have been incurred during such period by Landlord if it had at its own expense furnished such work or service to such tenant. If the Building or Project is not at least 100% occupied during all or a portion of the Base Year or any Expense Year or if Landlord is not supplying services to at least 100% of the total rentable square footage of the Building or Project at any time during all or a portion of the Base Year or any Expense Year, that portion of Operating Expenses that is allocated by Landlord solely to the Building shall be determined as if the Building had been 100% occupied and Landlord had been supplying services to 100% of the rentable square footage of the Building during that year, and that portion of Operating Expenses that is allocated by Landlord to the Project (and not solely to the Building) shall be determined as if the Project had been 100% occupied and Landlord had been supplying services to 100% of the rentable square footage of the Project during that year. The extrapolation of Operating Expenses under this Section shall be performed by appropriately adjusting the cost of those components of Operating Expenses that are impacted by changes in the occupancy of the Building or Project. Tenant acknowledges and agrees that since the Project will likely not be fully constructed until after the Base Year, Landlord's Base Year and Expense Year extrapolations of that portion of Operating Expenses allocated to the Project (and not solely the Building) may be equitably modified as the Project is constructed and leased and Landlord obtains more accurate information regarding the actual impacts of increased Project occupancy on such portion of Operating Expenses. Landlord shall operate the Project (or cause the same to be operated) in a commercially reasonable manner using Landlord's good faith business judgment, and Landlord shall not (i) make a profit by charging items to Operating Expenses that are otherwise also charged separately to others and (ii) collect Operating Expenses from Tenant and all other tenants/occupants in the Project in an amount in excess of what Landlord incurred for the items included in Operating Expenses. In no event shall the components of Direct Expenses for any Expense Year related to electrical costs be less than the components of Direct Expenses related to electrical costs in the Base Year, except to the extent the reduction is due to energy savings programs or devices that are adopted by the Landlord after the Base Year. Notwithstanding the foregoing, when calculating Operating Expenses for the Base Year, Operating Expenses shall not include market-wide labor-rate increases due to extraordinary circumstances, including, but not limited to, boycotts and strikes and utility rate increases due to extraordinary circumstances, including, but not limited to, conservation surcharges, boycotts, embargoes or other shortages. If Operating Expenses and/or Real Estate Taxes in any calendar year decrease below the amount of Operating Expenses and/or Real Estate Taxes for the Base Year, Tenant's Pro Rata Share of Operating Expenses and/or Real Estate Taxes, as the case may be, for that calendar year shall be \$0.

5.2.5. Taxes.

5.2.5.1 Subject to Section 5.6, "<u>Tax Expenses</u>" shall mean all federal, state, county, or local governmental or municipal taxes, fees, charges or other impositions of every kind and nature, whether general, special, ordinary or extraordinary, (including, without limitation, real estate taxes, general and special assessments, transit taxes, leasehold taxes or taxes based upon the receipt of rent, including gross receipts or sales taxes applicable to the receipt of rent, unless required to be paid by Tenant, personal property taxes imposed upon the fixtures, machinery, equipment, apparatus, systems and equipment, appurtenances, furniture and other personal property used in connection with the Project, or any portion thereof), which shall be paid or accrued during any Expense Year (without regard to any different fiscal year used by such governmental or municipal authority)

because of or in connection with the ownership, leasing and operation of the Project, or any portion thereof. The parties hereby acknowledge and agree that the Tax Expenses for the Base Year and any Expense Year shall include all Tax Expenses assessed or levied as a result of the construction of the Project, Building or Premises even if assessed or levied after the Base Year or such Expense Year.

5.2.5.2 Subject to Section 5.6, Tax Expenses shall include, without limitation: (i) any tax on the rent, right to rent or other income from the Project, or any portion thereof, or as against the business of leasing the Project, or any portion thereof; (ii) any assessment, tax, fee, levy or charge in addition to, or in substitution, partially or totally, of any assessment, tax, fee, levy or charge previously included within the definition of real property tax, it being acknowledged by Tenant and Landlord that Proposition 13 was adopted by the voters of the State of California in the June 1978 election ("Proposition 13") and that assessments, taxes, fees, levies and charges may be imposed by governmental agencies for such services as fire protection, street, sidewalk and road maintenance, refuse removal and for other governmental services formerly provided without charge to property owners or occupants, and, in further recognition of the decrease in the level and quality of governmental services and amenities as a result of Proposition 13, Tax Expenses shall also include any governmental or private assessments or the Project's contribution towards a governmental or private cost-sharing agreement for the purpose of augmenting or improving the quality of services and amenities normally provided by governmental agencies; (iii) any assessment, tax, fee, levy, or charge allocable to or measured by the area of the Premises or the Rent payable hereunder, including, without limitation, any business or gross income tax or excise tax with respect to the receipt of such rent, or upon or with respect to the possession, leasing, operating, management, maintenance, alteration, repair, use or occupancy by Tenant of the Premises, or any portion thereof; and (iv) any assessment, tax, fee, levy or charge, upon this transaction or any document to which Tenant is a party, creating or transferring an interest or an estate in the Premises.

5.2.5.3 Any costs and expenses (including, without limitation, reasonable attorneys' fees) incurred in attempting to protest, reduce or minimize Tax Expenses shall be included in Tax Expenses in the Expense Year such expenses are paid. Tax refunds shall be credited against Tax Expenses and refunded to Tenant regardless of when received, based on the Expense Year to which the refund is applicable, provided that in no event shall the amount to be refunded to Tenant for any such Expense Year exceed the total amount paid by Tenant as Additional Rent under this Section 5 for such Expense Year. If Tax Expenses for any Expense Year during the Term or any extension thereof are increased after payment thereof, or if Tax Expenses for the Base Year are decreased after payment thereof, in either event for any reason, including, without limitation, error or reassessment by applicable governmental or municipal authorities, Tenant shall pay Landlord upon demand any resulting increase in the amount of Tenant's Share of Direct Expenses for any Expense Year affected by such change.

Notwithstanding anything to the contrary contained in this Section 5.2.5 (except as set forth in Section 5.2.5.1, above), there shall be excluded from Tax Expenses (i) all excess profits taxes, franchise taxes, gift taxes, capital stock taxes, inheritance and succession taxes, estate taxes, federal and state income taxes, and other taxes to the extent applicable to Landlord's general or net income (as opposed to rents, receipts or income attributable to operations at the Project), (ii) any items included as Operating Expenses, (iii) any items paid by Tenant under Section 5.4 of this Lease, and (iv) any Tax Expense that are attributable to a period prior to or after the Term.

5.3 Allocation of Direct Expenses. "Tenant's Share" shall mean the percentage set forth in Section 1.11. Tenant's Share is calculated by dividing the rentable square footage of the Premises by the total rentable square footage of the Building, as set forth in Section 1.3, above. Tenant's Share shall be modified appropriately to reflect any changes in the rentable square footage of the Premises. Notwithstanding the foregoing, the parties acknowledge that the Building is expected to be part of a multi-building project and that the Direct Expenses incurred in connection with the Project as a whole should be shared among the tenants of the Building and the tenants or other occupants of the other Project Buildings as and when the same are completed. Accordingly, as set forth in Section 5.2 above, Direct Expenses (which consists of Operating Expenses and Tax Expenses) shall be determined annually for the Project as a whole, and a portion of the Direct Expenses, which portion shall be determined by Landlord on an equitable basis, shall be allocated to the Building (as to which Tenant's Share shall apply, as lessee of the Premises) and such portion shall be the Direct Expenses for purposes of this Lease. Such portion of Direct Expenses allocated to the Building shall include all Direct Expenses attributable solely to the Building (including, without limitation, all Tax Expenses attributable solely to the Building) and an equitable portion of the Direct Expenses reasonably determined by Landlord to benefit or relate substantially to the entire Project rather than any specific Building (for example, Direct Expenses on the Project Common Areas will generally be considered to benefit or relate substantially to the entire Project rather than any specific Project Building, whereas Direct Expenses attributable solely to any of the other Project Buildings will generally not be considered to benefit or relate substantially to the entire Project). Such equitable portion shall be predicated on the ratio of rentable square feet of the Building to the rentable square feet of all of the substantially completed Project Buildings (including the Building) existing from time to time in the Project, except where material factors require a different equitable treatment. If Landlord reasonably determines that any premises (including the Premises) or any portion thereof within any of the Project Buildings, or any of the Project Buildings, incurs a non-proportional benefit from any Direct Expense, or is the non-proportional cause of any such Direct Expense, Landlord may allocate a greater percentage of such Direct Expenses to such premises or such Project Building(s), as applicable, using Landlord's good faith business judgment.

5.4 <u>Calculation and Payment of Additional Rent.</u> If for any Expense Year ending or commencing within the Term, Tenant's Share of Direct Expenses for such Expense Year exceeds Tenant's Share of Direct Expenses applicable to the Base Year, then Tenant shall pay to Landlord, in the manner set forth in Section 5.4.1, below, and as Additional Rent, an amount equal to the excess (the "<u>Excess</u>").

5.4.1. <u>Statement of Actual Direct Expenses and Payment by Tenant</u>. Landlord shall use commercially reasonable efforts to give to Tenant within one hundred twenty (120) days following the end of the Base Year and each subsequent Expense Year, a statement (the "<u>Statement</u>") itemized on a line-item by line-item basis, which shall state, on a line-item by line-item basis, the Direct Expenses incurred or accrued for the Base Year or such preceding Expense Year, as applicable, and which shall indicate the amount of the Excess. Upon receipt of the Statement for

each Expense Year commencing or ending during the Term, if an Excess is present, Tenant shall pay, with its next installment of Base Rent due, the full amount of the Excess for such Expense Year, less the amounts, if any, paid during such Expense Year as Estimated Excess (as defined in Section 5.4.2), and if Tenant paid more as Estimated Excess than the actual Excess, Tenant shall receive a credit in the amount of Tenant's overpayment against Rent next due under this Lease. The failure of Landlord to timely furnish the Statement for any Expense Year shall not prejudice Landlord or Tenant from enforcing its rights under this Section 5. Even though the Term has expired and Tenant has vacated the Premises, when the final determination is made of Tenant's Share of Direct Expenses for the Expense Year in which this Lease terminates, if an Excess is present, Tenant shall immediately pay to Landlord such amount, and if Tenant paid more as Estimated Excess than the actual Excess, Landlord shall, concurrently with the delivery of the Statement, deliver a check payable to Tenant in the amount of the overpayment. The provisions of this Section 5.4.1 shall survive the expiration or earlier termination of the Term. Notwithstanding the immediately preceding sentence, Tenant shall not be responsible for Tenant's Share of any Direct Expenses attributable to any Expense Year which are first billed to Tenant more than two (2) calendar years after the earlier of the expiration of the applicable Expense Year or the Termination Date, provided that in any event Tenant shall be responsible for Tenant's Share of Direct Expenses levied by any governmental authority or by any public utility companies at any time following the Lease Expiration Date which are attributable to any Expense Year (provided that Landlord delivers Tenant a bill (a "Supplemental Statement") for such amounts within sixty (60) days following Landlord's receipt of the bill therefor).

5.4.2. Statement of Estimated Direct Expenses. In addition, Landlord shall use commercially reasonable efforts to give Tenant, within one hundred twenty (120) days following the end of the prior Expense Year, a yearly expense estimate statement (the "Estimate Statement") itemized on a line-item by line-item basis, which shall set forth Landlord's reasonable estimate (the "Estimate") of what the total amount of Direct Expenses for the then-current Expense Year shall be and the estimated excess (the "Estimated Excess") as calculated by comparing the Direct Expenses for such Expense Year, which shall be based upon the Estimate, to the amount of Direct Expenses for the Base Year. The failure of Landlord to timely furnish the Estimate Statement for any Expense Year shall not preclude Landlord from enforcing its rights to collect any Estimated Excess under this Section 5, nor shall Landlord be prohibited from revising any Estimate Statement or Estimated Excess theretofore delivered to the extent necessary; provided, however, that any such subsequent revision shall set forth on a reasonably specific basis any particular expense increase. Upon receipt of an Estimate Statement, Tenant shall thereafter pay, upon the later to occur of its next installment of Base Rent due or thirty (30) days after receipt of the Estimate Statement, a fraction of the Estimated Excess for the then-current Expense Year (reduced by any amounts paid pursuant to the next to last sentence of this Section 5.4.2). Such fraction shall have as its numerator the number of months that have elapsed in such current Expense Year, including the month of such payment, and twelve (12) as its denominator. Until a new Estimate Statement is furnished (which Landlord shall have the right to deliver to Tenant at any time), Tenant shall pay monthly, with the monthly Base Rent installments, an amount equal to one-twelfth (1/12) of the total Estimated Excess set forth in the previous Estimate Statement delivered by Landlord to Tenant.

5.5 <u>Taxes and Other Charges for Which Tenant Is Directly Responsible.</u>

- 5.5.1 Tenant shall be liable for and shall pay before delinquency, taxes levied against Tenant's equipment, furniture, fixtures and any other personal property located in or about the Premises. If any such taxes on Tenant's equipment, furniture, fixtures and any other personal property are levied against Landlord or Landlord's property or if the assessed value of Landlord's property is increased by the inclusion therein of a value placed upon such equipment, furniture, fixtures or any other personal property and if Landlord pays the taxes based upon such increased assessment, which Landlord shall have the right to do, after notice to Tenant, regardless of the validity thereof but only under proper protest if requested by Tenant, Tenant shall upon demand repay to Landlord the taxes so levied against Landlord or the proportion of such taxes resulting from such increase in the assessment, as the case may be.
- 5.5.2 If the tenant improvements in the Premises, whether installed and/or paid for by Landlord or Tenant and whether or not affixed to the real property so as to become a part thereof, are assessed for real property tax purposes at a valuation higher than the valuation at which tenant improvements conforming to Landlord's "building standard" in other space in the Project are assessed, then the Tax Expenses levied against Landlord or the property by reason of such excess assessed valuation shall be deemed to be taxes levied against personal property of Tenant and shall be governed by the provisions of Section 5.5.1, above.
- 5.5.3 Notwithstanding any contrary provision herein, Tenant shall pay prior to delinquency any (i) rent tax or sales tax, service tax, transfer tax or value added tax, or any other applicable tax on the rent or services herein or otherwise respecting this Lease, (ii) taxes assessed upon or with respect to the possession, leasing, operation, management, maintenance, alteration, repair, use or occupancy by Tenant of the Premises or any portion of the Project, including the Project Parking Facility; or (iii) taxes assessed upon this transaction or any document to which Tenant is a party creating or transferring an interest or an estate in the Premises.
- 5.6 <u>Tenant's Payment of Certain Tax Expenses.</u> Notwithstanding anything to the contrary contained in this Lease, in the event that, at any time during the initial Term, any sale or change in ownership of the Building is consummated, and as a result thereof, and to the extent that in connection therewith, the Building is reassessed (the "<u>Reassessment</u>") for real estate tax purposes by the appropriate governmental authority pursuant to the terms of Proposition 13 (as adopted by the voters of the State of California in the June, 1978 election, or any successor statute), then the terms of this Section 5.6 shall apply.
- 5.6.1. The Tax Increase. For purposes of this Section 5.6, the term "Tax Increase" shall mean that portion of the Tax Expenses, as calculated immediately following the Reassessment, which is attributable solely to the Reassessment. Accordingly, the term Tax Increase shall not include any portion of the Tax Expenses, as calculated immediately following the Reassessment, which (i) is attributable to the initial assessment of the value of the Building, the Base Building or the tenant improvements located in the Building, (ii) is attributable to assessments which were pending immediately prior to the Reassessment, which assessments were conducted during, and

included in, such Reassessment, or which assessments were otherwise rendered unnecessary following the Reassessment, (iii) is attributable to the annual inflationary increase of real estate taxes (as such increases are determined by statute from time to time), or (iv) is attributable to any Tax Expenses incurred during the Base Year.

5.6.2. <u>Protection</u>. With respect to a Reassessment during the initial Term resulting from a change in ownership of the Building, the Tax Increase attributable to such Reassessment shall be excluded from Tax Expenses during the initial Term.

5.6.3. Landlord's Right to Purchase the Proposition 13 Protection Amount Attributable to a Particular Reassessment. The amount of Tax Expenses which Tenant is not obligated to pay or will not be obligated to pay during the initial Term in connection with a particular Reassessment pursuant to the terms of this Section 5.6, shall be sometimes referred to hereafter as a "Proposition 13 Protection Amount." If the occurrence of a Reassessment is reasonably foreseeable by Landlord and the Proposition 13 Protection Amount attributable to such Reassessment can be reasonably quantified or estimated for each Lease Year commencing with the Lease Year in which the Reassessment will occur, the terms of this Section 5.6.3 shall apply to such Reassessment. Upon notice to Tenant, Landlord shall have the right to purchase the Proposition 13 Protection Amount relating to the applicable Reassessment (the "Applicable Reassessment"), by paying to Tenant an amount equal to the "Proposition 13 Purchase Price," as that term is defined below, provided that the right of any successor of Landlord to exercise its right of repurchase hereunder shall not apply to any Reassessment which results from the event pursuant to which such successor of Landlord became the Landlord under this Lease. As used herein, "Proposition 13 Purchase Price" shall mean the present value of the Proposition 13 Protection Amount remaining during the initial Term, as of the date of payment of the Proposition 13 Purchase Price by Landlord. Such present value shall be calculated (i) by using the portion of the Proposition 13 Protection Amount attributable to each remaining Lease Year (as though the portion of such Proposition 13 Protection Amount benefited Tenant at the end of each Lease Year), as the amounts to be discounted, and (ii) by using discount rates for each amount to be discounted equal to the annual "prime rate" quoted by the Los Angeles branch of Bank of America, N.A. (or such other comparable lender as Landlord and Tenant shall reasonably agree upon if such rate ceases to be quoted) as of the date of payment of the Proposition 13 Protection Amount by Landlord. Upon such payment of the Proposition 13 Purchase Price, the provisions of Section 5.6.2 shall not apply to any Tax Increase attributable to the Reassessment. Since Landlord would be estimating the Proposition 13 Purchase Price because a Reassessment has not yet occurred, then when such Reassessment occurs, if Landlord has underestimated the Proposition 13 Purchase Price, upon notice by Landlord to Tenant, Tenant's Rent next due shall be credited with the amount of such underestimation, and if Landlord overestimates the Proposition 13 Purchase Price, then upon notice by Landlord to Tenant, Rent next due shall be increased by the amount of the overestimation.

5.7 Landlord's Books and Records.

5.7.1. In General. If Tenant disputes the amount of Additional Rent set forth in any annual Statement or Supplemental Statement delivered by Landlord (including the allocation of Direct Expenses as described in Section 5.3), then subject to the provisions of Section 5.7.2, below, Tenant shall have the right to provide notice to Landlord that it intends to cause an independent, certified public accountant or other qualified professional acceptable to Landlord in Landlord's reasonable discretion (a "Third Party Auditor") to inspect Landlord's accounting records for the Base Year and/or Expense Year covered by such Statement or Supplemental Statement during normal business hours (the "Tenant Review"); provided, however, that, as a condition precedent to any such inspection, Tenant shall deliver to Landlord a copy of Tenant's written agreement with such Third Party Auditor, which agreement shall include provisions which state that (i) such Third Party Auditor will not in any manner solicit or agree to represent any other tenant of the Project with respect to an audit or other review of Landlord's accounting records at the Project, and (ii) such Third Party Auditor shall maintain in strict confidence any and all information obtained in connection with the Tenant Review and shall not disclose such information to any person or entity other than to the legal representatives and management personnel of Tenant. Any Tenant Review shall take place in Landlord's office at the Project or at such other location in Orange County, California as Landlord may reasonably designate, and Landlord will provide Tenant with reasonable accommodations for such Tenant Review and reasonable use of such available office equipment, but may charge Tenant for telephone calls and photocopies at Landlord's actual cost. Tenant shall provide Landlord with not less than two (2) weeks' prior written notice of its desire to conduct such Tenant Review. In connection with the foregoing review, Landlord shall furnish Tenant with such reasonable supporting documentation relating to the subject Statement as Tenant may reasonably request. In no event shall Tenant have the right to conduct such Tenant Review if Tenant is then in Default under this Lease with respect to any of Tenant's monetary obligations, including, without limitation, the payment by Tenant of all Additional Rent amounts described in the Statement which is the subject of Tenant's Review, which payment, at Tenant's election, may be made under dispute. In the event that following Tenant's Review, Tenant and Landlord continue to dispute the amounts of Additional Rent shown on Landlord's Statement or Supplemental Statement and Landlord and Tenant are unable to resolve such dispute, then either Landlord or Tenant may submit the matter to arbitration pursuant to Section 27.28 and the proper amount of the disputed items and/or categories of Direct Expenses to be shown on such Statement shall be determined by such proceeding producing an Arbitration Award (as defined in Section 27.28). The Arbitration Award shall be conclusive and binding upon both Landlord and Tenant. If the resolution of the parties' dispute with regard to the Additional Rent shown on the Statement, pursuant to the Arbitration Award reveals an error in the calculation of Tenant's Share of Direct Expenses to be paid for such Expense Year, the parties' sole remedy shall be for the parties to make appropriate payments or reimbursements, as the case may be, to each other as are determined to be owing. Any such payments shall be made within thirty (30) days following the resolution of such dispute. At Tenant's election, Tenant may treat any overpayments resulting from the foregoing resolution of such parties' dispute as a credit against Rent until such amounts are otherwise paid by Landlord. Tenant shall be responsible for all costs and expenses associated with Tenant's Review, and Tenant shall be responsible for all reasonable audit fees, attorney's fees and related costs of Tenant relating to an Arbitration Award (collectively, the "Costs"), provided that if the parties' final resolution of the dispute involves the overstatement by Landlord of Direct Expenses for such Expense Year in excess of five percent (5%), then Landlord shall be responsible for the Costs. Subject to the terms of Section 5.7.2, below, this provision shall survive the termination of this Lease to allow the parties to enforce their respective rights hereunder.

5.7.2. <u>Termination of Rights</u>. If, within two (2) years following receipt of any particular Statement or Supplemental Statement, as applicable, Tenant or Landlord shall fail to either (i) fully and finally settle any dispute with respect to such Statement or Supplemental Statement, as applicable, or (ii) submit the dispute to arbitration in accordance with the terms of Section 5.7.1, above, then Tenant shall have no further right to conduct a Tenant Review or to dispute the amount of Additional Rent set forth in the applicable Statement or Supplemental Statement, as applicable.

6. Security Deposit; Letter of Credit. The Security Deposit shall be delivered to Landlord upon the execution of this Lease by Tenant and shall be held by Landlord without liability for interest (unless required by Law) as security for the performance of Tenant's obligations. The Security Deposit is not an advance payment of Rent or a measure of Tenant's liability for damages. Landlord may, from time to time, without prejudice to any other remedy, use all or a portion of the Security Deposit to satisfy past due Rent or to cure any uncured default by Tenant. If Tenant leases additional space pursuant to any of the provisions of Section 3, the Security Deposit for the Other Lease covering such additional space, unless the terms of the applicable offer notice provide otherwise, shall be an amount equal to the monthly Base Rent applicable to such additional space during the last full calendar month of the initial term for such additional space. If Landlord uses the Security Deposit, Tenant shall on demand restore the Security Deposit to its original amount (as the same may be increased pursuant to this Lease). Landlord shall return any unapplied portion of the Security Deposit to Tenant within 30 days after the later to occur of: (1) the date Tenant surrenders possession of the Premises to Landlord in accordance with this Lease, or (2) the Termination Date. If Landlord transfers its interest in the Premises, Landlord may assign the Security Deposit to the transferee and, following the assignment, Landlord shall have no further liability for the return of the Security Deposit. Landlord shall not be required to keep the Security Deposit separate from its other accounts. Tenant hereby waives the provisions of Section 1950.7 of the California Civil Code, or any similar or successor Laws now or hereinafter in effect which (i) establish the time frame by which Landlord must refund collateral or security for performance of a tenant's obligations under a lease, and/or (ii) provide that Landlord may claim from collateral or security for performance of a tenant's obligations under a lease only those sums reasonably necessary to remedy defaults in the payment of rent, to repair damage caused by Tenant or to clean the Premises. In that regard, Tenant agrees that Landlord may, in addition, claim those sums specified hereinabove and/or those sums reasonably necessary to compensate Landlord for any loss or damage caused by Tenant's breach of this Lease or the acts or omission of Tenant or any employee, agent, contractor or invitee of Tenant. Landlord agrees that in lieu of a cash Security Deposit as required in this Section 6, Tenant may deliver to Landlord within ten (10) Business Days after execution of this Lease a letter of credit in the amount stated in Section 1.14, which letter of credit shall be in form and with the substance of Exhibit M attached hereto or such other form as may be reasonably acceptable to Landlord. The letter of credit shall be issued by a financial institution reasonably acceptable to Landlord with a branch in Orange County, California, at which draws on the letter of credit will be accepted. The letter of credit shall provide for automatic yearly renewals throughout the Term of this Lease and shall have an outside expiration date (if any) that is not earlier than thirty (30) days after the expiration of the Term, as the same may be extended pursuant to Section 3.4. The letter of credit shall be held by Landlord as security for the faithful performance by Tenant of all the terms, covenants, and conditions of this Lease to be kept and performed by Tenant during the Term. The letter of credit shall not be mortgaged, assigned or encumbered in any manner whatsoever by Tenant without the prior written consent of Landlord. If the letter of credit is not continuously renewed through the period set forth above, or upon any breach under this Lease by Tenant (and after the expiration of any applicable notice and cure period unless Landlord is precluded by law from delivering a default notice), Landlord shall be entitled to draw upon said letter of credit by the issuance of Landlord's sole written demand to the issuing financial institution. Any such draw shall be without waiver of any rights Landlord may have under this Lease or at law or in equity as a result of any breach hereunder by Tenant, and shall not prevent Landlord from exercising any other right or remedy provided by this Lease or by law, it being intended that Landlord shall not first be required to proceed against the letter of credit and that the letter of credit shall not operate as a limitation on any recovery to which Landlord may otherwise be entitled. Any amount of the letter of credit that is drawn upon by Landlord, but is not used or applied by Landlord, shall be held by Landlord and deemed a security deposit (the "Letter of Credit Security Deposit"). If the letter of credit is drawn upon and used or applied by Landlord, Tenant shall, within five (5) Business Days after written demand therefor, either (i) deposit cash with Landlord (which cash shall be applied by Landlord to the Letter of Credit Security Deposit) in an amount sufficient to cause the sum of the Letter of Credit Security Deposit and the amount of the remaining letter of credit, if any, to be equivalent to the amount of the letter of credit required under this Lease or (ii) reinstate the letter of credit to the amount required under this Lease, and Tenant's failure to do so shall be a Default under this Lease. If Tenant, at Tenant's option, elects to reinstate the letter of credit to the amount then required under this Lease, Landlord shall return the remaining balance, if any, of the Letter of Credit Security Deposit to Tenant within thirty (30) days following Landlord's receipt of such reinstated letter of credit. Tenant acknowledges that Landlord has the right to transfer or mortgage its interest in the Building or Project and in this Lease and Tenant agrees that in the event of any such transfer or mortgage, Landlord shall have the right to transfer or assign the Letter of Credit Security Deposit and/or the letter of credit to the transferee or mortgagee; in such event, Landlord shall be responsible for the first \$2,000 of any Letter of Credit transfer fee and Tenant shall be responsible for any Letter of Credit transfer fee in excess of \$2,000.

7. Services to be Furnished by Landlord.

7.1 <u>Utilities and Services</u>. Landlord agrees to furnish Tenant with the following utilities and services:

7.1.1 Subject to limitations imposed by all governmental rules, regulations and guidelines applicable thereto, Landlord shall provide heating and air conditioning ("HVAC") when necessary for normal comfort for normal office use in the Premises during Normal Business Hours, except for Holidays. If Tenant desires to use HVAC during hours other than those for which Landlord is obligated to supply same pursuant to the terms of this Section 7.1.1, Landlord shall supply such "after hours" HVAC to Tenant at a cost which will, as of the date of this

Lease, be equal to \$50.00 per floor (or partial floor) per hour, and which cost may be increased only to the extent that Landlord's Actual Cost (hereinafter defined) of providing such "after hours" HVAC increases from time to time. The minimum time period for after hours HVAC usage shall be one (1) hour. For purpose of this Lease, "Actual Cost" shall mean the actual cost incurred by Landlord, as reasonably determined by Landlord, but without charge for depreciation, profit, overhead or administration, provided that, notwithstanding the foregoing, any amount actually charged by any unrelated third party to Landlord for the supply of HVAC shall be deemed Landlord's "Actual Cost". When determining the Actual Cost of Tenant's utility usage pursuant to the terms of this Article 7, Landlord agrees that it shall use the monthly average rate paid by Landlord for a particular utility.

- 7.1.2 Subject to Section 7.2, Landlord shall provide adequate electrical wiring and facilities for connection to Tenant's lighting fixtures and incidental use equipment sufficient to accommodate an electrical connected load of eight (8) watts per rentable square foot and an electrical demand load of an average of eight (8) watts per rentable square foot of the Premises and an average lighting load of one and one-half (1.5) watts per rentable square foot of the Premises, with such averages determined on a monthly basis per Normal Building Hours (and not including the electricity required to run the Building HVAC system) (the "Standard Electrical Usage"), which electrical usage shall be subject to applicable laws and regulations, including Title 24. Tenant will design Tenant's electrical system serving any equipment producing nonlinear electrical loads to accommodate such nonlinear electrical loads, including, but not limited to, oversizing neutral conductors, derating transformers and/or providing power-line filters. Engineering plans shall include a calculation of Tenant's fully connected electrical design load with and without demand factors and shall indicate the number of watts of unmetered and submetered loads. Electrical service to the Premises may be furnished by one or more companies providing electrical generation, transmission and distribution services, and the cost of electricity may consist of several different components or separate charges for such services, such as generation, distribution and stranded cost charges. Landlord shall have the exclusive right to select any company providing electrical service to the Premises, to aggregate the electrical service for the Premises with other buildings, to purchase electricity through a broker and/or buyers group and to change the providers and manner of purchasing electricity. Landlord shall be entitled to receive a fee (if permitted by Law) for the selection of utility companies and the negotiation and administration of contracts for electricity, provided that the amount of such fee shall not exceed 50% of any savings obtained by Landlord. Tenant shall bear the cost of replacement of lamps, starters and ballasts for non-Building standard lighting fixtures within the Premises. Landlord, as part of Operating Expenses, shall bear the cost of lamps, starters and ballasts for Building standard lighting fixtures within the Premises.
 - 7.1.3 Landlord shall provide city water from the regular Building outlets for drinking, lavatory and toilet purposes.
- 7.1.4 Landlord shall provide janitorial services to the Premises, in material conformance with the specifications attached hereto as **Exhibit E**, except the date of observation of the Holidays, in and about the Premises and window washing services in a manner consistent with other Comparable Buildings. If Tenant's use, floor covering or other improvements require special services in excess of the standard services for the Building, Tenant shall pay the additional cost attributable to the special services.
- 7.1.5 Landlord shall provide nonexclusive, non-attended automatic passenger elevator service during Normal Business Hours, and, subject to closures for routine maintenance or repair, shall have at least one elevator available at all other times, to provide service to the Premises; provided further, however, except to the extent prevented by Force Majeure Events (as defined in Section 27.4), Tenant shall always have access to one (1) elevator servicing the Premises; provided, however, that Landlord shall not reduce the number of elevators existing in the Building as of the Commencement Date.
- 7.1.6 Landlord shall provide nonexclusive freight elevator service at all times, subject to scheduling by Landlord, and free of direct charge to Tenant. Landlord shall use commercially reasonable efforts to provide Tenant, on occasion, with exclusive freight elevator service as reasonably necessary, subject to availability and reasonable scheduling by Landlord.
- 7.1.7 Landlord shall provide reasonable access control for the Building and the Parking Facilities seven (7) days per week, twenty-four (24) hours per day, in a manner consistent with the specifications therefor provided in the approved Base Building Plans as described in the Work Letter.
 - 7.18 Landlord shall provide maintenance and repair of the Building and the Project as described in Section 9.2.
- 7.19 Landlord shall furnish periodic exterior window washing and pest control in a manner (and during times) consistent with the practices of Comparable Buildings.
 - 7.20 Landlord shall provide such other services as Landlord reasonably determines are necessary or appropriate for the Project.

Tenant shall cooperate fully with Landlord at all times and abide by all regulations and requirements that Landlord may reasonably prescribe for the proper functioning and protection of the HVAC, electrical, mechanical and plumbing systems that do not interfere with Tenant's business operations. Subject to Laws, as that term is defined in Article 24, below, Landlord shall not prohibit Tenant Parties or any of Tenant's visitors, patrons, guests, invitees, licensees or subtenants from smoking in the designated smoking areas of the Project.

7.2 Excess Usage; Tenant's Computer Room. The Building shall be designed to accommodate the Standard Electrical Usage. Electricity used by Tenant in the Premises shall be included in Operating Expenses as set forth in Section 5.1 above (except as provided in this Section 7.2 for excess usage). Tenant's use of electrical service shall not exceed, either in voltage, rated capacity, use beyond the Standard Electrical Usage. If Tenant requests permission to consume reasonable amounts of excess electrical service, Landlord may condition its consent upon

conditions that Landlord reasonably elects (including, without limitation, the installation of utility service upgrades, meters, submeters, air handlers or cooling units, all at Tenant's sole cost and expense), and the additional usage, installation and maintenance costs shall be paid by Tenant. In such case, Landlord shall have the right to measure electrical usage by survey or other commonly accepted methods. Tenant shall not, without Landlord's prior written consent, use heat-generating machines, machines other than normal office machines, or equipment or lighting other than Building standard lights in the Premises, to the extent same may materially adversely affect the temperature otherwise maintained by the HVAC system or materially increase the water (unless Tenant agrees to pay for such excess water) normally furnished for the Premises by Landlord pursuant to the terms of Section 7.1 of this Lease. If such consent is given, Landlord shall have the right to install supplementary air conditioning units or other facilities in the Premises, including supplementary or additional metering devices, and the Actual Cost, including the cost of installation, operation and maintenance, and other similar charges, shall be paid by Tenant to Landlord upon billing by Landlord.

Landlord and Tenant acknowledges and agree that Tenant's computer room (including any supplemental air conditioning and equipment) will be designed and constructed as part of the Tenant Improvements to be separately metered and billed for electricity based on actual usage (the "Computer Room Electricity"). Tenant shall pay the cost of installation, metering and use of the Computer Room Electricity, with usage charges paid directly to the appropriate supplier (unless Landlord pays the supplier, in which case Tenant shall reimburse Landlord), within thirty (30) days of delivery to Tenant of the statement or invoice therefor.

- 7.3 Interruption of Use. Tenant agrees that Landlord shall not be liable for damages, by abatement of Rent (except as provided in Section 17.6.2) or otherwise, for failure to furnish or delay in furnishing any service (including telephone and telecommunication services), or for any diminution in the quality or quantity thereof, when such failure or delay or diminution is occasioned, in whole or in part, by breakage, repairs, replacements, or improvements, by any strike, lockout or other labor trouble, by inability to secure electricity, gas, water, or other fuel at the Building or Project after reasonable effort to do so, by any riot or other dangerous condition, emergency, accident or casualty whatsoever, by act or Default of Tenant or other parties, or by any other cause, including without limitation any utility service provider initiated "brown-out" or "black-out"; and such failures or delays or diminution shall never be deemed to constitute an eviction or disturbance of Tenant's use and possession of the Premises or relieve Tenant from paying Rent (except as provided in Section 17.6.2) or performing any of its obligations under this Lease. Furthermore, Landlord shall not be liable under any circumstances for a loss of, or injury to, property or for injury to, or interference with, Tenant's business, including, without limitation, loss of profits, however occurring, through or in connection with or incidental to a failure to furnish any of the services or utilities as set forth in this Section 7. Landlord may comply with voluntary controls or guidelines promulgated by any governmental entity relating to the use or conservation of energy, water, gas, light or electricity or the reduction of automobile or other emissions ("Voluntary Compliance") without creating any liability of Landlord to Tenant under this Lease, provided that the Premises are not thereby rendered untenantable and/or not fit for its Permitted Use by Tenant, and provided, further, that such Voluntary Compliance shall not increase Tenant's mon
- 7.4 Tenant HVAC System. Subject to the provisions of Section 7.6, Tenant, at its sole expense, may install a supplemental HVAC system in the Premises, for the purpose of servicing the Premises during Building Hours or other than Building Hours (the "Tenant HVAC System"). If required for such purpose, Tenant may connect into the Building's chilled water system, if and to the extent that (i) Tenant's use of chilled water pursuant to this Section 7.4 will not materially adversely affect the chilled water system or the use thereof by other tenants of the Project, and (ii) such connection is otherwise approved by Landlord, which approval shall not be unreasonably withheld, conditioned or delayed unless a Design Problem exists. If Tenant connects into the Building's chilled water system pursuant to the terms of the foregoing sentence, (x) Landlord may install, at Tenant's expense, a meter to measure Tenant's use of chilled water, and (y) Tenant shall reimburse Landlord for Tenant's use of chilled water at Landlord's Actual Cost therefor. Tenant shall be permitted, at Tenant's sole cost and expense, to access 277/480 volts of electricity from the bus duct riser in connection with the Tenant HVAC System. In connection with the foregoing (a) Landlord shall, at Tenant's sole cost and expense, separately meter the electricity utilized by the Tenant HVAC System, and (b) Tenant shall be responsible for the cost of all electricity utilized by the Tenant HVAC System. Landlord hereby agrees that, at Tenant's sole option, Tenant shall be permitted to remove the Tenant HVAC System, and repair any damages to the Building caused by such removal, or leave same in the Premises, in which event the same shall become a part of the realty and belong to Landlord and shall be surrendered with the Premises upon the expiration or earlier termination of this Lease. Tenant's plans and specifications regarding utilities and services, as contained in the "Tenant Improvement Plans," as that term is defined in the Work Letter, and as approved by
- 7.5 <u>Emergency Generators.</u> Subject to the terms of this Section 7.5, Landlord hereby grants Tenant the right to install, at Tenant's sole cost and expense, up to two (2) emergency generators (including day fuel tank, and primary fuel tank) at the Project, in an area reasonably designated by Landlord, and supplemental fuel lines and related connections in the Common Areas of the Building (collectively, the "<u>Generator</u>").
- 7.5.1. <u>Auxiliary Space</u>. The areas within the Building and Project occupied by the Generator, as well as the conduits used by Tenant in connection therewith, are referred to herein as the "<u>Auxiliary Space</u>". The precise amount and location of the Auxiliary Space shall be reasonably designated by Landlord. The Auxiliary Space and the Generator, shall be deemed to be a part of the Premises for purposes of the indemnification and insurance provisions of this Lease; provided, however, Tenant shall not be charged Base Rent or Tenant's Share of Direct Expenses for such Auxiliary Space.
- 7.5.2. <u>Manner of Installation</u>. The installation of the Generator shall be subject to Landlord's prior approval of all plans and specifications, and shall be at Tenant's sole cost and expense (including without limitation, the cost of any structural reinforcements or other work necessary to provide the required floor loads). Any and all such installations shall be made in compliance with all applicable law, insurance policies, and any roof warranty or

other equipment warranties maintained for the Building or Project. Tenant shall submit the specifications for design, operation, installation and maintenance of the Generator and facilities related thereto to Landlord for Landlord's consent, which consent may be conditioned on Tenant complying with such reasonable requirements imposed by Landlord, based on the advice of Landlord's engineers and consultants, so that the Building's systems and equipment are not adversely affected by the installation and operation of the Generator. The cost of design (including engineering costs and review by Landlord's engineers and consultants) and installation of the Generator and the costs of the Generator itself shall be Tenant's sole responsibility.

- 7.5.3. Repair and Maintenance of Generator. All repairs and maintenance of the Generator shall be the sole responsibility of Tenant. Landlord shall not have any obligations with respect to the Generator, nor shall Landlord be responsible for any damage that may be caused to Tenant or the Generator by any other tenant in the Project. Notwithstanding the foregoing, at Landlord's option, Landlord may require that Tenant remove the Generator and all related facilities upon the expiration or earlier termination of the Term and repair all damage to the Building or Project resulting from the installation or removal of the Generator, at Tenant's sole cost and expense. Notwithstanding the use of the Auxiliary Space by Tenant, Tenant acknowledges that Landlord retains the right to use the roof and internal conduits of the Building for any purpose whatsoever. Tenant agrees that Landlord shall have the right, during the Term, after notice and subject to reasonable mutual scheduling, to relocate the Generator to a reasonably alternative location at Landlord's sole cost. Tenant shall maintain, at Tenant's cost, industry standard "boiler and machinery" insurance coverage with respect to the Generator and related facilities. In no event shall a delay in the installation, testing or operation of the Generator be deemed to be a delay in the Lease Commencement Date.
- 7.5.4. <u>Generator Use</u>. The Generator shall be used by Tenant only during (i) testing and regular maintenance (which testing shall be done on Sundays only), and (ii) any period of electrical power outage or power reduction in the Building. Tenant shall be entitled to operate the Generator for testing and regular maintenance only upon notice to Landlord and at times reasonably approved by Landlord and scheduled in advance through the Project office.

7.6 LEED Requirements.

- 7.6.1. <u>LEED Standards</u>. Tenant acknowledges that the Building is currently contemplated to be constructed and, at Landlord's option, shall be constructed and operated in accordance with the U.S. Green Building Council's Leadership in Energy and Environmental Design Program's standards, as the same are amended, supplemented or replaced from time to time, and, at Landlord's option, any similar standards ("<u>LEED</u>"). Landlord currently intends that the Building, upon completion, will be certified as a "certified" (as opposed to a LEED standard higher than "certified") level structure and will be operated in an environmentally responsible and energy-efficient manner.
- 7.6.2. Compliance with LEED Conditions. If Landlord builds and operates a "certified" level LEED building, Tenant shall cooperate with Landlord in the design and construction of the Tenant Improvements (and any Alterations), including agreeing to use environmentally friendly materials in the Tenant Improvements that are reasonably comparable to the materials for which they are substituted and with such elements and in such a manner as will assist Landlord in obtaining points toward such certification (the "LEED Conditions"), and Tenant shall not seek approval of any Tenant Improvements or construct any Alterations or operate in the Premises in a way that would result in the loss of such certification or the deduction of any certification points. Notwithstanding the foregoing, Tenant's cooperation shall in no event require Tenant to incur costs that, on an overall, net basis, would be substantially greater than the costs that Tenant would have incurred but for such cooperation. Moreover, Tenant shall have no obligation to comply with changes to the LEED requirements made after the Effective Date if such changes would materially adversely increase Tenant's obligations or decrease its rights under this Lease or materially adversely impact its business operations.
- 7.6.3. Reimbursement. If Tenant fails to observe, perform or otherwise comply with Section 7.6.2 above, within thirty (30) days after written notice from Landlord (unless a shorter period is necessary in order for Landlord to comply with the LEED Conditions), Landlord may enter the Premises to perform such act or replace such materials as may be reasonably necessary to keep the Building in compliance with the LEED Conditions. Tenant shall pay Landlord, as Additional Rent, all costs (including reasonable attorneys' fees) incurred by Landlord in connection with any action taken by Landlord to remedy Tenant's noncompliance with Section 7.6.2, together with interest thereon at the Interest Rate, within thirty (30) days after Landlord's demand therefor.

8. Use; Occupancy Level.

- 8.1 <u>Permitted Use</u>. Tenant shall use the Premises solely for the Permitted Use set forth in Section 1.15 and Tenant shall not use or permit the Premises or the Project to be used for any other purpose or purposes whatsoever without the prior written consent of Landlord, which shall not be unreasonably withheld, conditioned or delayed (provided such use is not a prohibited use as described in Section 8.2), with consideration given to the first-class and office use nature of the Project. Landlord agrees that any dispute with respect to the Permitted Use may be submitted to arbitration in accordance with the terms of Section 27.28 below.
- 8.2 <u>Prohibited Uses</u>. The uses prohibited under this Lease shall include, without limitation, use of the Premises or a portion thereof for (i) offices of any agency or bureau of the United States or any state or political subdivision thereof; (ii) offices or agencies of any foreign governmental or political subdivision thereof; (iii) offices of any health care professionals or service organization; (iv) schools or other training facilities which are not ancillary to corporate, executive or professional office use; or (v) retail or restaurant uses; provided, however, that nothing in this Section 8.2 shall prohibit or restrict Tenant from conducting inhouse food service and retail service from the Premises solely for the benefit of Tenant and Tenant's employees. Tenant further covenants and agrees that Tenant shall not use, or suffer or permit any person or persons to use, the Premises or any part thereof for any use in

violation of Laws. Tenant shall not do or permit anything to be done in or about the Premises that will unreasonably interfere with the rights of other tenants or occupants of the Project, or use or allow the Premises to be used for any unlawful purpose, nor shall Tenant cause, maintain or permit any nuisance in, on or about the Premises. Tenant shall comply with the Rules and Regulations attached as Exhibit B and such other reasonable rules and regulations adopted by Landlord from time to time (the "Rules and Regulations") and with all recorded covenants, conditions and restrictions now or hereafter affecting the Building or the Project (collectively, the "CC&Rs"); provided, however, that Tenant shall not be bound by any CC&Rs placed against the Building or Project after the date hereof to the extent the same materially adversely increase Tenant's monetary or non-monetary obligations with respect to the Premises, the Project or the Parking Facilities or materially adversely interfere with Tenant's use of, or ingress to or egress from, the Premises, the Project or the Parking Facilities. Without limitation on the foregoing, Tenant acknowledges that the CC&R's may provide for some or all of the Project Common Areas to be transferred to a property owner's association which will assume the obligation to cause to be operated and maintained some or all of the Project Common Areas, including, without limitation the landscaping, driveways, walkways and Parking Facilities (typically, through a property management/maintenance company retained by the property owners' association in respect of such obligations). All costs and expenses required to be paid by the owner or occupant of the Building under the CC&R's for any work performed during the Term shall be included in Operating Expenses for purposes of this Lease. Tenant shall also cause its agents, contractors, subcontractors, employees, customers, and subtenants to comply with all rules and regulations and CC&Rs. Landlord shall not knowingly discriminate against Tenant in Landlord's enforcement of the rules and regulations. Landlord shall have the right to require Tenant to execute and acknowledge, within 15 days of request by Landlord, a "Recognition of Covenants, Conditions and Restrictions" in a form substantially similar to the Form attached hereto as Exhibit G, agreeing to and acknowledging the CC&Rs. Tenant, within 10 days after receipt, shall provide Landlord with copies of any notices it receives regarding a violation or alleged violation of any Laws. Without limitation on the foregoing, Tenant acknowledges and agrees that (i) Landlord has granted to another tenant in the Project the exclusive right to operate a health and fitness center offering fitness programs, recreational facilities and other services (which may include, but are not limited to, a jogging track, racquetball courts, other racquet sports, basketball courts, gymnasiums, jacuzzi, whirlpools, swimming pool, saunas, steam rooms, aerobics and/or floor exercise, strength training, nutritional counseling, personal training, cardio fitness training, free weights, exercise machinery and equipment, martial arts, physical therapy and rehabilitative services, sports medicine, weight loss advising, nutritional counseling, and related programs, therapeutic massage and related spa-type services, treatments and bodywork therapies, chiropractic care, swim lessons, racquetball lessons, tanning salon, babysitting services for members (while the member is utilizing the facility), and vitamin and nutritional supplement sales, facilities and services commonly found in a high quality day spa or wellness salon (including, without limitation, skin treatments, facials, and other specialty beauty treatments, as well as the retail sale of products for body and skin care) in the Project, (ii) Landlord intends to lease space within the Project to a full service restaurant and that Landlord may grant such tenant exclusive rights in connection with such lease, and (iii) Tenant shall not use the Premises or the Project in any way that would violate any such exclusives; provided, however, that the foregoing exclusives shall not prohibit Tenant from operating an ancillary food and beverage service for its employees as permitted by this Lease.

8.3 Occupancy Level. Landlord and Tenant acknowledge and agree that the Premises shall be designed for a maximum occupancy level of five (5) persons per one thousand (1,000) rentable square feet of space comprising the Premises (the "Occupancy Threshold"). Tenant therefore covenants and agrees that the number of persons occupying the Premises shall not exceed the Occupancy Threshold at any given time during any day of the week. Tenant acknowledges further that Tenant's parking privileges hereunder are expressly limited to the parking spaces specified in Section 24 and Tenant shall limit the occupancy of the Building so as to not utilize more than the number of parking spaces permitted in the Parking Agreement, unless Tenant implements measures reasonably satisfactory to Landlord (e.g., securing additional offsite parking, carpooling or rotational shifts) to ensure that Tenant does not utilize more than its allotted parking spaces under the Parking Agreement. If Landlord at any time determines (in its reasonable judgment) that the Occupancy Threshold is being exceeded on a day-to-day basis or that the parking spaces are being exceeded on a day-to-day basis, Landlord shall notify Tenant and Tenant shall have 10 Business Days to reduce the occupancy level to below the Occupancy Threshold (as respects occupancy in the Premises) or reduce the number of parking spaces being used (as respects parking usage). If the Premises occupancy level remains in excess of the Occupancy Threshold (as respects occupancy) or remains in excess of the permitted parking spaces as of the 10th Business Day following the date of Landlord's notice, Tenant shall automatically be deemed in default of this Lease and Landlord shall be entitled to exercise any and all remedies set forth in this Lease or at law or in equity by reason of such default. Tenant shall be solely responsible for the fees, costs and expenses for any and all improvements, alterations or additions required to be made by any governmental authority or other third party to the Premises, the Building, the Common Areas and/or the Project directly resulting from any use of the Premises by Tenant in excess of the Occupancy Threshold regardless of when imposed by any such party and regardless of whether or not the improvement, alteration or addition is of a capital nature (hereinafter, an "Additional Occupancy Cost"). Tenant shall reimburse Landlord within 20 days of written demand for any Additional Occupancy Cost.

9. Repairs and Alterations.

9.1 <u>Tenant's Repair Obligations</u>. Tenant shall, at its sole cost and expense, promptly perform all maintenance and repairs to the Premises that are not Landlord's express responsibility under this Lease, and shall keep the Premises, including all improvements, fixtures and furnishings therein, and the floor or floors of the Building on which the Premises are located, in good condition and repair, reasonable wear and tear excepted (but such obligation shall not extend to the Building Structure and the Building Systems except pursuant to the BS/BS Exception as described in Section 9.2). Tenant's repair obligations include, without limitation, repairs to: (1) the interior side of demising walls; (2) doors; (3) floor covering; (4) interior partitions; (5) Lines (as defined in Section 27.9) installed by or on behalf of Tenant and located in or exclusively serving the Premises; (6) supplemental air conditioning units, private kitchens, including hot water heaters, plumbing, and similar facilities serving Tenant exclusively (it being agreed that Tenant shall provide Landlord with written copies of all maintenance contracts for such work); (7) all Tenant Improvements, and (8) Alterations performed by contractors retained by Tenant, including related HVAC

balancing. All work shall be performed in accordance with the rules and procedures described in Section 9.4. If Tenant fails to make any repairs to the Premises for more than fifteen (15) days after notice from Landlord (although notice shall not be required if there is an emergency), Landlord may make the repairs, and Tenant shall pay the reasonable cost of the repairs to Landlord within thirty (30) days after receipt of an invoice, together with an administrative charge in an amount equal to 10% of the cost of the repairs.

9.2 Landlord's Repair Obligations. Landlord shall maintain or cause to be maintained in good condition and operating order and keep or cause to be kept in good repair and condition, as part of Operating Expenses, the structural portions of the Building, including the foundation, floor/ceiling slabs, roof, curtain wall, exterior glass and mullions, columns, beams, shafts (including elevator shafts), elevator cabs, stairs and parking areas, men's and women's washrooms, mechanical, electrical and telephone closets (collectively, "Building Structure") and the Building's mechanical, electrical, life safety, plumbing, sprinkler systems, access systems, security systems (if any) and HVAC systems which were not constructed or installed by Tenant or Tenant Parties or Tenant's contractors or subtenants (collectively, the "Building Systems"), as well as all Project landscaping, exterior Project signage, Project parking areas and parking structures, Project mechanical, electrical, life safety, plumbing, sewer, storm drain, sprinkler and HVAC systems, plazas, art work, fountains and sculptures and all other Common Areas. Notwithstanding anything in this Lease to the contrary, Tenant shall be required to repair (or, at Landlord's election, to reimburse Landlord for the reasonable cost of such repair, together with an administrative charge in an amount equal to 10% of the cost thereof) the Building Structure, Building Systems and Common Areas to the extent required due to the negligence or willful misconduct of Tenant or because of use of the Premises for other than other than normal and customary business operations, unless and to the extent such damage is covered by insurance carried or required to be carried by Landlord pursuant to Section 13.1 and to which the waiver of subrogation is applicable (such obligation to the extent applicable to Tenant as qualified and conditioned will hereinafter be defined as the "BS/BS Exception"). Landlord may, but shall not be required to, enter the Premises at all reasonable times to make such repairs, alterations, improvements or additions to the Premises or to the Project or to any equipment located in the Project as Landlord shall desire or deem necessary or as Landlord may be required to do by Law; provided, however, except for emergencies, (i) Landlord shall provide Tenant with at least twenty-four (24) hours notice (which may be oral) prior to any such entry; (ii) any such entry shall be performed in a manner designed to minimize, to the extent reasonably practicable, interference with Tenant's use of, or ingress to or egress from, the Premises or the Parking Facilities; and (iii) subject to Tenant's rights under Section 17.6.2, to the extent the work requires Landlord to close the Premises, Landlord shall perform such work on the weekends or after Normal Business Hours if the work can reasonably be completed on the weekends or after Normal Business Hours. Without limitation on the other terms and conditions of this Lease, Tenant hereby waives any and all rights under and benefits of Section 1 of Section 1932 and Sections 1941 and 1942 of the California Civil Code or under any similar law, statute, or ordinance now or hereafter in effect.

9.3 Tenant's Right to Make Repairs. Notwithstanding any of the provisions set forth in this Lease to the contrary, if Tenant provides written notice (or, in the case of an emergency, oral notice followed by written notice within twenty-four (24) hours of such oral notice) to Landlord of an event or circumstance which requires the action of Landlord with respect to repair and/or maintenance of the Building, including the Building Structure and/or Building Systems, which event or circumstance with respect to the Building Structure or Building Systems materially or adversely affects the conduct of Tenant's business from the Premises, and Landlord fails to commence corrective action within a reasonable period of time, given the circumstances, after the receipt of such notice, but in any event not later than thirty (30) days after receipt of such notice, then Tenant may proceed to take the required action upon delivery of an additional ten (10) Business Days' notice to Landlord and any Lender whose address Landlord shall have provided to Tenant specifying that Tenant is taking such required action, and if such action was required under the provisions of this Lease to be taken by Landlord and was not commenced by Landlord within such ten (10) Business Day period and thereafter diligently pursued to completion, then Tenant shall be entitled to prompt reimbursement by Landlord of Tenant's reasonable costs and expenses in taking such action plus an administrative charge in the amount of 10% of the cost of the repairs. In the event Tenant takes such action, Tenant shall use only those contractors used by Landlord in the Building for work unless such contractors are unwilling or unable to perform, or timely perform, such work, in which event Tenant may utilize the services of any other qualified contractor which normally and regularly performs similar work in Comparable Buildings. In the event of repairs of an emergency nature which are necessary, in Tenant's reasonable opinion, to allow Tenant to continue to use the Premises for critical business operations or to protect persons or property from imminent risk of damage or injury, then Tenant may immediately make such repairs (using the services of any qualified contractor which normally and regularly performs similar work in the Project or in Comparable Buildings) without regard to the time periods or additional notice set forth above and in connection therewith, and, if such repairs were required to be made by Landlord under this Lease, then Landlord shall reimburse Tenant for the cost of such repairs, to the extent Landlord would have been required to do so pursuant to this Lease (but only to the extent that the cost thereof does not exceed the cost that Landlord would reasonably have had to incur if Landlord had made such repairs). Promptly following completion of any work taken by Tenant pursuant to the provisions of this Section, Tenant shall deliver a detailed invoice of the work completed, the materials used and the costs relating thereto. Promptly following completion of any work taken by Tenant pursuant to the provisions of this Section 9.3, Tenant shall deliver a detailed invoice of the work completed, the materials used and the costs relating thereto. If Landlord does not deliver a detailed written objection to Tenant within thirty (30) days after receipt of an invoice from Tenant, then Tenant shall be entitled to payment from Landlord of the amount set forth in such invoice. If, however, Landlord delivers to Tenant, within thirty (30) days after receipt of Tenant's invoice, a written objection to the payment of such invoice, setting forth with reasonable particularity Landlord's reasons for its claim that such action did not have to be taken by Landlord pursuant to the provisions of this Lease or that the charges are excessive (in which case Landlord shall pay the amount it contends would not have been excessive), then Tenant may proceed to claim a default by Landlord and, if elected by either Landlord or Tenant, the matter shall proceed to resolution pursuant to the procedures set forth in Section 27.28. If Tenant prevails in the arbitration and Landlord fails to pay the Award within thirty (30) days thereafter, the amount of the Award (which shall include interest at the Interest Rate from the time of each expenditure by Tenant until the date Tenant receives such amount by payment or offset and attorneys' fees and related costs) may be deducted by Tenant from the Rent next due and owing under this Lease.

9.4 Alterations. Except as permitted in this Section 9.4, Tenant shall not make alterations, additions or improvements to the Premises or install Lines in the Premises or other portions of the Building (collectively referred to as "Alterations") without first obtaining the written consent of Landlord in each instance. Landlord shall not unreasonably withhold, condition or delay its consent to any proposed Alterations, unless such Alteration will (i) affect the exterior appearance of the Building; (ii) adversely affect the Building Structure; (iii) adversely affect the Building Systems; (iv) unreasonably interfere with any other occupant's normal and customary office operation, (v) fail to comply with Law, or (vi) involve the use of hazardous materials (items (i)-(vi) may be collectively referred to in this Lease as a "Design Problem"). Landlord's consent shall not be required for any Alteration that satisfies all of the following criteria (a "Minor Alteration"): (1) is not visible from the exterior of the Premises or Building; (2) will not create a Design Problem; (3) is of a cosmetic nature such as painting, wallpapering, hanging pictures and installing carpeting; and (4) does not require work to be performed inside the walls or above the ceiling of the Premises. Even though consent is not required, the performance of Minor Alterations shall be subject to all the other provisions of this Section 10.3. Regardless of whether Landlord's consent is required to an Alteration, prior to starting work, Tenant shall furnish Landlord with plans and specifications reasonably acceptable to Landlord; names of contractors reasonably acceptable to Landlord (provided that Landlord may designate specific contractors with respect to any work affecting the Building Systems); copies of contracts; necessary permits and approvals; evidence of contractor's and subcontractor's insurance in form and amounts reasonably required by Landlord (it being also understood and agreed that all of such Alterations shall be insured by Tenant pursuant to this Lease immediately upon completion thereof); and any security for performance that is reasonably required by Landlord. Changes to the plans and specifications must also be submitted to Landlord for its approval. Alterations shall be constructed in a good and workmanlike manner using materials of a quality that is at least equal to the quality designated by Landlord as the minimum standard for the Building and in accordance with reasonable construction rules and regulations that Landlord may designate for the performance of work in the Building. In performing the work of any such Alterations, Tenant shall have the work performed in such manner so as not to obstruct access to the Project or any portion thereof, by any other tenant of the Project, and so as not to obstruct the business of Landlord or other tenants in the Project. Tenant shall reimburse Landlord within 30 days after receipt of an invoice for sums paid by Landlord for third party examination of Tenant's plans for non-Minor Alterations. In addition, within 30 days after receipt of an invoice from Landlord, Tenant shall pay Landlord a fee for Landlord's oversight and coordination of any non-Minor Alterations equal to 5% of the cost of the non-Minor Alterations. Upon completion, Tenant shall furnish (except for Minor Alterations), completion affidavits, full and final waivers of lien in recordable form, and receipted bills covering all labor and materials. Tenant shall assure that the Alterations comply with all insurance requirements and Laws. Landlord's approval of an Alteration shall not be a representation by Landlord that the Alteration complies with applicable Laws or will be adequate for Tenant's use. Tenant shall not use (and upon notice from Landlord shall cease using) contractors, services, workmen, labor, materials or equipment that, in Landlord's reasonable judgment, would disturb labor harmony with the workforce or trades engaged in performing other work, labor or services in or about the Building or the Common Areas. Tenant shall reimburse Landlord for Landlord's reasonable out-of-pocket costs and expenses reasonably incurred in connection with Landlord's review of any Alterations to the extent such Alterations could materially adversely affect the Building Structure or the Building Systems. Upon completion of any Alterations (except for Minor Alterations), at Landlord's request, Tenant agrees to prepare and Landlord shall execute if factually correct, and Tenant shall cause a Notice of Completion to be recorded in the office of the Recorder of the County of Orange in accordance with Section 3093 of the Civil Code of the State of California or any successor statute, and Tenant shall deliver to the Project management office and Landlord will sign a reproducible copy of the "as built" drawings of the Alterations as well as all permits, approvals and other documents issued by any governmental agency in connection with the Alterations. Any Alterations installed in the Premises may remain in the Premises upon the expiration or earlier termination of this Lease, except (1) if upon providing approval of the Alteration Landlord delivers Tenant written notice that such Alteration must be removed upon the expiration or earlier termination of this Lease, or (2) if Landlord's consent is not required to an Alteration, Landlord gives written notice to Tenant within 30 days of receiving written notice of such Alteration, that such Alteration must be removed upon the expiration or earlier termination of the Lease. The construction of the initial Tenant Improvements (as defined in the Work Letter) shall be governed by the terms of the Work Letter and not the terms of this Section 9.4.

9.5 <u>Landlord's Property.</u> All improvements, Alterations and appurtenances which may be installed or placed in or about the Premises, from time to time, shall be at the sole cost of Tenant and shall be the property of Landlord to the Premises shall be owned by Landlord and shall remain upon the Premises without compensation to Tenant; provided, however, that (i) Tenant may remove any of its furniture, fixtures and/or equipment (regardless of whether the same is built-in or free-standing), provided Tenant repairs any damage to the Premises caused by any such removal (except that Tenant may leave floor and wall coverings in their then-existing "as is" condition and shall have no obligation to repaint, install new floor covering or to patch normal, immaterial, wall penetrations), and (ii) Tenant shall remove the "Required Removables" as provided in the Work Letter, and (iii) Tenant shall remove any Alteration or other improvements which Landlord has notified Tenant at the time of plan approval are atypical for normal and customary office purposes and, at Landlord's request, any or all internal stairways, other than those included in the initial Tenant Improvements (unless the same are Required Removables), all at Tenant's expense, upon the expiration or early termination of the Term, and shall repair any damage to the Premises and Building caused by such removal. If Tenant fails to complete such removal and/or to repair any damage caused by the removal of such Alterations or improvements in the Premises to the extent required hereunder, Landlord may do so and may charge the actual, reasonable and documented cost thereof to Tenant.

10. **Liens**. Notwithstanding anything in this Lease to the contrary, if any liens or encumbrances are placed against the Project, Building or the Premises, or against Tenant's leasehold interest, arising out of or alleged to arise out of the work performed, materials furnished or obligations incurred by or on behalf of Tenant (a "<u>Lien</u>"), the Tenant shall protect, defend, indemnify and hold Landlord harmless from and against any claims, liabilities, judgments or costs (including, without limitation, reasonable attorneys' fees and costs) arising out of same or in connection therewith. Tenant shall give Landlord notice at least twenty (20) days prior to the commencement of any such work on the Premises (or such additional time as may be necessary under applicable laws) to afford Landlord the opportunity of posting and recording appropriate notices of non-responsibility. If a Lien occurs, Tenant reserves the right to contest such Lien by notice to Landlord of such contest given within ten (10) days after Tenant is

notified of the Lien, provided that in the event that (a) Landlord is in the process of refinancing or selling its interest in the Building or Project, or (b) Landlord's Lender requires the removal of the Lien, or (c) the Lien is the subject of a foreclosure action (any of items a, b and c to be referred to herein as a "Bonding Condition"), then Tenant shall, at its sole cost and expense, provide a bond in accordance with the California Civil Code, Section 3143 within ten (10) days after notice from Landlord that a Bonding Condition exists. If no Bonding Condition exists, a bond shall not be required if, unless Tenant fails to actively and diligently pursue the removal of the Lien. If Tenant does not timely exercise its right to contest such Lien, Tenant shall remove any such Lien by bond or otherwise within fifteen (15) days after notice by Landlord, and if Tenant shall fail to do so, Landlord may pay the amount necessary to remove such lien or encumbrance, without being responsible for investigating the validity thereof. The amount so paid including, without limitation, reasonable attorneys' fees (if and to the extent permitted by Law) shall be deemed Additional Rent under this Lease payable within 15 days after receipt of an invoice from Landlord, without limitation as to other remedies available to Landlord under this Lease. Nothing contained in this Lease shall authorize Tenant to do any act that shall subject Landlord's title to the Building or Premises to any liens or encumbrances whether claimed by operation of law or express or implied contract. Any claim to a lien or encumbrance upon the Building or Premises arising in connection with any such work or respecting the Premises not performed by or at the request of Landlord's title to the Project, Building and Premises.

11. Entry by Landlord. Subject to Tenant's reasonable security and confidentiality requirements, Landlord reserves the right at all reasonable times and upon at least twenty-four (24) hours' prior notice to Tenant, which may be given orally (and except that no notice is required in the case of an emergency) to enter the Premises to (i) inspect them; (ii) show the Premises to prospective purchasers, mortgagees, or to current or prospective mortgagees, ground or underlying lessors or insurers, or during the last twelve (12) months of the Term, to prospective tenants; (iii) post notices of nonresponsibility; or (iv) alter, improve or repair the Premises or the Building, or for alterations, repairs or improvements to the Base Building as required or permitted by this Lease. Notwithstanding anything to the contrary contained herein, Landlord may enter the Premises at any time to (A) perform services required of Landlord, including janitorial service; (B) take possession due to any Default under this Lease in the manner provided herein; and (C) subject to the provisions of this Lease, perform any covenants of Tenant which Tenant fails to perform. Subject to Tenant's rights under Section 17.6.2, Landlord may make any such entries without the abatement of Rent and may take such reasonable steps as required to accomplish the stated purposes; provided, however, except for emergencies, any such entry shall be designed to minimize, to the extent reasonably practicable, interference with Tenant's use of the Premises. For each of the above purposes, Landlord shall at all times have a key with which to unlock all the doors in the Premises, excluding Tenant's vaults, safes and special security areas designated in advance by Tenant. In an emergency, Landlord shall have the right to use any means that Landlord may deem proper to open the doors in and to the Premises. Any entry into the Premises by Landlord in the manner hereinbefore described shall not be deemed to be a forcible or unlawful entry into, or a detainer of, the Premises, or an actual or constructive eviction of Tenant from any portion of the Premises. No provision of this Lease shall be construed as obligating Landlord to perform any repairs, alterations or decorations except as otherwise expressly agreed to be performed by Landlord herein. If reasonably necessary for the protection and safety of Tenant and its employees, Landlord shall have the right, subject to Tenant's rights under Section 17.6.2, to temporarily close all or a portion of the Premises to perform repairs, alterations and additions. However, except in emergencies, Landlord will not close the Premises if the work can reasonably be completed on weekends and after Normal Business Hours.

12. Assignment and Subletting.

12.1 Transfers. Tenant shall not, without the prior written consent (except as otherwise provided in Section 12.2 or 12.7) of Landlord, which consent will not be unreasonably withheld, conditioned or delayed, assign, mortgage, pledge, hypothecate, encumber, or permit any lien to attach to, or otherwise transfer, this Lease or any interest hereunder, permit any assignment, or other transfer of this Lease or any interest hereunder by operation of law, sublet the Premises or any part thereof, or enter into any license or concession agreements or otherwise permit the occupancy or use of the Premises or any part thereof by any persons other than Tenant and its employees and contractors (other than on a temporary, occasional basis by persons or entities having a business relationship with Tenant, as set forth in Section 12.5), (all of the foregoing are hereinafter sometimes referred to collectively as "Transfers" and any person to whom any Transfer is made or sought to be made is hereinafter sometimes referred to as a "Transferee"). Except as provided in Section 12.7, if Tenant is a corporation, limited liability company, partnership, or similar entity, and if the entity which owns or controls a majority of the voting shares/rights at any time changes for any reason (including but not limited to a merger, consolidation or reorganization), such change of ownership or control shall constitute a Transfer for purposes of this Lease. The foregoing shall not apply so long as Tenant is an entity whose outstanding stock is listed on a recognized security exchange, or if at least 80% of its voting stock is owned by another entity, the voting stock of which is so listed. If Tenant desires Landlord's consent to any Transfer, Tenant shall notify Landlord in writing, which notice (the "Transfer Notice") shall include (i) the proposed effective date of the Transfer, which shall not be less than ten (10) days after the date of delivery of the Transfer Notice, (ii) a description of the protion of the Premises to be transferred (the "Subject Space"), (iii) all of the terms of the proposed Transfer and the consideration therefor, including calculation of the Transfer Premium (as defined in Section 12.3), in connection with such Transfer, the name and address of the proposed Transferee, and a copy of all existing executed and/or proposed documentation pertaining to the proposed Transfer, including all existing operative documents to be executed to evidence such Transfer or the agreements incidental or related to such Transfer, and (iv) current financial statements of the proposed Transferee certified by an officer, partner or owner thereof, business credit and personal references and history of the proposed Transferee and any other information reasonably required by Landlord which will enable Landlord to determine the financial responsibility, character, and reputation of the proposed Transferee, nature of such Transferee's business and proposed use of the Subject Space. Landlord shall approve or disapprove of the proposed Transfer within ten (10) days (the "Review Period") after Landlord's receipt of the applicable Transfer Notice. If Landlord fails to notify Tenant in writing of such approval or disapproval within such Review Period, Landlord shall be deemed to have approved such Transfer.

If Landlord disapproves a Transfer as provided in Section 12.2, Landlord shall specify the reasons for such disapproval. Any Transfer requiring Landlord's consent hereunder which is made without Landlord's prior written consent shall, at Landlord's option, be null, void and of no effect, and shall, at Landlord's option, constitute a Default by Tenant under this Lease. Whether or not Landlord consents to any proposed Transfer, Tenant shall pay Landlord's review and processing fees, as well as any reasonable professional fees (including, without limitation, attorneys', accountants', architects', engineers' and consultants' fees) incurred by Landlord, within fifteen (15) days after written request by Landlord, provided that (i) such fees shall not exceed One Thousand and No/100 Dollars (\$1,000.00) for a Transfer in the ordinary course of business; in the event of a non-ordinary course Transfer, Tenant shall reimburse Landlord for all reasonable and actual out-of-pocket costs and expenses incurred by Landlord in connection with its review of the proposed Transfer, and (ii) no such fees shall be charged in connection with any Transfer Request seeking a Transfer of Subject Space within Building II if the rentable square footage of such Subject Space, when combined with the rentable square footage of all previous Transfers of space within Building II, does not exceed twenty-five percent (25%) of the Premises leased by Tenant within Building II at the time of the Transfer Request.

- 12.2 <u>Landlord's Consent</u>. Landlord shall not unreasonably withhold its consent to any proposed Transfer of the Subject Space to the Transfere on the terms specified in the Transfer Notice. The parties hereby agree that the only reasonable reasons under this Lease and under any applicable Law for Landlord to withhold consent to any proposed Transfer are where one or more of the following apply:
 - 12.2.1 The Transferee is of a character or reputation or engaged in a business that is not consistent with the quality of the Building or the Project;
- 12.2.2 The Transferee intends to use the Subject Space for purposes which are not permitted by or would violate this Lease, or such Transferee requires improvements to the Subject Space that adversely affect the Base Building;
- 12.2.3 The Transferee is a governmental or quasi-governmental entity or an agency, department or instrumentality of a governmental or quasi-governmental agency;
- 12.2.4 The proposed Transfer would cause a violation of an exclusive right granted by Landlord in good faith in another lease for space in the Project, or would give an occupant of the Project a right to cancel its lease as a result of the proposed use to be made of the space by the sublessee or assignee, provided that upon request from Tenant, Landlord shall provide notice of all applicable exclusive rights;
 - 12.2.5 Tenant is in Default at the time of the Transfer Request;
- 12.2.6 The Transferee is not a party of reasonable financial worth and/or financial stability in light of the responsibilities to be undertaken in connection with the Transfer on the date consent is requested;
- 12.2.7 If Tenant's or Guarantor's net worth at the time of the Transfer request is less than 50% of Tenant's net worth at the time of execution of this Lease, the Transferee does not have the creditworthiness and financial strength then generally required by Landlord and landlords of the Comparable Buildings of a new tenant who is leasing space of a rentable area comparable to the rentable area of the Subject Space for a term equal to the term of the Transfer;
- 12.2.8 The Transferee does not intend to occupy the entire Subject Space and conduct its business therefrom for a substantial portion of the term of the Transfer;
- 12.2.9 Landlord and the Transferee (or an affiliate of the Transferee) have been opposing parties in litigation at any time prior to the Transfer Request; or
- 12.2.10 The Transferee has been subject to an unlawful detainer proceeding resulting in a termination of such Transferee's lease for space in southern California during the 5 years prior to the Transfer Request.

Notwithstanding the foregoing, items 12.2.6 through 12.2.10 above shall not be considered reasonable grounds for Landlord to withhold its consent with respect to any Transfer Request seeking a Transfer of Subject Space within Building II if the rentable square footage of such Subject Space, when combined with the rentable square footage of all previous Transfers of space within Building II, does not exceed twenty-five percent (25%) of the Premises leased by Tenant within Building II at the time of the Transfer Request.

If Landlord consents to any Transfer pursuant to the terms of this Section 12.2, Tenant may within six (6) months after Landlord's consent, but not later than the expiration of said six (6) month period, enter into such Transfer of the Premises or portion thereof, upon substantially the same terms and conditions as are set forth in the Transfer Notice furnished by Tenant to Landlord pursuant to Section 12.1, provided that if there are any changes in the terms and conditions from those specified in the Transfer Notice such that Landlord would initially have been entitled to refuse its consent to such Transfer under this Section 12.2, Tenant shall again submit the Transfer to Landlord for its approval and other action under this Section 12. Notwithstanding anything to the contrary in this Lease, if Tenant or any proposed Transferee claims that Landlord has unreasonably withheld, conditioned or delayed its consent under Section 12.2 or otherwise has breached or acted unreasonably under this Section 12, their sole remedies shall be an arbitration action pursuant to Section 27.28, which action shall be limited to (i) a determination of whether Landlord has acted unreasonably (and, if Landlord is held to have acted unreasonably, Landlord's consent to the Transfer shall be deemed given), and (ii) a claim for actual damages (not including any punitive, special, unforeseeable or speculative damages) caused by Landlord's breach under this Section 12.2, and Tenant hereby waives all other remedies, including, without limitation, any right to terminate this Lease, on its own behalf and, to the extent permitted under all applicable Laws, on behalf of the proposed Transferee.

12.3 Transfer Premium. If Landlord consents to a Transfer, as a condition thereto which the parties hereby agree is reasonable, Tenant shall pay to Landlord fifty percent (50%) of any Transfer Premium (hereinafter defined) actually received by Tenant from such Transferee; provided, however, that Tenant shall not be required to pay to Landlord any Transfer Premium until such time as Tenant has recovered all applicable Transaction Costs (hereinafter defined). "Transfer Premium" shall mean all rent, additional rent or other consideration payable by such Transferee in connection with the Transfer in excess of the Rent and Additional Rent payable by Tenant under this Lease during the term of the Transfer (on a per rentable square foot basis if less than all of the Premises is transferred), after deducting the reasonable expenses incurred by Tenant in connection with such Transfer for (i) any improvement allowance or other economic concessions or economic benefits (space planning allowance, moving expenses, free rent) paid by Tenant to Transferee in connection with such Transfer; (ii) any brokerage commissions incurred by Tenant in connection with the Transfer; (iii) reasonable attorneys' fees incurred by Tenant in connection with the Transfer or paid to Landlord pursuant to Section 12.1; (iv) any lease takeover incurred by Tenant in connection with the Transfer; and (v) out-of-pocket costs of advertising the space subject to the Transfer (collectively, "Transaction Costs"). "Transfer Premium" shall also include, but not be limited to, key money, bonus money or other cash consideration for rent or in lieu of rent paid by Transferee to Tenant in connection with such Transfer, and any payment in excess of fair market value for services rendered by Tenant to Transferee or for assets, fixtures, inventory, equipment, furniture, or specialized, non-general office improvements paid for by Tenant and transferred by Tenant to Transferee in connection with such Transfer. The determination of the amount of Landlord's applicable share of the Transfer Premium shall be made on a monthly basis as rent or other consideration is received by Tenant under the Transfer. For purposes of calculating the Transfer Premium on a monthly basis, Tenant's Transaction Costs shall be deemed expended by Tenant in equal monthly amounts over the entire term of the Transfer. Tenant shall pay Landlord Landlord's share of any Transfer Premium within fifteen (15) days after Tenant's receipt thereof.

12.4 Effect of Transfer. If Landlord consents to a Transfer, (i) the terms and conditions of this Lease shall in no way be deemed to have been waived or modified, (ii) such consent shall not be deemed consent to any further Transfer by either Tenant or a Transferee, (iii) Tenant shall deliver to Landlord, promptly after execution, an original executed copy of all documentation pertaining to the Transfer, (iv) Tenant shall furnish upon Landlord's request a complete statement, certified by an independent certified public accountant, or Tenant's chief financial officer, setting forth in detail the computation of any Transfer Premium Tenant has derived and shall derive from such Transfer, and (v) no Transfer relating to this Lease or agreement entered into with respect thereto, whether with or without Landlord's consent, shall relieve Original Tenant from any liability under this Lease, including, without limitation, in connection with the Subject Space. Landlord or its authorized representatives shall have the right at all reasonable times to audit the books, records and papers of Tenant relating to any Transfer, and shall have the right to make copies thereof. If the Transfer Premium respecting any Transfer shall be found understated by more than five percent (5%), Tenant shall, within thirty (30) days after demand, pay the deficiency and Landlord's reasonable costs of such audit.

12.5 Permitted Occupants. Notwithstanding any contrary provision of this Section 12, Tenant shall have the right without the payment of a Transfer Premium, and without the receipt of Landlord's consent, and provided Tenant complies with the Permitted Occupant Conditions (hereinafter defined), to allow employees of any wholly-owned (directly or indirectly) subsidiary of Tenant or Guarantor ("Related Personnel"), or of one or more entities with whom Tenant has an ongoing business relationship ("Tenant Venturers") (collectively, Related Personnel and Tenant Venturers are called "Permitted Occupants") to use portions of the Premises without Landlord's consent and without the requirement by Landlord of execution and approval of a separate sublease for such use. The "Permitted Occupant Conditions" shall mean all of the following: (a) with respect to Permitted Occupants that are employees of Tenant Venturers, such persons use, in the aggregate at any one time, no more than fifteen percent (15%) of the Premises and only on a temporary basis (in no event more than three (3) consecutive months); provided, however, that the foregoing "temporary basis" restriction shall not apply to any Tenant Venturer with whom Tenant has contracted to provide concessionaire food services to Tenant and its employees, (b) Tenant shall provide Landlord with prior written notice of the presence on the Premises of any such Permitted Occupants, (c) Tenant shall provide to Landlord, or cause the Permitted Occupants' employer(s) to provide to Landlord, satisfactory evidence of workers' compensation insurance covering the activities of their respective employees within the Premises, (d) such Permitted Occupants shall be considered "agents" of Tenant in connection with all of Tenant's obligations under this Lease and shall not be considered "third parties" for purposes of any indemnity obligations of Landlord under this Lease, (e) such occupancy shall not be a subterfuge by Tenant to avoid its obligations under this Lease or the restrictions on Transfers under this Section 12, (f) such individuals or entities shall not be permitted to occupy a separately demised portion of the Premises which contains an entrance to such portion of the Premises other than the primary entrance to the Premises, (g) all such individuals or entities shall be of a character and reputation consistent with the first-class quality of the Building and the Project, and (h) such right to use the Premises shall at all times be subject and subordinate to this Lease; provided, however, Tenant need not comply with items (b) and (c) above with respect to Related Personnel. Tenant shall promptly supply Landlord with any documents or information reasonably requested by Landlord regarding any such individuals or entities. Any occupancy permitted under this Section 12.5 shall not be deemed a Transfer under this Section 12. Notwithstanding the foregoing, no such occupancy shall relieve Tenant from any liability under this Lease.

12.6 Occurrence of Default. Any Transfer hereunder shall be subordinate and subject to the provisions of this Lease, and if this Lease shall be terminated during the term of any Transfer, Landlord shall have the right to: (i) treat such Transfer as canceled and repossess the Subject Space by any lawful means, or (ii) require that such Transferee attorn to and recognize Landlord as its landlord under any such Transfer. If Tenant shall be in Default under this Lease, Landlord is hereby irrevocably authorized, as Tenant's agent and attorney-in-fact, to direct any Transferee to make all payments under or in connection with the Transfer directly to Landlord (which Landlord shall apply towards Tenant's obligations under this Lease) until such Default is cured. Such Transferee shall rely on any representation by Landlord that Tenant is in Default hereunder, without any need for confirmation thereof by Tenant. Upon any assignment, the assignee shall assume in writing all obligations and covenants of Tenant thereafter to be performed or observed under this Lease. No collection or acceptance of rent by Landlord from any Transferee shall be deemed a waiver of any provision of this Section 12 or the approval of any Transferee or a

release of Tenant from any obligation under this Lease, whether theretofore or thereafter accruing. In no event shall Landlord's enforcement of any provision of this Lease against any Transferee be deemed a waiver of Landlord's right to enforce any term of this Lease against Tenant or any other person. If Tenant's obligations hereunder have been guaranteed, Landlord's consent to any Transfer shall not be effective unless the guaranter also consents to such Transfer.

12.7 Permitted Transfers. Notwithstanding anything to the contrary contained in this Section 12, a Transfer (i) to a successor to Tenant by purchase, merger, consolidation or reorganization (a "Successor"), or (ii) to an affiliate of Tenant (an entity which is controlled by, controls, or is under common control with, Tenant) (an "Affiliate"), shall not require Landlord's consent (and shall not be subject to the Transfer Premium) provided that Tenant notifies Landlord of any such Transfer and promptly supplies Landlord with any documents or information reasonably requested by Landlord regarding such Transfer or such entity, and further provided that (1) Tenant is not in Default under this Lease, and (2) in the case of a Successor by purchase, the Successor shall acquire all or substantially all of the stock or assets of Tenant's business or, in the case of a Successor by merger, consolidation or reorganization, the continuing or surviving entity shall own all or substantially all of the assets of Tenant, and (3) in the case of a Successor, the continuing or surviving entity shall have a net worth which is at least equal to Tenant's net worth at the date of this Lease. If requested by Landlord, Tenant's Successor shall sign a commercially reasonable form of assumption agreement. "Control," as used in this Section 12.7, shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a person or entity, whether by the ownership of voting securities, by contract or otherwise.

13. Insurance.

- 13.1 Landlord's Fire and Casualty Insurance. Landlord shall insure the Building (including the Building Structure and Building Systems) and the Project during the Term against loss or damage due to fire and other casualties covered within the classification of fire and extended coverage, vandalism coverage and malicious mischief, sprinkler leakage, water damage and special extended coverage. Such coverage shall be in such amounts, from such companies, and on such other terms, as Landlord may from time to time reasonably determine, provided that to the extent consistent with the practices of landlords of the Comparable Buildings, such coverage shall (i) be for full replacement of the Building and the Project in compliance with all then existing Laws; (ii) provide for rent continuation insurance equal to twelve months rent; and (iii) be with companies and have policies meeting the criteria set forth in Section 13.4(iii) in this Lease. Additionally, at the sole option of Landlord, such insurance coverage may include the risks of earthquakes and/or flood damage and additional hazards (including terrorism insurance), a rental loss endorsement and one or more loss payee endorsements in favor of the holders of any mortgages or deeds of trust encumbering the interest of Landlord in the Building or the ground or underlying lessors of the Building, or any portion thereof. Notwithstanding the foregoing terms of this Section 13.1, the coverage and amounts of insurance (including the amount of deductible) carried by Landlord in connection with the Building shall at a minimum be comparable to the coverage and amounts of insurance (including the amount of deductible) carried by reasonably prudent landlords of Comparable Buildings. Upon inquiry by Tenant, from time to time, Landlord shall inform Tenant of all such insurance carried by Landlord. Tenant shall, at Tenant's expense, except with respect to the Building Structure and Building Systems governed by Section 7.1 of this Lease, comply with all reasonable and customary insurance company requirements pertaining to the use of the Premises. If Tenant's conduct or use of the Premises other than for the Permitted Use causes any increase in the premium for any of Landlord's insurance policies then Tenant shall reimburse Landlord for any such increase. Tenant, at Tenant's expense, shall comply with all rules, orders, regulations or requirements of the American Insurance Association (formerly the National Board of Fire Underwriters) and with any similar body.
- 13.2 Tenant's Insurance. Tenant shall maintain the following coverages in the following amounts.
- 13.2.1 Commercial General Liability Insurance covering the insured against claims of bodily injury, personal injury and property damage (including loss of use thereof) arising out of Tenant's operations, and contractual liabilities (with respect to the contractual liabilities only, covering the performance by Tenant of its indemnity agreements under Section 14.2, relating to damage to property and/or injury or death to persons) including a Broad Form endorsement covering the insuring provisions of this Lease, for limits of liability not less than \$5,000,000 each occurrence and \$5,000,000 annual aggregate for bodily injury and property damage liability, and not less than \$5,000,000 each occurrence and \$5,000,000 annual aggregate for personal injury liability.
- 13.2.2 Physical Damage Insurance covering (i) all office furniture, business and trade fixtures, office equipment, free-standing cabinet work, movable partitions, merchandise and all other items of Tenant's property on the Premises installed by, for, or at the expense of Tenant, (ii) the Tenant Improvements (as defined in the Work Letter), excluding the Base Building, and (iii) all other improvements, alterations and additions to the Premises. Such insurance shall be written on an "all risks" of physical loss or damage basis, for the full replacement cost value (subject to reasonable deductible amounts) new without deduction for depreciation of the covered items and in amounts that meet any co-insurance clauses of the policies of insurance and shall include coverage for damage or other loss caused by fire or other peril including, but not limited to, vandalism and malicious mischief, theft, water damage of any type, including sprinkler leakage, bursting or stoppage of pipes, and explosion, and providing business interruption coverage for a period of one year.
- 13.2.3 Worker's Compensation pursuant to all applicable state and local statutes and regulations, and Employers Liability Coverage of at least \$1,000,000.00 per occurrence.
- 13.3 Form of Policies. The minimum limits of policies of insurance required of Tenant and Landlord under this Lease shall in no event limit the liability of Tenant or Landlord under this Lease. Tenant's insurance shall (i) name Landlord, and any other party the Landlord so specifies by written notice to Tenant, as an additional insured, including Landlord's managing agent, if any, (ii) be issued by an insurance company having a rating of not less than A-X in Best's Insurance Guide and licensed to do business in the State of California; (iii) be primary insurance as to

all claims thereunder and provide that any insurance carried by Landlord is excess and is non-contributing with any insurance requirement of Tenant; (iv) be in form and content reasonably acceptable to Landlord; and (v) provide that said insurance shall not be canceled or coverage reduced below the amounts required under this Section 13 unless thirty (30) days' prior written notice shall have been given to Landlord and Tenant and any Lender of Landlord, the identity of whom has been provided to Tenant in writing. Tenant shall deliver insurance certificates or other reasonable evidence that such insurance is in place to Landlord on or before the Commencement Date and at least fifteen (15) days before the expiration dates thereof. Tenant may fulfill its insurance obligations under Section 13.3.1, above, through "blanket" insurance. In the event Tenant shall fail to procure such insurance, or to deliver such policies or certificate, Landlord may, at its option upon five (5) Business Days written notice to Tenant, procure such policies for the account of Tenant unless Tenant provides same within such five (5) day period, and the cost thereof shall be paid to Landlord within five (5) days after delivery to Tenant of bills therefor.

13.4 <u>Additional Insurance Obligations</u>. Tenant shall carry and maintain during the entire Term, at Tenant's sole cost and expense, increased amounts of the insurance required to be carried by Tenant pursuant to this Section 13 and such other reasonable types of insurance coverage and in such reasonable amounts covering the Premises and Tenant's operations therein, as may be reasonably requested by Landlord, but in no event in excess of the amounts and types of insurance then being required by landlords of other Comparable Buildings. Notwithstanding anything to the contrary contained in this Lease, in the event of any termination of this Lease pursuant to Sections 15.1 or 15.2 below, Tenant shall assign and deliver to Landlord (or to any party designated by Landlord) all insurance proceeds payable to Tenant based upon the insurance carried by Tenant under Section 13.3.2 of this Lease for the value of the Tenant Improvements (to the extent paid for with the Tenant Improvement Allowance) ("Landlord's TI Proceeds").

13.5 Waiver of Claims and Subrogation. Landlord and Tenant intend that their respective property loss risks shall be borne by reasonable insurance carriers to the extent above provided, and Landlord and Tenant hereby agree to look solely to, and seek recovery only from, their respective insurance carriers in the event of a property loss to the extent that such coverage is agreed to be provided hereunder or if higher, to the extent such insurance has been obtained. The parties each hereby waive all rights and claims against each other for such losses, and waive all rights of subrogation of their respective insurers, including all rights of recovery, claims, actions or causes of action arising out of the negligence of Landlord or any Landlord Parties or the negligence of Tenant, any Tenant Parties or any of Tenant's contractors, visitors, patrons, guests, invitees, licensees or subtenants, provided such waiver of subrogation shall not affect the right to the insured to recover thereunder from the insurer. The parties agree that their respective insurance policies are now, or shall be, endorsed such that the waiver of subrogation shall not affect the right of the insured to recover thereunder. If Landlord or Tenant fail to carry the amounts and types of insurance required to be carried pursuant to this Section 13, in addition to any remedies Landlord or Tenant may have under this Lease, such failure shall be deemed to be a covenant and agreement by the party failing to carry such insurance to self-insure with respect to the type and amount of insurance such party so failed to carry, with full waiver of subrogation with respect thereto.

14. Indemnity and Waiver of Claims.

14.1 <u>Indemnification of Landlord</u>. To the fullest extent permitted by Law, Tenant shall defend (with attorneys reasonably acceptable to Landlord), indemnify, protect, save and hold harmless Landlord, its members, principals, affiliates, beneficiaries, partners, officers, directors, employees and agents (collectively, "<u>Landlord Parties</u>") from and against from and against any and all third party claims or liability for bodily injury to or death of any person or damage to any property arising before or after the Commencement Date from Tenant's use or occupancy of the Premises or the Project, or from the conduct of its business, or from any activity, work, or thing done, permitted or suffered by Tenant or its members, principals, beneficiaries, partners, officers, directors, employees and agents (collectively, "<u>Tenant Parties</u>") or any of Tenant's contractors, visitors, patrons, guests, invitees, licensees or subtenants in or about the Premises or the Project. Tenant's obligations under this Section 14.1 shall not apply to the extent that (i) the claim or liability is caused by the negligence or willful misconduct of Landlord or its agents or employees, or (ii) the claim or liability is for property damage addressed in Section 13.5. Such indemnity shall include all reasonable costs, attorneys' fees and expenses incurred in the defense of any such claim or any action or proceeding brought thereon, and will be applicable to a claim only if the Landlord: (i) notifies Tenant of the claim or liability in writing within one hundred one hundred eighty (180) days after Landlord receives notice of the claim or liability; (ii) permits Tenant to defend or settle against the claim or liability.

14.2 **Indemnification of Tenant**. To the fullest extent permitted by Law (but subject to the express limitations on liability contained in this Lease, including, without limitation, the provisions of Section 14.3), Landlord shall indemnify, defend (with attorneys reasonably acceptable to Tenant), protect, save and hold harmless Tenant and the Tenant Parties from and against any and all third party claims or liability arising before or after the Commencement Date for bodily injury to or death of any person or damage to any property arising out of Landlord's conduct, or from any activity, work, or thing done, permitted or suffered by Landlord or its agents or employees in or about the Project Common Areas, except: (i) claims and liabilities to the extent occasioned by the negligent acts or omissions of Tenant or any Tenant Parties or any of Tenant's Transferees, contractors, visitors, patrons, guests, invitees or licensees, or (ii) claims and liabilities for property damage addressed in Section 13.5. Such indemnity shall include all reasonable costs, attorneys' fees and expenses incurred in the defense of any such claim or any action or proceeding brought thereon. This indemnity will be applicable to a claim only if the Tenant: (i) notifies Landlord of the claim or liability in writing within one hundred eighty (180) days after the Tenant receives notice of the claim or liability; (ii) permits Landlord to defend or settle against the claim or liability.

14.3 <u>Landlord Not Liable</u>. Landlord and the Landlord Parties shall not be liable for, and Tenant waives, all claims for loss or damage to Tenant's business or loss, theft or damage to Tenant's Property or the property of any person claiming by, through or under Tenant resulting from: (1) wind or weather; (2) the failure of any sprinkler, heating or air-conditioning equipment, any electric wiring or any gas, water or steam pipes; (3) the backing up of any sewer pipe or downspout; (4) the bursting, leaking or running of any tank, water closet, drain or other pipe; (5) rain, water or other elements upon or coming through the roof, skylight, stairs, doorways, windows, walks or any other place upon or near the Building; and (6) any causes not reasonably within the control of Landlord. Tenant shall insure itself against such losses under Section 13 above. Should Landlord provide or Tenant elect to receive any service or products from a concessionaire, licensee or third party tenant of Landlord, Landlord shall have no liability for any services or products so provided or for any breach of contract by such third party provider. This Section 14.3 shall not limit any of Landlord's maintenance or repair obligations pursuant to this Lease or the provisions of Section 13.5 or Section 18.

14.4 Survival. The indemnification obligations set forth in this Section 14 shall survive the expiration or earlier termination of this Lease.

15. Casualty Damage.

15.1 Repair of Damage to Premises by Landlord. To the extent Landlord does not have actual knowledge of same, Tenant shall promptly notify Landlord of any damage to the Premises resulting from fire or any other casualty. If the Premises, the Building Structure, the Building Systems, or any Common Areas (including the Parking Facilities) serving or providing ingress/egress to the Premises shall be damaged by fire or other casualty, Landlord shall promptly and diligently, subject to reasonable delays for insurance adjustment or other matters beyond Landlord's reasonable control, and subject to all other terms of this Section 15, restore the Base Building (inclusive of the Building Structure and Building Systems) and such Common Areas. Such restoration shall be to substantially the same condition of the Base Building and the Common Areas prior to the casualty, except for modifications required by zoning and building codes and other Laws or by the holder of a mortgage on the Building or Project or any other modifications to the Common Areas deemed desirable by Landlord, which are consistent with the character of the Project, provided that ingress/egress to the Premises and the Parking Facilities shall not be materially impaired. Upon the occurrence of any damage to the Premises, if this Lease is not terminated, Tenant shall, at its sole cost and expense, repair any injury or damage to the Tenant Improvements and subsequent Alterations installed in the Premises and shall return such Tenant Improvements and subsequent Alterations to substantially the same condition as existed immediately prior to the casualty; provided that if the cost of such repair by Tenant exceeds the amount of insurance proceeds received by Tenant from Tenant's insurance carrier, the incremental cost differential of such repairs shall be paid by Tenant. As long as the Tenant Improvements in the Premises are rebuilt, Tenant shall be entitled to retain any portion of the proceeds of the insurance described in Section 13.2.2 (ii) and (iii) in excess of the cost of such restoration, or, if this Lease terminates, Tenant will receive all of such insurance proceeds to the extent such proceeds exceed Landlord's TI Proceeds. Prior to the commencement of reconstruction, if this Lease does not terminate pursuant to Section 15.2, or for any other reason, Tenant shall submit to Landlord, for Landlord's review and approval, all plans, specifications and working drawings relating thereto, and the provisions of Section 9.4 shall apply to all such improvement work. Landlord shall not be liable for any inconvenience or annoyance to Tenant Parties or any of Tenant's contractors, visitors, patrons, guests, invitees, licensees or subtenants, or injury to Tenant's business resulting in any way from such damage or the repair thereof; provided however, that if such fire or other casualty shall have damaged the Premises or Common Areas necessary for Tenant to reasonably conduct Tenant's Permitted Use, or if the repair of the Premises or Common Areas prevents Tenant from reasonably conducting Tenant's Permitted Use in the Premises, and the Premises are not occupied by Tenant as a result thereof, Landlord shall allow Tenant a proportionate abatement of Rent, during the time and to the extent the Premises are unfit for occupancy for the purposes permitted under this Lease, and are not occupied by Tenant as a result thereof. Tenant's right to rent abatement pursuant to the preceding sentence shall terminate as of the date Tenant should have completed repairs to the Premises assuming Tenant used reasonable due diligence in connection therewith (which date shall be subject to extension for delays attributable to Force Majeure Events).

15.2 Landlord's Option to Repair. Notwithstanding the terms of Section 15.1, Landlord may elect not to rebuild and/or restore the Premises or the Building, and instead terminate this Lease, by notifying Tenant in writing of such termination within sixty (60) days after the date of damage, such notice to include a termination date giving Tenant ninety (90) days to vacate the Premises, which ninety (90) day period is subject to extension due to Force Majeure Events, but Landlord may so elect only if the Building or the Project facilities serving the Building shall be damaged by fire or other casualty or cause, whether or not the Premises are affected, and one or more of the following conditions is present: (i) in Landlord's reasonable judgment, repairs cannot reasonably be completed within one (1) year after the date of discovery of the damage (when such repairs are made without the payment of overtime or other premiums); (ii) the holder of any mortgage on the Building or Project or ground lessor with respect to the Building or Project shall require that the insurance proceeds or any portion thereof in excess of Landlord's Contribution (hereinafter defined) be used to retire the mortgage debt; (iii) the costs of repair of the damage is not covered by Landlord's insurance (or by the insurance Landlord is required to carry under this Lease) in excess of the sum of the deductible plus Landlord's Contribution, and (a) Tenant does not agree to fund the amount in excess of Landlord's Contribution plus the deductible required to complete the appropriate repairs, (b) Landlord elects not to commence rebuilding or reconstructing within one (1) year from the date of such damage and destruction and (c) Landlord elects to terminate the leases of all other tenants of the Project similarly affected by the damage and destruction; or (iv) the damage occurs during the last twelve (12) months of the Lease Term; provided, however, that if Landlord does not elect to terminate this Lease pursuant to Landlord's termination right as provided above, and either (1) the repairs cannot, in the reasonable opinion of an architect or contractor mutually and reasonably agreed upon by Landlord and Tenant, be completed within one (1) year after the damage or destruction is discovered (which period shall not be subject to extension as a result of any Force Majeure Event), or (2) the damage occurs during the last twelve (12) months of the Lease Term, Tenant may elect, no earlier than forty-five (45) days after the date of the damage and not later than ninety (90) days after the date of such damage, to terminate this Lease by written notice to

Landlord effective as of the date specified in such notice (Tenant, however, shall not have the right to terminate this Lease if the damage was caused by the negligence or intentional misconduct of Tenant, Tenant Parties or any of Tenant's transferees, contractors or licensees). At any time, from time to time, after the date occurring forty-five (45) days after the date of the damage, Tenant may request that Landlord provide Tenant with a certificate from the architect or contractor described above setting forth such architect's or contractor's reasonable opinion of the date of completion of the repairs and Landlord shall respond to such request within fifteen (15) Business Days. For purposes of this Section 15.2, "Landlord's Contribution" shall mean Five Hundred Thousand and No/100 Dollars (\$500,000.00); provided, however, that such amount shall be reduced by an amount equal to \$8,333.33 on the first day of each month following the last day of the fifth (5th) year of the initial Term.

- 15.3 <u>Waiver of Statutory Provisions</u>. The provisions of this Lease, including this Section 15, constitute an express agreement between Landlord and Tenant with respect to any and all damage to, or destruction of, all or any part of the Premises, the Building or the Project, and any statute or regulation of the State of California, including, without limitation, Sections 1932(2) and 1933(4) of the California Civil Code, with respect to any rights or obligations concerning damage or destruction in the absence of an express agreement between the parties, and any other statute or regulation, now or hereafter in effect, shall have no application to this Lease or any damage or destruction to all or any part of the Premises, the Building or the Project.
- 16. Condemnation. If the whole or any part of the Premises, the Building, the Parking Facilities or the Project shall be taken by power of eminent domain or condemned by any competent authority for any public or quasi-public use or purpose, or if any adjacent property or street shall be so taken or condemned, or reconfigured or vacated by such authority in such manner as to require the use, reconstruction or remodeling of any part of the Premises, the Building or Project and if as a result thereof Tenant cannot conduct its business operations in substantially the same manner such business operations were conducted prior to such taking while still retaining substantially the same material rights and benefits it bargained to receive under this Lease, or if Landlord shall grant a deed or other instrument in lieu of such taking by eminent domain or condemnation as a result thereof, Landlord and Tenant shall each have the option to terminate this Lease on ninety (90) days written notice to the other party effective as of the date possession is required to be surrendered to the authority. Tenant shall not because of such taking assert any claim against Landlord or the authority for any compensation because of such taking and Landlord shall be entitled to the entire award or payment in connection therewith, except that Tenant shall have the right to file any separate claim available to Tenant for any taking of Tenant's personal property and fixtures belonging to Tenant and removable by Tenant upon expiration of the Lease Term pursuant to the terms of this Lease, for moving expenses, and for any other amount to which Tenant may be entitled pursuant to applicable Law that does not reduce Landlord's award, and provided such claim is payable separately to Tenant or is otherwise separately identifiable. All Rent shall be apportioned as of the date of such termination. If any part of the Premises shall be taken, and this Lease shall not be so terminated, the Rent shall be proportionately abated. Tenant hereby waives any and all rights it might otherwise have pursuant to Section 1265.130 of The California Code of Civil Procedure. Notwithstanding anything to the contrary contained in this Article 13, in the event of a temporary taking of all or any portion of the Premises for a period of one hundred and eighty (180) days or less, then this Lease shall not terminate but the Base Rent and the Additional Rent shall be abated for the period of such taking in proportion to the ratio that the amount of rentable square feet of the Premises taken bears to the total rentable square feet of the Premises. Landlord shall be entitled to receive the entire award made in connection with any such temporary taking.

17. Events of Default.

- 17.1 Events of Default. The occurrence of any of the following shall constitute a default of this Lease by Tenant (a "Default"):
- 17.1.1 Any failure by Tenant to pay any Rent or any other charge required to be paid under this Lease, or any part thereof, when due, or to provide any portion of the Security Deposit when required pursuant to the terms of this Lease, unless such failure is cured within five (5) Business Days after Tenant's receipt of written notice thereof; or
- 17.1.2 Any failure by Tenant to observe or perform any other provision, covenant or condition of this Lease to be observed or performed by Tenant where such failure continues for thirty (30) days after Tenant's receipt of written notice thereof; provided that if the nature of such default is such that the same cannot reasonably be cured within a thirty (30) day period, Tenant shall not be deemed to be in default if it diligently commences such cure within such period and thereafter diligently proceeds to rectify and cure such default; or
- 17.1.3 The failure by Tenant to observe or perform according to any of the material provisions of Sections 8, 10, 12, 13 or 21 of this Lease, where such failure continues for more than ten (10) Business Days after Tenant's receipt of written notice thereof; or
 - 17.1.4 Tenant defaults beyond any notice and cure period under any Other Lease covering space in excess of 5,000 rentable square feet; or
- 17.1.5 (i) Tenant or Guarantor makes a general assignment or general arrangement for the benefit of creditors; (ii) a petition for adjudication of bankruptcy or for reorganization or rearrangement is filed by or against Tenant or Guarantor and is not dismissed within ninety (90) days; (iii) a trustee or receiver is appointed to take possession of substantially all of Tenant's or Guarantor's assets or of Tenant's interest in this Lease and possession is not restored within ninety (90) days; or (iv) substantially all of Tenant's or Guarantor's assets or Tenant's interest in this Lease is subjected to attachment, execution or other judicial seizure which is not discharged within ninety (90) days.

17.1.6 Guarantor or any other guarantor of this Lease revokes, rescinds, denies or otherwise terminates or purports to revoke, rescind, deny or otherwise terminate, any guaranty of all or any portion of Tenant's obligations under this Lease.

The notice periods provided herein are in lieu of, and not in addition to, any notice periods provided by Law.

- 17.2 <u>Remedies Upon Default.</u> Upon the occurrence of a Default by Tenant, Landlord shall have, in addition to any other remedies available to Landlord at law or in equity (all of which remedies shall be distinct, separate and cumulative), the option to pursue any one or more of the following remedies, each and all of which shall be cumulative and nonexclusive, without any notice or demand whatsoever.
- 17.2.1 Terminate this Lease, in which event Tenant shall immediately surrender the Premises to Landlord, and if Tenant fails to do so, Landlord may, without prejudice to any other remedy which it may have for possession or arrearages in rent, and after due process of law, enter upon and take possession of the Premises and expel or remove Tenant and any other person who may be occupying the Premises or any part thereof, without being liable for prosecution or any claim or damages therefor; and Landlord may recover from Tenant the following:
 - (i) The worth at the time of award of any unpaid rent which has been earned at the time of such termination; plus
 - (ii) The worth at the time of award of the amount by which the unpaid rent which would have been earned after termination until the time of award exceeds the amount of such rental loss that Tenant proves could have been reasonably avoided; plus
 - (iii) The worth at the time of award of the amount by which the unpaid rent for the balance of the Lease Term after the time of award exceeds the amount of such rental loss that Tenant proves could have been reasonably avoided; plus
 - (iv) Any other amount necessary to compensate Landlord for all the detriment proximately caused by Tenant's failure to perform its obligations under this Lease or which in the ordinary course of things would be likely to result therefrom, specifically including but not limited to, brokerage commissions and advertising expenses incurred, expenses of remodeling the Premises or any portion thereof for a new tenant, whether for the same or a different use, and any special concessions made to obtain a new tenant; and
 - (v) At Landlord's election, but subject to the provisions of this Lease, such other amounts in addition to or in lieu of the foregoing as may be permitted from time to time by applicable law.

The term "rent" as used in this Section 17.2 shall be deemed to be and to mean all sums of every nature required to be paid by Tenant to Landlord pursuant to the terms of this Lease. As used in Paragraphs 17.2.1(i) and (ii), above, the "worth at the time of award" shall be computed by allowing interest at the rate set forth in Article 25 of this Lease, but in no case greater than the maximum amount of such interest permitted by law. As used in Paragraph 17.2.1(iii) above, the "worth at the time of award" shall be computed by discounting such amount at the discount rate of the Federal Reserve Bank of San Francisco at the time of award plus one percent (1%).

- 17.2.2 Landlord shall have the remedy described in California Civil Code Section 1951.4 (lessor may continue lease in effect after lessee's breach and abandonment and recover rent as it becomes due, if lessee has the right to sublet or assign, subject only to reasonable limitations). Accordingly, if Landlord does not elect to terminate this Lease on account of any Default by Tenant, Landlord may, from time to time, without terminating this Lease, enforce all of its rights and remedies under this Lease, including the right to recover all rent as it becomes due.
- 17.2.3 Landlord shall at all times have the rights and remedies (which shall be cumulative with each other and cumulative and in addition to those rights and remedies available under Sections 17.2.1 and 17.2.2, above, or any law or other provision of this Lease), without prior demand or notice except as required by applicable law, to seek any declaratory, injunctive or other equitable relief, and specifically enforce this Lease, or restrain or enjoin a violation or breach of any provision hereof.
- 17.3 <u>Subleases of Tenant</u>. Whether or not Landlord elects to terminate this Lease on account of any Default by Tenant, as set forth in this Section 17, Landlord shall have the right to terminate any and all subleases, licenses, concessions or other consensual arrangements for possession entered into by Tenant and affecting the Premises or may, in Landlord's sole discretion, succeed to Tenant's interest in such subleases, licenses, concessions or arrangements. In the event of Landlord's election to succeed to Tenant's interest in any such subleases, licenses, concessions or arrangements, Tenant shall, as of the date of notice by Landlord of such election, have no further right to or interest in the rent or other consideration receivable thereunder.
- 17.4 Waiver of Jury Trial. TENANT AND LANDLORD HEREBY WAIVE, TO THE FULLEST EXTENT PERMITTED BY LAW, THE RIGHT TO TRIAL BY JURY IN ANY LITIGATION ARISING OUT OF OR RELATING TO THIS LEASE.
- 17.5 <u>Landlord's Cure.</u> All covenants and agreements to be kept or performed by Tenant under this Lease shall be performed by Tenant at Tenant's sole cost and expense and without any reduction of Rent, except to the extent, if any, otherwise expressly provided herein. If Tenant shall fail to perform any obligation under this Lease, and such failure shall continue in excess of the time allowed under Section 17.1, above, then upon three (3) additional days notice from Landlord, Landlord may, but shall not be obligated to, make any such payment or perform any such act on Tenant's part without waiving its rights based upon any Default of Tenant and without releasing Tenant from any

obligations hereunder. Except as may be specifically provided to the contrary in this Lease, Tenant shall pay to Landlord, upon delivery by Landlord to Tenant of statements therefore, sums equal to expenditures reasonably made and obligations incurred by Landlord in connection with the remedying by Landlord of Tenant's Defaults pursuant to the provisions of this Lease. Tenant's obligations under this Section 17.5 shall survive the expiration or sooner termination of the Term

17.6 Landlord Default.

17.6.1. General. Notwithstanding anything to the contrary set forth in this Lease, Landlord shall be in default in the performance of any obligation required to be performed by Landlord pursuant to this Lease if (i) in the event a failure by Landlord is with respect to the payment of money, Landlord fails to pay such unpaid amounts within five (5) Business Days of written notice from Tenant that the same was not paid when due; (ii) if the failure by Landlord is other than (i) above, Landlord fails to perform such obligation within a reasonable time period with the expenditure of diligent efforts, but in no event more than thirty (30) days after the receipt of written notice from Tenant to Landlord (and to any Lenders whose address Landlord has supplied to Tenant) specifying in detail Landlord's failure to perform; provided, however, if the nature of Landlord's obligation is such that more than thirty (30) days are reasonably required for its performance, then Landlord shall not be in default under this Lease if Landlord commences such performance within such thirty (30) day period and thereafter diligently pursue the same to completion. Upon any such default by Landlord under this Lease, Tenant may, except as otherwise specifically provided in this Lease to the contrary, exercise any of its rights provided at law or in equity; provided, however, that in no event may Tenant terminate this Lease or abate rent except as specifically set forth in this Lease, and provided, further, that notwithstanding any contrary provision in this Lease, neither Landlord nor the Landlord Parties shall be liable under any circumstances for injury or damage to, or interference with, Tenant's business, including but not limited to, loss of profits, loss of rents or other revenues, loss of business opportunity, loss of goodwill or loss of use, in each case, however occurring, except to the extent a court of competent jurisdiction (or an arbitrator pursuant to Section 27.28) ultimately determines the same to be attributable to the gros

17.6.2. Abatement of Rent. If Tenant is prevented from using, and does not use, the Premises or any portion thereof, as a result of (i) any repair, maintenance or alteration which Landlord failed to perform after the Lease Commencement Date and required by Landlord to perform pursuant to this Lease, which substantially interferes with Tenant's use of or ingress to or egress from the Premises or Parking Facilities; (ii) any failure to provide services, utilities or ingress to and egress from the Premises or Parking Facilities as required by this Lease; or (iii) the cleanup or remediation activities in connection with Hazardous Materials not brought on the Premises by Tenant Parties, (iv) any closure of the Building pursuant to Section 22.1 below, or (v) any default by Landlord under Sections 22.2 or 27.20 (any such set of circumstances as set forth in items (i) through (v), above, to be known as an "Abatement Event"), then Tenant shall give Landlord written notice of such Abatement Event, and if such Abatement Event continues for three (3) consecutive Business Days after Landlord's receipt of any such notice (the "Eligibility Period"), then the Base Rent and Tenant's Share of Direct Expenses shall be abated or reduced, as the case may be, after expiration of the Eligibility Period for such time that Tenant continues to be so prevented from using, and does not use, the Premises, or a portion thereof, in the proportion that the rentable area of the portion of the Premises that Tenant is prevented from using, and does not use bears to the total rentable area of the Premises. Notwithstanding the foregoing, if Tenant is prevented from using the Premises due to an Abatement Event resulting from interference with Tenant's use of or ingress to or egress from the Parking Facilities as described above (a "Parking Abatement Event"), Landlord shall use commercially reasonable efforts to provide to Tenant, at Landlord's sole cost and expense, to the extent reasonably available, substitute parking spaces reasonably acceptable to Tenant (including, if such substitute parking is offsite, reasonably acceptable shuttle services between such offsite substitute parking and the Building for Tenant and Tenant's business invitees), and no Parking Abatement Event shall be deemed to exist during such times as Landlord provides such substitute parking. Such right to abate Base Rent and Tenant's Share of Direct Expenses shall be Tenant's sole and exclusive remedy at law or in equity for an Abatement Event; provided, however, that nothing in this Section 17.6.2 shall impair Tenant's rights under Section 17.6.1.

- 17.7 <u>Efforts to Relet</u>. No re-entry or repossession, repairs, maintenance, changes, alterations and additions, reletting, appointment of a receiver to protect Landlord's interests hereunder, or any other action or omission by Landlord shall be construed as an election by Landlord to terminate this Lease or Tenant's right to possession, or to accept a surrender of the Premises, nor shall same operate to release Tenant in whole or in part from any of Tenant's obligations hereunder, unless express written notice of such intention is sent by Landlord to Tenant.
- 18. Limitation of Liability. The liability of Landlord or the Landlord Parties to Tenant for any default by Landlord under this Lease or arising in connection herewith or with Landlord's operation, management, leasing, repair, renovation, alteration or any other matter relating to the Project or the Premises shall be limited solely and exclusively to an amount which is equal to the interest of Landlord in the Project (including any sales, financing, condemnation, or insurance proceeds which Landlord receives). Neither Landlord, nor any of the Landlord Parties shall have any personal liability therefor, and Tenant hereby expressly waives and releases such personal liability on behalf of itself and all persons claiming by, through or under Tenant. The limitations of liability contained in this Section 18 shall inure to the benefit of Landlord's and the Landlord Parties' present and future partners, beneficiaries, officers, directors, trustees, shareholders, agents and employees, and their respective partners, heirs, successors and assigns.
- 19. **Holding Over**. If Tenant holds over after the expiration of the Lease Term or earlier termination thereof, with or without the express or implied consent of Landlord, such tenancy shall be from month-to-month only, and shall not constitute a renewal hereof or an extension for any further term, and in such case Rent shall be payable at a monthly rate equal to (i) the Rent applicable during the last rental period of the Lease Term under this Lease, and (ii) during the first thirty (30) days of such holdover, one hundred ten percent (110%), and thereafter, one hundred fifty percent (150%) (the "Rent Premium"). Such month-to-month tenancy shall be subject to every other applicable

term, covenant and agreement contained herein. Nothing contained in this Section 19 shall be construed as consent by Landlord to any holding over by Tenant, and Landlord expressly reserves the right to require Tenant to surrender possession of the Premises to Landlord as provided in this Lease upon the expiration or other termination of this Lease. The provisions of this Section 19 shall not be deemed to limit or constitute a waiver of any other rights or remedies of Landlord provided herein or at law. If Tenant fails to surrender the Premises upon the termination or expiration of this Lease, in addition to any other liabilities to Landlord accruing therefrom, Tenant shall protect, defend, indemnify and hold Landlord harmless from all loss, costs (including reasonable attorneys' fees) and liability resulting from such failure, including, without limiting the generality of the foregoing, any claims made by any succeeding tenant founded upon such failure to surrender and any lost profits to Landlord resulting therefrom; provided that the foregoing shall not apply during or with respect to any period for which Landlord has accepted a Rent Premium. In addition to the payment of the amounts provided above, if Landlord is unable to deliver possession of the Premises to a new tenant, or to perform improvements for a new tenant, as a result of Tenant's holdover and Tenant fails to vacate the Premises within fifteen (15) days after Landlord notifies Tenant of Landlord's inability to deliver possession, or perform improvements, Tenant shall be liable to Landlord for all damages, including, without limitation, consequential damages, that Landlord suffers from the holdover.

20. Surrender of Premises.

20.1 <u>Surrender; Removal of Tenant's Property and Required Removables</u>. At the expiration or earlier termination of this Lease or Tenant's right of possession, Tenant shall remove Tenant's Property from the Premises, and quit and surrender the Premises to Landlord, broom clean, and in good order, condition and repair, ordinary wear and tear excepted. Tenant shall also be required to remove the "Required Removables" (as defined and in accordance with the Work Letter) and the Alterations that Landlord has notified Tenant will be required to be removed pursuant to Section 9.5. If Tenant fails to remove any of Tenant's Property within five (5) Business Days after the termination of this Lease or of Tenant's right to possession, Landlord, at Tenant's sole cost and expense, shall be entitled (but not obligated) to remove and store Tenant's Property. Landlord shall not be responsible for the value, preservation or safekeeping of Tenant's Property. Tenant shall pay Landlord, upon demand, the actual expenses and storage charges incurred for Tenant's Property. In addition, if Tenant fails to remove Tenant's Property from the Premises or storage, as the case may be, within 30 days after written notice, Landlord may deem all or any part of Tenant's Property to be abandoned, and title to Tenant's Property shall be deemed immediately vested in Landlord.

20.2 Removal of Lines...Unless Landlord requires the Lines (as defined in Section 27.9) to remain, Tenant shall have the right, but not the obligation, to remove such Lines prior to the expiration or earlier termination of this Lease. At least 60 days prior to the expiration of the end of the Term, Tenant shall advise Landlord if it desires to remove the Lines. If Tenant provides such notice, then upon receipt of 24-hour advance notice from Landlord to Tenant at least 14 days prior to the end of the Term, the Lines will be made available for inspection by Landlord to determine if Landlord will require Tenant to leave the Lines at the Premises upon the end of the Term. If Landlord elects not to require the Lines to remain, Tenant may (but shall not be obligated) to remove, at Tenant's expense, all of the Lines. Such Lines shall be removed by Tenant before the Termination Date, provided that upon prior written notice to Landlord, Tenant may remain in the Premises for up to five (5) days after the Termination Date for the sole purpose of removing the Lines. Tenant's possession of the Premises shall be subject to all of the terms and conditions of this Lease, including the obligation to pay Rent on a per diem basis at the rate in effect for the last month of the Term. Tenant shall repair damage caused by Tenant's installation or removal of Lines and, if Tenant elects to remove Lines as provided herein, Tenant shall cause the Lines to be removed and disposed of in accordance with all applicable Laws. If Tenant fails to make such repairs in a timely manner, Landlord, at Tenant's expense, may perform the required repairs and Tenant, within thirty (30) days after receipt of an invoice, shall reimburse Landlord for the reasonable costs incurred by Landlord.

21. Subordination; Estoppel Certificate.

21.1 <u>Subordination</u>. Landlord shall deliver to Tenant a commercially reasonable subordination and non-disturbance agreement ("SNDA", whether one or more) in favor of Tenant from all ground lessors, mortgage holders or lien holders of Landlord encumbering the Building ("Lender", whether one or more), or who later come into existence with respect to the Building at any time prior to the expiration of the Term, and such delivery shall be in consideration of, and a condition precedent to Tenant's agreement to be bound by the terms of this Section 2.1. Tenant shall, within ten (10) days of request by Landlord from time to time, execute and deliver to Landlord any such commercially reasonable SNDA delivered by Landlord. Tenant acknowledges and agrees that any such SNDA shall, among other things, recognize that: (i) Lender (which term shall include any purchaser at a foreclosure sale) shall not be liable for, nor subject to, any offsets or defenses which Tenant may have by reason of any act or omission of Landlord under this Lease; (ii) Lender shall not be liable for the return of any sums which Tenant may have paid to Landlord under this Lease as and for security deposits, advance rentals or otherwise, except to the extent that such sums are actually delivered by Landlord to Lender; and (iii) Lender shall not be bound by any amendment to this Lease made without Lender's consent. Tenant and Landlord acknowledge that the form of SNDA attached hereto as **Exhibit H** is a commercially reasonable SNDA. Subject to Tenant's receipt of an SNDA from the applicable Lender as described above, (i) this Lease shall be subject and subordinate to the ground lease, mortgage, trust deed or other encumbrances and to all renewals, extensions, modifications, consolidations and replacements thereof, and to all advances made upon the security thereof, described in the applicable SNDA, unless the applicable Lender requires in writing that this Lease be superior thereto, (ii) Tenant covenants and agrees in the event any proceedings are brought for the foreclosure of any such mortgage or deed in lieu thereof (or if any ground lease is terminated), to attorn to the Lender or purchaser or any successors thereto upon any such foreclosure sale or deed in lieu thereof, if so requested to do so by such purchaser or Lender, and to recognize such purchaser or Lender as the lessor under this Lease, provided such Lender, purchaser or successor shall agree to accept this Lease and not disturb Tenant's occupancy so long as Tenant timely pays the Rent and observes and performs the terms, covenants and conditions of this Lease to be observed and performed by Tenant, and (iii) Tenant waives the provisions of any current or future statute, rule or law which may give or purport to give Tenant any right or election to terminate or

otherwise adversely affect this Lease and the obligations of the Tenant hereunder in the event of any foreclosure proceeding or sale. Landlord's interest herein may be assigned as security at any time to any Lender. Tenant shall, within ten (10) Business Days of request by Landlord, execute such further instruments or assurances as Landlord may reasonably deem necessary to evidence or confirm the subordination or superiority of this Lease to any such mortgages, trust deeds, ground leases or underlying leases in accordance with the provisions of this Section 21.1. If Tenant fails to return to Landlord the executed SNDA (or other commercially reasonable instrument/assurances described in the preceding sentence) within the required ten (10) day period, Landlord shall make a second request to Tenant, and if Tenant fails to deliver the executed SNDA (or other commercially reasonable instrument/assurances described in the preceding sentence) within five (5) Business Days thereafter, in addition to Landlord's other rights and remedies provided in this Lease, Tenant shall pay to Landlord a late fee of \$500.00 for each day beyond such five (5) Business Day period that Tenant fails to deliver the executed SNDA (or other commercially reasonable instrument/assurances described in the preceding sentence) to Landlord.

21.2 Estoppels. Landlord and Tenant shall each, within ten (10) days after receipt of a written request from the other, execute and deliver an estoppel certificate to those parties as are reasonably requested by the other (including a Lender or prospective purchaser). The estoppel certificate shall include a statement certifying that this Lease is unmodified (except as identified in the estoppel certificate) and in full force and effect, describing the dates to which Rent and other charges have been paid, representing that, to such party's actual knowledge, there is no default (or stating the nature of the alleged default) and indicating other matters with respect to the Lease that may reasonably be requested. If Tenant fails to return to Landlord an executed estoppel certificate within the foregoing ten (10) day period, Landlord shall make a second request to Tenant, and if Tenant fails to deliver an executed estoppel certificate within five (5) days thereafter, in addition to Landlord's other rights and remedies provided in this Lease, Tenant shall pay to Landlord a late fee of \$500.00 for each day beyond such 5 day period, that Tenant fails to deliver an executed estoppel certificate within the foregoing ten (10) day period, Tenant shall make a second request to Landlord, and if Landlord fails to deliver an executed estoppel certificate within five (5) days thereafter, in addition to Tenant's other rights and remedies provided in this Lease, Landlord shall pay to Tenant a late fee of \$500.00 for each day beyond such 5 day period, that Landlord fails to deliver an executed estoppel certificate to Tenant.

22. Landlord's Development Rights.

22.1 <u>Development of Project</u>. Except as otherwise expressly set forth in this Lease, Landlord excepts and reserves exclusively to itself the use of: (1) roofs (subject to Section 22.3), (2) telephone, electrical and janitorial closets, (3) equipment rooms, Building risers or similar areas that are used by Landlord for the provision of Building services, (4) rights to the land and improvements below the floor of the Premises, (5) the improvements and air rights above the Premises, (6) the improvements and air rights outside the demising walls of the Premises, and (7) the areas within the Premises used for the installation of utility lines and other installations serving occupants of the Building. Notwithstanding the foregoing, at Tenant's request, and subject to the other terms and conditions of this Lease and such reasonable conditions as Landlord may impose, Landlord shall permit Tenant the non-exclusive use of the telephone, electrical and janitorial closets and the equipment rooms, Building risers or similar areas as reasonably required for Tenant's business operations. Any such use, and any additional installation, maintenance, insurance or repair costs resulting from such use, shall be at Tenant's sole cost and expense. Landlord also has the right to make such other changes to the Project and the Building as Landlord deems appropriate, provided the changes do not materially affect Tenant's ability to use the Premises for the Permitted Use. Landlord shall also have the right (but not the obligation) to temporarily close the Building if Landlord reasonably determines that there is an imminent danger of significant damage to the Building or of personal injury to Landlord's employees or the occupants of the Building. The circumstances under which Landlord may temporarily close the Building shall include, without limitation, electrical interruptions, hurricanes and civil disturbances. A closure of the Building under such circumstances shall not constitute a constructive eviction nor entitle Tenant to an abatem

22.2 <u>Subdivision of Project</u>. Landlord reserves the right to further subdivide all or a portion of the Project. Tenant agrees to execute and deliver, upon demand by Landlord and in the form requested by Landlord, any additional documents needed to conform this Lease to the circumstances resulting from such subdivision. If portions of the Project or property adjacent to the Project (collectively, the "<u>Other Improvements</u>") are owned by an entity other than Landlord, Landlord, at its option, may enter into an agreement with the owner or owners of any or all of the Other Improvements to provide (i) for reciprocal rights of access and/or use of the Project and the Other Improvements, (ii) for the common management, operation, maintenance, improvement and/or repair of all or any portion of the Project and the Other Improvements, (iii) for the allocation of a portion of the Operating Expenses to the Other Improvements and the operating expenses and taxes for the Other Improvements to the Project, and (iv) for the use or improvement of the Other Improvements and/or the Project in connection with the improvement, construction, and/or excavation of the Other Improvements and/or the Project; provided, however, that the foregoing shall not (A) materially adversely increase or affect Tenant's monetary or non-monetary obligations with respect to the Premises or the Project, or (B) materially adversely interfere with Tenant's use of, or ingress to or egress from the Premises or the Parking Facilities. Nothing contained herein shall be deemed or construct to limit or otherwise affect Landlord's right to convey all or any portion of the Project or any other of Landlord's rights described in this Lease. Tenant acknowledges that portions of the Project and/or the Other Improvements will be under construction following Tenant's occupancy of the Premises, and that such construction will result in levels of noise, dust, obstruction of access, etc. which are in excess of that present in a fully constructed project. Tenant hereby w

22.3 <u>Telecommunications Equipment</u>. At any time during the Lease Term, Tenant may install, at Tenant's sole cost and expense, telecommunication equipment upon the roof of the Building in up to two thousand (2,000) square feet of space on the roof of the Building, without the payment of Base Rent or Operating Expenses or any charge by Landlord therefor, except utilities necessary to Tenant's operation of the telecommunication equipment. The

physical appearance and location of any such installation and the size of the equipment shall be subject to Landlord's reasonable approval, and Landlord may require Tenant to install screening around such equipment, at Tenant's sole cost and expense, as reasonably designated by Landlord. Tenant shall maintain such equipment, at Tenant's sole cost and expense. If Tenant elects to exercise its right to install telecommunication equipment as set forth in this Section 22.3, then Tenant shall give Landlord prior written notice thereof and Landlord and Tenant shall execute a Telecommunication Agreement in substantially the form as **Exhibit K**, covering the installation and maintenance of such equipment, Tenant's indemnification of Landlord with respect thereto, Tenant's obligation to remove such equipment upon the expiration or earlier termination of this Lease, and other related matters.

23. Signage.

- 23.1 <u>Project Name and Project Signage</u>. Subject to the later provisions hereof, from and after the Commencement Date, the name of the Project (the "<u>Project Name</u>") shall be "Impac Center," or such other name as Tenant may reasonably request in writing that is not an Objectionable Name (hereinafter defined), and if during such time Landlord refers to the Project by name in its marketing materials, Landlord shall use the Project Name. Tenant shall be entitled to modify, at Tenant's sole cost and expense, but not more than two (2) times during the Term, the Project name so long as Tenant's New Name is not an Objectionable Name (hereinafter defined). Notwithstanding the foregoing, if at any time during the Term (as the same may be extended), Tenant ceases to lease either (i) all of the rentable space in the Building, or (ii) an aggregate of at least 200,000 rentable square feet in the Project, Landlord shall have the right to change the name of the Project from time to time to such name as Landlord may elect in its sole discretion, and Tenant's rights under this Section 23.1 shall expire and Tenant shall have no further Project naming rights under this Lease. Except as otherwise set forth in this Section 23, Landlord shall have the right to install, affix and maintain any and all signs on the exterior and on the interior of the Project or Project Buildings as Landlord may, in Landlord's sole discretion, desire. Tenant shall not use pictures or illustrations of the Project or the Building in advertising or other publicity or for any purpose other than as the address of the business to be conducted by Tenant in the Premises, without the prior written consent of Landlord, which consent shall not be unreasonably withheld, conditioned or delayed. Landlord shall have the right to change the address of the Project and/or the Project Buildings from time to time upon prior written notice to Tenant; provided, however, that Landlord shall reimburse Tenant for all out-of-pocket costs reasonably incurred by Tenant in connecti
- 23.2 <u>Full Floors</u>. Subject to Landlord's prior written approval, in its reasonable discretion, and provided all signs are in keeping with the quality, design and style of the Building and the Project and applicable Law, Tenant, if the Premises comprise one or more entire floors, at its sole cost and expense, may install identification signage anywhere in the Premises, including in the elevator lobby of such floor(s), provided that such signs must not be visible from the exterior of the Building.
- 23.3 <u>Multi-Tenant Floors</u>. If other tenants occupy space on the floor on which the Premises is located, Tenant's identifying signage shall be provided by Landlord, at Tenant's cost, and such signage shall be comparable to that used by Landlord for other similar floors in the Building and shall comply with Landlord's Building standard signage program.
- 23.4 <u>Prohibited Signage and Other Items.</u> Any signs, notices, logos, pictures, names or advertisements which are installed and that have not been separately approved by Landlord may be removed upon ten (10) Business Days notice by Landlord at the sole expense of Tenant. Tenant may not install any signs on the exterior or roof of the Project or the Common Areas, except as provided in this Section 23. Any signs, window coverings, or blinds (even if the same are located behind the Landlord-approved window coverings for the Building), or other items visible from the exterior of the Premises or Building, shall be subject to the prior approval of Landlord, in its reasonable discretion.
- 23.5 <u>Building Directory.</u>..If Landlord elects to construct a lobby directory board for the Building, Tenant shall be entitled to display on the Building's lobby directory board, at no additional cost to Tenant, its name and location in the Building.
- 23.6 Monument Signage. Tenant shall be entitled, at its sole cost and expense, to non-exclusive identification signage on the monument sign that Landlord may construct for the Project (the "Project Monument"), as well as non-exclusive identification signage on the monument sign that Landlord may construct for the Building (the "Building Monument"), and which shall be located in the Common Area as selected by Landlord (individually and collectively, the "Monument"). The space offered to Tenant for Tenant's signage on each Monument shall not be smaller than the space offered on such Monument to any other tenant at the Project. Landlord shall have the right, at Landlord's sole cost and expense, at any time to replace, refurbish, redesign, or relocate (collectively, "Change") the Monument, subject to the prior approval of Tenant, which approval shall not be unreasonably withheld, conditioned or delayed. All aspects of Tenant's signage for the Monument or Change monument, including, but not limited to, quality, design, style, lighting and size, as applicable, shall be (a) subject to Landlord's prior written approval, which approval shall not be unreasonably withheld, conditioned or delayed, and (b) in compliance with all applicable Laws. Tenant shall be responsible, at its sole cost and expense, for the maintenance, repair, and replacement of Tenant's signage on the Monument or Change monument. Landlord shall be responsible, at its sole cost and expense, for the maintenance and repair of the Monument or Change monument. Should the name of Tenant be legally changed to another name (the "New Name"), Tenant shall be entitled to modify, at Tenant's sole cost and expense, Tenant's name on the Monument to reflect Tenant's New Name, so long as Tenant's New Name is not an Objectionable Name. The term "Objectionable Name" shall mean any name that relates to an entity that is of a character or reputation, or is associated with a political orientation or faction, which is inconsistent with the quality of the Project, or which would otherwise reasonably offend a landlord of Comparable Buildings. Landlord agrees that the area granted to Tenant upon the Monument or Change monument shall be greater than or equal to the area granted to any other tenant upon the applicable Monument or Change monument. Upon the expiration or earlier termination of this Lease, or upon Tenant's loss of its Project Monument rights or Building Monument Rights as

hereinafter provided, Tenant shall promptly, at Tenant's sole cost and expense, remove Tenant's signage from the applicable Monument or Change monument and repair any damage resulting therefrom. Notwithstanding the foregoing, (A) if at any time during the Term (as the same may be extended), Tenant ceases to lease either (i) all of the rentable space in the Building, or (ii) an aggregate of at least 200,000 rentable square feet in the Project, Tenant's rights under this Section 23.1 with respect to the Project Monument shall expire and Tenant shall have no further Project Monument rights under this Lease; and (B) if at any time during the Term (as the same may be extended), Tenant ceases to lease at least 25% of all of the rentable space in the Building, Tenant's rights under this Section 23.1 with respect to the Building Monument shall expire and Tenant shall have no further Building Monument rights under this Lease.

23.7 <u>Building Top Signage</u>. Tenant shall be entitled, at its sole cost and expense, to exclusive identification signage on the top of the Building (the "<u>Building Top Signage</u>"). All aspects of Tenant's Building Top Signage, including, but not limited to, quality, design, style, lighting and size, as applicable, shall be (a) subject to Landlord's prior written approval, which shall not be unreasonably withheld, conditioned or delayed, and (b) in compliance with all applicable Laws. The Building Top Signage may be lighted and include other features as provided in **Exhibit I**. Tenant shall be responsible, at its sole cost and expense, for the construction, installation, maintenance, repair and replacement of Tenant's Building Top Signage (which shall be installed in accordance with Section 9.4). Should the name of Tenant be legally changed to a New Name, Tenant shall be entitled to modify, at Tenant's sole cost and expense, Tenant's name on the Building Top Signage to reflect Tenant's New Name, so long as Tenant's New Name is not an Objectionable Name. Upon the expiration or earlier termination of this Lease, Tenant shall, at Tenant's sole cost and expense, remove Tenant's Building Top Signage from the Building and repair any resulting damage. Tenant's signage rights contained in this Section 23.7 may only be exercised by Original Tenant (and any approved Transferee) and only if Original Tenant or such Transferee occupies at least fifty percent (50%) of the rentable square feet of the Building.

23.8 <u>Transferability.</u> Tenant's signage rights contained in this Section 23 may only be exercised by Original Tenant (and any approved Transferee) if Original Tenant or such Transferee occupies at least fifty percent (50%) of the Building, provided that any such transfer of Tenant's signage rights must be a transfer of all of the rights contained in Sections 23.6 and 23.7, together, and such rights may not be divided among entities; provided further, that such Transferee's signage may not contain any Objectionable Name; and provided further that the rights under Section 23.1 shall be restricted solely to Original Tenant and shall be subject to the additional restrictions of Section 23.1.

23.9 <u>Signage Maintenance</u>. Tenant's maintenance and use of the Building Top and Monument Signage as set forth in this Section 23 shall be subject to the terms of **Exhibit I**.

24. Parking. Tenant shall have the right to rent from Landlord, from time to time during the Term, commencing on the Commencement Date, the amount of parking passes (the "Parking Passes") determined as set forth in Section 1.20, on a monthly basis throughout the Term, which Parking Passes shall pertain to the parking facility located in the Building (the "Building Parking Facility") and the parking facility to be constructed at the Project as generally shown on the Site Plan (the "Project Parking Facility") in the following manner: Tenant shall be entitled to rent five (5) parking passes for every one thousand (1,000) rentable square feet of the Premises, a portion of which shall be allocated to all of the parking spaces available in the Building Parking Facility (not including the handicapped spaces or any spaces retained by Landlord for service vehicles), which parking spaces shall be reserved to Tenant (the "Reserved Parking Passes") and the remainder of which shall be allocated to the unreserved parking in the Project Parking Facility (the "Unreserved Parking Passes"). For purposes of this Lease, the Building Parking Facilities. As of the Commencement Date, Tenant has elected to rent all of the Unreserved Parking Passes and all of the Reserved Parking Passes. Tenant may change the number of parking passes rented pursuant to this Section 24 upon at least thirty (30) days prior written notice to Landlord, provided that in no event shall Tenant be entitled to rent more than the amount of parking passes determined pursuant to this Section 24 in the Building Parking Facility. Tenant shall pay to Landlord for the Parking Passes at the following rates:

Unreserved Parking Passes: Months 1-48 \$0

Months 49-60 \$50 per Parking Pass per month Months 61-120 \$65 per Parking Pass per month

Reserved Parking Passes: Months 1-48 \$0

Months 49-60 \$120 per Parking Pass per month Months 61-120 \$140 per Parking Pass per month

The rates for Parking Passes during any renewal Term shall be on a monthly basis at the prevailing rate charged by Landlord or the parking operator from time to time at the Project, except that such rates shall be not less than the rate charged in the last month of the initial Term (provided, however, that such parking charges shall be considered in determining the Fair Market Rental Rate for the extended Term(s)). Notwithstanding anything in this Lease to the contrary, if Tenant leases any additional space pursuant to any of the provisions of Section 3, then (i) if Rent for the additional space is determined at the Fair Market Rental Rate, then the parking ratios and parking rates for such space shall be determined as part of the fair market rate process, (ii) if Rent for the additional space is to be on the same terms as this Lease, then Tenant shall be entitled to the same parking pass ratio and parking rates as specified above. During the Term and any Option Term, Tenant shall have the right to purchase parking validation stickers (or Tenant may validate visitor parking by such other method or methods as Landlord may establish from time to time) at the prevailing validation rate charged by Landlord or the parking operator from time to time at the Project generally applicable to visitor parking.

In addition, Tenant shall be responsible for the full amount of any taxes imposed by any governmental authority in connection with the renting of such Parking Passes by Tenant or the use of the Parking Facilities by Tenant. Tenant shall comply, and shall cooperate with Landlord to attempt to require that Tenant Parties and any of Tenant's visitors, patrons, guests, invitees, licensees or subtenants comply, with the Parking Rules and Regulations set forth in **Exhibit B** and any reasonable modifications thereto which are prescribed by Landlord from time to time for the orderly operation and use of the parking facility. Landlord specifically reserves the right to change the size, configuration, design, layout and all other aspects of the Parking Facilities and any other parking areas at the Project and improvements (provided that Tenant's parking rights are not reduced as a result thereof and as long as Tenant's obligations are not materially or unreasonably increased as a result thereof and such change(s) do not create a material security risk for Tenant) at any time and Tenant acknowledges and agrees that Landlord may, without incurring any liability to Tenant and without any abatement of Rent under this Lease (except as specifically set forth in Section 17.6.2), from time to time, close-off or restrict access to the Parking Facilities and any other parking areas at the Project for purposes of permitting or facilitating any such construction, alteration or improvements, provided that such modifications may only be made by Landlord to the extent required by applicable Law, or as reasonably necessary on a temporary basis in the event of damage, destruction or new construction. Landlord shall use commercially reasonable efforts to cause any such work to be conducted in a manner that will minimize any inconvenience to Tenant and to provide alternative parking during such periods. Landlord may delegate its responsibilities hereunder to a parking operator in which case such parking operator shall have all the rights of control and obligations attributed hereby to the Landlord but such delegation shall not relieve Landlord of its liabilities hereunder. The Parking Passes rented by Tenant pursuant to this Section 24 are provided to Tenant solely for use by Tenant's own personnel and such Parking Passes may not be transferred, assigned, subleased or otherwise alienated by Tenant, except to any Transferee permitted by Section 12 of this Lease. Landlord shall provide handicapped parking in the Parking Facilities and any other parking areas at the Project separate and apart from the Parking Passes rented by Tenant pursuant to this Section 24.

25. Compliance With Laws. Tenant shall not do anything or suffer anything to be done in or about the Premises or the Project that will in any way conflict with any Laws, provided that no violation of Law regarding Tenant's employment practices shall be a Default under this Lease. At its sole cost and expense, Tenant shall promptly comply with all such governmental measures to the extent that such governmental measures relate to Tenant's use of the Premises, the Tenant Improvements located in the Premises, or any Alterations thereof, provided that Tenant shall have the right, to the extent allowed at law, to postpone such compliance during any period in which Tenant is contesting the validity or applicability of such Law, subject to reasonable bonding or other requirements or Landlord. Should any standard or regulation now or hereafter be imposed on Landlord or Tenant by a state, federal or local governmental body charged with the establishment, regulation and enforcement of occupational, health or safety standards for employers, employees, landlords or tenants, then Tenant agrees, at its sole cost and expense, to comply promptly with such standards or regulations to the extent such standards or regulations relate to Tenant's use of the Premises, the Tenant Improvements located in the Premises, or any Alterations thereof; provided that Landlord shall comply with any standards or regulations which relate to the Base Building or the Building Systems, unless such compliance obligations are triggered by the Tenant Improvements or Alterations in the Premises or required because of Tenant's use of the Premises for other than normal and customary business office operations, in which event such compliance obligations shall be at Tenant's sole cost and expense The judgment of any court of competent jurisdiction or the admission by either party hereto in any judicial action, regardless of whether this other party is a party thereto, that such party has violated any of said governmental measures, shall be conclusive of that fact as between Landlord and Tenant. Landlord shall comply with all Laws relating to the Project, Base Building and Building Systems, provided that compliance with such Laws is not the responsibility of Tenant under this Lease, and provided further that Landlord's failure to comply therewith would prohibit Tenant from obtaining or maintaining a certificate of occupancy for the Premises, or would unreasonably and materially affect the safety of Tenant's Parties or create a significant health hazard for Tenant's Parties or otherwise materially interfere with or materially affect Tenant's Permitted Use and enjoyment of the Premises. Landlord shall be permitted to include in Operating Expenses any costs or expenses incurred by Landlord under this Section 25 to the extent consistent with, and amortized to the extent required by, the provisions of Section 5.2.

26. Notices. If a demand, request, approval, consent or notice (collectively referred to as a "notice") shall or may be given to either party by the other, the notice shall be in writing and shall be (A) sent by United States certified or registered mail, postage prepaid, return receipt requested ("Mail"), (B) transmitted by telefacsimile, if such telefacsimile is promptly followed by a notice sent by Mail, (C) delivered by an overnight or same day courier, or (D) delivered by hand. Any Notice shall be sent, transmitted, or delivered, as the case may be, to the party's respective Notice Address(es) set forth in Section 1.1, except that if Tenant has vacated the Premises (or if the Notice Address for Tenant is other than the Premises, and Tenant has vacated such address) without providing Landlord a new Notice Address, Landlord may serve notice in any manner described in this Section or in any other manner permitted by Law. Any notice will be deemed to have been received or given (i) three (3) days after the date it is posted if sent by Mail, (ii) the date of transmission of the telex or fax transmission or other form of electronic transmission, if such transmission is prior to 3:00 p.m. in the recipient's time zone (if after 3:00 p.m., the notice shall be effective at 9:00 a.m. on the next business day after such transmission), or (iii) on the earlier to occur of actual delivery or the date on which delivery is refused, if sent by courier or by hand. Either party may, at any time, change its Notice Address by giving the other party written notice of the new address in the manner described in this Section. If Tenant is notified of the identity and address of Landlord's Lender, Tenant shall give to such Lender written notice of any default by Landlord under the terms of this Lease by registered or certified mail.

27. Miscellaneous.

27.1 <u>Governing Law</u>. This Lease and the rights and obligations of the parties shall be interpreted, construed and enforced in accordance with the Laws of the State of California and Landlord and Tenant hereby irrevocably consent to the jurisdiction and proper venue of such state. If any term or provision of this Lease shall to any extent be invalid or unenforceable, the remainder of this Lease shall not be affected, and each provision of this Lease shall be valid and enforced to the fullest extent permitted by Law. The headings and titles to the Sections of this Lease are for convenience only and shall have no effect on the interpretation of any part of the Lease.

- 27.2 <u>Recordation</u>. Upon Tenant's delivery of the first month's Base Rent and Security Deposit pursuant to this Lease, Landlord shall execute, in recordable form, and deliver to Tenant a Memorandum of Lease in the form of **Exhibit L**, and Tenant shall have the right to record such Memorandum at Tenant's expense in the Official Records of Orange County, California. Tenant shall, from time to time, execute and deliver to Landlord, within ten (10) days after Landlord's request therefor, such documents and instruments as Landlord may reasonably require to reflect changes in circumstances that may have occurred regarding the Lease (including, upon termination of this Lease, a quitclaim from Tenant releasing the Memorandum of Lease and quitclaiming all right, title and interest in and to the Premises and Building)
- 27.3 Attorneys' Fees. In the event of any arbitration or suit under this Lease, reasonable attorneys' fees and costs shall be awarded by a court or arbitrator to the prevailing party (as determined by the trier of fact) and are to be included in any judgment or award. In addition, the prevailing party shall be entitled to recover reasonable attorneys' fees and costs incurred in enforcing any judgment arising from a suit or arbitration under this Lease including but not limited to post judgment motions, contempt proceedings, garnishment, levy and debtor and third party examinations, discovery and bankruptcy litigation, without regard to schedule or rule of court purporting to restrict such award. This post judgment or award of attorneys' fees and costs provision shall be severable from any other provisions of this Lease and shall survive any judgment/award on such suit or arbitration and is not to be deemed merged into the judgment/award or terminated with the Lease. For the purpose of this provision, the term "attorneys' fees" or "attorneys' fees and costs" shall mean the fees and expenses of legal counsel (including external counsel and, to the extent fulfilling functions of an attorney at law, excluding de minimis administrative activities of, in-house counsel) to the parties hereto, which include printing, photocopying, duplicating, mail, overnight mail, messenger, court filing fees, cost of discovery, fees billed for law clerks, paralegals, investigators and other persons not admitted to the bar but performing services under the supervision or direction of an attorney. For the purpose of determining in-house counsel fees, the same shall be considered as those fees normally applicable to a partner in a law firm with like experience in such field.
- 27.4 Force Majeure. Whenever a period of time is prescribed for the taking of an action by Landlord or Tenant, the period of time for the performance of such action shall be extended by the number of days that the performance is actually delayed due to strikes, acts of God, shortages of labor or materials, war, civil disturbances and other causes beyond the reasonable control of the performing party ("Force Majeure Events"). However, Force Majeure Events shall not extend any period of time for the payment of Rent or other sums payable by either party or any period of time for the written exercise of an option or right by either party.
- 27.5 <u>Landlord Transfer</u>. Landlord shall have the right to transfer and assign, in whole or in part, all of its rights and obligations under this Lease and/or in the Building and/or Project referred to herein, and upon such transfer and if the transferee assumes the applicable obligations, Landlord shall automatically be released from all liability (to the extent such obligations are assumed by the transferee) under this Lease not accrued as of the date of the transfer, and Tenant agrees to look solely to such transferee for the performance of Landlord's obligations hereunder after the date of transfer, and Tenant shall attorn to such transferee.
- 27.6 Brokers. Landlord and Tenant hereby warrant to each other that they have had no dealings with any real estate broker or agent in connection with the negotiation of this Lease, excepting only the real estate brokers or agents specified in Section 1.16 (the "Brokers"), and that they know of no other real estate broker or agent who is entitled to a commission in connection with this Lease. Landlord shall pay all fees due the Brokers pursuant to a separate written agreements between Landlord and the Brokers (each, a "Written Agreement"). Each party agrees to indemnify and defend the other party against and hold the other party harmless from any and all claims, demands, losses, liabilities, lawsuits, judgments, costs and expenses (including without limitation reasonable attorneys' fees) with respect to any leasing commission or equivalent compensation alleged to be owing on account of any dealings with any real estate broker or agent, other than the Brokers, occurring by, through, or under the indemnifying party. The provisions of this Section 27.6 shall survive the expiration or earlier termination of the Term.
- 27.7 <u>Confidentiality</u>. Landlord and Tenant each hereby acknowledge that the contents of this Lease and any related documents are confidential information. Each of the parties shall keep such confidential information strictly confidential and shall not disclose such confidential information to any person or entity other than Landlord's or Tenant members, officers, directors, administrators, consultants, financial, legal, and space planning consultants a prospective or current purchaser, mortgagee, or ground or underlying lessor of the Building or the Project, a prospective Transferee, and except as required by Law or in connection with a dispute or litigation hereunder or as required by a subpoena. Any press releases or announcements made in connection with the contents of this Lease and/or any related documents shall be subject to the mutual approval of Landlord and Tenant.
- 27.8 <u>Hazardous Materials</u>. For purposes of this Lease, the following definitions shall apply: "<u>Hazardous Material(s)</u>" shall mean any substance or material that is described as a toxic or hazardous substance, waste, material, pollutant, contaminant or infectious waste, or any matter that in certain specified quantities would be injurious to the public health or welfare, or words of similar import, in any of the Environmental Laws (hereinafter defined), or any other words which are intended to define, list or classify substances by reason of deleterious properties such as ignitability, corrosivity, reactivity, carcinogenicity, toxicity or reproductive toxicity and includes, without limitation, asbestos, petroleum (including crude oil or any fraction thereof, natural gas, natural gas liquids, liquefied natural gas, or synthetic gas usable for fuel, or any mixture thereof), petroleum products, polychlorinated biphenyls, urea formaldehyde, radon gas, radioactive matter, medical waste, and chemicals which may cause cancer or reproductive toxicity. "Environmental Laws" shall mean all federal, state, local and quasi-governmental laws (whether under common law, statute or otherwise), ordinances, decrees, codes, rulings, awards, rules, regulations and guidance documents now or hereafter be enacted or promulgated as amended from time to time, in any way relating to or regulating Hazardous Materials. Landlord covenants that during the Term, Landlord shall comply with

all Environmental Laws in accordance with, and as required by, the terms of this Lease. Landlord shall indemnify, defend, protect and hold harmless the Tenant Parties from any and all claims of liability asserted against Tenant by a third party, including without limitation any agency or instrumentality of the federal, state or local government, for bodily injury, including death, physical damage to or loss of use of property or cleanup activities to the extent required by applicable law (remedial or removal), arising from any Hazardous Materials (collectively, "Hazardous Materials Claims") to the extent placed in, on, under or about the Project by Landlord or Landlord Parties or Landlord's contractors or consultants, or to the extent such materials existed in the Premises prior to the Commencement Date, which an applicable governmental authority, if it had knowledge of such materials prior to the Commencement Date and if such materials were not subject to a variance or a grandfathered/grandmothered code waiver exception, would have then deemed such materials to be Hazardous Materials pursuant to then-current applicable Law, including, but not limited to then-current Environmental Laws, in their form existing as of the Commencement Date. Tenant agrees to indemnify, defend, protect and hold harmless Landlord and the Landlord Parties from any and all Hazardous Materials Claims arising from any Hazardous Materials to the extent placed in, on, under or about the Premises or the Project by Tenant or Tenant Parties and any of Tenant's visitors, patrons, guests, invitees, licensees or subtenants.

- 27.9 <u>Communications and Computer Lines</u>. Tenant may install, maintain, replace, remove or use any electronic, phone, data and communication cabling and lines and related switches and transmission lines (collectively, "<u>Lines</u>") at the Project in or serving only the Premises or space leased by Tenant pursuant to Other Leases, provided that (i) Tenant shall obtain Landlord's prior written consent which shall be granted unless a Design Problem arises, use an experienced and qualified contractor approved in writing by Landlord, and comply with all of the other provisions of Section 9.4 of this Lease, (ii) an acceptable number of Lines and space for additional Lines shall be maintained for existing and future occupants of the Project, as determined in Landlord's reasonable opinion, (iii) the Lines therefor (including riser cables) shall be appropriately insulated to prevent excessive electromagnetic fields or radiation, and shall be surrounded by a protective conduit reasonably acceptable to Landlord, (iv) any new or existing Lines servicing the Premises shall comply with all applicable Laws, (v) as a condition to permitting the installation of new Lines, Landlord may require that Tenant remove existing Lines located in or serving the Premises and repair any damage in connection with such removal, and (vi) Tenant shall pay all costs in connection therewith. Landlord reserves the right to require that Tenant remove any Lines located in or serving the Premises which are installed in violation of these provisions, or which are at any time in violation of any Laws or represent a dangerous or potentially dangerous condition.
- 27.10 <u>Binding Effect</u>. Subject to all other provisions of this Lease, each of the covenants, conditions and provisions of this Lease shall extend to and shall, as the case may require, bind or inure to the benefit not only of Landlord and of Tenant, but also of their respective heirs, personal representatives, successors or assigns, provided this clause shall not permit any assignment by Tenant contrary to the provisions of Section 12 of this Lease.
- 27.11 No Air Rights. No rights to any view or to light or air over any property, whether belonging to Landlord or any other person, are granted to Tenant by this Lease. If at any time any windows of the Premises are temporarily darkened or the light or view therefrom is obstructed by reason of any repairs, improvements, maintenance or cleaning in or about the Project, the same shall be without liability to Landlord and without any reduction or diminution of Tenant's obligations under this Lease.
- 27.12 <u>Modification of Lease</u>. Should any current or prospective Lender for the Building or Project require a modification of this Lease, which modification will not cause an increased cost or expense to Tenant or in any other way materially and adversely change the rights or obligations of Tenant hereunder, then and in such event, Tenant agrees that this Lease may be so modified and agrees to execute whatever documents are reasonably required therefor and to deliver the same to Landlord within twenty (20) days following a request therefor. At the request of Landlord or any Lender, Tenant agrees to execute a short form of Lease and deliver the same to Landlord within twenty (20) days following the request therefor.
- 27.13 <u>Relationship of Parties</u>. Nothing contained in this Lease shall be deemed or construed by the parties hereto or by any third party to create the relationship of principal and agent, partnership, joint venturer or any association between Landlord and Tenant.
- 27.14 <u>Application of Payments</u>. Unless Tenant, upon written notice to Landlord, specifically restricts how such payment is to be applied, Landlord shall have the right to apply payments received from Tenant pursuant to this Lease, regardless of Tenant's designation of such payments, to satisfy any obligations of Tenant hereunder, in such order and amounts as Landlord, in its sole discretion, may elect. Notwithstanding the foregoing, however, while Landlord must accept payments made under protest, Landlord is not obligated to accept a payment made with other restrictions, and such payment, if not accepted by Landlord, shall be returned to Tenant within five (5) Business Days of demand by Tenant.
- 27.15 <u>Time of Essence</u>. Time is of the essence with respect to the performance of every provision of this Lease in which time of performance is a factor. Whenever in the Lease a payment is required to be made by one party to the other, but a specific date for payment is not set forth or a specific number of days within which payment is to be made is not set forth, or the words "immediately," "promptly," and/or "on demand," or their equivalent, are used to specify when such payment is due, then such payment shall be due fifteen (15) days after the date that the party which is entitled to such payment sends written notice to the other party demanding such payment.
- 27.16 <u>Right to Lease</u>. Landlord reserves the absolute right to effect such other tenancies in the Project as Landlord in the exercise of its sole business judgment shall determine to best promote the interests of the Project, and Landlord may seek to lease space in the Project to a prospective Transferee of Tenant even if Tenant is then negotiating with such Transferee. Tenant does not rely on the fact, nor does Landlord represent, that any specific tenant or type or number of tenants shall, during the Lease Term, occupy any space in the Building or Project.

27.17 Intentionally Omitted.

- 27.18 <u>Counterparts</u>. This Lease may be executed in counterparts with the same effect as if both parties hereto had executed the same document. Both counterparts shall be construed together and shall constitute a single lease.
- 27.19 <u>Transportation Management</u>. Tenant shall fully comply with all present or future governmentally mandated programs intended to manage parking, transportation or traffic in and around the Project, and in connection therewith, Tenant shall take responsible action for the transportation planning and management of all employees located at the Premises by working directly with Landlord, any governmental transportation management organization or any other transportation-related committees or entities.
- 27.20 Building Renovations. It is specifically understood and agreed that Landlord has made no representation or warranty to Tenant and has no obligation and has made no promises to alter, remodel, improve, renovate, repair or decorate the Premises, the Building, or any part thereof and that no representations respecting the condition of the Premises, the Building have been made by Landlord to Tenant except as specifically set forth herein or in the Work Letter. However, Tenant hereby acknowledges that Landlord may during the Lease Term renovate, improve, alter, or modify (collectively, the "Renovations") the Project, the Building and/or the Premises including without limitation the Parking Facilities, Common Areas, systems and equipment, roof, and structural portions of the same, which Renovations may include, without limitation, (i) modifying the Common Areas and tenant spaces to comply with applicable Laws, including regulations relating to the physically disabled, seismic conditions, and building safety and security, and (ii) installing new floor covering, lighting, and wall coverings in any Building common areas, and in connection with any Renovations, Landlord may, among other things, erect scaffolding or other necessary structures in the Building, limit or eliminate access to portions of the Project, including portions of the Common Areas, or perform work in the Building, which work may create noise, dust or leave debris in the Building. Tenant hereby agrees that such Renovations and Landlord's actions in connection with such Renovations shall in no way constitute a constructive eviction of Tenant nor entitle Tenant to any abatement of Rent; provided, however, that nothing in this Section 27.20 shall impair Tenant's rights under Section 17.6.2. Landlord shall have no responsibility or for any reason be liable to Tenant for any direct or indirect injury to or interference with Tenant's business arising from the Renovations, nor shall Tenant be entitled to any compensation or damages from Landlord for loss of the use of the whole or any part of the Premises or of Tenant's personal property or improvements resulting from the Renovations or Landlord's actions in connection with such Renovations, or for any inconvenience or annoyance occasioned by such Renovations or Landlord's actions. Notwithstanding the foregoing, Landlord shall perform all Renovations in a commercially reasonable manner and whenever appropriate, after Building Hours, and in a manner, whenever reasonably possible, to minimize any material, adverse or unreasonable interference with Tenant's business operations and Permitted Uses and, whenever reasonably possible, Landlord shall provide Tenant with reasonable prior notice of such Renovations. If any Renovations will materially affect Tenant's use of the Premises, such Renovation shall only be performed in accordance with a schedule mutually agreed upon by Landlord and Tenant.
- 27.21 No Violation. Tenant and Landlord hereby respectively warrant and represent to the other that neither its execution of nor performance under this Lease shall cause Tenant or Landlord, as appropriate, to be in violation of any agreement, instrument, contract, law, rule or regulation by which Tenant or Landlord, as appropriate, is bound, and Tenant or Landlord, as appropriate, shall protect, defend, indemnify and hold the other harmless against any claims, demands, losses, damages, liabilities, costs and expenses, including, without limitation, reasonable attorneys' fees and costs, arising from Tenant's or Landlord's, as appropriate, breach of this warranty and representation.
- 27.22 <u>Calendar Days</u>. All references made in this Lease to the word "days," whether for Notices, schedules or other miscellaneous time limits, shall at all times herein be deemed to mean calendar days, unless specifically references as "Business Days."
- 27.23 <u>Assistance in Litigation</u>. Upon written request from the other party, Landlord or Tenant, as the case may be, shall disclose to the requesting party any cause of action arising under this Lease which is likely to have a material affect upon the requesting party. Such disclosure shall contain the name and address of the claimant, place and circumstances of the alleged incident and the addresses and identities of any known witnesses.
- 27.24 <u>Survival of Provisions Upon Termination of Lease</u>. Any term, covenant or condition of this Lease that requires the performance of obligations or forbearance of an act by either party hereto after the termination of this Lease shall survive such termination of this Lease. Such survival shall be to the extent reasonably necessary to fulfill the intent thereof, or if specified, to the extent of such specification, as the same is reasonably necessary to perform the obligations and/or forbearance of an act set forth in such term, covenant or condition. Notwithstanding the foregoing in the event a specific term, covenant or condition is expressly provided for in such a clear fashion as to indicate that such performance of an obligation or forbearance of an act is no longer required, then the specific shall govern over this general provision of this Lease.
- 27.25 <u>Financial Statements</u>. Tenant, within 15 days after request, shall provide Landlord with Tenant's and Guarantor's current financial statement (which may be consolidated) and such other information as Landlord may reasonably request. Landlord, however, shall not require Tenant to provide such information unless Landlord is requested to produce the information in connection with a proposed financing or sale of the Building or unless required by Landlord's Lender. Upon written request by Tenant, Landlord shall enter into a commercially reasonable confidentiality agreement covering any confidential information that is disclosed by Tenant. So long as Guarantor is a publicly traded company, Guarantor's most recent 10-K and 10-Q reports filed with the Securities and Exchange Commission, consolidated with Tenant, shall serve to satisfy the requirement for a current financial statement from Tenant and Guarantor.
- 27.26 <u>Independent Covenants</u>. This Lease shall be construed as though the covenants herein between Landlord and Tenant are independent and not dependent and Tenant hereby expressly waives the benefit of any statute to the

contrary and agrees that if Landlord fails to perform its obligations set forth herein, Tenant shall not be entitled to make any repairs or perform any acts hereunder at Landlord's expense or to any setoff of the Rent or other amounts owing hereunder against Landlord except as specifically set forth in this Lease.

27.27 Quiet Enjoyment. Tenant shall, and may peacefully have, hold and enjoy the Premises, subject to the terms of this Lease, provided Tenant pays the Rent and fully performs all of its covenants and agreements under this Lease. This covenant and all other covenants of Landlord shall be binding upon Landlord and its successors only during its or their respective periods of ownership of the Building, and shall not be a personal covenant of Landlord or the Landlord Parties.

27.28 Arbitration

27.28.1. General Submittals to Arbitration. The submittal of all matters to arbitration in accordance with the provisions of this Section 27.28 is the sole and exclusive method, means and procedure to resolve any and all claims, disputes or disagreements arising under this Lease, including, but not limited to any matter relating to Landlord's failure to approve a Transfer under Section 12 of this Lease, any other defaults by Landlord, or any Tenant Default, except for (i) all claims by either party which (A) seek anything other than enforcement of rights under this Lease, or (B) are primarily founded upon matters of fraud, willful misconduct, bad faith or any other allegations of tortious action, and seek the award of punitive or exemplary damages, (ii) claims relating to Landlord's exercise of any unlawful detainer rights pursuant to California law or rights or remedies used by Landlord to gain possession of the Premises or terminate Tenant's right of possession to the Premises, which disputes shall be resolved by suit filed in the Superior Court of Orange County, California, the decision of which court shall be subject to appeal pursuant to applicable Law, or (iii) claims by either party for injunctive relief. The parties hereby irrevocably waive any and all rights to the contrary and shall at all times conduct themselves in strict, full, complete and timely accordance with the provisions of this Section 27.28 and all attempts to circumvent the provisions of this Section 27.28 shall be absolutely null and void and of no force or effect whatsoever. As to any matter submitted to arbitration (except with respect to the payment of money) to determine whether a matter would, with the passage of time, constitute a Default, such passage of time shall not commence to run until any such affirmative arbitrated determination, as long as it is simultaneously determined in such arbitration that the challenge of such matter as a potential Tenant Default or Landlord default was made in good faith. As to any matter submitted to arbitration with respect to the payment of money, to determine whether a matter would, with the passage of time, constitute a Default, such passage of time shall not commence to run in the event that the party that is obligated to make the payment does in fact make the payment to the other party. Such payment can be made "under protest," which shall occur when such payment is accompanied by a good faith notice stating the reasons that the party has elected to make a payment under protest. Such protest will be deemed waived unless the subject matter identified in the protest is submitted to arbitration as set forth in this Section 27.28.

27.28.2. <u>JAMS</u>. Any dispute to be arbitrated pursuant to the provisions of this Section 27.28 shall be determined by binding arbitration before a retired judge of the Superior Court of the State of California (the "<u>Arbitrator</u>") under the auspices of Judicial Arbitration & Mediation Services, Inc. ("<u>JAMS</u>"). Such arbitration shall be initiated by the parties, or either of them, within ten (10) days after either party sends written notice (the "<u>Arbitration Notice</u>") of a demand to arbitrate to the other party and to JAMS. The Arbitration Notice shall contain a description of the subject matter of the arbitration, the dispute with respect thereto, the amount involved, if any, and the remedy or determination sought. The parties may agree on a retired judge from the JAMS panel. If they are unable to promptly agree, JAMS will provide a list of three available judges and each party may strike one. The remaining judge (or if there are two, the one selected by JAMS) will serve as the Arbitrator. In the event that JAMS shall no longer exist or if JAMS fails or refuses to accept submission of such dispute, then the dispute shall be resolved by binding arbitration before the American Arbitration Association ("<u>AAA</u>") under the AAA's commercial arbitration rules then in effect. Each party shall split the costs and expenses of the Arbitration until such time as an Arbitration Award is made, at which time all or any portion of the Arbitrator's fees and costs shall be paid by the non-prevailing party as determined by the Arbitrator in his discretion, as further provided in Section 27.28.3(b) below.

27.28.3. Arbitration Procedure.

- (a) <u>Pre-Decision Actions</u>. The Arbitrator shall schedule a pre-hearing conference to resolve procedural matters, arrange for the exchange of information, obtain stipulations, and narrow the issues. The parties will submit proposed discovery schedules to the Arbitrator at the pre-hearing conference. The scope and duration of discovery will be within the sole discretion of the Arbitrator. The Arbitrator shall have the discretion to order a pre-hearing exchange of information by the parties, including, without limitation, production of requested documents, exchange of summaries of testimony of proposed witnesses, and examination by deposition of parties and third-party witnesses. This discretion shall be exercised in favor of discovery reasonable under the circumstances.
- (b) The Decision. The arbitration shall be conducted in Orange County, California. Any party may be represented by counsel or other authorized representative. In rendering a decision(s), the Arbitrator shall determine the rights and obligations of the parties according to the substantive and procedural laws of the State of California and the provisions of this Lease. The Arbitrator's decision shall be based on the evidence introduced at the hearing, including all logical and reasonable inferences therefrom. The Arbitrator may make any determination, and/or grant any remedy or relief (an "Arbitration Award") that is just and equitable. The decision must be based on, and accompanied by, a written statement of decision explaining the factual and legal basis for the decision as to each of the principal controverted issues. The decision shall be conclusive and binding, and it may thereafter be confirmed as a judgment by the Superior Court of the State of California, subject only to challenge on the grounds set forth in the California Code of Civil Procedure Section 1286.2. The validity and enforceability of the Arbitrator's decision is to be determined exclusively by the California courts pursuant to the provisions of this Lease. The Arbitrator shall award costs, including without limitation attorneys' fees, and expert and witness costs, to the

prevailing party as defined in California Code of Civil Procedure Section 1032, if any, as determined by the Arbitrator in his discretion. All or any portion of the Arbitrator's fees and costs shall be paid by the non-prevailing party as determined by the Arbitrator in his discretion. A party shall be determined by the Arbitrator to be the prevailing party if its proposal for the resolution of dispute is the closer to that adopted by the Arbitrator.

- 27.29 Entire Agreement. This Lease and the following exhibits and attachments constitute the entire agreement between the parties and supersede all prior agreements and understandings related to the Premises, including all lease proposals, letters of intent and other documents. Neither party is relying on any warranty, statement or representation not contained in this Lease.
- 27.30 Modification. This Lease may be modified only by a written agreement signed by Landlord and Tenant.
- 27.31 No Offer to Lease. Landlord has delivered a copy of this Lease to Tenant for Tenant's review only, and the delivery of it does not constitute an offer to Tenant or an option. This Lease shall not be effective against any party hereto until an original copy of this Lease has been signed and delivered by such party.
- 27.32 <u>Anti-Terrorism Representations</u>. Tenant is not, and shall not during the Term of this Lease become, a person or entity with whom Landlord is restricted from doing business with under the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001. H.R. 3162, Public Law 107-56 (commonly known as the "<u>USA Patriot Act</u>") and Executive Order Number 13224 on Terrorism Financing, effective September 24, 2001 and regulations promulgated pursuant thereto (collectively, "<u>Anti-Terrorism Laws</u>"), including without limitation persons and entities named on the Office of Foreign Asset Control Specially Designated Nationals and Blocked Persons List (collectively "<u>Prohibited Persons</u>"). Tenant will not during the Term of this Lease knowingly engage in any transactions or dealings, or be otherwise associated with, any Prohibited Persons in connection with the use or occupancy of the Premises. A breach of these representations by Tenant shall constitute a material breach of this Lease and shall entitle Landlord to any and all remedies available hereunder, or at law or in equity.
- 27.33 <u>Guaranty of Lease</u>. As a material inducement to Landlord to enter into this Lease, Tenant shall cause to be delivered to Landlord (and it shall be a condition precedent to the effectiveness of this Lease, for the benefit of Landlord, that Landlord be delivered) concurrently with the execution and delivery of this Lease, a fully executed Guaranty of Lease from the Guarantor set forth in Section 1.21, which Guaranty of Lease shall be in the form of **Exhibit J**.

[No Further Text On This Page]

Landlord and Tenant have executed this Lease as of the day and year first above written.

LANDLORD:

SCHOLLE JAMBOREE PROPERTY DEVELOPMENT I LLC, a California limited liability company

By: /s/ Robert W. Scholle

Name: Robert W. Scholle

Title: Manager

TENANT:

IMPAC FUNDING CORPORATION, a California corporation

By: /s/ R. Morrison

Name: R. Morrison Title: Exec. V.P.

*NOTE:

Concurrently with the execution of this Lease, each party shall deliver to the other a certified copy of a corporate resolution in a form reasonably acceptable to the other party authorizing the signatory(ies) to execute this Lease.

EXHIBIT A
SITE PLAN

[image]

EXHIBIT A

-1-

<u>EXHIBIT B</u> RULES AND REGULATIONS

Subject to the terms and conditions of the Lease, Tenant shall faithfully observe and comply with the following Rules and Regulations. Landlord shall not be responsible to Tenant for the nonperformance of any of said Rules and Regulations by or otherwise with respect to the acts or omissions of any other tenants or occupants of the Building.

Building Rules and Regulations

- 1. Tenant shall not alter any lock or install any new or additional locks or bolts on any doors or windows of the Premises without obtaining Landlord's prior written consent. All re-keying or new locks or bolts must be done by Landlord's locksmith and must be consistent with the master keying system for the Building. Tenant shall bear the cost of any lock changes or repairs required by Tenant. Two keys or electronic key cards will be furnished by Landlord for the Premises, and any additional keys or electronic key cards required by Tenant must be obtained from Landlord at a reasonable cost to be established by Landlord; provided, however, that Landlord shall provide Tenant with a \$5,000 credit to be used toward the cost of such additional keys or electronic key cards required by Tenant.
- 2. All doors opening to public corridors shall be kept closed at all times except for normal ingress and egress to the Premises, unless electrical hold backs have been installed.
- 3. Landlord reserves the right to close and keep locked all entrance and exit doors of the Building during such hours as are customary for Comparable Buildings. Tenant, its employees and agents must be sure that the doors to the Building are securely closed and locked when leaving the Premises if it is after the normal hours of business for the Building. Any tenant, its employees, agents or any other persons entering or leaving the Building at any time when it is so locked, or any time when it is considered to be after normal business hours for the Building, may be required to sign the Building register when so doing. Access to the Building may be refused unless the person seeking access has proper identification or has a previously arranged pass for access to the Building. The Landlord and his agents shall in no case be liable for damages for any error with regard to the admission to or exclusion from the Building of any person. In case of invasion, mob, riot, public excitement, or other commotion, Landlord reserves the right to prevent access to the Building during the continuance of same by any means it deems appropriate for the safety and protection of life and property.
- 4. Landlord shall have the right to prescribe the weight, size and position of all safes and other heavy property brought into the Building. Safes and other heavy objects shall, if considered necessary by Landlord, stand on supports of such thickness as is necessary to properly distribute the weight. Landlord will not be responsible for loss of or damage to any such safe or property in any case. All damage done to any part of the Building, its contents, occupants or visitors by moving or maintaining any such safe or other property shall be the sole responsibility of Tenant and any expense of said damage or injury shall be borne by Tenant.
- 5. No furniture, freight, packages, supplies, equipment or merchandise will be brought into or removed from the Building or carried up or down in the elevators, except upon prior notice to Landlord, and in such manner, in such specific elevator, and between such hours as shall be designated by Landlord. Tenant shall provide Landlord with not less than 24 hours prior notice of the need to utilize an elevator for any such purpose, so as to provide Landlord with a reasonable period to schedule such use and to install such padding or take such other actions or prescribe such procedures as are appropriate to protect against damage to the elevators or other parts of the Building. If the Building is a multi-tenant building, in no event shall Tenant's use of the elevators for any such purpose be permitted during the hours of 7:00 a.m. 9:00 a.m., 11:30 a.m. 1:30 p.m. and 4:30 p.m. 6:30 p.m.
- 6. Landlord shall have the right to control and operate the public portions of the Building, the public facilities, the heating and air conditioning, and any other facilities furnished for the common use of tenants, in such manner as is customary for Comparable Buildings.
- 7. The requirements of Tenant will be attended to only upon application at the office location designated by Landlord. Employees of Landlord shall not perform any work or do anything outside their regular duties unless under special instructions from Landlord.
 - 8. Tenant shall not disturb, solicit, or canvass any occupant of the Building and shall cooperate with Landlord or Landlord's agents to prevent same.
- 9. The toilet rooms, urinals, wash bowls and other apparatus shall not be used for any purpose other than that for which they were constructed, and no foreign substance of any kind whatsoever shall be thrown therein. The expense of any breakage, stoppage or damage resulting from the violation of this rule shall be borne by the tenant who, or whose employees or agents, shall have caused it.
- 10. Tenant shall not overload the floor of the Premises, nor mark, drive nails or screws, or drill into the partitions, woodwork or plaster or in any way deface the Premises or any part thereof without Landlord's consent first had and obtained.

- 11. Except for vending machines intended for the sole use of Tenant's employees and invitees, no vending machine or machines of any description other than fractional horsepower office machines shall be installed, maintained or operated upon the Premises without the written consent of Landlord.
 - 12. Tenant shall not use or keep in or on the Premises or the Building any kerosene, gasoline or other inflammable or combustible fluid or material.
- 13. Tenant shall not use any method of heating or air conditioning other than that which may be supplied by Landlord, without the prior written consent of Landlord.
- 14. Tenant shall not use, keep or permit to be used or kept, any foul or noxious gas or substance in or on the Premises, or permit or allow the Premises to be occupied or used in a manner offensive or objectionable to Landlord or other occupants of the Building by reason of noise, odors, or vibrations, or interfere in any way with other Tenants or those having business therein.
 - 15. Tenant shall not bring into or keep within the Building or the Premises any animals, birds, bicycles or other vehicles.
- 16. No cooking shall be done or permitted by any tenant on the Premises, nor shall the Premises be used for the storage of merchandise, for lodging or for any improper, objectionable or immoral purposes. Notwithstanding the foregoing, Underwriters' laboratory-approved equipment and microwave ovens may be used in the Premises for heating food and brewing coffee, tea, hot chocolate and similar beverages, provided that such use is in accordance with all applicable federal, state and city laws, codes, ordinances, rules and regulations, and does not cause odors which are objectionable to Landlord and other Tenants.
- 17. Landlord will approve where and how telephone and telegraph wires are to be introduced to the Premises. No boring or cutting for wires shall be allowed without the consent of Landlord. The location of telephone, call boxes and other office equipment affixed to the Premises shall be subject to the approval of Landlord.
- 18. Landlord reserves the right to exclude or expel from the Building any person who, in the judgment of Landlord, is intoxicated or under the influence of liquor or drugs, or who shall in any manner do any act in violation of any of these Rules and Regulations.
- 19. Tenant, its employees and agents shall not loiter in the entrances or corridors, nor in any way obstruct the sidewalks, lobby, halls, stairways or elevators, and shall use the same only as a means of ingress and egress for the Premises.
- 20. Tenant shall not waste electricity, water or air conditioning and agrees to cooperate fully with Landlord to ensure the most effective operation of the Building's heating and air conditioning system, and shall refrain from attempting to adjust any controls. This includes the closing of exterior blinds, disallowing the sun rays to shine directly into areas adjacent to exterior windows.
- 21. Tenant shall store all its trash and garbage within the interior of the Premises. No material shall be placed in the trash boxes or receptacles if such material is of such nature that it may not be disposed of in the ordinary and customary manner of removing and disposing of trash and garbage in the city in which the Building is located without violation of any law or ordinance governing such disposal. All trash, garbage and refuse disposal shall be made only through entry-ways and elevators provided for such purposes at such times as Landlord shall designate.
 - 22. Tenant shall cooperate with Landlord's trash recycling programs and the orderly sorting of trash materials to facilitate such programs.
 - 23. Tenant shall comply with all safety, fire protection and evacuation procedures and regulations established by Landlord or any governmental agency.
- 24. Tenant shall assume any and all responsibility for protecting the Premises from theft, robbery and pilferage, which includes keeping doors locked and other means of entry to the Premises closed, when the Premises are not occupied.
- 25. Landlord may waive any one or more of these Rules and Regulations for the benefit of any particular tenant or tenants, but no such waiver by Landlord shall be construed as a waiver of such Rules and Regulations in favor of any other tenant or tenants, nor prevent Landlord from thereafter enforcing any such Rules or Regulations against any or all tenants of the Building.
- 26. No awnings or other projection shall be attached to the outside walls of the Building without the prior written consent of Landlord. No curtains, blinds, shades or screens shall be attached to or hung in, or used in connection with, any window or door of the Premises without the prior written consent of Landlord. All electrical ceiling fixtures hung in offices or spaces along the perimeter of the Building must be fluorescent and/or of a quality, type, design and bulb color approved by Landlord.
- 27. The sashes, sash doors, skylights, windows, and doors that reflect or admit light and air into the halls, passageways or other public places in the Building shall not be covered or obstructed by Tenant, nor shall any bottles, parcels or other Sections be placed on the windowsills.

- 28. The washing and/or detailing of or, the installation of windshields, radios, telephones in or general work on, automobiles shall not be allowed on the Project, except by concessionaires of Landlord.
- 29. Food vendors appropriately licensed by the appropriate authorities shall be allowed in the Building upon twenty-four (24) hour advance receipt of a written request from the Tenant. The food vendor shall service only the tenants that have a written request on file in the Project management office. Under no circumstance shall the food vendor display their products in a public or common area including corridors and elevator lobbies. Any failure to comply with this rule shall result in immediate permanent withdrawal of the vendor from the Building.
 - 30. Tenant must comply with requests by the Landlord concerning the informing of their employees of items of importance to the Landlord.
- 31. Tenant shall comply with any non-smoking ordinance adopted by any applicable governmental authority. There is no smoking permitted in any of the buildings comprising the Project. In addition, Landlord reserves the right to designate, in Landlord's sole discretion, the only outside areas in the Project where smoking shall be permitted.
- 32. No antenna, aerial, discs, dishes or other such device shall be erected or placed on the roof of the Building or anywhere else outside the Premises, without the written consent of the Landlord in each instance. Any device so installed without such written consent shall be subject to removal without notice at any time.
- 33. Landlord reserves the right at any time to change or rescind any one or more of these Rules and Regulations, or to make such other and further reasonable Rules and Regulations as in Landlord's judgment may from time to time be necessary for the management, safety, care and cleanliness of the Premises and Building, and for the preservation of good order therein, as well as for the convenience of other occupants and tenants therein. Landlord shall not be responsible to Tenant or to any other person for the nonobservance of the Rules and Regulations by another tenant or other person. Tenant shall be deemed to have read these Rules and Regulations and to have agreed to abide by them as a condition of its occupancy of the Premises.

Parking Rules and Regulations

- 1. Tenant shall at all times comply with all applicable Laws respecting the use of the Parking Facilities. Landlord may refuse to permit any person who violates such rules to park in the Parking Facilities, and any violation of the rules shall subject the car to removal from the Parking Facilities.
- 2. Unless specified to the contrary in the Lease, the parking spaces hereunder shall be provided on a non-designated "first-come, first-served" basis. Tenant acknowledges that Landlord has no liability for claims arising through acts or omissions of any independent operator of the Parking Facilities. Landlord shall have no liability whatsoever for any damage to items located in the Parking Facilities, nor for any personal injuries or death arising out of any matter relating to the Parking Facilities, and in all events, Tenant agrees to look first to its insurance carrier and to require that Tenant's employees look first to their respective insurance carriers for payment of any losses sustained in connection with any use of the Parking Facilities. Tenant hereby waives on behalf of its insurance carriers all rights of subrogation against Landlord or Landlord's agents. Landlord reserves the right to assign specific parking spaces, and to reserve parking spaces for visitors, small cars, handicapped persons and for other tenants, guests of tenants or other parties, which assignment and reservation or spaces may be relocated as determined by Landlord from time to time, and Tenant and persons designated by Tenant hereunder shall not park in any location designated for such assigned or reserved parking spaces. Tenant acknowledges that the Parking Facilities may be closed entirely or in part in order to make repairs or perform maintenance services, or to alter, modify, re-stripe or renovate the Parking Facilities, or if required by casualty, strike, condemnation, act of God, governmental law or requirement or other reason beyond the operator's reasonable control. In such event, Landlord shall refund any prepaid parking rent hereunder, prorated on a per diem basis.
- 3. If Tenant shall breach any of the Parking Rules and Regulations, the operator shall have the right to remove from the Parking Facilities any vehicles hereunder which shall have been involved or shall have been owned or driven by parties involved in causing such default, without liability therefor whatsoever.
- 4. Parking Facilities hours shall be 7:00 a.m. to 8:00 p.m. Tenant shall not store or permit its employees to store any automobiles in the Parking Facilities without the prior written consent of the operator. Except for emergency repairs, Tenant and its employees shall not perform any work on any automobiles while located in the Parking Facilities, or on the Project. If it is necessary for Tenant or its employees to leave an automobile in the Parking Facilities overnight, Tenant shall provide the operator with prior notice thereof designating the license plate number and model of such automobile.
- 5. Cars must be parked entirely within the stall lines painted on the floor, and only small cars may be parked in areas reserved for small cars.
- 6. All directional signs and arrows must be observed.
- 7. The speed limit shall be 5 miles per hour.
- 8. Parking spaces reserved for spaces for handicapped persons must be used only by vehicles properly designated.

- 9. Parking is prohibited in all areas not expressly designated for parking, including without limitation: (a) areas not striped for parking, (b) aisles, (c) where "no parking" signs are posted, (d) ramps, loading zones.
- 10. Parking stickers, key cards or any other devices or forms of identification or entry supplied by the operator shall remain the property of the operator. Such device must be displayed as requested and may not be mutilated in any manner. The serial number of the parking identification device may not be obliterated. Parking spaces and devices are not transferable and any pass or device in the possession of an unauthorized holder will be void.
- 11. Monthly fees shall be payable in advance prior to the first day of each month. Failure to do so will automatically cancel parking privileges and a charge at the prevailing daily parking rate will be due. No deductions or allowances from the monthly rate will be made for days on which the Parking Facilities is not used by Tenant or its designees.
- 12. Parking Facilities managers or attendants are not authorized to make or allow any exceptions to these Rules.
- 13. Every parker is required to park and lock his/her own car.
- 14. Loss or theft of parking pass, identification, key cards or other such devices must be reported to Landlord and to the Parking Facilities manager immediately. Any parking devices reported lost or stolen found on any authorized car will be confiscated and the illegal holder will be subject to prosecution. Lost or stolen spaces and devices found by Tenant or its employees must be reported to the office of the garage immediately.
- 15. Washing, waxing, cleaning or servicing of any vehicle by the customer and/or his agents is prohibited except by Landlord or Landlord's vendors or concessionaires. Parking spaces may be used only for parking automobiles.
- 16. Tenant agrees to acquaint all persons to whom Tenant assigns a Parking Pass with these Rules.
- 17. TENANT ACKNOWLEDGES AND AGREES THAT, TO THE FULLEST EXTENT PERMITTED BY LAW, LANDLORD SHALL NOT BE RESPONSIBLE FOR ANY LOSS OR DAMAGE TO TENANT OR TENANT'S PROPERTY (INCLUDING, WITHOUT LIMITATIONS, ANY LOSS OR DAMAGE TO TENANT'S AUTOMOBILE OR THE CONTENTS THEREOF DUE TO THEFT, VANDALISM OR ACCIDENT) ARISING FROM OR RELATED TO TENANT'S USE OF THE PARKING FACILITY OR EXERCISE OF ANY RIGHTS UNDER THIS PARKING AGREEMENT, WHETHER OR NOT SUCH LOSS OR DAMAGE RESULTS FROM LANDLORD'S ACTIVE NEGLIGENCE OR NEGLIGENT OMISSION. THE LIMITATION ON LANDLORD'S LIABILITY UNDER THE PRECEDING SENTENCE SHALL NOT APPLY HOWEVER TO LOSS OR DAMAGE ARISING DIRECTLY FROM LANDLORD'S WILLFUL MISCONDUCT.

EXHIBIT B

EXHIBIT C COMMENCEMENT LETTER (EXAMPLE)

Date: _____

Tenant Addre	SS:			
Re:	Commencement Letter with respect to that certain, as Landlord, and	Lease dated as of the	day of, , as Tenant, for	, by and between, a
	, as Landlord, and (usable square feet) on the	floor of the Building located	l at	
Dear	:			
In accordance	with the terms and conditions of the above referenced	Lease, Tenant accepts possession	of the Premises and ag	rees:
1.	The Commencement Date of the Lease is	;		
2.	The Termination Date of the Lease is	; and		
3.	The Premises contain rentable square feet.			
	wledge your acceptance of possession and agreement to www.and.returning 2 fully executed originals to my attent		ing 3 originals of this C	Commencement Letter in the space
Sincerely,				
Agreed and A	accepted:			
Tenant:				
By:				
Name:				
Title:				
Date:				

EXHIBIT C

EXHIBIT D WORK LETTER

This Work Letter (this "Work Letter") sets forth the terms and conditions relating to the planning, construction and completion of Landlord's Work at the Project and the Tenant Improvements at the Premises. All references in this Work Letter to "the Lease" or Sections of "the Lease" shall mean the relevant portions of the body of the Lease to which this Work Letter is attached as Exhibit D and of which this Work Letter forms a part, and all references in this Work Letter to Sections of "this Work Letter" shall mean the relevant portion of this Work Letter. Capitalized terms used and not otherwise defined herein shall have the meaning given in the Lease.

SECTION 1 LANDLORD'S WORK

- 1.1 <u>Landlord's Work</u>. Pursuant to this Work Letter, Landlord shall cause to be constructed at the Project, at Landlord's sole cost and expense, the following (collectively, "<u>Landlord's Work</u>"):
 - (A) the "Base Building Improvements", which for purposes hereof means the Building Structure and Building Systems of Building II (a seven-story office building containing approximately 200,000 rentable square feet), as more particularly described in **Schedule 1** attached hereto. The Base Building Improvements shall be constructed by Landlord at the Project in the "Building II" area generally depicted on the Site Plan and in substantial accordance with the Base Building Plans (hereinafter defined) prepared by architect(s)/engineer(s) selected by Landlord (collectively, "Landlord's Architect") and approved by the applicable city, county and other governmental agencies, departments and authorities with jurisdiction over the Project ("Governmental Entities") and by Tenant as provided herein; and
 - (B) the "Essential Site Improvements", which for purposes hereof means those certain portions of the exterior Common Area improvements at the Project as described in **Schedule 1** attached hereto. The Essential Site Improvements shall be constructed by Landlord at the Project in the areas generally depicted on the Site Plan (subject to Landlord's Site Plan modification rights set forth in the Lease) and in substantial accordance with plans and specifications prepared by Landlord's Architect and approved by the Governmental Entities and by Tenant as provided herein.
- 1.2 Cooperation. Landlord and Tenant shall cooperate in good faith and shall use reasonable efforts to accommodate the other's reasonable comments and proposed revisions, if any, to the various drawings, plans and specifications contemplated by this Work Letter for the Base Building Improvements and the Tenant Improvements (hereinafter defined). To that end, Tenant shall cause Tenant's selected architect or space planner reasonably approved by Landlord ("Tenant's Architect"), and Landlord shall cause Landlord's Architect, to collaborate and cooperate with each other and with Landlord and Tenant and each of their contractors and consultants to the extent reasonably necessary to properly perform the work required of them hereunder, consistent with good design and construction practices, and to keep the other parties informed of the progress in completing the space plans, drawings, construction documents, changes to plans, meetings and communications with the applicable Governmental Entities and assisting each other in obtaining all required information and approvals. In that regard, Landlord's Architect and Tenant's Space Planner have met and have developed an estimated schedule for the architectural design of the Base Building Improvements, a copy of which is attached hereto as **Schedule 2** (as the same may be amended from time to time, the "Design Schedule"). To the extent the time periods set forth in this Work Letter for the preparation, approval and submittal of plans and specifications conflict with the time periods for such items set forth in the Design Schedule, the time periods set forth in the Design Schedule shall control.
- 1.3 D&D Drawings. By or before the date that is forty-five (45) days after execution of the Lease, Landlord shall deliver to Tenant a full set of developed design drawings ("D&D Drawings") for Landlord's Work, prepared by Landlord's Architect in substantial accordance with Schedule 1 for Tenant's review and comment. The D&D Drawings shall include such Base Building elements as structure, perimeter wall design, column spacing detail, mechanical, electrical and plumbing point of connection and schematic layout of the above (Landlord and Tenant hereby agreeing that the HVAC units, chiller plant, and other mechanical, electrical and plumbing systems serving the Premises shall be installed in locations reasonably determined by Landlord to the extent consistent with other Comparable Buildings and do not materially and adversely interfere with Tenant's use of, or ingress to or egress from, the Premises or the Parking Facilities), and such design elements as wall sections, bracings and elevations. Within ten (10) days following receipt by Tenant, Tenant shall either approve the D&D Drawings or disapprove same, in which latter case, Tenant shall concurrently with such disapproval provide Landlord with a full set of comments and proposed revisions, if any, to the D&D Drawings based on Tenant's preliminary space planning,

EXHIBIT D

including its programmatic layout, utilities points of connection requirements, and location of any Tenant supplemental HVAC system. If Tenant fails to either approve or disapprove the D&D Drawings within the ten (10) day period, Tenant shall be deemed to have approved the D&D Drawings. Landlord shall revise the D&D Drawings within ten (10) days following its receipt of Tenant's comments and proposed revisions, and the above cooperative process shall be continued until the D&D Drawings have been finally completed and approved.

- 1.4 Base Building Plans. Within sixty (60) days from the finalizing of the D&D Drawings, Landlord shall cause Landlord's Architect to prepare and deliver to Tenant Base Building construction plans and specifications ("Base Building Plans"). Within ten (10) days of Tenant's receipt of the Base Building Plans, Tenant shall either approve the Base Building Plans or disapprove same, in which latter case, Tenant shall concurrently with such disapproval provide Landlord with a full set of comments and proposed revisions, if any, to the Base Building Plans based on the D&D Drawings; provided, however, that Tenant's comments and proposed revisions shall be limited to any non-conformity of the Base Building Plans with the final D&D Drawings. If Tenant fails to either approve or disapprove the Base Building Plans within the ten (10) day period, Tenant shall be deemed to have approved the Base Building Plans. Landlord shall use reasonable efforts to accommodate Tenant's reasonable comments and proposed revisions, if any, to the Base Building Plans. Landlord shall revise the Base Building Plans within ten (10) days following its receipt of Tenant's comments and proposed revisions and the above process shall be continued until the Base Building Plans have been finally completed and approved.
- 1.5 Changes to Base Building Plans. Once the Base Building Plans are finalized in accordance with the process set forth above, all subsequent revisions requested by Tenant will be accommodated by Landlord only to the extent accommodation would be reasonably practicable and only if such revisions would not materially delay the completion of Landlord's Work or the Tenant Improvements or any of the remainder of the Project or constitute a Material Change (hereinafter defined), and provided that Tenant shall bear all of the expenses and increased costs associated with such revisions. Unless (a) required by applicable Governmental Entities, or (b) approved by Tenant (which approval shall not be unreasonably withheld), Landlord shall make no material changes to the approved Base Building Plans. Changes that would delay or increase the cost of the Tenant Improvements in any material respect, that would alter the facade design, including materials and structure, and/or alter the layout of the Building or the Essential Site Improvements in any significant respect, shall be deemed "material" for purposes hereof. If material changes are required by applicable Governmental Entities, Landlord shall consult with Tenant with respect to such required changes and the parties shall use all reasonable efforts to cooperate in effectuating such changes in a mutually agreeable manner.
- Performance of Landlord's Work; Schedule of Milestones. Landlord shall cause Landlord's Work to be performed by one or more duly licensed contractors selected by Landlord in substantial accordance with the approved final plans and specifications therefor, pursuant to validly issued permits, and in compliance with all applicable Laws. Landlord's Work shall be performed in a good and workmanlike manner, with reasonable diligence, and Landlord shall use commercially reasonable diligent good faith efforts to complete the various components of the Base Building Improvements in accordance with the "Schedule of Milestones" attached hereto as Schedule 3 (the parties hereby agreeing that all milestone dates set forth in Schedule 3 shall be subject to extension for delays attributable to Force Majeure Events and Tenant Delays) and otherwise in accordance with good construction practices. If Landlord fails to perform and/or substantially complete those items of Landlord's Work set forth in Schedule 3 by the applicable milestone date for such item of Landlord's Work set forth in Schedule 3, and Landlord has not consistently and diligently performed Landlord's Work in good faith after receipt of all necessary permits, Tenant shall be entitled to: (i) deliver a written notice to Landlord specifying those items of Landlord's Work which Tenant believes have not been substantially completed by the applicable milestone date and (ii) if such items are not substantially completed within fifteen (15) days after delivery of such notice (or, if such work cannot reasonably be completed with fifteen (15) days, then within a reasonable time provided that Landlord promptly commences work on such items and diligently prosecutes the same to completion), then Tenant shall have the right, as Tenant's sole remedy (subject to Tenant's right to terminate as set forth in Section 3.1 of the Lease), to pursue an action or actions for specific performance of Landlord's obligations hereunder, and/or for actual damages caused by Landlord's failure to perform and/or substantially complete Landlord's Work, cumulatively or alternatively, singularly or in combination; provided, however, in no event shall Landlord be liable to Tenant for any speculative, punitive or special damages.

SECTION 2

TENANT IMPROVEMENTS

2.1 <u>Description; General Requirements</u>. As more fully set forth below, Landlord shall cause to be constructed, at Tenant's sole cost and expense, all improvements in and to the Premises, other than Landlord's Work and Tenant's Work (as hereinafter defined), required pursuant to this Work Letter to prepare the same for Tenant's initial use and occupancy (collectively, the "<u>Tenant Improvements</u>"), including, by way of example and not by limitation, interior walls and movable floor to ceiling partitioning; interior doors and door hardware; fire sprinkler system drops and heads; interior drop ceilings; interior plumbing; distribution of HVAC extensions beyond the trunk

EXHIBIT D

line distribution/main HVAC shaft; HVAC pneumatic or DDC controls; electrical distribution from the main electrical room on each floor; elements and requirements necessary to satisfy the project LEED requirements for tenant improvements; emergency generators and UPS system (if any); carpeting; floor covering; interior painting; security systems for the Premises; data and telephone distribution beyond the main riser on each floor; acoustic insulation; signage of any type (including exit signage except at exit doors into exit stairwells, restrooms and elevator lobbies); panic hardware (except at exit doors into exit stairwells or into elevator lobbies, which Landlord shall provide as part of the Base Building to the extent required by code); contractor's fees; architect's fees; engineer's fees; building permit fees; utility connection fees (not including sewer connection "tap in" fees) for utilities to or serving the Premises. Attached hereto as **Schedule 4** are specifications for the Building standard components to be used in the construction of the Tenant Improvements in the Premises (collectively, the "Standard Improvement Package"). The Standard Improvement Package shall be reasonably consistent with the quality and type of building standard improvements provided at other Comparable Buildings. The quality of Tenant Improvements shall be equal to or of greater quality than the quality set forth in the Standard Improvement Package; provided, however, that Landlord may, at Landlord's option, require the Tenant Improvements to comply with certain aspects of the Standard Improvement Package and Landlord may make reasonable changes to the Standard Improvement Package from time to time, so long as such changes are with the quality and type of building standard improvements provided at other Comparable Buildings and that such changes do not materially and adversely affect Tenant's use of, or ingress to and egress from, the Premises or the Parking Facilities. In that regard, Tenant acknowledges that Building II is currently contemplated to be constructed and, at Landlord's option, shall be constructed and operated, in accordance with the U.S. Green Building Council's Leadership in Energy and Environmental Design Program's standards, as the same are amended, supplemented or replaced from time to time, and, at Landlord's option, any similar standards ("LEED"). Landlord currently intends that Building II, upon completion, will be certified as a "certified" level structure and will be operated in an environmentally responsible and energyefficient manner in accordance with LEED. Tenant shall cooperate with Landlord in obtaining and maintaining the LEED "certified" level as provided in the Lease.

- Space Planning. On or before the "Space Plan Delivery Date" set forth in the Design Schedule, Tenant shall cause Tenant's Architect to prepare a final space plan for the Tenant Improvements in the Premises (collectively, the "Space Plan") and shall deliver the Space Plan to Landlord for Landlord's approval. The Space Plan shall include a description of the Tenant Improvements, as well as a layout and designation of all offices, rooms and other partitioning, their intended use, and equipment to be contained therein, all of which shall be consistent with the D&D Drawings and Section 2.1 of this Work Letter. Within ten (10) days of Landlord's receipt of the Space Plan, Landlord shall either approve the Space Plan or disapprove same, in which latter case, Landlord shall concurrently with such disapproval provide Tenant with a full set of comments and proposed revisions, if any, to the Space Plan. Landlord may disapprove the Space Plan only if it will (i) affect the exterior appearance of the Building; (ii) adversely affect the Building Structure; (iii) adversely affect the Building Systems; (iv) unreasonably interfere with any other potential occupant's normal and customary office operation, if any, (v) fail to comply with Law, (vi) involve the use of hazardous materials, (vii) be materially inconsistent with the D&D Drawings, (viii) have a material adverse affect on the quality or efficiency of Landlord's Work, (ix) reduce the quality of the Tenant Improvements to a quality lower than set forth in the Standard Improvement Package, or (x) materially delay completion of Landlord's Work, all as determined by Landlord in its reasonable discretion (each, a "Material Change"). If Landlord fails to either approve or disapprove the Space Plan within the ten (10) day period, Landlord shall be deemed to have approved the Space Plan. Tenant shall use reasonable efforts to accommodate Landlord's reasonable comments and proposed revisions, if any, to the Space Plan. Tenant shall revise the Space Plan within ten (10) days follo
- 2.3 Preliminary Cost Estimate. Within ten (10) Business Days after the Space Plan has been approved, Landlord shall provide Tenant with a cost estimate in accordance with the approved Space Plan, which cost estimate shall include, as nearly as reasonably possible, the estimated cost of all Tenant Improvement Allowance items to be incurred by Tenant in connection with the design and construction of the Tenant Improvements (the "Preliminary Cost Estimate"). Within ten (10) days of Tenant's receipt of the Preliminary Cost Estimate, Tenant shall deliver to Landlord in writing any comments or disapprovals Tenant has relating to the Preliminary Cost Estimate. Tenant's failure to deliver any comments or disapprovals within such time shall be deemed Tenant's approval thereof. If Tenant reasonably disapproves the Preliminary Cost Estimate, Landlord and Tenant shall negotiate in good faith to reach agreement on the items contained in the Preliminary Cost Estimate, provided that Landlord shall not be obligated to approve any change proposed by Tenant to the Preliminary Cost Estimate (or the approved Space Plan) that constitutes a Material Change. Upon approval of the Preliminary Cost Estimate, Tenant shall promptly modify the Space Plan as necessary to conform to any agreed changes therein and Landlord shall be released by Tenant to purchase any of the items set forth in the Preliminary Cost Estimate and to commence the construction relating to such items; provided, however, that the parties acknowledge that the Preliminary Cost Estimate is an estimate only and shall not limit Tenant's obligations to pay for the design and construction of the Tenant Improvements as provided in this Work Letter.

- 2.4 Tenant Improvement Plans. Within sixty (60) days after approval of the Preliminary Cost Estimate and final Space Plan, Tenant shall cause Tenant's Architect to prepare construction plans and specifications for the Tenant Improvements in a form which is sufficient to allow contractors to bid on the work and to obtain all applicable permits ("Tenant Improvement Plans") and shall deliver same to Landlord for approval. The Tenant Improvement Plans shall be based on the final Space Plan and the requirements of Section 2.1 of this Work Letter. Tenant and Tenant's Architect shall verify, in the field, the dimensions and conditions as shown on the relevant portions of the Base Building plans, and Tenant and Tenant's Architect shall be solely responsible for verifying the same, and Landlord shall have no responsibility in connection therewith. Within ten (10) days of Landlord's receipt of the Tenant Improvements Plans, Landlord shall either approve the Tenant Improvement Plans or disapprove same, in which latter case, Landlord shall concurrently with such disapproval provide Tenant with a full set of comments and proposed revisions, if any, to the Tenant Improvement Plans. If Landlord fails to either approve or disapprove the Tenant Improvement Plans within the (10) days period, Landlord shall be deemed to have approved the Tenant Improvement Plans. Tenant shall revise the Tenant Improvement Plans within ten (10) days following its receipt of Landlord's comments and proposed revisions and the above process shall be continued until the Tenant Improvement Plans have been finally completed and approved. Landlord's review of the Tenant Improvement Plans shall not imply Landlord's review of the same, or obligate Landlord to review the same, for quality, design, code compliance or other like matters. Accordingly, notwithstanding that the Tenant Improvement Plans are reviewed by Landlord or its architect, engineers and consultants, Landlord shall not be responsible for any omissions or errors contained in the Tenant Improv
- 2.5 Changes to Tenant Improvement Plans. Once the Tenant Improvement Plans are finalized in accordance with the process set forth above, all subsequent revisions requested by Tenant will be accommodated by Landlord only to the extent that such revisions will not constitute a Material Change, and provided that Tenant shall bear all of the expenses and increased costs and delays associated with such revisions. Landlord shall use reasonable efforts to provide Tenant with an estimate of such increased costs and delays within ten (10) days after receipt of Tenant's request for such revisions, and Tenant shall either direct Landlord to proceed with such revisions or withdraw its request for such revisions within ten (10) days after receipt of Landlord's cost increase/delay estimate. Landlord shall not be required to suspend work pending the resolution of the revision request process described above. Unless (a) required by applicable Governmental Entities, or (b) approved by Tenant (which approval shall not be unreasonably withheld), Landlord shall make no material changes to the approved Tenant Improvement Plans. Changes that would materially delay or increase the cost of the Tenant Improvements in any material respect or that would alter the materials and/or the layout of the Tenant Improvements in any significant respect shall be deemed "material" for purposes hereof. If material changes are required by applicable Governmental Entities, Landlord shall consult with Tenant with respect to such required changes and the parties shall use all reasonable efforts to cooperate in effectuating such changes in a mutually agreeable manner.
- Bid Process; Construction of Tenant Improvements. As soon as reasonably possible after approval of the Tenant Improvement Plans, Landlord shall solicit bids for the Tenant Improvement Work from the base building contractor (or an affiliate of same) selected by Landlord (the "First Chance Contractor"). Landlord shall prepare a bid package, the contents and substance of which shall be subject to Tenant's prior written approval (not to be unreasonably withheld), and which shall, at a minimum, contain the approved Tenant Improvement Plans and the form of proposed contract that the successful bidder will be asked to sign (the "TI Contract"). In that regard, Landlord shall deliver a draft of the proposed TI Contract to Tenant prior to submittal of the bid package. Within ten (10) days of Tenant's receipt of the TI Contract, Tenant shall deliver to Landlord in writing any comments or disapprovals Tenant has relating to the TI Contract. Tenant's failure to deliver any comments or disapprovals within such time shall be deemed Tenant's approval thereof. If Tenant reasonably disapproves the TI Contract, Landlord and Tenant shall negotiate in good faith to reach agreement on the disapproved items. Landlord may specify acceptable subcontractors for the major trades (electrical, mechanical, fire and life safety, and plumbing), so long as such subcontractors are reasonably acceptable to Tenant and charge competitive market rates. The First Chance Contractor shall have not more than thirty (30) days from receipt of the bid package to submit its bid to Landlord, with a copy to Tenant. Upon receipt of the First Chance Contractor's bid, Landlord shall meet with Tenant to review the bid and to determine whether to proceed with the First Chance Contractor. If the First Chance Contractor's bid is not acceptable to Landlord or Tenant, then Landlord shall solicit bids from not less than two (2) other responsible contractors, one of which shall be selected by Landlord (subject to Tenant's reasonable approval) and one of which may be selected by Tenant (subject to Landlord's reasonable approval; Landlord hereby approves Burge as a Tenant-selected contractor for such purpose). Each proposed contractor shall have not more than thirty (30) days after receipt of the bid package to submit its bid to Landlord, with a copy to Tenant. Landlord and Tenant shall meet to compare and reconcile the bids, and Landlord shall select the lowest qualified bidder therefrom (including any re-bid by the First Chance Contractor) to construct the Tenant Improvements, after consultation with Tenant (except that Landlord may, with Tenant's prior written consent, which shall not be unreasonably withheld, conditioned or delayed, select any other bidder whose bid is not more than 5% higher than the lowest qualified bidder if Landlord has reasonable grounds to do so). Landlord shall notify Tenant of the identity of the contractor so selected by Landlord to perform the Tenant Improvement work (the "TI Contractor"), together with a revised cost estimate for the design and construction of the

- Tenant Improvements based on the bid (the "<u>Final Cost Estimate</u>"). Upon approval of the TI Contract and upon Tenant's delivery of the Over-Allowance Amount (as described in Section 2.7 of this Work Letter), Landlord shall enter into the TI Contract and shall cause the TI Contract to construct and complete the Tenant Improvement Work in accordance with the approved final Tenant Improvement Plans, the TI Contract and this Work Letter.
- Tenant Improvement Allowance; Over-Allowance Amount. Tenant shall be entitled to a one-time tenant improvement allowance (the "Tenant Improvement Allowance") in the amount of Forty-Three Dollars (\$43.00) per rentable square foot of the Premises for the costs relating to the design and construction of the Tenant Improvements. In no event shall Landlord be obligated to make disbursements with respect to the Tenant Improvements pursuant to this Work Letter in a total amount that exceeds the Tenant Improvement Allowance. All Tenant Improvements for which the Tenant Improvement Allowance has been made available shall be deemed Landlord's property under the terms of this Lease. Within ten (10) days after Landlord notifies Tenant of the Final Cost Estimate, Tenant shall deliver to Landlord cash in an amount (the "Over-Allowance Amount") equal to the difference between (i) the Final Cost Estimate, and (ii) the amount of the Tenant Improvement Allowance. The Over-Allowance Amount shall be disbursed by Landlord prorata with the disbursement of any then remaining portion of the Tenant Improvement Allowance, and such disbursement shall be pursuant to the same procedure as the Tenant Improvement Allowance (by way of example only, if the Over-Allowance Amount is 10% of the Final Cost Estimate, then each disbursement will be 90% from the Tenant Improvement Allowance and 10% from the Over-Allowance Amount). If any revisions, changes, or substitutions shall be made to the Tenant Improvement Plans or the Tenant Improvements as permitted by this Work Letter, any additional costs which arise in connection with such revisions, changes or substitutions or any other additional costs shall be paid by Tenant to Landlord within ten (10) days after Landlord's request as an addition to the Over-Allowance Amount. As soon as reasonably possible following Substantial Completion, Landlord shall provide Tenant with the final breakdown of the actual cost of the Tenant Improvement work (the "Actual Cost") together with reasonable supporting documentation (the "Actual Cost") Notice"). If the Actual Cost is less than the Tenant Improvement Allowance plus any Over-Allowance Amounts paid by Tenant, then Landlord shall refund that difference to Tenant within twenty (20) Business Days following delivery of the Actual Cost Notice. If, however, the Actual Cost is greater than the Tenant Improvement Allowance plus any Over-Allowance Amounts paid by Tenant, then Tenant shall pay that deficiency to Landlord as additional Rent within twenty (20) Business Days following delivery of the Actual Cost Notice.
- Disbursement of the Tenant Improvement Allowance. Except as otherwise set forth in this Work Letter, the Tenant Improvement Allowance shall be disbursed by Landlord (each of which disbursements shall be made pursuant to Landlord's disbursement process) for costs related to the construction of the Tenant Improvements and for the following items and costs (collectively, the "Tenant Improvement Allowance Items"): (i) payment of the fees of Tenant's Architect and the fees of Landlord's Architect and other fees incurred by, and the cost of documents and materials supplied by, Landlord and Landlord's consultants in connection with the preparation and review of the Space Plans and the Tenant Improvement Plans; (ii) the cost of obtaining any and all permits for the construction of the Tenant Improvements; (iii) the cost of any changes in the Tenant Improvements when such changes are required by the Tenant Improvement Plans; and (iv) the cost of any changes to the Tenant Improvements required by applicable Laws. Tenant agrees to pay Landlord for the costs of design and construction of the Tenant Improvements incurred in excess of the Tenant Improvement Allowance; provided, however, Tenant shall in no event be required to pay for:
 - Construction costs in excess of amounts payable under the approved TI Contract or in excess of change orders approved in writing by Tenant;
 - (ii) Any cost or expense incurred in connection with or arising out of the construction of the Landlord Work;
 - (iii) Any supervision or administration fees or charges of Landlord for supervising the Tenant Improvement work, and any salaries, payroll, taxes, unemployment insurance contributions, social security taxes and other assessments, fringe benefits, and related compensation for Landlord's employees for their time spent supervising or administering the Tenant Improvement work;
 - (iv) Expenses of Landlord's offices, including, but not limited to, Landlord's overhead or general expenses of any kind;
 - (v) Any part of the Landlord's financing expenses, including interest on the Landlord's capital employed in constructing the Tenant Improvement Work;
 - (vi) Costs due to the negligence, willful acts or misconduct of Landlord or Landlord's Architect or the TI Contractor, including, but not limited to, the correction of any defective or nonconforming Tenant Improvement work, disposal of materials and equipment wrongly supplied, or making good any damage to property;

- (vii) Any and all expenses, including, but not limited to attorneys' fees and costs, incurred in defending against mechanic's lien, stop notice, and/or payment bond claims (other than those caused by Tenant); or
- (viii) Any and all costs incurred by Landlord's Architect to prepare the Base Building Plans or the development of drawings and specifications for the construction of any of the Landlord Work.
- Required Removables. Landlord, by written notice to Tenant given at least twelve months prior to the Termination Date, may require Tenant to remove, at Tenant's expense, any Tenant Improvements that are performed by or for the benefit of Tenant and, in Landlord's reasonable judgment, are of a nature that would require removal and repair costs that are materially in excess of the removal and repair costs associated with standard office improvements (collectively referred to as "Required Removables"). Without limitation, it is agreed that Required Removables include internal stairways, raised floors, personal baths and showers, vaults, rolling file systems and structural alterations and modifications of any type. The Required Removables designated by Landlord shall be removed by Tenant before the Termination Date, provided that upon prior written notice to Landlord, Tenant may remain in the Premises for up to five (5) days after the Termination Date for the sole purpose of removing the Required Removables. Tenant's possession of the Premises shall be subject to all of the terms and conditions of this Lease, including the obligation to pay Rent on a per diem basis at the rate in effect for the last month of the Term. Tenant shall repair damage caused by the installation or removal of Required Removables. If Tenant fails to remove any Required Removables or perform related repairs in a timely manner, Landlord, at Tenant's expense, may remove and dispose of the Required Removables and perform the required repairs. Tenant, within fifteen (15) days after receipt of an invoice, shall reimburse Landlord for the reasonable costs incurred by Landlord.
- 2.10 Contractor's Warranties and Guaranties. Landlord shall obtain customary warranties from the TI Contractor to the extent provided in the TI Contract and from the manufacturers of equipment included as part of the Tenant Improvements, provided such warranties are provided at no extra cost (unless Tenant agrees to pay the extra cost). If repairs are required to be made to any portion of the Tenant Improvements, Tenant shall notify Landlord (and Tenant may also, at its election, notify the TI Contractor), and if such repairs are items for which Landlord may have obtained a warranty, Landlord shall assign such warranty to Tenant (or, at Landlord's option, enforce such warranty for Tenant's benefit) upon request. Tenant hereby waives all claims against Landlord relating to, or arising out of the construction of, the Tenant Improvements, except for defaults by Landlord of its obligations under this Work Letter or the Lease. Following completion of the Tenant Improvements, Landlord shall assemble and provide to Tenant copies of all warranties relating to the Tenant Improvements.

SECTION 3

SUBSTANTIAL COMPLETION

- Ready for Occupancy. The Premises shall be deemed "Ready for Occupancy" upon the earlier of the date upon which Tenant commences business operations from the Premises or the date of Substantial Completion. For purposes of this Lease, "Substantial Completion" shall be deemed to mean the date when all of the following shall have occurred: (i) Landlord shall have delivered to Tenant a certificate of occupancy issued by the City of Irvine for the Premises, or Tenant is legally entitled to occupy the Premises for the operation of its business without violating any law or regulation or voiding or adversely affecting its insurance coverage, whether pursuant to a temporary certificate of occupancy or otherwise (such as complete sign off on all building inspection cards); and (ii) Landlord's Work and the Tenant Improvements have been completed in accordance with the terms hereof, except for normal so-called "punch-list" or "pick-up list" items which do not unreasonably interfere with Tenant's use of the Premises or the Project (which shall be completed as set forth in Section 3.2 of this Work Letter); (iii) all sewer, water, electrical and other utility services have been connected, are available and in good working order, and have sufficient capacity to service the Premises are required by the Lease, (iv) Tenant is provided with reasonable access to the Premises, with the Essential Site Improvements in a reasonably clean and safe condition and open and available for use by Tenant, its employees, guests, customers and invitees; and (v) Landlord's Architect delivers its statement to Tenant that Landlord's Work and the Tenant Improvements have been substantially completed in accordance with the approved plans and specifications. Landlord shall use commercially reasonable efforts to give Tenant not less than thirty (30) days advance written notice of the likely date upon which Substantial Completion is to be achieved. Landlord currently anticipates that Substantial Completion will occur on or before March 1, 2006 (the "Est
- 3.2 <u>Punch List</u>. Within twenty (20) days after Substantial Completion, Tenant shall prepare and submit to Landlord a list of those items of Landlord's Work and the Tenant Improvements requiring correction and/or completion (a "<u>Punch List</u>"). If Landlord shall object to any item set forth on such Punch List, Landlord shall give Tenant written notice thereof within five (5) business days after Landlord's receipt of the Punch List, setting forth in reasonable detail the item or items to which Landlord objects and the grounds for such objection. If Landlord shall fail to so

notify Tenant within such five (5) business day period, then Landlord shall be deemed to have accepted the Punch List and agreed to correct and/or complete (as the case may be) all items set forth thereon as provided below. Conversely, if Landlord shall so notify Tenant within such five (5) business day period, then: (i) if Landlord shall fail to object to all of the items set forth in the Punch List, then Landlord shall be deemed to have accepted those items on the Punch List as to which Landlord shall not have objected and agreed to correct and/or complete (as the case may be) such items as provided in below; and (ii) with respect to those items set forth on the Punch List as to which Landlord has so objected: (x) Landlord and Tenant shall attempt in good faith to resolve the dispute within ten (10) days thereafter, and, to the extent that the parties shall agree as to any item(s) to which Landlord has objected, they shall confirm such agreement in writing; and (y) to the extent that the parties are unable to resolve their dispute as to one or more items on the Punch List and join in such a confirmatory writing within such ten (10) day period, either party shall have the right, at any time thereafter prior to such a resolution, to submit the dispute to arbitration as set forth in Section 27.28 of the Lease. Landlord shall correct and/or complete all items on each Punch List with reasonable diligence but, in any event, within thirty (30) days after Landlord's receipt of the same or, if Landlord shall object to one or more items on the Punch List as and when provided in paragraph (a) above, as to those items within sixty (60) days (subject to extension for Force Majeure Events) after Landlord's objection has been resolved as provided therein. If Landlord fails to correct and/or complete any of such items within such sixty (60) day period, and Landlord has not been diligently working to correct and/or complete such Punch List items, then (i) in the case of any portion of such Punch List items which are within the Premises, have the same done for the account of Landlord without any further notice or cure period, the cost of which shall be paid by Landlord to Tenant within fifteen (15) days after Tenant's written demand therefor and delivery of copies of invoices and other reasonable back up documentation, or (ii) in the case of all other unperformed Punch List Items, Tenant shall be entitled to pursue an arbitration action or actions pursuant to Section 27.28 of the Lease for specific performance of Landlord's obligations hereunder, and/or for actual damages caused by Landlord's failure to perform and/or substantially complete Landlord's Work, cumulatively or alternatively, singularly or in combination; provided, however, in no event shall Landlord be liable to Tenant for any speculative, punitive or special damages.

- 3.3 <u>Delay in Substantial Completion of the Premises</u>. Except as provided in this Section 5.2, the Commencement Date shall occur as set forth in the Lease and Section 5.1, above. If there shall be a delay or there are delays in the Substantial Completion of the Premises or in the occurrence of any of the other conditions precedent to the Commencement Date, as set forth in the Lease, to the extent resulting from the following (collectively, "Tenant Delays"):
 - (a) Tenant's failure to comply with the Design Schedule;
 - (b) Tenant's failure to timely approve any matter requiring Tenant's approval (provided Tenant receives all reasonable information to make such determination);
 - (c) A material breach by Tenant of the terms of this Work Letter or the Lease;
 - (d) Changes in any of the Base Building Plans or Tenant Improvement Plans due to Tenant's acts or omissions after approval of the same by Landlord and Tenant:
 - (e) Tenant's request for changes in the Approved Tenant Improvement Plans (which actually delays substantial completion);
 - (f) Tenant's requirement for materials, components, finishes or improvements which are not available in a commercially reasonable time given the anticipated date of Substantial Completion of the Premises, as set forth in the Lease, or which are different from, or not included in, the Standard Improvement Package;
 - (g) Changes to the Base Building Plans required by the approved Tenant Improvement Plans; or
 - (h) Any other acts or omissions of Tenant, or its agents, or employees;

then, notwithstanding anything to the contrary set forth in the Lease or this Work Letter and regardless of the actual date of Substantial Completion of the Premises, the Commencement Date shall be deemed to be the date the Commencement Date would have occurred if no Tenant Delay or Delays, as set forth above, had occurred; provided that any such delays are not due to the acts, delays or omissions of Landlord. Landlord shall notify Tenant of any act or omission that Landlord considers a Tenant Delay within a reasonable time after Landlord actually becomes aware of such Tenant Delay.

SECTION 4 MISCELLANEOUS

4.1 <u>Tenant's Work In Premises Prior to Substantial Completion</u>. Landlord and Tenant acknowledge and agree that certain work required for Tenant's occupancy of the Premises, including but not

EXHIBIT D

limited to the procurement and installation of furniture, fixtures, equipment, artwork and interior signage are beyond the scope of the Tenant Improvements and shall be performed by Tenant or its contractors at Tenant's sole cost and expense. Provided that Tenant and its agents do not interfere with Contractor's work in the Building and the Premises and complies with the requirements set forth below, Landlord shall allow (and shall direct the TI Contractor to allow) Tenant access to the Premises prior to Substantial Completion for the purpose of performing such work ("Tenant's Work"). Landlord shall not charge Tenant for utilities or elevator service used by Tenant in connection with Tenant's Work. Prior to Tenant's entry into the Premises as permitted by the terms hereof, Tenant shall submit a schedule to Landlord and the TI Contractor, for their approval, which schedule shall detail the timing and purpose of Tenant's entry, and shall perform Tenant's Work in such a way as not to hinder or delay the operations of Landlord or the TI Contractor in the Building. Tenant shall promptly pay to Landlord any costs incurred by Landlord as a result of any interference with Landlord's operations by Tenant or its contractors. Landlord shall make all reasonable efforts to notify Tenant of any such interference of which Landlord has actual knowledge, but failure to provide such notice shall in no way limit Landlord's right to demand payment for such costs. Tenant's contractors shall be subject to Landlord's prior written approval, which shall not be unreasonably withheld, conditioned or delayed, and to the administrative supervision of the TI Contractor. All items of Tenant's Work shall constitute Tenant's Property. Tenant's Work shall comply with all of the following requirements:

- (a) Tenant's Work shall not proceed until Landlord has approved in writing: (i) Tenant's contractors, (ii) proof of the amount and coverage of public liability and property damage insurance carried by Tenant's contractors in the form of an endorsed insurance certificate naming Landlord, the Contractor, and the agents of Landlord and the Contractor as additional insureds, in an amount not less than two million dollars, and (iii) the plans and specifications for Tenant's Work.
- (b) Tenant's Work shall be performed in conformity with a valid permit when required, a copy of which shall be furnished to Landlord before such work is commenced. In any event, all Tenant's Work shall comply with all applicable Laws. Landlord shall have no responsibility for Tenant's failure to comply with such applicable Laws. Any and all delay in obtaining a certificate of occupancy due to Tenant's vendors is the responsibility of Tenant and shall be a Tenant Delay. In connection with Tenant's Work, Tenant or its contractors shall arrange for any necessary hoisting or elevator service with Landlord and shall pay such reasonable costs for such services as may be charged by Landlord.
- (c) Tenant shall promptly pay Landlord upon demand for any extra expense incurred by Landlord by reason of faulty work done by Tenant or its contractors, by reason of damage to existing work caused by Tenant or its contractors, or by reason of inadequate cleanup by Tenant or its contractors.
- 4.2 <u>Freight Elevators</u>. Landlord shall make the freight elevator reasonably available to Tenant, at no additional cost to Tenant, in connection with initial decorating, furnishing and moving into the Premises.
- 4.3 Tenant's Representative. Tenant has designated Kathy Murray as its sole representative ("Tenant's Representative") with respect to the matters set forth in this Work Letter, who, until further notice to Landlord, shall have full authority and responsibility to act on behalf of the Tenant as required in this Work Letter. Tenant shall have the right, at Tenant's sole cost and expense, to keep Tenant's Representative at the Project at such times as it may deem necessary during the period of construction of Landlord's Work and the Tenant Improvement work, to assist in the supervision thereof, and to verify that all work is being done in accordance with the approved plans and specifications. If Tenant's Representative determines that the construction is not proceeding in accordance with the approved plans and specifications and the requirements of this Work Letter, Tenant's Representative shall have the right, but not the obligation, to require Landlord to immediately take all steps necessary to correct the non-complying work, at Landlord's sole cost and expense. If Landlord fails to correct defective work, Tenant, upon five (5) days' prior written notice, may order Landlord to stop the work, or any portion thereof, until the cause for such order has been eliminated; however, this right of Tenant to stop the work shall not give rise to any duty on the part of Tenant to exercise this right for the benefit of Landlord or any other person or entity. Neither Tenant's failure to inspect the work, nor Tenant's failure to require the correction of any defective work, shall result in any liability to Tenant whatsoever.
- 4.4 <u>Landlord's Representative</u>. Landlord has designated Greg Galletly as its sole representative ("<u>Landlord's Representative</u>") with respect to the matters set forth in this Work Letter, who, until further notice to Tenant, shall have full authority and responsibility to act on behalf of the Landlord as required in this Work Letter. Landlord shall have the right, at Landlord's sole cost and expense, to keep Landlord's Representative at the Premises at such times as it may deem necessary during the period of Tenant's Work, to assist in the supervision thereof, and to verify that all work is being done in accordance with the approved plans and specifications. If Landlord's Representative determines that Tenant's Work is not proceeding in accordance with the approved plans and specifications and the requirements of this Work Letter, Landlord's Representative shall have the right, but not the obligation, to require Tenant to immediately take all steps necessary to correct the non-complying work, at Tenant's sole cost and expense. If Tenant fails to correct

- defective work, Landlord, upon five (5) days' prior written notice, may order Tenant to stop the work, or any portion thereof, until the cause for such order has been eliminated; however, this right of Landlord to stop the work shall not give rise to any duty on the part of Landlord to exercise this right for the benefit of Tenant or any other person or entity. Neither Landlord's failure to inspect the work, nor Landlord's failure to require the correction of any defective work, shall result in any liability to Landlord whatsoever.
- 4.5 <u>Builder's Risk Insurance</u>. Landlord shall obtain, at its cost, and maintain in full force and effect during the course of construction of Landlord's Work and the Tenant Improvement work, "all risk" property insurance covering Landlord's Work and the Tenant Improvements for the full cost of replacement thereof at the time of any loss.
- 4.6 <u>As-Built Plans; Notice of Completion</u>. Upon completion of the Tenant Improvements, Landlord shall deliver to Tenant a set of "as-built" plans and specifications of the Tenant Improvements, and within ten (10) days after completion, Landlord shall cause a Notice of Completion to be recorded in the office of the Recorder of the County of Orange in accordance with Section 3093 of the Civil Code of the State of California or any successor statute
- 4.7 Tenant's Lease Default. Notwithstanding any provision to the contrary contained in the Lease or this Work Letter, if a Default by Tenant under the Lease or a default under this Work Letter has occurred at any time on or before the Substantial Completion, then (i) in addition to all other rights and remedies granted to Landlord pursuant to the Lease, Landlord shall have the right to withhold payment of all or any portion of the Tenant Improvement Allowance and/or Landlord may cause the applicable contractor(s) to cease construction (in which case, Tenant shall be responsible for any delay in Substantial Completion caused by such work stoppage as set forth in this Work Letter), and (ii) all other obligations of Landlord under the terms of this Work Letter shall be forgiven until such time as such default is cured pursuant to the terms of the Lease.
- 4.8 <u>Lease Provisions</u>. The indemnification provisions set forth in the Lease shall apply to activities by Landlord and Tenant under this Work Letter.
- 4.9 <u>Books and Records</u>. Landlord shall prepare and maintain or cause to be prepared and maintained detailed and accurate operating and financial records for the Tenant Improvements. Tenant shall have the right to inspect such records at the offices of Landlord in Southern California during normal business hours and to make copies thereof at Tenant's sole cost and expense. Landlord shall preserve, or cause its agents to preserve, copies of all books and records relating to the Tenant Improvements for a period of three (3) years after the completion of the Tenant Improvements.
- 4.10 <u>Governmental Approvals</u>. Landlord shall be responsible for obtaining all necessary agreements with governmental agencies and utility companies relating to access, traffic control, availability of utility services and other similar entitlements affecting the construction of Landlord's Work and the Tenant Improvements.
- Permits; Compliance with Law. In addition to Landlord's obligations under Section 4.10 above, Landlord shall be responsible for obtaining, or causing the TI Contractor to obtain, all building permits and other governmental approvals, permits and land use entitlements required to complete Landlord's Work and the Tenant Improvements. Landlord shall perform or cause Landlord's Work to be performed in compliance with all current applicable Laws governing completion of the Landlord Work, including, without limitation, the requirements of the ADA and Title 24 in effect as of the Commencement Date, except to the extent such compliance results from Tenant's unique design or requirements not typical of a large general office user or from Tenant's particular manner of use of the Premises. If Landlord's Work does not comply with current applicable Laws as described above and the cost to complete the design and/or construction of the Tenant Improvements increases above that which would have been incurred had Landlord's Work been constructed in compliance with the then current applicable Laws, then such costs shall not be charged against the Tenant Improvement Allowance and Landlord shall be responsible for the same. In furtherance of the foregoing, the parties acknowledge and agree that the installation of the Tenant Improvements in the Premises will require certain work within the Premises (including work to the Base Building) that is not required to be completed until the installation of such Tenant Improvements (such as, for example, the fire sprinkler distribution work) that will be required as the Tenant Improvements are being installed in the Premises; this type of work shall not be considered to be Landlord's responsibility pursuant to the foregoing sentence.
- 4.12 Status Reports. Landlord and Tenant shall schedule weekly status conferences to discuss the progress of the design and construction of Landlord's Work and the Tenant Improvements. Landlord shall inform Tenant of any problems or developments with respect to such design or construction that may have an adverse impact on the completion of Landlord's Work or the Tenant Improvements promptly after Landlord becomes aware of such problems or developments.

Schedules to be attached:

SCHEDULE 1 Landlord's Work Description (Base Building and Essential Site Improvements)

SCHEDULE 2 Design Schedule

SCHEDULE 3 Schedule of Milestones for Base Building

SCHEDULE 4 Building Standard Tenant Improvement Specifications

EXHIBIT D

-10-

SCHEDULE 1 TO WORK LETTER

BASE BUILDING AND ESSENTIAL SITE IMPROVEMENTS

A. GENERAL

The "Base Building" and the "Essential Site Improvements" as referenced in the Work Letter to which this Schedule is attached shall be constructed materially in accordance with the architectural drawings prepared by Ware & Malcomb Architects and mutually approved by Landlord and Tenant, including the improvements described below.

B. STRUCTURE; CONCRETE FLOORS

- 1. The Building shall be a seven-story structural steel building with floor plates ranging from approximately 24,800 rentable square feet to 31,200 rentable square feet. The foundations for the Building shall be typical for a Building of this size and character and shall have been professionally designed in accordance with the soil and geological conditions of the site.
- 2. A subterranean parking area shall be provided beneath the Building providing between fifty and sixty-five total full size vehicle parking stalls serving the Building (including handicap spaces and handicap access as required by code).
- 3. The majority of each floor shall accommodate a minimum 12' finished ceiling height, except as otherwise deemed necessary by Landlord to accommodate stairwells or Building systems or equipment.
- 4. The concrete floors of the Premises within the Building shall be finished in accordance with the plans and specifications for the Building and shall be ready to accept Tenant's carpeting and hard surface floor covering with ordinary floor preparation by the tenant improvement contractor. The concrete floors shall be designed and constructed to support the following loads:
 - (a) Partition Load (20) twenty lbs./sf.
- (b) Live Load (50) fifty lbs./sf. in accordance with the current Uniform Building Code. Subject to live load redistribution calculations, Landlord's architect will work with Tenant to identify and provide areas of the Building that will accommodate heavier loads to accommodate additional loading requirements of special areas (e.g., File Room).

C. EXTERIOR GLASS

- 1. Exterior glass shall be energy efficient, tinted and shall be free from nicks, cracks and marring.
- 2. The glass and glass framing system shall be designed, constructed and sealed to assure the integrity of the exterior wall system and protect against moisture penetration.
- 3. The interior window mullions and or metal frames shall be free from adhesive, punctures, screw holes and dents and shall be painted or finished so as to provide a like new appearance.

D. ROOFING MEMBRANE AND STRUCTURE

- 1. The roof type and design shall incorporate a single ply roof membrane.
- 2. Landlord shall utilize the services of a professional roofing consultant to provide mutually approved formal specifications and to assure the integrity of the installation.
 - 3. Landlord shall provide in accordance with local code, rooftop equipment screening adequate for initial and future Building rooftop equipment.

E. COMMON AREAS, COLUMNS, PERIMETER CONDITIONS AND BUILDING CORE

- 1. Service core shall be complete with finished elevators, vertical stairwell, exit shafts as required by code, ventilation shafts, electrical/telephone rooms, and mechanical room(s). Exit stairwells to be painted. Exterior wall and columns shall be finished by Tenant.
 - 2. All exterior doors, door closer and locking devices as necessary.

3. Tenant shall be responsible for the cost of Building main and floor lobby area improvements (e.g., flooring, wall finishes, ceiling and lighting treatments), which shall be installed by Landlord as part of the Tenant Improvement work and paid for out of the Tenant Improvement Allowance.

F. WASTE WATER AND VENT SYSTEM (PLUMBING)

Landlord shall provide a minimum of two points of access to domestic water, sanitary waste and vent for Tenant's distribution shall be provided on each floor.

G. HVAC SYSTEM

- 1. Landlord shall provide a quality heating, ventilating and air conditioning (HVAC) System with service available on a year round basis. Tenant shall be permitted to use after-hours HVAC service. The Base Building shall be provided with the "main loop" in place. All distribution shall be installed by Landlord at Tenant's cost as part of the Tenant Improvement work and paid for out of the Tenant Improvement Allowance.
- 2. Base Building HVAC system to furnish conditioned clean supply air through the medium pressure base building air ducts, in sufficient capacities and quantities and at temperatures and pressures as required to maintain conditions as noted below:
 - (a) Indoor winter conditions: 72 dg. F. Dry bulb; Indoor summer conditions: 78 dg F. Dry bulb; with 50% Maximum Relative Humidity.
 - (b) Outdoor summer and winter design conditions in accordance with ASHRAE publication #SPCDX, Climatic Data for Region Ten, 5th Edition, May 1982.
 - (c) Summer Design Condition 0.5% Column, Winter Design Condition Median of Extremes.
 - (d) Population Density 1/100 usf.
 - (e) Cooling Load: 3 watts/usf dissipated heat for lighting and equipment (not including rooms with supplemental cooling such as computer and print rooms or Base Building equipment).
 - (f) Minimum zoning requirements:
 - (1) Open areas one (1) zone per 1,500 usable square feet.
 - (2) Private offices five (5) per zone.
 - (3) Separate zones for (i) special rooms with 2 exposures, and (ii) conference/meeting rooms.
 - 3. The Base Building HVAC system furnished to the Premises shall be a digitally controlled water source DX/VAV system.
 - 4. Base Building HVAC controls to be of adequate capacity and size to accommodate the criteria specified herein.
 - 5. Supply air requirements shall have a maximum efficiency rating value ("MERV") of 14 in accordance with ASHRAE 52.2.
 - 6. Outside air requirements shall be consistent with ASHRAE Standard 62-1989.
- 7. The energy management system shall be capable of allowing Tenant to request floor by floor after hours HVAC and lighting via direct telephone interface, 24 hours daily.

H. ELECTRICAL AND POWER SYSTEM

- 1. Electrical services shall be in accordance with serving utility requirements including but not limited to all substructures, backboards, and switchgear, for complete operating systems.
- 2. Landlord shall provide the main power service from utility provider to the entry point of the Building, which shall be at 277/480 volts, three-phase, four wire located in a room within the Building.
- 3. Landlord shall furnish on the floors occupied by Tenant all electrical feeders, transformers, distribution panels and branch circuit breaker panels to support the following load requirements:
 - (a) Connected Load / 208V/120V Utility Power: 8 watts/rsf with the number of branch circuits provided on the basis of one (1) 20 ampere, single pole, 120-volt circuit breaker per 100 rsf.

- (b) 227V/480V lighting power: 1.5 watts/rsf with the number of branch circuits based on a load of 3000 watts per 20 ampere, 277 volt branch circuit.
- (c) Power for HVAC: As required by Building to support stated electrical demand load.
- 4. Demand Load (Consumed): An average of 8 watts/rsf] consumed power on a monthly basis, such average calculated based on Normal Business Hours (excluding power for HVAC).
 - 5. Emergency Power: Landlord shall provide emergency power to operate all essential building services per codes including but not limited to:
 - (a) Emergency elevator service;
 - (b) Egress lighting;
 - (c) Fire life safety systems;
 - (d) Mechanical Smoke Control Equipment.

I. TOILET ROOMS

- 1. Women's and men's toilet rooms, on the floors occupied by Tenant, shall be designed and constructed in compliance, as to the Base Building, with current (as of the Commencement Date) code requirements and laws for size and quantity, including the Americans with Disabilities Act. The minimum level of design and finish (subject to Tenant's approval rights in the Work Letter) shall not be less than the following:
 - (a) City water shall be provided for all toilet rooms.
 - (b) All vertical wetwalls shall be finished with full height ceramic tile.
 - (c) Floors shall be finished with ceramic tile and include recessed floor drains.
 - (d) The ceilings shall be painted drywall.
 - (e) Metal toilet partitions shall be floor mounted with concealed latch and coat hooks.
 - (f) Urinal partitions shall be wall mounted.
 - (g) Urinals shall be wall mounted in all rest rooms.
 - (h) All fixtures and accessories shall be stainless steel and meet current governmental laws, codes and regulations, including without limitation, the Americans with Disabilities Act and shall include, but shall not be limited to:
 - (1) Recessed seat cover dispenser.
 - (2) Recessed paper towel dispenser/waste receptacles.
 - (3) Recessed feminine napkin vendor.
 - (4) Partition mounted roll toilet tissue dispensers.
 - (5) Handicap grab bar as required by code.
 - (6) Lavatory soap dispensers.
 - 2. The layout of each restroom facility within the Building shall be in accordance with applicable laws and codes.

J. FIRE AND LIFE SAFETY SYSTEMS

- 1. Base Building fire and life safety systems shall meet all applicable code requirements, including but not limited to California State Title 24 and Americans with Disabilities Act, as of the Commencement Date, including the following improvements for the Building Common Areas (exit stairwells, exit corridors, lobbies):
 - (a) Exit signs.

- (b) Fire extinguishers and cabinets.
- (c) Life safety speakers/strobe lights and smoke detectors.
- 2. Fire sprinkler loop with points of connection within Tenant space including a main loop and branch distribution piping, including mains and laterals (drops and heads to be at Tenant's cost as part of the Tenant Improvement work), as required by local code on each floor.
 - 3. Standpipe branch connection into Premises for code-required tenant hose valves stations within Premises.
- 4. Capacity only for fire alarm inputs and control outputs and for fire alarm sounding devices as required to be audible throughout the Premises and strobe light capacity in accordance with current code (as of the Commencement Date).

K. SECURITY ACCESS SYSTEMS

- 1. Card access devices provided at all Building entrances and exits, Building elevators, and entrances and exits to parking areas. Such system shall include recording and monitoring devices located within a central location, to be determined by Landlord and Tenant, in the Building.
- 2. All Building security equipment and devices may be separable from Tenant's security system(s), and Landlord and Tenant will work cooperatively to design their security systems to be compatible.

L. COMMUNICATION SYSTEMS & TELECOMMUNICATIONS ACCESS

- 1. A main telephone terminal room in the Building shall be provided with multiple feeder ducts and service from the telephone company. Landlord shall provide conduits from this terminal room in the Building to the main telephone risers and conduits that service the Premises. Fire resistant plywood telephone backboards shall be provided in each telephone room and/or riser closet.
 - 2. Fire sealed floor openings shall be provided in the telephone room on each floor of the Building for additional risers, conduits and cables.

M. ELEVATORS

- 1. 4 passenger elevators. Elevator cars in the Building shall be equipped with security card readers. All elevator cars, lobby call lanterns and call buttons shall be in compliance with all applicable Laws in effect on the Commencement Date.
- 2. One (1) separate swing service elevator (deeper than it is wide) with adequate size and weight capabilities to accommodate the efficient loading and unloading of building materials and standard office equipment. This elevator shall provide access to all floors of the Building and basement levels of the Building and have a rear access door on each floor into a service lobby.
 - 3. Passenger elevator speed shall be a minimum average of 350' per minute.

N. LOADING DOCK; TRASH AREAS

Tenant shall have full use of the Building's loading dock bays. All loading dock activity shall be operated and maintained by Landlord on a nondiscriminatory basis.

O. WALLS AND WINDOWS

- 1. Curtain walls shall be installed and sealed.
- 2. Exterior windows shall be installed and sealed.
- 3. Landlord shall provide "Mecco" window blinds or the equivalent.

P. SOUND ATTENUATION

Floor assemblies and the Building structure shall be constructed with sufficient provisions for thermal insulation and acoustic control to limit noise transmission generated from Building equipment.

Q. ESSENTIAL SITE IMPROVEMENTS

The "Essential Site Improvements" as referenced in the Work Letter shall consist of the following:

- 1. Building I as shown on the Site Plan;
- 2. 6-Level Parking Structure with a minimum capacity for 1,000 stalls;
- 3. A main driveway from the signalized intersection at Jamboree Road, crossing the project and connecting to Fairchild Road;
- 4. A secondary driveway (which may include a valet station);
- 5. A service driveway with access to the sub-grade parking below the Building (the service drive will include a trash enclosure, loading dock, emergency generator and pads for Tenant supplied back-up generators);
- 6. Landscape and hardscape of common areas surrounding the Building, including outdoor smoking areas away from the primary Building entries, seating and waste containers and assembly area (amphitheatre). All landscaped areas to be irrigated with electrically controlled automatic systems.
- 7. Storm drainage connected to the City storm system main, underground sanitary sewer laterals connected to the City sewer main utility service laterals, domestic water mains connected to the city water main, gas and electrical lines connected to the utility mains, curbs, gutters, sidewalks and exterior lighting per code.

Schedule 2 to Work Letter Design Schedule

DESIGN DEVELOPMENT

•	Design Development	3/4/05-4/18/05
•	Tenant Review	4/18/05-4/27/05
•	Complete Design Development	4/27/05-5/6/05

CONSTRUCTION DOCUMENTS

•	Construction Documents	5/6/05-7/6/05
•	Tenant Review	7/6/05-7/18/05
•	Complete Construction Documents	7/18/05-7/28/05

PERMITS

•	Architectural Plan Check	6/6/05-7/21/05
•	Plan Check Corrections	7/21/05-8/4/05
•	Shell Permit	8/4/05-8/15/05

EXHIBIT D

-16-

Schedule 3 to Work Letter Schedule of Milestones

GRADING

•	Grading Permit	4/6/05-6/1/05
•	Grading/Basement Excavation	6/1/05-6/13/05

FOUNDATION

•	Foundation Permit	5/6/05-6/13/05
•	Foundation/Grade Beams	6/13/05-7/11/05

<u>STEEL</u>

•	Complete Mill Order Design	3/15//05-4/15/05
•	Steel Fabrication	5/23/05-7/8/05
•	Steel Erection	7/11/05-8/15/05

EXTERIOR SKIN

Precast/Glazing
 Roof deck
 8/15/05-10/26/05
 8/15/05-10/3/05

EXHIBIT D -17-

SCHEDULE 4 to EXHIBIT D

TENANT IMPROVEMENT STANDARDS

NOTE: THESE ITEMS HAVE NOT YET BEEN REVIEWED FOR LEED COMPLIANCE ISSUES AND THUS ARE SUBJECT TO CHANGE.

TENANT AREA

Demising Partitions

- A. 35/8" x 20 ga metal studs at 16" oc. extended full height to underside of floor or roof deck with deflection slip track at top. Fire caulk top track to deck.
- B. 5/8" type 'X' gypsum wallboard full height, one layer each side of wall.
- C. Full height R-13 batt insulation.
- D. Partitions to be taped smooth and sanded to receive paint or wallcovering finish.
- E. Straight line termination at building columns and window mullions with sound seal neoprene gasket closure/fillers.
- F. Stagger electrical outlets and other gang box wall penetrations.
- G. Caulk entire partition perimeter along floor line, top of wall to deck intersection and at perimeter columns and window mullion intersected surfaces.
- H. Sound boots at HVAC penetrations.
- I. Full height tenant demising walls to be reviewed by Owner's smoke management engineer for compatibility with Base Building smoke management report and event matrix.

2. Interior Partitions

- A. 35/8 x 20 ga. metal studs at 24" or 16" oc (L/240 deflection criteria), braced to structure above.
- B. 5/8" type 'X' gypsum wallboard, one layer each side of wall.
- C. Partition taped smooth to receive paint or wall covering.
- D. Sound sealed gasket closure at mullion terminations. <u>Do not</u> anchor to window wall system.

Perimeter Wall Furring

- A. $2^{1/2}$ " x 20 ga. metal studs at 24" oc, braced to structure above.
- B. 5/8" type 'X' gyp-board taped smooth to receive paint or wall covering.
- C. R-8 batt insulation
- D. Sound sealed gasket closure at mullion terminations. <u>Do not</u> anchor to window wall system.

4. Perimeter Window Wall Sill Furring

A. Custom gypsum wallboard adapter with clear anodized finish to match window wall system.

Tenant to purchase custom adapter, if needed, from Building Owner's supplier.

- B. $1^{5}/8$ " x 20 ga in-fill furring at 24" oc punched for utility conduits. R-8 batt insulation.
- C. One layer of 5/8" type 'X' gyp-board taped smooth to receive paint or wall covering.

Doors

- A. Size: 3'-0" x 8'-10"
- B. Finish: Quarter sliced cherry (or equivalent) wood veneer with clear gloss finish.
- C. 20 minute label where required.

Door Frames

- A. Interior Tenant Space: Western Integrated clear anodized aluminum frame, factory finish, 1 ³/₄" profile.
- B. 20 minute label where required.

7. Door Hardware

- A. Lockset / Latchset / Cylinder / Strike: Schlage L9000 series, 626 satin chromium finish.
- B. Lever: Schlage L–17A, 626 satin chromium finish. 40" mtg. Ht.
- C. Hinges: $4^{1/2}$ " x $4^{1/2}$ " Hager BB1279, satin finish (4 per door).
- D. Closers: Norton 8500 series, 652 satin chromium finish, with ADA calibrator.
- E. Stops: Trimco W1276 CCS, satin finish.

8. Ceilings

A. Ceiling Tiles: Armstrong Cirrus Open Plan #558 w/ beveled tegular edge profile NRC - 0.75, AC - 170 (or the equivalent), 24" x 24" x 7/8" tile. Color: white finish.

B. Ceiling Grid: Armstrong Silhouette XL 9/16" bolt-slot system w/ 1/8" reveal (or the equivalent). Color: white finish.

C. Ceiling Heights: 1st Floor: 15'-0" 2nd thru 7th Floor: 12'-0"

9. Flooring

A. Carpeting: Shaw, Cypress Point II 32oz (or the equivalent). Installation: Over 3/8" Nova pad (or the equivalent). Color: Selected by Tenant from Mfr.'s standard range.

B. VCT: Armstrong Excelon (or the equivalent). Color: Selected by Tenant from Mfr.'s standard range.

C. Base: Burke 2 1/2" (or the equivalent) coveless in carpeted areas, coved in hard surface areas. Color: Selected by Tenant from Mfr.'s standard range.

10. Exterior Window Wall Coverings

A. Mechanism: MechoShade Systems Mecho III Bracket, manually chain operated, maximum shade height 12'-0" (ground floor 15'). Use MechoShade Extended Bracket (where required) for shade heights exceeding 12'-0" (ground floor 15') or shades with blackouts exceeding 8'-0". (or equivalent)

Installation: Recessed / Flush ceiling installation with removable Mecho Snap-Loc Fascia enclosure plates and end fillers. Bracket assembly anchored to metal stud framing suspended from structure above.

Fascia Type: Closure assembly as required for gyp-board or lay-in ceiling tile installation.

Fascia / Closure Assembly Finish: Clear anodized to match window wall system.

Shade Material: MechoShade EuroVeil, Dense Basket Weave, 5% openness. Color / Pattern: # 5306 Silver Birch. (or equivalent)

B. Fry-Reglet, 'Z' edge molding (where required) between Mecho Bracket and window wall system for Bracket alignment. Clear anodized finish to match window wall system. <u>Do not</u> anchor to window wall system. Z molding to be mounted integral with Mecho shade assembly. Where used, 'Z' molding to run continuous between partitions or columns

11. Painting

A. Two coats of finish paint over primer. Dunn-Edwards (or the equivalent). Color selected by Tenant from Mfr's. standard color range.

12. Lighting Fixtures

A. Peerless – Peerlite – Cerra 7 suspended light fixtures with two (2), 28 watt T5 lamps. (or equivalent). Fixtures at perimeter building wall spaces (alignment and direction) to be approved by Building Owner. Lighting fixtures to meet title 24 requirements.

13. Lighting Controls

- A. Motion Sensors: Novitas model 01-100 series. White finish cover plate.
- B. Wall Switch Light Controls: General Electric or equal, white finish cover plate. Vertically mounted 40" AFF. Switched / paired in double gang box to meet title 24 requirements.

14. Electrical Wall Outlets

A. General Electric or equal, self grounding duplex type, white finish cover plate. Vertically mounted 15" AFF

15. Telephone, Computer Data and Cable Wall Outlets

A. Single gang box, white finish cover plate. Vertically mounted 15" AFF

16. HVAC

- A. 2' x 2' perforated diffusers in accessible ceiling areas, linear diffusers in drywall ceiling areas. Color: White finish.
- B. Thermostats located per title 24 zone requirements. Mounted 40" AFF. White finish cover.
- C. Tenant's HVAC design to be compatible with Base Building system and smoke management requirements. Tenant's design to be reviewed and approved by Building Owner's mechanical and smoke control engineers.

17. Fire Extinguisher Cabinets

A. Potter-Roemer, satin stainless steel finish frame, clear glass view.

18. Sprinklers / Smoke & Fire Detectors / Life Safety Devices

- A. Sprinklers: Semi-recessed chrome heads in ceiling areas. Type: Quick response compatible with Base Building smoke management and event matrix report.
- B. Smoke / Fire Detectors: Compatible with Building smoke management and event matrix report. White finish cover housing where visible.
- C. All life / safety features to be reviewed by Owner's smoke control engineer for compatibility with Base Building design.

19. Exit Signs

A. Isolite, 2040-50-20, self-illuminated emergency type. White finish frame with green letters.

EXHIBIT E JANITORIAL SPECIFICATIONS

Janitorial Requirements Lobby

SERVICE]	Daily	Weekly	Monthly	Quarterly	Yearly
Vacuum all carpeted areas		•				
Spot clean carpets		•				
Sweep, damp mop all floored surfaces						
Buff floored areas to insure high luster appearance				•		
Clean ledges and moldings		•				
Clean security station						
Clean chrome/anodized metal finishes						
Clean entry glass (inside and out) up to 10'						
Clean entry thresholds		•				
Dust mullions and sills		•				
Clean baseboards with detergent and water			•			
Empty waste receptacles						
Police entrance (int and ext) for trash and debris						
Sweep all stairwells and corridors						
Clean and dust stairwell railings						
Strip and wax tile floors				•		
Dust furnishings		•				
Dust artwork			•			
Polish wood furnishings				•		

EXHIBIT E

-1-

Janitorial	Requirements

Work Areas

SERVICE	Daily	Weekly	Monthly	Quarterly	Yearly
Empty waste baskets and replace liners as needed	.			· 	
Vacuum carpeted areas, to include around and beneath desks	•				
Spot clean doors, frames and switchplates	•				
Dust horizontal, including artwork and furnishings, surfaces (up to 6')	•				
Dust baseboards	•				
Spot clean carpet	•				
Dust all surfaces (higher than 6')					
Spot clean partitions, walls glass and windows					
Clean chrome/anodized metal finishes	•				
Wash vinyl and metal door kick plates	•				
Vacuum/clean ceiling air supply and exhaust diffusers/grills				•	
Clean ledges and moldings	•				
Polish wood furnishings					
Dust mullions and sills	•				
FYHIRIT F					

EXHIBIT E

-2-

Janitoriai Requirements	Janitorial	Requirements
-------------------------	------------	--------------

Rest Rooms

SERVICE	Daily	Weekly	Monthly	Quarterly	Yearly
Clean, disinfect and polish toilets and urinals					
Damp wipe/disinfect ledges, counters, stalls, and smooth surfaces	•				
Sweep and mop with germicide floored surfaces	•				
Clean, disinfect and polish mirrors, soap dispensers and chrome fixtures	•				
Clean, disinfect and polish sinks and faucets	•				
Clean and disinfect dispensers	•				
Furnish/refill soap, paper and product dispensers					
Clean, disinfect and replace liners in trash and product waste receptacles					
Clean all baseboards with a germicidal cleanser					
Remove and clean urinal screens*					
Clean and flush floor drains*					
Vacuum ceiling air supply and exhaust diffusers/grills					
Strip, disinfect and wax floored surfaces					
Vacuum/clean ceiling air supply and exhaust diffusers/grills*					
Polish chrome and porcelain surfaces			•		

^{*} Replaced as needed.

EXHIBIT E

Janitorial Requirements

Lounge, Copy and Conference Rooms

SERVICE	Daily	Weekly	Monthly	Quarterly	Yearly
Dust/polish conference tables	-				
Vacuum chairs			•		
Empty trash receptacles					
Clean, remove any remaining food or beverage/trash					
Vacuum carpeted areas	•				
Clean/disinfect counter surfaces	•				
Mop/disinfect floored surfaces	•				
Spot clean doors, frames and switchplates	•				
Empty waste receptacles and replace liners as needed	•				
Dust horizontal surfaces (up to 6')	•				
Clean baseboards with detergent and water	•				
Spot clean carpet	•				
Sweep, damp mop floored surfaces	•				
Buff floored areas				•	
Strip, disinfect and wax floors			•		
Clean refrigerator*			•		
Vacuum ceiling air supply and exhaust diffusers and grills				٠	

[to be supplied]

EXHIBIT E

-4-

EXHIBIT F

INTENTIONALLY OMITTED

EXHIBIT F

EXHIBIT G

RECORDING REQUESTED BY AND WHEN RECORDED RETURN TO:

RECOGNITION OF COVENANTS, CONDITIONS, AND RESTRICTIONS

This Recognition of Covenants, Conditions, and Restrictions (this "Agreement") is entered into as of the day of, 200, by an ("Landlord"), and ("Tenant"), with reference to the following facts:	d between
A. Landlord and Tenant entered into that certain Office Lease (Building II) dated, 2005 (the "Lease"). Pursuant to the Lease, Land to Tenant and Tenant leased from Landlord space (the "Premises") located in an office building on certain real property described in Exhibit A attack and incorporated herein by this reference (the "Property").	
B. The Premises are located in an office building located on real property which is part of an area owned by Landlord containing approximate () acres of real property located in the City of Irvine, California (the "Project"), as more particularly described in Exhibit B attached hereto and in herein by this reference.	-
C. Landlord or its predecessor in interest, as declarant, has previously recorded, or proposes to record concurrently with the recordation of this a Declaration of Covenants, Conditions, and Restrictions (the "Declaration"), dated,, in connection with the Project.	Agreement,
D. Tenant is agreeing to recognize and be bound by the terms of the Declaration, and the parties hereto desire to set forth their agreements consame.	cerning the
NOW THE PERPERPER.	1 11

NOW, THEREFORE, in consideration of (a) the foregoing recitals and the mutual agreements hereinafter set forth, and (b) for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

1. <u>Tenant's Recognition of Declaration</u>. Notwithstanding that the Lease has been executed prior to the recordation of the Declaration, Tenant agrees to recognize and by bound by all of the terms and conditions of the Declaration.

Miscellaneous.

- 2.1 This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective heirs, estates, personal representatives, successors, and assigns.
- 2.2 This Agreement is made in, and shall be governed, enforced and construed under the laws of, the State of California.
- 2.3 This Agreement constitutes the entire understanding and agreements of the parties with respect to the subject matter hereof, and shall supersede and replace all prior understandings and agreements, whether verbal or in writing. The parties confirm and acknowledge that there are no other promises, covenants, understandings, agreements, representations, or warranties with respect to the subject matter of this Agreement except as expressly set forth herein.
- 2.4 This Agreement is not to be modified, terminated, or amended in any respect, except pursuant to any instrument in writing duly executed by both of the parties hereto.
- 2.5 In the event that either party hereto shall bring any legal action or other proceeding with respect to the breach, interpretation, or enforcement of this Agreement, or with respect to any dispute relating to any transaction covered by this Agreement, the losing party in such action or proceeding shall reimburse the prevailing party therein for all reasonable costs of litigation, including reasonable attorneys' fees, in such amount as may be determined by the court or other tribunal having jurisdiction, including matters on appeal.
- 2.6 All captions and heading herein are for convenience and ease of reference only, and shall not be used or referred to in any way in connection with the interpretation or enforcement of this Agreement.
- 2.7 If any provision of this Agreement, as applied to any party or to any circumstance, shall be adjudged by a court of competent jurisdictions to be void or unenforceable for any reason, the same shall not affect any other provision of this Agreement, the application of such provision under circumstances different form those adjudged by the court, or the validity or enforceability of this Agreement as a whole.

EXHIBIT G

- 2.8 Time is of the essence of this Agreement.
- 2.9 The Parties agree to execute any further documents, and take any further actions, as may be reasonable and appropriate in order to carry out the purpose and intent of this Agreement.
- 2.10 As used herein, the masculine, feminine or neuter gender, and the singular and plural numbers, shall each be deemed to include the others whenever and whatever the context so indicates.

EXHIBIT G

SIGNATURE PAGE OF RECOGNITION OF COVENANTS, CONDITIONS AND RESTRICTIONS

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the day and year first above written.

"Landlo	·d":		
By:			
Name:			
Title:			
"Tenant'	' :		
By:			
Name:			
Title:			
D			
By:			
Name:			
Title:			

EXHIBIT G

EXHIBIT H

FORM OF SUBORDINATION, NON-DISTURBANCE AND ATTORNMENT AGREEMENT

RECORDING REQUESTED BY	
AND WHEN RECORDED MAIL	
TO:	
VALLEY INDEPENDENT BANK	
74-199 El Paseo Drive, Suite 102	
Palm Desert, California 92260	
Attn: Steve Stearman, Regional Vice President	

Space above for Recorder's Use

SUBORDINATION, NONDISTURBANCE, AND ATTORNMENT AGREEMENT

This Subordination, Nondisturbance, and Attornment Agreement ("Agreement") is entered into as of _______, 2005, by and among Valley Independent Bank, a California corporation ("Bank"), Scholle Jamboree Property Development I LLC, a California limited liability company, as landlord ("Landlord"), and Impac Funding Corporation, a California corporation ("Tenant").

RECITALS

- A. Landlord and Tenant have entered into a certain Office Lease (Building II) dated February ___, 2005 (the "Lease") covering certain premises ("Demised Premises"), located at the corner of Jamboree and Fairchild Roads, Irvine, California ("Property"). The Property is more particularly described in Exhibit A attached hereto and incorporated herein.
- B. Bank is or will be the beneficiary of a deed of trust ("Deed of Trust") encumbering the Property, which secures a note payable to Bank in the original principal amount of Twenty Nine Million Dollars (\$29,000,000.00).
- C. On the terms and conditions in this Agreement, the parties desire to subordinate Tenant's leasehold interest in the Demised Premises to the lien of the Deed of Trust and to assure Tenant possession of the Demised Premises for the entire term of the Lease, even though Bank may foreclose the lien of the Deed of Trust before expiration of the Lease.

Therefore, in consideration of the mutual covenants and agreements contained in this Agreement, the parties agree as follows:

Section 1. Subordination

The Lease (including any rights of first refusal or purchase options contained therein) shall be subject and subordinate to the lien of the Deed of Trust and to any renewals, modifications, consolidations, replacements and extensions of the Deed of Trust to the full extent of the principal sum secured by the Deed of Trust, including any interest.

Section 2. Nondisturbance.

So long as Tenant is not in default, beyond any period given to Tenant to cure a default, in the payment of rent or in the performance of any of the terms, covenants, or conditions of the Lease, Tenant's possession of the Demised Premises and Tenant's rights and privileges under the Lease, including any extensions or renewals, shall not be diminished or interfered with by Bank during the term of the Lease or any extensions or renewals. So long as Tenant is not in default, beyond any period given Tenant to cure such default, in the payment of rent or in the performance of any of the terms, covenants, or conditions of the Lease, Bank will not join Tenant as a party for the purpose of termination or otherwise affecting Tenant's interest under the Lease, in any action of foreclosure or other proceeding brought by Bank to enforce any rights arising because of any default under the Deed of Trust. Bank may, however, join Tenant as a party if joinder is necessary under any statute or law to secure the remedies available to Bank under the Deed of Trust, but joinder shall be for that purpose only and not for the purpose of terminating the Lease or affecting Tenant's right to possession of the Demised Premises.

Section 3. Attornment

Tenant hereby agrees that if Bank elects at any time to have the Lease superior to its Deed of Trust and gives notice of its election to Tenant, then the Lease shall be superior to the lien of any such Deed of Trust and all renewals, modifications, extensions, substitutions, replacements and/or consolidations thereof, whether the Lease is dated or recorded before or after the Deed of Trust. If Bank shall become the owner of the Demised Premises, or if the Demised Premises shall be sold by reason of foreclosure or other proceedings brought to enforce the Deed of Trust, or if the Demised Premises shall be transferred by deed in lieu of foreclosure. Tenant shall attorn to Bank or

any other such owner as its Landlord, said attornment to be effective and self operative without the execution of any further instruments. From and after Bank's or other such owner's succession to the interest of Landlord under the Lease, Tenant shall have the same remedies against Bank or such other owner for the breach of any covenant contained in the Lease that Tenant might have had under the Lease against Landlord, except that neither Bank nor any other such owner shall be:

- (a) liable for any act or omission of, or for the performance of the obligation of, any prior lessor (including Landlord) under the Lease (except to the extent Bank continues such act or omission after Bank succeeds to Landlord's interest in the Demised Premises; provided, however, that the foregoing exception shall not apply to any unperformed construction obligations of Landlord under the Lease, as to which Bank shall have no responsibility even after succeeding to Landlord's interest in the Property); or
 - (b) subject to any offsets or defenses which Lessee might have against any prior landlord (including Landlord); or
- (c) bound by any prepayment of rent or additional rent which Tenant might have paid for more than the current month or by payment of any security deposits to any prior landlord (including Landlord), except such security deposits as have actually been received by Bank; or
- (d) bound by any amendment or modification of the Lease or by any waiver or forbearance on the part of any prior tenant (including Tenant) made or given without the written consent of Bank or any subsequent holder of the Deed of Trust; or
 - (e) bound by any representations or warranties of Landlord under the Lease.

Section 4. Bankruptcy of Landlord

In the event of the bankruptcy of Landlord and the termination of the Lease in connection with such proceeding, Tenant shall, upon the request of Bank (or any other owner of the Demised Premises), enter into a new lease for the Demised Premises on substantially the same terms and conditions as the Lease.

Section 5. Definitions.

The term "Bank" or any similar term shall include Bank, the trustee under any deed of trust affecting the Demised Premises, and any agents, heirs, successors or assigns, including any party that succeeds to Landlord's interest by foreclosure of the Deed of Trust, of the deed in lieu of foreclosure, or of a sale under a private power contained in the Deed of Trust or by any other proceeding. The term "Deed of Trust" or any similar term shall include the Deed of Trust and any amendments or addenda. The term "Landlord" shall include Landlord and its successors and assigns. The term "Tenant" shall include Tenant and the successors, assigns, and sublessees of Tenant. This Agreement shall inure to the benefit of and be binding upon all successors, assigns and sublessees. The term "Lease" shall include the Lease and all amendments, addenda, extensions, and renewals.

Section 6. No Change in Lease.

Landlord and Tenant agree not to materially change, alter, amend or otherwise modify the Lease without the prior written consent of Bank, which consent shall not be unreasonably withheld or delayed. Any material change, alteration, amendment, or other modification to the Lease without the prior written consent of Bank shall be void as to Bank.

Section 7. Insurance and Condemnation Proceeds.

In the event of a casualty to any improvements on the Demised Premises or a condemnation or taking under a power of eminent domain of all or any portion of the Demised Premises or any of the improvements on the Demised Premises all payments and settlements of insurance claims or condemnation awards (except for any payments and settlements derived from Tenant's insurance policies that Tenant is entitled to retain pursuant to the Lease) shall be paid to the Bank for the use and application pursuant to the terms and provisions of the Deed of Trust.

Section 8. Assignment of Lease or Rents.

Landlord will by a separate assignment ("Assignment") assign its interest in the rents and payments due under the Lease to Bank as security for repayment of the Loan. If in the future there is a default by the Landlord in the performance and observance of the terms of the Deed of Trust, the Bank may, at its option under the Assignment, require that all rents and other payments due under the Lease be paid directly to it. Upon notification to that effect by the Bank, Landlord hereby authorizes and directs Tenant and the Tenant agrees to pay the rent and any payments due under the terms of the Lease to Bank. Landlord hereby acknowledges and agrees that Tenant shall have the right to rely on any notice by Bank that all rents and other payments due under the terms of the Lease shall be paid to Bank. The Assignment does not diminish any obligations of the Landlord under the Lease nor impose any such obligations on the Bank.

Section 9. Landlord Default; Notice to Bank

Tenant shall promptly provide Bank at its address set forth below with a written notice of any default on the part of the Landlord under the Lease. Bank shall have the option to cure such default within the time allotted to Landlord under the Lease plus ten (10) days in the case of a monetary default and thirty (30) days in the case of a non-monetary default. Tenant shall not invoke any of its remedies under this Lease or any other remedies available to Tenant at law or in equity during any period that Bank is proceeding to cure any such default with due diligence or (if possession of the Demised Premises is necessary for such cure to be effectuated) during any period that Bank is taking steps with due diligence to obtain the legal right to enter the Demised Premises and cure any such default.

Section 10. Bank Consent Required

Without the prior written consent of Bank, Tenant shall not (a) pay the rent or any other sums becoming due under the terms of the Lease more than one month in advance; (b) accept Landlord's waiver of or release from the performance of any material obligation under the Lease, (c) assign the Lease as collateral security or mortgage or otherwise encumber its leasehold interest, or (d) agree with Landlord to terminate the Lease except as permitted pursuant to the express terms of the Lease. Landlord's consent as to items (b) and (c) above shall not be unreasonably withheld or delayed.

Section 11. Notices.

In this Agreement, whenever it is required or permitted that notice and demand be given by any party to another party, that notice or demand shall be given in writing and forwarded by certified mail, addressed as follows:

For Landlord: Scholle Jamboree Property Development I LLC

19500 Jamboree Road Irvine, California 92612 Attn: Robert W. Scholle

For Tenant: Impac Funding Corporation

1401 Dove Street

Newport Beach, CA 92660 Attention: General Counsel

For Bank: Valley Independent Bank

74-199 El Paseo Drive, Suite 102 Palm Desert, California 92260

Attn: Steve Stearman, Regional Vice President

Any party may change an address given for notice by giving written notice of that change by certified mail to all other parties.

Section 12. Authority

If any party is a corporation, limited liability company or a partnership, all individuals executing this Agreement on behalf of a corporation, limited liability company or partnership represent and warrant that they are authorized to execute and deliver this Agreement on behalf of the corporation, limited liability company, or partnership and that this Agreement is binding upon the corporation, limited liability company, or partnership.

Section 13. Miscellaneous.

This Agreement may not be modified other than by an agreement in writing signed by the parties or by their respective successors in interest. If any party commences any action against any other party based on this Agreement, the prevailing party shall be entitled to recover reasonable attorney fees, expenses, and costs of suit. This Agreement shall be binding on and inure to the benefit of the parties and their respective heirs, successors and assigns. The headings of this Agreement are for reference only and shall not limit or define any meaning of this Agreement. This Agreement may be executed in one or more counterparts, each of which is an original, but all of which shall constitute one and the same instrument. This Agreement shall be construed in accordance with and governed by California law.

[SIGNATURES FOLLOW THIS PAGE]

The parties have duly executed this Agreement as of the date	first above written.	
BANK:	VALLEY INDEPENDENT BANK	
	Ву:	
	Steve Stearman Regional Vice President	
LANDLORD:	SCHOLLE JAMBOREE PROPERTY DEVELOPMENT I LLC a California limited liability company	
	Ву:	
	Robert W. Scholle Its: Manager	
TENANT:	IMPAC FUNDING CORPORATION, a California corporation	
	Ву:	
	Name:	
	Title:	
	EXHIBIT H 4	

EXHIBIT A PROPERTY DESCRIPTION

The real property referred to in this Exhibit is situated in the State of California, County of Orange, described as follows:

PARCEL A:

Parcel 2, in the City of Irvine, County of Orange, State of California, as shown on a Parcel Map filed in Book 47, Page 22 of Parcel Maps, in the office of the County Recorder of said County.

Except all oil, oil rights, minerals, minerals rights, natural gas, natural gas rights, and other hydrocarbons by whatsoever name known that may be within or under the Parcel of land hereinabove described. Together with the perpetual right of drilling, mining, exploring and operating therefor and storing in and removing the same from said land or any other land, including the right to whipstock or directionally drill and mine from lands other than those hereinabove described, oil or gas wells, tunnels and shafts into, through or across the subsurface of the land hereinabove described, and to bottom such whipstocked or directionally drilled wells, tunnels and shaft under and beneath or beyond the exterior limits thereof, and to redrill, retunnel, equip, maintain, repair, deepen, and operate any such wells or mines, without however, the right to drill, mine, store, explore and operate through the surface or the upper five hundred (500) feet of the subsurface of the land hereinabove described, as reserved by Irvine Industrial Complex, a California Corporation in deed recorded January 31, 1977 as Instrument No. 40776 in Book 12052, Page 1304 of Official Records.

PARCEL B:

An easement for the purposes of ingress and egress and sanitary sewer over and under that portion of Block 50 of Irvine's Subdivision, in the City of Irvine, County of Orange, State of California, shown as Parcel 1 on map filed in Book 22, Page 28 of Parcel Maps, in the office of the County Recorder of said County, described as follows:

Commencing at the most Northerly corner of Parcel 1 as shown on a map filed in Book 22, Page 28 of Parcel Maps, records of Orange County; thence South 49° 21' 34" East 120.0 feet along the Northeast Lot Line of said Parcel 1 to the true point of beginning; thence South 40° 38' 26" West 30.00 feet; thence South 49° 21' 34" East 189.37 feet to a point on the Easterly lot line of said Parcel 1; thence along said lot line North 6° 59' 16" West 44.52 feet to the Northeast corner of said Parcel 1; thence North 49° 21' 34" West 156.48 feet along the Northeast lot line of said parcel 1 to the true point of beginning.

APN: 445-132-06

On, 2005, before me,, personally appeared, personally known to me (or proved to me on the basis of satisfactory evidence) to be the person whose name is subscribed to the within instrument and acknowledged to me that he executed the same in his authorized capacity, and that by his signature on the instrument the person, or the entity upon behalf of which the person acted, executed the instrument. WITNESS my hand and official seal. NOTARY PUBLIC STATE OF CALIFORNIA) ss. COUNTY OF) personally appeared, personally known to me (or proved to me on the basis of satisfactory evidence) to be the person whose name is subscribed to the within instrument and acknowledged to me that he executed the same in his authorized capacity, and that by his signature on the instrument the person, or the entity upon behalf of which the person acted, executed the instrument. WITNESS my hand and official seal.
capacity, and that by his signature on the instrument the person, or the entity upon behalf of which the person acted, executed the instrument. WITNESS my hand and official seal. NOTARY PUBLIC STATE OF CALIFORNIA) ss. COUNTY OF) personally appeared, personally known to me (or proved to me on the basis of satisfactory evidence) to be the person whose name is subscribed to the within instrument and acknowledged to me that he executed the same in his authorized capacity, and that by his signature on the instrument the person, or the entity upon behalf of which the person acted, executed the instrument.
NOTARY PUBLIC STATE OF CALIFORNIA) ss. COUNTY OF, personally appeared, personally known to me (or proved to me on the basis of satisfactory evidence) to be the person whose name is subscribed to the within instrument and acknowledged to me that he executed the same in his authorized capacity, and that by his signature on the instrument the person, or the entity upon behalf of which the person acted, executed the instrument.
STATE OF CALIFORNIA)) ss. COUNTY OF) ss. On, 2005, before me,, personally appeared, personally known to me (or proved to me on the basis of satisfactory evidence) to be the person whose name is subscribed to the within instrument and acknowledged to me that he executed the same in his authorized capacity, and that by his signature on the instrument the person, or the entity upon behalf of which the person acted, executed the instrument.
On, 2005, before me,, personally appeared, personally known to me (or proved to me on the basis of satisfactory evidence) to be the person whose name is subscribed to the within instrument and acknowledged to me that he executed the same in his authorized capacity, and that by his signature on the instrument the person, or the entity upon behalf of which the person acted, executed the instrument.
On, 2005, before me,, personally appeared, personally known to me (or proved to me on the basis of satisfactory evidence) to be the person whose name is subscribed to the within instrument and acknowledged to me that he executed the same in his authorized capacity, and that by his signature on the instrument the person, or the entity upon behalf of which the person acted, executed the instrument.
of satisfactory evidence) to be the person whose name is subscribed to the within instrument and acknowledged to me that he executed the same in his authorized capacity, and that by his signature on the instrument the person, or the entity upon behalf of which the person acted, executed the instrument.
WITNESS my hand and official seal.
NOTARY PUBLIC
STATE OF CALIFORNIA)) ss.
COUNTY OF
On, 2005, before me,, personally appeared, personally known to me (or proved to me on the basis of satisfactory evidence) to be the person whose name is subscribed to the within instrument and acknowledged to me that he executed the same in his authorized capacity, and that by his signature on the instrument the person, or the entity upon behalf of which the person acted, executed the instrument.
WITNESS my hand and official seal.
NOTARY PUBLIC

EXHIBIT I

SIGNAGE MAINTENANCE REQUIREMENTS

- 1. Tenant shall not install the Building Top Signage or thereafter replace or make alterations to the Building Top Signage until: (a) Landlord has approved in writing, which approval shall not be unreasonably withheld or delayed, the sign planner, engineer and installation company and professionally prepared sign plans submitted by Tenant showing the design, size, content, color, illumination and quality of materials and placement of the sign, all required engineering, and (b) Tenant has obtained and submitted to Landlord evidence of the insurance required hereunder and under the Lease and any permits or approvals required by law. The original installation work for the Building Top Signage shall be performed pursuant to a design-build contract between Tenant and a contractor approved by Landlord, which approval shall not be unreasonably withheld or delayed. Such work shall be performed in a manner so as to minimize damage to the Building. Without limiting the generality of the foregoing, Landlord shall have the right to approve all staging and other construction procedures, which approval shall not be unreasonably withheld or delayed, and Tenant shall be responsible for assuring that such installation or other work does not affect any of Landlord's then existing warranties and does not damage the Building (including, without limitation, the roof membrane). All installation or other work hereunder shall be performed in a good and workmanlike manner, in accordance with all governmental requirements, and at Tenant's sole cost and expense. If Tenant is required to remove any glass or other material from the Building, Tenant will obtain Landlord's prior written approval to such removal, which approval shall not be unreasonably withheld or delayed, (unless such removal is shown on the plans previously approved by Landlord) and shall store all such material with appropriate care and replace it when the Building Top Signage is removed. If the window washing equipment for the Building or any other Building equipment requires modification to accommodate installation or repair of the Building Top Signage, Tenant shall perform such modification at its sole cost and expense, subject to Landlord's approval of the nature and scope of such modification. Tenant shall promptly repair, to the extent reasonably practical, at its sole cost and expense, any and all damage to the Building and the Project (including damage to any landscaping) caused by the installation or subsequent removal of the Building Top Signage and Monument Signage.
- 2. Once installation of the Building Top Signage has commenced, it shall be completed as soon as possible, and (subject to Force Majeure Events and delays caused by Landlord) in no event later than one (1) month thereafter.
- 3. The Building Top Signage shall, at all time, be fully and completely illuminated each night between sunset and sunrise, at Tenant's sole cost and expense, based on separate meters (to be installed at Tenant's sole cost and expense), and Tenant shall pay any meter-reading charges in connection therewith. Tenant shall not be permitted to illuminate the Building Top Signage until the foregoing described separate meters have been installed. Such illumination shall be in compliance with applicable governmental requirements.
- 4. Tenant shall maintain the Building Top Signage in good, sightly and first-class appearance, condition and repair, and so as not to detract from the appearance of the Building. Landlord shall have the right to reasonably approve the maintenance personnel. If Tenant shall fail to maintain, repair, or illuminate the Building Top Signage in the condition required hereunder within five (5) business days after written notice by Landlord, Landlord may, after written notice to Tenant, so repair, maintain and illuminate the Building Top Signage, at Tenant's sole cost and expense (which Tenant shall pay to Landlord as Additional Rent when billed by Landlord), without limiting Landlord's other rights and remedies.
- 5. Landlord does not represent or warrant that installation of the Building Top Signage and/or Monument Signage hereunder will comply with any applicable federal, state, county or local law or ordinances or the regulations of any of their agencies or any quasi-governmental requirements or any other applicable agreements. Landlord shall use its good faith efforts (without the expenditure of any money unless paid by Tenant in advance of the required expenditure) to assist Tenant in obtaining such approvals. Tenant shall at all times comply with any applicable laws, ordinances, regulations and requirements pertaining to the Building Top Signage and/or Monument Signage.
- 6. Except to the extent arising out of the negligence or willful misconduct of Landlord, its agents or employees, Tenant shall defend, indemnify and hold Landlord harmless from and against any and all loss, cost, claim, damage, liability or expense which Landlord may incur as a result of the Building Top Signage and Monument Signage or Tenant's installation, maintenance or other activities in connection therewith, including but not limited to Tenant's use of the roof. Tenant shall maintain commercial general liability insurance covering risks of bodily injury, death or property damage arising directly or indirectly out of the Building Top Signage and

Monument Signage or Tenant's installation, maintenance or other activities in connection therewith, including but not limited to, Tenant's use of the roof. Tenant shall provide a certificate of such insurance to Landlord prior to commencing the installation work for the Building Top Signage and Monument Signage, and such insurance policy shall not be cancelable without at least fifteen (15) days' written notice to Landlord. Except to the extent arising out of the negligence or willful misconduct of Landlord, its agents and employees, Landlord shall not be responsible for the Building Top Signage or Monument Signage in the event of loss or damage thereto from any cause whatsoever.

7. Landlord shall have the right to use photographs of the Building, including the Building Top Signage and Monument Signage, in Landlord's brochures and other materials without compensation to Tenant.

8. Upon the earlier of termination of the Lease or the termination of Tenant's sign rights hereunder, by expiration of the Lease or otherwise, Tenant shall (and may at any time during the Lease Term upon reasonably prior written notice) disconnect and remove the Building Top Signage and Monument Signage and repair and restore the Building and monuments. Tenant shall promptly and properly repair (or at Landlord's option, pay Landlord's reasonable charges for repairing) during the Lease Term and upon termination of the Lease or the sign rights hereunder, any roof leaks or other damage or injury to the roof, or the Building (or any portion thereof, contents thereof or equipment associated therewith) caused by the Building Top Signage or Monument Signage or their installation, use, maintenance or removal, except to the extent arising out of the negligence or willful misconduct of Landlord, its agents or employees. If Tenant does not commence to repair any such leaks, damage or injury, or does not commence to remove (after Tenant's sign rights hereunder have terminated) the Building Top Signage and Monument Signage within seven (7) business days after written request, or if Tenant does not thereafter proceed to diligently complete such work, Tenant hereby authorizes Landlord, after written notice to Tenant, to make such repairs or remove the Building Top Signage and Monument Signage, and Tenant shall promptly pay Landlord's reasonable charges for doing so as Additional Rent. With respect to any property so removed by Landlord, Landlord shall comply with applicable law.

EXHIBIT I

EXHIBIT J FORM OF GUARANTY OF LEASE

THIS LEASE GUARANTY (this "Guaranty") is made on ______, 2005, by IMPAC MORTGAGE HOLDINGS, INC., a Maryland corporation ("Guarantor"), for the benefit of SCHOLLE JAMBOREE PROPERTY DEVELOPMENT I LLC, a California limited liability company ("Landlord"), with reference to the following facts:

A. A certain Office Lease (Building II) (the "Lease") dated ______, 2005 has been, or will be, executed between Landlord and IMPAC FUNDING CORPORATION, a California corporation ("Tenant"), covering certain premises located at 19500 Jamboree Road in the City of Irvine, County of Orange, State of California (the "Premises"). For purposes hereof and as used herein, the term "Tenant" shall also mean, as the case may be, any successor to Impac Funding Corporation, succeeding to its interest as the "Tenant" under the Lease;

- B. Guarantor has a financial interest in Impac Funding Corporation and, for business reasons significantly beneficial to Guarantor has a financial interest in Tenant, desires that Landlord enter into the Lease with Tenant.
- C. Landlord requires as a condition to entering into the Lease that Guarantor guarantee Tenant's full and timely performance under the Lease on terms and conditions acceptable to Landlord.
- D. Guarantor has requested that Landlord accept this Guaranty to induce Landlord to enter into the Lease, and Landlord is willing to accommodate such request.

NOW, THEREFORE, in consideration of Landlord entering into the Lease, and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties agree as follows:

- (1) Guarantor, acting by and through its duly authorized corporate officers, unconditionally, absolutely and irrevocably guarantees to Landlord (i) the punctual payment and performance of all of Tenant's present and future indebtedness and obligations to Landlord under the Lease, and under all modifications and amendments thereof, including, without limitation, payment to Landlord of all rent, additional rent and all other financial, indemnification and monetary obligations and liabilities due and to become due to Landlord from Tenant under the Lease in the amounts and at the times provided in the Lease, and performance and observance of all the terms, covenants, agreements and conditions of the Lease to be performed by Tenant. All such indebtedness and obligations are referred to in this Guaranty as the "Indebtedness" and will be payable and performable by Guarantor to Landlord, or order, immediately on demand in the event of any default by Tenant with respect to the Indebtedness or any part thereof, without deduction, offset, defense or counterclaim. All payments shall be made to Landlord in lawful money of the United States. Time is of the essence in the performance by Guarantor under this Guaranty. This Guaranty is an absolute, unconditional and irrevocable guaranty of payment and performance of the Indebtedness (and not merely of collection) pursuant to the terms, conditions and covenants herein.
- (2) Guarantor assumes the full responsibility for being and keeping itself informed of the financial condition and assets of Tenant and of all other circumstances bearing upon the risk of nonpayment of the Indebtedness which diligent inquiry would reveal. Landlord shall have no duty to advise Guarantor of information known to Landlord regarding such condition or any such circumstances, regardless of whether Landlord has reason to believe that (i) any such information materially increases the risk beyond that which Guarantor intends to assume or (ii) such information is unknown to Guarantor, and regardless of whether Landlord has a reasonable opportunity to communicate such information to Guarantor. By executing this Guaranty, Guarantor knowingly accepts the full range of risks encompassed within a contract of continuing guaranty, which risks Guarantor acknowledges include the possibility that Tenant will incur additional Indebtedness for which Guarantor will be liable after Tenant's financial condition or ability to pay such Indebtedness has deteriorated.
- (3) Guarantor authorizes Landlord, without notice or demand and without affecting or impairing Guarantor's liability hereunder, from time to time to do one or more of the following: (a) renew, compromise, settle, extend, increase, accelerate or otherwise change the time for payment, discharge or performance of the Indebtedness or any part thereof; (b) change the terms of the Indebtedness or any part thereof, including increase or decrease the rate of interest thereon or the rent under the Lease; (c) take and hold security for the payment of the Indebtedness or any part thereof or any guarantee thereof, and subordinate, exchange, modify, vary, enforce, waive, release (intentionally or unintentionally), renew or abstain from perfecting or taking advantage of any security; or (d) release, substitute or add any one or more cosigners, endorsers or other guarantors of the Indebtedness.
- (4) Guarantor waives: (a) all presentments, demands for performance, notices of nonperformance, protests, and all other notices, including notices of all of the following: protest, dishonor, acceptance of this Guaranty, any default, partial payment or nonpayment of all or any part of the Indebtedness and the existence, creation or incurring of new or additional Indebtedness; (b) any right to require Landlord to proceed against Tenant or any other person, to proceed against or exhaust any security held from Tenant or any other person for the Indebtedness, to proceed against or exhaust any security held from Guarantor or any other person for this

Guaranty or to pursue any other remedy in Landlord's power whatsoever; (c) the benefits of any laws which provide that the obligation of a guarantor must neither be larger in amount nor in other respects more burdensome than that of the principal or which reduce a guarantor's obligation in proportion to the principal obligation; (d) any defense arising by reason of the invalidity, illegality or lack of enforceability of the Indebtedness or any part thereof, or by reason of any incapacity, lack of authority, death, disability or other defense of Tenant or any other person, or by reason of the failure of Landlord to file or enforce a claim against the estate (in administration, bankruptcy or any other proceeding) of Tenant or any other person, or by reason of the cessation from any cause whatsoever of the liability of Tenant or any other person with respect to all or any part of the Indebtedness, or by reason of any act or omission of Landlord or others which directly or indirectly results in the discharge or release of Tenant or any other person or any Indebtedness or any security therefor, whether by operation of law or otherwise; (e) any defense based upon: (i) any failure of Landlord to give Tenant or Guarantor notice of any sale or other disposition of any property securing any or all of the Indebtedness or any guarantee thereof, or (ii) any modifications or amendments to the Lease (regardless of how extensive and regardless of whether those modifications or amendments materially increase Tenant's rent or other obligations) or extension of the Lease or change in the size or location of the space leased; (f) any defense based upon or arising out of any bankruptcy, insolvency, reorganization, arrangement, readjustment of debt, liquidation or dissolution proceeding commenced by or against Tenant or any other person, including any discharge of, or bar against collecting, any of the Indebtedness (including any interest thereon), in or as a result of any such proceeding; (g) any defense based upon termination of t

- (5) Landlord may, at its election, but without obligation to do so, exercise any right or remedy it may have against Tenant or any other person or any real or personal property security it holds for the Indebtedness or any part thereof or any guarantee thereof, including foreclosing a trust deed, mortgage or other instrument judicially or nonjudicially or taking a deed, assignment or transfer in lieu of foreclosure as to any such property, without affecting or impairing in any way the liability of Guarantor hereunder except to the extent the Indebtedness has been paid, even if the effect of such action is to destroy or diminish Guarantor's subrogation rights, Guarantor's right to proceed against Tenant for reimbursement, Guarantor's right to recover contribution from any other guarantor or any other right or remedy or any such security, and Guarantor waives all rights and defenses arising out of an election of remedies by Landlord, even though that election of remedies, such as a nonjudicial foreclosure with respect to any security that may be given for the Indebtedness, has destroyed Guarantor's rights of subrogation and reimbursement against Tenant by the operation of Section 580d of the California Code of Civil Procedure or otherwise.
- (6) Until all Indebtedness of Tenant to Landlord shall have been paid and performed in full, even though such Indebtedness may be in excess of Guarantor's liability hereunder, and all periods within which any amount paid on account of the Indebtedness may be required to be restored or returned by Landlord upon the bankruptcy, insolvency or reorganization of Tenant, any other guarantor or any other person have expired, Guarantor shall have no right of subrogation, and waives any right to enforce any remedy which Landlord now has or may hereafter have against Tenant, and waives any benefit of, and any right to participate in, any security now or hereafter held by Landlord.
- (7) Without in any manner limiting the generality of any other provision of this Guaranty, Guarantor waives all benefits and defenses under the provisions of Civil Code sections 2809 (which provides generally that the liability of a surety must be neither larger in amount nor more burdensome than that of the principal), 2810 (which provides generally that a surety is not liable if the principal had no liability at the time of the execution of a contract or thereafter ceases to be liable), 2819 (which provides generally that a surety is exonerated if the original obligation of the principal is altered in any respect, or the remedies or rights of a creditor against the principal are in any way impaired or suspended without the consent of the surety), 2839 (which provides generally that performance of the principal obligation, or an offer of such performance, exonerates a surety), 2845 (which provides generally that a surety may require a creditor to proceed against the principal, or to pursue any other remedy in the creditor's power which the surety cannot pursue; and if the creditor neglects to do so, the surety is exonerated to the extent the surety is thereby prejudiced), 2847, 2848 and 2849 (which generally provide a surety with rights of subrogation or reimbursement against a principal, rights of subrogation or reimbursement against any collateral or security provided to secure the obligations of the principal, and rights of contribution against any other guarantor or pledgor), 2850 (which provides generally that whenever property of a surety is hypothecated with property of the principal, the surety is entitled to have the property of the principal first applied to the discharge of the obligation), 2899 and 3433 (which relate generally to marshalling assets and the order in which liens may be foreclosed), and any similar or analogous statutes of this or any other jurisdiction.
- (8) The obligations of Guarantor hereunder are independent of the obligations of Tenant. A separate action or actions may be brought and prosecuted against Guarantor without first proceeding against Tenant or any other person or any security held by Landlord and without pursuing any other remedy and without joining Tenant or any other person in any such action or actions. Guarantor waives the benefit of any statute of limitations affecting his liability hereunder or the enforcement thereof. Any payment of any Indebtedness or other act which shall toll any statute of limitations applicable thereto shall also operate to toll such statute of limitations applicable to Guarantor's liability hereunder.
- (9) It is not necessary for Landlord to inquire into the powers of Tenant or of the officers, directors, partners or agents acting or purporting to act on its behalf, and any Indebtedness made or created in reliance upon the professed exercise of such powers shall be guaranteed hereunder.
- (10) Guarantor agrees to pay reasonable attorneys' fees and all other costs and expenses which may be incurred by Landlord in any effort to collect or enforce the Indebtedness or this Guaranty.

- (11) The liability of Guarantor hereunder shall be reinstated and revived, and the rights of Landlord shall continue, with respect to any amount at any time paid on account of the Indebtedness which shall thereafter be required to be restored or returned by Landlord upon the bankruptcy, insolvency or reorganization of Tenant, or otherwise, all as though such amount had not been paid.
- (12) On Landlord's request, Guarantor will provide complete and current financial statements, in a form reasonably satisfactory to Landlord, and such other financial information about Guarantor as Landlord may reasonably require. Upon written request by Guarantor, Landlord shall enter into a commercially reasonable confidentiality agreement covering any confidential information that is disclosed by Guarantor. So long as Tenant is a publicly traded company, Guarantor's most recent 10-K and 10-Q reports filed with the Securities and Exchange Commission shall serve to satisfy the requirement for a current financial statement.
- (13) This Guaranty shall inure to the benefit of Landlord, its successors and assigns, and shall bind the heirs, executors, administrators, successors and assigns of Guarantor. This Guaranty is assignable by Landlord without notice with respect to all or any portion of the Indebtedness hereby guaranteed, and when so assigned, Guarantor shall be liable to the assignees under this Guaranty without in any manner affecting the liability of Guarantor hereunder with respect to any Indebtedness retained by Landlord.
- (14) This Guaranty shall be governed by and construed according to the laws of the State of California. Whenever the context so requires, the singular includes the plural. As used herein, "any other person" shall include, without limitation, any other guarantor of the Indebtedness or any part thereof, any endorser thereof or any co-maker thereof, "including" shall not be limiting, and "or" shall not be exclusive. If this Guaranty is executed by more than one Guarantor, their obligations hereunder shall be joint and several. The invalidity or unenforceability of any one or more provisions of this Guaranty will not affect any other provision. Any legal action with respect to this Guaranty may be brought in the courts of the State of California, with venue in the County of Orange, or in the United States District Court in California, with venue in the County of Orange, as Landlord may elect. Guarantor submits to each such jurisdiction and waives any and all objections that it may have as to jurisdiction or venue in either of the above courts, and Guarantor hereby irrevocably appoints Tenant as its agent in Orange County, California for service of process.
- (15) This Guaranty constitutes the entire agreement of Guarantor and Landlord with respect to the subject matter hereof and there are no promises, statements or representations of any kind or nature whatsoever other than those herein contained. No delay or failure by Landlord to exercise any right or remedy against Guarantor will be construed as a waiver of that right or remedy. The terms and provisions of this Guaranty may not be waived, altered, modified or amended except by written instrument executed by Guarantor and Landlord. All rights and remedies of Landlord against Guarantor are cumulative and not exclusive and may be exercised successively or concurrently. No exercise of any right or remedy shall be deemed to be an election of remedies and preclude exercise of any other right or remedy.
- (16) If a demand, request, approval, consent or notice (collectively referred to as a "<u>notice</u>") shall or may be given to either party by the other, the notice shall be in writing and shall be (A) sent by United States certified or registered mail, postage prepaid, return receipt requested ("<u>Mail</u>"), (B) transmitted by telefacsimile, if such telefacsimile is promptly followed by a notice sent by Mail, (C) delivered by an overnight or same day courier, or (D) delivered by hand. Any Notice shall be sent, transmitted, or delivered, as the case may be, to the party's respective notice address(es) set forth below. Any notice will be deemed to have been received or given (i) three (3) days after the date it is posted if sent by Mail, (ii) the date of transmission of the telex or fax transmission or other form of electronic transmission, if such transmission is prior to 3:00 p.m. in the recipient's time zone (if after 3:00 p.m., the notice shall be effective at 9:00 a.m. on the next business day after such transmission), or (iii) on the earlier to occur of actual delivery or the date on which delivery is refused, if sent by courier or by hand. Either party may, at any time, change its Notice Address by giving the other party written notice of the new address in the manner described in this Section. If Tenant is notified of the identity and address of Landlord's Lender, Tenant shall give to such Lender written notice of any default by Landlord under the terms of this Lease by registered or certified mail.
 - (17) Guarantor represents and warrants to Landlord as follows:
 - (i) Guarantor is the parent company of Tenant, and expects to derive a direct and substantial benefit from the Lease/
 - (ii) Guarantor has the full power, authority and legal right to execute and deliver this Guaranty and to perform, fulfill and observe all the provisions of this Guaranty, including but not limited to payment of all monies which Guarantor is obligated to pay hereunder.
 - (iii) This Guaranty constitutes the legal, valid and binding obligation of Guarantor, enforceable in accordance with its terms.
 - (iv) Neither the execution, nor delivery of this Guaranty, nor the compliance with or the performance of the terms and provisions hereof, conflict or will conflict with or result in a breach of any of the terms, conditions or provisions of any law, order, writ, injunction or decree of any court or governmental authority, or of any agreement or instrument to which Guarantor is a party or by which it is bound.
 - (v) Guarantor has read the Lease and is fully familiar with its contents.

	(18) It is an express condition precedent to the granting, execution and delivery of the Lease by Landlord that Guarantor execute and deliver this
Guaranty.	
-	
	"GUARANTOR":

	AC MORTGAGE HOLDINGS, INC., ryland corporation
By:	
Its:	
By:	
Its:	

Guarantor's Addresses for Notice:

Impac Mortgage Holdings, Inc. 1401 Dove Street Newport Beach, CA 92660 Attention: General Counsel Fax: 949-475-3967

Note: Guarantor shall deliver to Landlord a certified copy of a corporate resolution in a form reasonably acceptable to Landlord authorizing the signatory(ies) to execute this Guaranty.

EXHIBIT J

4

EXHIBIT K

FORM OF TELECOMMUNICATIONS AGREEMENT

This Telecommunications Agreement (this "Agreement") is entered into as of("Landlord"), and, a("Tenant").	, 200_ by and between, a	,
<u>R E C I T A L S</u>		
This Agreement is made with regard to the following facts:		
A. Landlord and Tenant entered into that certain Office Lease (Building II) dated	, 2005 (the " Lease "), under which Tenant leases	

B. In connection with the Lease, Tenant desires to use up to 2,000 square feet of an additional area located on the roof area of the Building for the purpose of constructing, installing, operating, replacing (subject to Section 3 of this Agreement) and maintaining a [INSERT APPROPRIATE EQUIPMENT] microwave, radio, satellite and other telecommunications equipment as set forth on Schedule 1, attached hereto (individually, a "Transmission Device", and collectively, the "Transmission Devices"), and an additional area within the Building's riser system for the telecommunications conduits, devices, fiber optics and electrical, coaxial, and other connections (collectively, the "Connections") necessary to connect the Transmission Devices to the Premises. The Transmission Devices and Connections are sometimes referred to in this Agreement collectively as the "Telecommunication Devices". Landlord has agreed to permit Tenant to use those areas and to construct, install, operate, repair, replace, and maintain the Telecommunication Devices at Tenant's sole cost and expense.

approximately 200,000 rentable square feet (the "Premises") in that certain office building located at 19500 Jamboree Road, Irvine, California (the "Building").

AGREEMENT

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Landlord and Tenant hereby agree as follows:

- 1. License of Transmission Devices Area.
- 1.1 <u>Designation Notice</u>; <u>Transmission Devices Area</u>. On thirty (30) days prior notice from Tenant to Landlord (the "**Designation Notice**"), Landlord shall designate areas on the roof of the Building (the "**Roof Location**") that Tenant may use for the purpose of constructing, installing, operating, replacing (subject to <u>Section 3</u> of this Agreement) and maintaining the Transmission Devices (collectively, the "**Transmission Devices Area**"). For purposes of this Agreement, the Building riser system as used by Tenant for the Connections shall also be referred to as a part of the Transmission Devices Area.
- 1.2 <u>Notice of Exercise</u>. Tenant may exercise its right to use that Transmission Devices Area by written notice delivered to Landlord (the "**Notice of Exercise**") at any time after the delivery of the Designation Notice. The terms of this Agreement shall be effective upon the date of this Agreement and shall continue in effect until the expiration or earlier termination of this Agreement as set forth in Section 1.3, below, or the failure of Tenant to timely deliver a Notice of Exercise.
- 1.3 <u>License to use the Transmission Devices Area; Exclusive Use</u>. Five (5) business days following the delivery of the Notice of Exercise, Tenant's license to use the Transmission Devices Area to construct, install, operate, repair, replace (subject to <u>Section 3</u> of this Agreement) and maintain the Telecommunication Devices shall commence and shall continue until the earlier of (i) the expiration or earlier termination of the Lease, (ii) any termination of this Agreement required by law, governmental authority or quasi-governmental authority, or (iii) the effective date set forth in a written notice from Tenant to Landlord electing to terminate this Agreement. Subject to the rights of Landlord to maintain, operate and repair the Building, Tenant shall have the exclusive right to use the Roof Location. Landlord shall have the right to use, and to grant to third parties the right to use, the Building riser system, and portions of the roof of the Building, other than the Roof Location, so long as such subsequent use does not unreasonably interfere with Tenant's use.

EXHIBIT K

- 1.4 Access to Telecommunication Devices. During the term of this Agreement, and subject to Landlord's reasonable procedures regarding Building access and security, Tenant, its agents, employees and contractors, will have the right of access to the Transmission Devices, the Transmission Devices Area, and the Connections, 24-hours a day. In the event that the Connections are only accessible through space in the Building leased to other tenants, Tenant may access such Connections only (i) after giving Landlord at least three (3) business days' prior written notice, (ii) if such access is reasonably necessary, and (iii) if accompanied by an agent of Landlord. In the event of an emergency, Tenant shall notify Landlord of such emergency and, thereafter, Landlord shall use its commercially reasonable efforts to respond more quickly than as set forth in this Section 1.4 to the access needs of Tenant.
- 1.5 Ownership and Removal of Telecommunication Devices. The Telecommunication Devices shall at all times remain the property of Tenant. Tenant shall have the right to remove the Telecommunication Devices, or any part thereof, at any reasonable time upon at least thirty (30) days' prior written notice to Landlord; provided that in the event of an emergency, Landlord shall use its commercially reasonable efforts to allow Tenant to remove such Telecommunication Devices upon less notice. On or before the expiration or earlier termination of this Agreement, Tenant will remove, at its own cost and expense, the Telecommunication Devices and all related facilities in the Transmission Devices Area (specifically including, but not limited to, (i) any fencing and barriers securing the Telecommunication Devices, and (ii) any Connections installed by or on behalf of Tenant in the Building riser system), and return the Transmission Devices Area (specifically including, but not limited to, the Building riser system) to its condition existing prior to Tenant's installation of the Telecommunication Devices. If Tenant fails to complete such removal or fails to repair any damage caused by such removal, Landlord may complete such removal and repair such damage and charge the cost thereof to Tenant, which amounts shall be immediately payable by Tenant.
- 2. Rent. Tenant is not obligated to pay any base rent for the use of the Transmission Devices Area. In addition, Tenant shall, however, pay, as Additional Rent, all costs incurred by Landlord or Tenant for Tenant's use of Building utilities in connection with the Telecommunication Device (including the cost of any separate metering requested by Landlord), including, without limitation, any electricity, water, gas, or heating, ventilation or air conditioning. In addition, Tenant shall directly pay for all costs in connection with the construction, installation, operation, maintenance, repair, replacement, and insurance of the Telecommunication Devices and the Transmission Devices Area.
 - 3. Installation, Maintenance and Operation of Telecommunication Devices.
- 3.1 <u>Approvals and Permits</u>. During the term of this Agreement and subject to the terms of Section 3.2, below, Tenant may install and operate the Telecommunication Devices in the Transmission Devices Area, in the particular locations as indicated on <u>Schedule 2</u> for the particular Telecommunication Devices, provided that: (a) Tenant has obtained Landlord's prior written approval, which approval shall not be unreasonably withheld, conditioned or delayed, of the plans and specifications for the Telecommunication Devices and all working drawings for the installation of the Telecommunication Devices, (b) Tenant has obtained all required permits and governmental or quasi-governmental approvals (including satisfying any applicable Federal Communications Commission and Federal Aviation Administration requirements) to install and operate the Telecommunication Devices, and (c) Tenant complies with all applicable governmental and quasi-governmental laws, regulations and building codes in connection with the Transmission Devices Area and the Telecommunication Devices. Landlord shall have the right to condition its approval of any Telecommunication Devices proposed to be installed by Tenant on Tenant, among other things, erecting fencing or other barriers to secure such devices as required by applicable Law or as determined by Landlord to be good operational practice. With regard to Tenant obtaining all required permits and approvals set forth in Section 3.1(b) above, Landlord shall reasonably cooperate, at Tenant's sole cost, with Tenant; provided, however, that Landlord shall not be responsible for any such approvals. Once Landlord has given its requisite approval, Tenant may not alter or modify the working drawings, or the actual installation of the Telecommunication Devices without Landlord's prior written consent, which consent shall be granted or withheld in Landlord's reasonable discretion.
- 3.2 <u>Compatibility with Building Systems and Operations</u>. All Telecommunication Devices shall be compatible with the Building systems and equipment and shall not impair window washing or the use of chiller units, the cooling tower, the emergency generator, elevators, machine rooms, helipads, ventilation shafts, if any, or any other parts of the Building. If the installation, maintenance, repair, operation or removal of the Telecommunication Devices require any changes or modifications to any structural systems or components of the Building or any of the Building's systems or equipment, Landlord shall have the right, after reasonable notice to Tenant, to either (i) perform such changes or modifications and Tenant shall pay for the costs thereof upon demand or (ii) require Tenant to perform such changes or modifications at Tenant's sole cost and expense. If required by Landlord, in its sole discretion, or any governmental agency or authority, Tenant shall fully enclose the Transmission

Devices Area with suitable fencing or other required enclosures, subject to the terms of Section 3.1, above. Landlord shall have the right to post notices of non-responsibility in connection with any work performed by Tenant or its agents or contractors in connection with this Agreement. The terms and conditions of Section 9.4 of the Lease shall specifically be applicable in connection with any work performed by Tenant or its agents or contractors in connection with the Telecommunication Devices or this Agreement.

- 4. Use of Transmission Devices Area. Tenant shall have the right to use the Building electricity, water, gas and heating, ventilation and air conditioning located on the roof of the Building for the operation of the Telecommunication Devices. Tenant will not store any materials in the Transmission Devices Area. Tenant will use the Transmission Devices Area solely for the Telecommunication Devices and not for any other purpose. Landlord and its agents may enter and inspect the Transmission Devices Area after reasonable notice to Tenant (except in the event of an emergency) and subject to Tenant's reasonable security requirements. Concurrently with Tenant's installation of any locks for the Transmission Devices Area, Tenant will deliver to Landlord a key for any such lock. Tenant will not interfere with the mechanical, electrical, heating, ventilation and air conditioning, or plumbing systems of the Building or the operation, or transmission of any other satellite, microwave, or other broadcasting or receiving devices that are, or will be, located on the roof of, or in, the Building.
- 5. Indemnification and Insurance. Tenant agrees and acknowledges that it shall use the Transmission Devices Area at its sole risk, and, except to the extent due to the negligence or wilful misconduct of Landlord or any Landlord Parties, Tenant absolves and fully releases Landlord and the "Landlord Parties," as that term is defined in the Lease, from (i) any and all cost, loss, damage, expense, liability, and cause of action, whether foreseeable or not, arising from any cause, that Tenant may suffer to its personal property located in the Transmission Devices Area, or (ii) that Tenant or Tenant's officers, agents, employees, or independent contractors Landlord or the Landlord Parties may suffer as a direct or indirect consequence of Tenant's use of the Transmission Devices Area, the Telecommunication Devices or access areas to the Transmission Devices Area, or (iii) any other cost, loss, damage, expense, liability, or cause of action arising from or related to this Agreement. Landlord and Tenant agree that the mutual indemnification and waiver of claims provisions in Section 14 of the Lease shall apply with respect to the Telecommunication Devices and the Transmission Devices Area. In addition, Tenant will procure and maintain, at Tenant's sole expense, insurance in connection with the Transmission Devices Area, the Telecommunication Devices and the obligations assumed by Tenant under this Agreement, in the same amounts and with the same types of coverage as required to be procured by Tenant under the Lease.
- 6. Defaults. If Tenant fails to cure the breach of any of the covenants set forth in this Agreement within thirty (30) days following notice from Landlord, or fails to commence to cure such breach, and diligently pursue such cure to completion, Landlord shall have the right to terminate this Agreement upon written notice to Tenant. In addition, at the option of Landlord, breach of any of the covenants under this Agreement by Tenant will also constitute a Default by Tenant under the Lease, and a Default by Tenant under the Lease will also constitute a Default by Tenant under this Agreement (in which event Landlord may terminate this Agreement concurrently with any early termination of the Lease).
 - 7. Notices. Any notice required or permitted to be given under this Agreement by Tenant or Landlord will be given under the terms of the Lease.
- 8. Incorporation of Lease Provisions. All applicable provisions of the Lease apply to Tenant's payment of charges pursuant to this Agreement, and the Transmission Devices Area and Tenant's use thereof in the same manner as those provisions apply to the Premises and are incorporated into this Agreement by this reference as though fully set forth in this Agreement. In the event of any conflicts between the provisions of this Agreement and the Lease, in connection with the interpretation of this Agreement only, the provisions of this Agreement shall govern.
- 9. No Warranty. Landlord has made no warranty or representation that the Telecommunication Devices are permitted by law and Tenant assumes all liability and risk in obtaining all permits and approvals necessary for the installation and use of the Telecommunication Devices. Landlord does not warrant or guaranty that Tenant will receive unobstructed transmission or reception to or from the Telecommunication Devices and Tenant assumes the liability for the transmission and reception to and from the Telecommunication Devices.
- 10. Assignment. Notwithstanding any contrary provision set forth in this Agreement above, this Agreement, and Tenant's rights contained herein, may not be transferred or assigned to any other person or entity other than an assignee of the Lease permitted pursuant to the terms of the Lease, and no person or entity other than Tenant and its employees shall be entitled to use the Telecommunication Devices or the Transmission Devices Area.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first above written.

SCHOLLE JAMBOREE PROPERTY DEVELOPMENT I LLC, a California limited liability company
By:
Name:
Title:
TENANT: IMPAC FUNDING CORPORATION, a California corporation By: Name: Title:

LANDLORD:

Schedule 1 and Schedule 2

[to be attached]

EXHIBIT K

4

EXHIBIT L FORM OF MEMORANDUM OF LEASE

RECORDING REQUESTED BY AND WHEN RECORDED RETURN TO:

MEMORANDUM OF OFFICE LEASE (BUILDING II)

THIS MEMORANDUM OF OFFICE LEASE (BUILDING II) (this "Memorandum") is entered into as of ______, 2005, by and between SCHOLLE JAMBOREE PROPERTY DEVELOPMENT I LLC, a California limited liability company ("Landlord"), and IMPAC FUNDING CORPORATION, a California corporation ("Tenant").

RECITALS:

A. Landlord and Tenant hereby covenant and agree that Landlord has leased and does hereby lease to Tenant and Tenant has leased and does hereby rent from Landlord, the Premises, as defined in the Lease, located on that certain real property in the City of Irvine, County of Orange, State of California, more particularly described in Exhibit A attached hereto and incorporated herein by this reference (the "Property") on the terms and conditions and for the consideration set forth in that certain unrecorded document dated as of March 1, 2005 and entitled "Office Lease (Building II)," which terms and conditions are incorporated herein by reference (the "Lease"). All capitalized terms not otherwise defined herein shall have the meaning assigned thereto in the Lease.

B. Landlord and Tenant desire to provide notice (i) of Tenant's lease of the Premises within that certain building to be constructed at the Property commonly known as Building II, and (ii) that Tenant has certain rights of first offer to lease space in those certain buildings constructed or that may be constructed at the Property commonly known as Building I, Building III and Building IV, (iii) that Tenant has the option to lease space in Building III, and (iv) that Tenant has a right of first offer to purchase some or all of the Property, all subject to and on the terms and conditions as more fully set forth in the Lease.

NOW, THEREFORE, in consideration of the fact hereinabove set forth, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Landlord and Tenant hereby agree as follows:

AGREEMENT:

- 1. <u>Demise of Premises</u>. Landlord leases to Tenant, and Tenant leases from Landlord, subject to the terms and conditions set forth in the Lease, the Premises as defined in the Lease. The Lease Commencement Date shall be as set forth in the Lease, and the term of the Lease shall continue for 120 months thereafter.
- 2. <u>Extension Option</u>. Landlord has granted to Tenant, subject to the terms and conditions as set forth in the Lease, two (2) consecutive options of five (5) years each to extend the term of the Lease, provided the option(s) are exercised in writing at least twelve (12) months prior to the applicable expiration date.
- 3. <u>Rights to Lease</u>. Landlord has granted to Tenant, subject to the terms and conditions as set forth in the Lease, (i) certain rights of first offer to lease space within Building I, Building III and Building IV, and (ii) an option to lease Building III, provided that such rights and option are exercised by the time periods and in the manner set forth in the Lease.

EXHIBIT L

4. <u>Right of First Offer to Purchase</u> . Landlord has granted to Tenant, subject to the terms and cosome or all of the entire Property, provided that such right is exercised by the time periods and	
5. <u>Incorporation by Reference; No Modification of Lease</u> . The terms and conditions of the Leaprepared and recorded for the purpose of putting the public on notice of the Lease, and this M Lease. In the event of any inconsistency between the terms and conditions of this Memorandu conditions of the Lease shall control.	emorandum in no way modifies the terms and conditions of the
IN WITNESS WHEREOF, Landlord and Tenant have caused this Memorandum to be e	executed the day of, 2005.
	LANDLORD:
	SCHOLLE JAMBOREE PROPERTY DEVELOPMENT I LLC a California limited liability company
	Ву:
	Name:
	Title:
	TENANT:
	IMPAC FUNDING CORPORATION, a California corporation
	Ву:
	Name:

EXHIBIT L

Title:

STATE OF)	
COUNTY OF) ss.	
(or proved to me on the basis of satisfactory evid	, a Notary Public in and for said state, personally appeared, personally known to me dence) to be the person whose name is subscribed to the within instrument and acknowledged to me that he/she and that by his/her signature on the instrument, the person, or the entity upon behalf of which the person acted,
WITNESS my hand and official seal.	
	Notary Public in and for said State
STATE OF) ss. COUNTY OF)	
(or proved to me on the basis of satisfactory evid	, a Notary Public in and for said state, personally appeared, personally known to me dence) to be the person whose name is subscribed to the within instrument and acknowledged to me that he/she and that by his/her signature on the instrument, the person, or the entity upon behalf of which the person acted,
WITNESS my hand and official seal.	
	Notary Public in and for said State
	EXHIBIT L

EXHIBIT A TO EXHIBIT L

LEGAL DESCRIPTION OF THE PROPERTY

The Property is situated in the State of California, County of Orange, and is described as follows:

The real property referred to in this Exhibit is situated in the State of California, County of Orange, described as follows:

PARCEL A:

Parcel 2, in the City of Irvine, County of Orange, State of California, as shown on a Parcel Map filed in Book 47, Page 22 of Parcel Maps, in the office of the County Recorder of said County.

Except all oil, oil rights, minerals, minerals rights, natural gas, natural gas rights, and other hydrocarbons by whatsoever name known that may be within or under the Parcel of land hereinabove described. Together with the perpetual right of drilling, mining, exploring and operating therefor and storing in and removing the same from said land or any other land, including the right to whipstock or directionally drill and mine from lands other than those hereinabove described, oil or gas wells, tunnels and shafts into, through or across the subsurface of the land hereinabove described, and to bottom such whipstocked or directionally drilled wells, tunnels and shaft under and beneath or beyond the exterior limits thereof, and to redrill, retunnel, equip, maintain, repair, deepen, and operate any such wells or mines, without however, the right to drill, mine, store, explore and operate through the surface or the upper five hundred (500) feet of the subsurface of the land hereinabove described, as reserved by Irvine Industrial Complex, a California Corporation in deed recorded January 31, 1977 as Instrument No. 40776 in Book 12052, Page 1304 of Official Records.

PARCEL B:

An easement for the purposes of ingress and egress and sanitary sewer over and under that portion of Block 50 of Irvine's Subdivision, in the City of Irvine, County of Orange, State of California, shown as Parcel 1 on map filed in Book 22, Page 28 of Parcel Maps, in the office of the County Recorder of said County, described as follows:

Commencing at the most Northerly corner of Parcel 1 as shown on a map filed in Book 22, Page 28 of Parcel Maps, records of Orange County; thence South 49° 21' 34" East 120.0 feet along the Northeast Lot Line of said Parcel 1 to the true point of beginning; thence South 40° 38' 26" West 30.00 feet; thence South 49° 21' 34" East 189.37 feet to a point on the Easterly lot line of said Parcel 1; thence along said lot line North 6° 59' 16" West 44.52 feet to the Northeast corner of said Parcel 1; thence North 49° 21' 34" West 156.48 feet along the Northeast lot line of said parcel 1 to the true point of beginning.

APN: 445-132-06

EXHIBIT L

EXHIBIT M FORM OF LETTER OF CREDIT

IRREVOCABLE STANDBY LETTER OF CREDIT Number: Date: Amount: Expiration: _ **BENEFICIARY** ACCOUNT PARTY We hereby issue our Irrevocable Letter of Credit No. in favor of SCHOLLE JAMBOREE PROPERTY DEVELOPMENT I LLC, a California limited liability company ("Beneficiary"), its successors and assigns, for the account of IMPAC FUNDING CORPORATION, a California corporation. We , California, for any sum or sums not to exceed a total of (\$ undertake to honor your sight draft, upon presentation at our office in) in favor of Beneficiary when accompanied by the original of this Letter of Credit and a written statement signed by an officer of Beneficiary, stating that such moneys are due and owing to Beneficiary. Partial and multiple drawings are permitted under this Letter of Credit. In the event of a partial draw, the amount of the draft shall be endorsed on the reverse side hereof by the negotiating bank. This Letter of Credit is transferable in its entire undrawn balance to a successor beneficiary upon presentation by Beneficiary of the original of this Letter of Credit, together with a written request for transfer executed by Beneficiary. The amount of each draft must be endorsed on the reverse hereof by the negotiating bank. We hereby agree that this Letter of Credit shall be duly honored upon presentation and delivery of the certification specified above It is a condition of this Letter of Credit that it shall remain enforceable against us for a period of from this date and further, that it shall be deemed automatically extended for successive one-year periods without amendment thereafter unless thirty (30) days prior to the expiration date set forth above, or within thirty (30) days prior to the end of any yearly anniversary date thereafter, you shall receive our notice in writing by certified mail, return receipt requested, that we elect not to renew this Letter of Credit for any subsequent year. The draft must be marked "Drawn under Letter of Credit No. dated There are no other conditions of this letter of credit. Except so far as otherwise stated, this credit is subject to the International Standby Practices 1998, International Chamber of Commerce Publication No. 590, and is otherwise governed by the law of the State of California.

EXHIBIT M

1

To: Gretchen Verdugo
From: Richard J. Johnson
Date: January 20, 2005

RE: Proposed Compensation Agreement Terms for Chief Accounting Officer Position

In conjunction with the transition from your current position of Executive Vice President, Impac Warehouse Lending Group, Inc. to Executive Vice President, Chief Accounting Officer ("CAO") of Impac Mortgage Holdings, Inc. ("IMH") and Impac Funding Corporation, ("IFC"). We are proposing the following modification to your existing compensation plan. The following is a summary of the proposed terms:

Effective Date: February 1, 2005.

Term: 3-years from the Effective Date
Title: Executive Vice President, CAO

Base Salary: \$250,000 Per Annum

Bonus Incentive: Up to 120% of Base Salary paid Quarterly

Bonus Incentive Terms: The Bonus Incentive will be paid within 30 days of each calendar quarter and will be paid based upon mutually

agreed upon goals/objectives. Goals and objectives will be established each year in conjunction with the preparation of the Company's Annual Business Plan that is approved my [sic] the Company's board of directors. Individual Management Objectives will be established at the beginning of each quarter. The following are [sic] a

list of goal [sic] and objectives that will determine the actual Bonus Incentive calculation:

50% Based upon the Company attaining its business plan 50% Based upon the Individual Management Objectives

Each Goal and Objective, when calculating the quarterly bonus will have a percentage completion based upon each

goal or objective. For example, IMH's business plan target will have either a 0%, 50%, 75% or 100% factor if IMH meets its quarterly business plan of taxable income. The same applies to the completion of Individual

Management Objectives.

Stock Options: Eligible for Stock Options Based upon IMH's board approved Company Stock Option Plan

Vacation: 4 Weeks

Term without Cause: 12 months trailing Base salary and Bonus Incentive along with health benefits to be paid out proportionally over a

12 month period of time. During the 12 month period, Gretchen will remain as an employee during this period of

time and will continue to vest in her stock option but will not receive any new grants.

The purpose of this position is to train and transition Gretchen into the Chief Financial ("CFO") position of IMH. The follow is a list of benchmarks and training programs that Gretchen needs to complete before being considered for the position of CFO. There are no guarantees that upon completion of these benchmarks and training programs that you will be promoted to CFO. However, after successful completion of all your training objectives to both parties mutual satisfaction and Gretchen is notified that she is not being promoted to CFO of IMH and IFC, Gretchen will have 30 days from that date to elect out of this agreement as a Term without Cause and receive her Base Salary plus health benefits for a period of 12 months subsequent to her last date of employment. This clause is only valid if Gretchen notifies the Company in writing that she is electing out of her arrangement and is only valid for a period of 30 days from completing her training program to both parties mutual satisfaction and being notified that she will not be promoted to CFO at that time.

Training Programs: Specific training classes on FAS 133, FAS 140 and FIN 46

REIT Training classes or attends REIT conferences to update skills and knowledge on managing REIT compliance

measures.

SEC training classes on 10K, 10Q and SOX 404 requirements

Executive Coach: Obtain an executive coach to assist Gretchen in preparing her for becoming IMH's CFO

Benchmarks: Complete and oversee the closing of IMH's books and the preparation of the Company's SEC reporting for at least

two quarters.

Complete the transition of Jim Dickinson to head up the Warehouse Lending Division

Please confirm that these terms are acceptable for your new employment to CAO. Upon getting your sign-off, I will forward to legal counsel to prepare an employment agreement.

Agreed:

/s/ Gretchen Verdugo

Gretchen Verdugo IWLG, Executive Vice President

/s/ Richard J. Johnson

Richard J. Johnson

Executive Vice President, CFO

CERTIFICATION

I, Joseph R. Tomkinson, certify that:

- 1. I have reviewed this report on Form 10-K of Impac Mortgage Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and we have:
 - a. designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation;
 - d. disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting.
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a. all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ Joseph R. Tomkinson

Joseph R. Tomkinson Chief Executive Officer March 31, 2005

CERTIFICATION

I, Richard J. Johnson, certify that:

- 1. I have reviewed this report on Form 10-K of Impac Mortgage Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and we have:
 - a. designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation;
 - d. disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting.
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a. all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ Richard J. Johnson

Richard J. Johnson Chief Financial Officer March 31, 2005

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the annual report of Impac Mortgage Holdings, Inc. (the "Company") on Form 10-K for the period ending December 31, 2004 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), each of the undersigned, in the capacities and on the dates indicated below, hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that to his knowledge:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, except that the Report does not include the following; and
 - (a) the Company's audited consolidated financial statements and the audit report of the Company's independent registered public accounting firm thereon;
 - (b) the audit report of an independent registered public accounting firm relating to management's assessment of the effectiveness, of the Company's internal control over financial reporting:
 - (c) a statement that the independent registered public accounting firm that audited the consolidated financial statements included in the annual report (which, however, are not included in this Report) has issued an audit report on management's assessment of the effectiveness of the Company's internal control over financial reporting;
 - (d) Management's Report on Internal Control Over Financial Reporting; and
 - (e) the consent of the Company's independent registered public accounting firm.
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ JOSEPH R. TOMKINSON

Joseph R. Tomkinson Chief Executive Officer March 31, 2005

/s/ RICHARD J. JOHNSON

Richard J. Johnson Chief Financial Officer March 31, 2005

A signed original of this written statement required by Section 906 will be provided to Impac Mortgage Holdings, Inc. and will be retained by Impac Mortgage Holdings, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.

Note: Upon receipt of the items set forth above, the Company plans to file a new certification to include these items with an amended annual report on Form 10-K/A.